Replication of Bunyamwera virus in *Aedes albopictus* C6/36 cells: establishment and maintenance of a persistent infection

by

Martina Frances Scallan[©]

A Thesis Presented for the Degree of Doctor of Philosophy

in

The Faculty of Science, University of Glasgow

Institute of Virology, Church Street, Glasgow G11 5JR April, 1993

ProQuest Number: 11007693

All rights reserved

INFORMATION TO ALL USERS The quality of this reproduction is dependent upon the quality of the copy submitted.

In the unlikely event that the author did not send a complete manuscript and there are missing pages, these will be noted. Also, if material had to be removed, a note will indicate the deletion.



ProQuest 11007693

Published by ProQuest LLC (2018). Copyright of the Dissertation is held by the Author.

All rights reserved. This work is protected against unauthorized copying under Title 17, United States Code Microform Edition © ProQuest LLC.

> ProQuest LLC. 789 East Eisenhower Parkway P.O. Box 1346 Ann Arbor, MI 48106 – 1346



TABLE OF CONTENTS

Page No.

.

Introduction

1	The	Bunyaviridae	1
	1.1	 General biology 1.1.1 Taxonomy and classification of the Bunyaviridae 1.1.2 Characteristics of the Bunyaviridae 1.1.3 Host range and transmission 1.1.4 Disease and geographical distribution 1.1.5 Invertebrate infection 1.1.5.1 Oral infection 1.1.5.2 Transovarial transmission 1.1.5.3 Venereal transmission 1.1.5.4 Transtadial transmission 1.1.5.5 Effect of infection on the invertebrate host 1.1.6 Evolution of the Bunyaviridae 	1
	1.2	Molecular Biology of the Bunyaviridae 1.2.1 The virion 1.2.2 Coding assignments 1.2.3 Genetics	9
	1.3	 Replication of the Bunyaviridae 1.3.1 Attachment and entry 1.3.2 RNA synthesis 1.3.2.1 Transcription 1.3.2.2 Genome replication 1.3.3 Protein expression strategies and protein processing 1.3.3.1 Expression strategies 1.3.3.2 The kinetics of viral protein synthesis in infected cells 1.3.3.3 Protein processing 1.3.4 Intracellular transport and targetting of the viral glycoproteins 1.3.4 Assembly and virus release 1.3.5 Effects of viral replication on host cells 	14
2		Persistent viral infections	31
		 2.1 Involvement of persistent viral infections in human disease 2.1.1 Mechanisms by which persistent viral infections could cause disease 	32
		 2.2 Characteristics of persistently infected cell cultures 2.2.1 Evolution of viral variants during persistent infections 2.2.2 The role of the host cell in the establishment and maintenance of persistent viral infections 2.2.3 Inherent resistance of cells to virus 	34

- 2.2.4 Co-evolution of cells and virus2.2.5 Interferon

 2.6 Defective interfering particles 2.2.6.1 Generation of DI RNAs 2.2.6.2 Interference 2.2.6.3 Influence of host cell type 2.2.6.4 Cell protection, disease attenuation and persistence 	
ersistent infections established by arboviruses .3.1 Defective interfering particles associated with arboviruses 2.3.1.1 The Bunyaviridae 2.3.1.2 The Alphaviridae	47
 .3.2 Antiviral activity associated with persistent arbovirus infections of invertebrate cell lines .3.3 The value of studying arbovirus persistence 	
ims of this study	54
rials and Methods	55
	55
Cells and virus .1.1 Cells .1.2 Virus	55
Sissue culture materials.2.1Tissue culture media.2.2Tissue culture solutions.2.3Overlay for plaque titrations.2.4Neutral red overlay.2.5Tissue culture plasticware	55
acteria	57
acterial culture materials	57
Plasmids and vectors .5.1 Plasmids .5.2 Cloning vectors .5.3 Recombinant M13mp18 bacteriophage	57
Chemicals, radiochemicals, miscellaneous eagents and enzymes .6.1 Chemicals .6.2 Radiochemicals .6.3 Miscellaneous materials .6.4 Enzymes	58
 Commonly used solutions .7.1 Reaction buffers .7.2 Buffers for RNA extractions .7.3 Buffers for direct RNA sequencing .7.4 Sample buffers (containing dyes to mark progress of electrophoresis) .7.5 Electrophoresis buffers 	59
	 2.6 Defective interfering particles 2.2.6.1 Generation of DI RNAs 2.2.6.2 Interference 2.2.6.3 Influence of host cell type 2.2.6.4 Cell protection, disease attenuation and persistence ersistent infections established by arboviruses 3.1 Defective interfering particles associated with arboviruses 2.3.1.1 The Bunyaviridae 3.3.1.2 The Alphaviridae 3.3 The value of studying arbovirus persistent arbovirus infections of invertebrate cell lines 3.3 The value of studying arbovirus persistence ims of this study ials and Methods ells and virus 1.1 Cells 2.2 Tissue culture materials 2.1 Tissue culture motia 2.2 Tissue culture motia 2.3 Overlay for plaque titrations 2.4 Neutral red overlay 2.5 Tissue culture plasticware acterial acterial acterial and vectors 5.1 Plasmids 5.2 Cloning vectors 5.3 Recombinant M13mp18 bacteriophage hemicals, radiochemicals, miscellaneous agents and enzymes 6.1 Chemicals 6.2 Radiochemicals 6.3 Miscellaneous materials 6.3 Miscellaneous materials 6.4 Enzymes ommonly used solutions 7.3 Buffers for RNA extractions 7.3 Buffers of electrophoresis) 7.5 Electrophoresis buffers

.

3.8 Synthetic oligonucleotides61**3.9** Cell manipulations
3.9.1 Maintenance of cells61

	3.9.2	Establishment of an <i>Aedes albopictus</i> C6/36 cell line (C6/36/BUN) persistently infected with Bunyamwera virus	
	3.9.3 3.9.4	Cloning of cells from the C6/36/BUN cell line Cryopreservation of cell cultures	
	3.9.5	Revival of liquid-nitrogen cell stocks	
3.10	Virus	manipulations	63
	3.10.1	A same of wimit information by plague titration	
	5.10.2 2 10 2	Assay of viral infectivity by plaque tiration Droparation of elite stocks of virus	
	3.10.3	Preparation of virus working stocks	
	3 10 5	Assay for resistance to superinfection	
	3.10.6	Infectious centres assay	
3.11	Bacter	ial manipulations	65
	3.11.1	Preparation of competent bacteria	
	3.11.2	Transformation of <i>E. coli</i> strain JM101	
3.12	DNA 1	manipulations	66
	5.12.1	cDNAs to the Bunyamwera virus M segment	
	3.12.2	Preparation of single-stranded template DNA	
	3.12.3	Quantitation of nucleic acid	
	3.12.4	Assessment of the purity of nucleic acid preparations	
	3.12.5	Ethanol precipitation	
	3.12.6	Restriction enzyme digestion	
	3.12.7	Gel purification of oligonucleotides	
3.13	DNA	sequencing	68
	3.13.1	Sequencing and size determination of recombinant M13mp18 cDNAs	
	3.13.2	Single-stranded DNA sequencing (using Sequenase)	
3.14	Radiol	labelling procedures	69
	3.14.1	Prime cut probes	
	3.14.2	Nick translation	
	3.14.3	Generation of single-stranded α -[³² P] labelled cDNAs	
	3.14.4	End-labelling of oligonucleotides with γ -[³² P]-dATP	
	3.14.5	Synthesis of riboprobes	
	3.14.6	Pulse-labelling with [³⁵ S]-methionine	
3.15	RNA	isolation	72
	3.15.1	Preparation of virion RNA	
	3.15.2	Preparation of fractionated infected cell RNA	
	3.15.3	Preparation of total infected cell RNA	
3.16	RNA	manipulations	74
	3.10.1	Slot blot analysis	
	5.10.2 2.16.2	Northern blot analysis Hybridization conditions	
	3.10.5	Post hybridization washes	
	3.16.5	In vitro translation	
3.17	Direct	RNA sequencing	76
3.18	Electr	ophoresis	77
	3.18.1	SDS-PAGE	
	3.18.2	Agarose gels for electrophoresis of DNA	

3.18.3 Agarose gels containing 5mM methylmercury hydroxide

	3.18.4 Polyacrylamide/urea denaturing gels3.18.5 Treatment of gels after electrophoresis	
3.19	Autoradiography and fluorography	80
3.20	Densitometric analyses	80

4-7 **Results**

4

5

6

Bunyamwera virus wtL₉BUN: plaque-purification and determination of the consensus population sequence of the genomic S RNA segment

4.1	Production of a triply plaque-purified elite stock	
	of Bunyamwera virus	81
4.2	Direct RNA sequence determination of the small (S)	
	genomic RNA segment of wtLoBUN virus	81
4.3	Summary	83

81

84

Comparative study of macromolecular syntheses in Bunyamwera virus infected *Aedes albopictus* C6/36 and BHK cells

5.1	Protein synthesis in Bunyamwera virus	
	infected BHK and C6/36 cells	84
5.2	S segment mRNA levels in infected C6/36 cells	85
5.3	In vitro translation	86
5.4	Accumulation of viral RNA species in Bunyamwera	
	virus-infected Aedes albopictus C6/36 and BHK cells	88
5.5	Summary	90
Stud	lies on Aedes albopictus C6/36 cell lines	
pers	istently infected with Bunyamwera virus	92
6.1	Screening of viruses released from two persistently infected	
	cell lines established by Elliott and Wilkie (1986)	92
6.2	Determination of the temperature-sensitivity of variants	
	arising in the C6/36-PI LO cell line	94
6.3	Sequence analysis of the genomic S RNA segments of	
	viruses released from the C6/36-PI LO cell line	95
6.4	Establishment of an Aedes albopictus C6/36 cell line	
	persistently infected with wtLoBUN virus	97
6.5	Virus release from the C6/36/BUN cell line	97
6.6	Protein profiles in BHK cells infected with culture fluid from	
	the persistently infected C6/36/BUN cell line	98
6.7	Levels of intracellular viral RNAs in the persistently	
	infected cell line	99
6.8	Defective viral RNAs in the persistently infected cells	100
6.9	Encapsidation of viral RNA species in	
	the C6/36/BUN cell line	101
6.10	Summary	102

7

8

Analysis of cell lines cloned from the C6/36/BUN culture

104

7.1	The ability of cells cloned from p12, p20, p30 and p58 of the	
	C6/36/BUN culture to give rise to stable cell lines	104
7.2	Virus release from the cloned cell lines	105
7.3	Determination of the susceptibility of the cloned cell lines to	
	superinfection with wtLoBUN virus	105
7.4	Presence/absence of Bunyamwera virus RNA in	
	the cloned cell lines	106
7.5	Summary	108
	•	
n:	· · · · · · · · · · · · · · · · · · ·	
DIS	cussion	110
0.1		
8.1	Establishment versus maintenance of persistent	
~ •	viral infections	110
8.2	Molecular events underlying the establishment of	
	persistent viral infections	111
8.3	Analysis of Aedes albopictus C6/36 cells persistently	
	infected with Bunyamwera virus	118
8.4	Bunyaviridae DI particles	119
8.5	Cell heterogeneity within persistently infected cell cultures	122
0 /		

References

Acknowledgements

I thank Professor John H. Subak-Sharpe for providing me with the opportunity to work in the Institute of Virology. I appreciated the hard work, helpfulness and cheerfulness of the ladies of the media and wash-up rooms of the Institute. I also thank Dr. John McLauchlan for synthesising the oligonucleotides used during this project.

My thanks go to Dr. Richard M. Elliott for the use of the facilities in his laboratory and for his supervision of my research and proof reading of this thesis. Thanks also to my contemporaries in lab. 104 for their companionship.

I am grateful for the support received from, and the good times shared with my fellow Ph.D. students: Paddy, Valgerdur, Janice, Ruth, John, Graham and Gordon. Thanks to Hazel for being such a good friend, and to the other friendly faces in the Institute.

Heartfelt thanks to the many friends who have helped me to persevere with this thesis, by providing both moral and practical support. These include Anna, Callum, Frances, Shona, Bruce and Marina. Thanks to Elizabeth, Bruce and Helen for lending me computers at various stages and in various locations during the preparation of this thesis. Special thanks to Paddy for persevering with me over the past year and for his considerable contribution to the putting together of this thesis.

Most importantly I would like to thank my family for their constant support and encouragement during my academic pursuits.

This thesis is dedicated to the memory of my grandfather *Mr Patrick Parnell*. His wisdom and hard work were an example to me and he provided advice and financial support throughout my education.

I would also like to thank my new colleagues in Cambridge for their support and patience over the past year. Special thanks to Dr. John Fazakerley for his understanding.

The author was the recipient of a research studentship from the Medical Research Council. Unless otherwise stated, all the results described in this thesis were obtained by the author's own efforts.

Summary

Bunyamwera virus is a member of the Bunyaviridae, a large family of mainly arthropod-borne viruses which possess a tri-segmented RNA genome comprising three segments of single-stranded RNA, designated L (large), M (medium) and S (small). There are five recognised genera within the Bunyaviridae-Bunyavirus, Hantavirus, Nairovirus, Phlebovirus and Tospovirus. Bunyamwera virus is the prototype virus of the Bunyavirus genus and each of its genome segments are of negative polarity.

In common with other arboviruses Bunyamwera virus replicates in its vector species (the mosquito) without causing overt detrimental effects and establishes a persistent infection which lasts for the life time of the mosquito. Upon transmission to a susceptible vertebrate host, Bunyamwera virus causes an acute infection. These disparate host-dependent outcomes of infection can be reproduced in tissue culture. Bunyamwera virus undergoes a lytic replication cycle in mammalian cells which results in cell death, but establishes a long-term, non-cytocidal, persistent infection in cultures of mosquito cells. In this thesis, the molecular mechanisms leading to the establishment of persistent Bunyamwera virus infection of a cloned mosquito cell line, *Aedes albopictus* C6/36 (C6/36) cells, and the factors responsible for the maintenance of the persistent infection, were investigated.

Viral protein and RNA synthesis were compared in BHK and C6/36 cells infected with Bunyamwera virus. In BHK cells, host protein synthesis was inhibited and viral protein synthesis was detected until the cells died. In C6/36 cells, host protein synthesis continued throughout the infection, but viral protein synthesis declined from maximum levels at 24 h pi to barely detectable levels at 36 h pi, despite the presence of translatable S mRNA. The levels of encapsidated L and M RNAs in the C6/36 cells declined after 24 h pi, but encapsidated S RNA species continued to accumulate. In the BHK cell infection, encapsidated L, M and S RNA species continued to accumulate up to 48 h pi. Overall, relatively more S segment RNA than L or M segment RNA accumulated in infected C6/36 cells compared to BHK cells. The decline in viral protein synthesis and genome replication observed in the C6/36 cells after 24 h pi, was paralleled by an accumulation of encapsidated S mRNA. Encapsidation of S mRNA may have a role to play in preventing the cytopathic effects of Bunyamwera virus infection in C6/36 cells by limiting translation of this viral mRNA.

A persistent infection of C6/36 cells was established with a biologically cloned stock of Bunyamwera virus (wtL₉BUN virus) and the persistently infected cell line (C6/36/BUN) was passaged at weekly intervals for over a year without cytopathic effects. The titre of virus released from the cells, the susceptibility of the cells to superinfection with homologous virus, and the level of viral RNA in the cells at different passages fluctuated markedly, but there was no simple relationship between virus titre, superinfectibility and

levels of viral RNA. There was an alteration in the plaque-phenotype of virus released from the persistently infected culture with increasing passage level. By passage 15 the virus released from the C6/36/BUN cell line produced small, cloudy plaques on BHK cell monolayers, in contrast to the clear lytic plaques produced by the wtL₉BUN virus.

Direct RNA sequence analysis was carried out on the S RNA segments of viruses plaque-purified from the supernatant culture fluid of the C6/36-PI LO cell line established by Elliott and Wilkie (Virology 150, p21-32, 1986). Only one nucleotide substitution, in a non-conserved region of the viral genome, was detected in viral isolate BUN18.10 (which was isolated from the persistently infected culture after it had been passaged for four months) suggesting that genetic drift during persistent Bunyamwera virus infection of C6/36 cells was minimal. However, Northern blot analysis of viral RNA extracted from different passage levels of the C6/36/BUN cell line revealed that polymerase errors which generated subgenomic L RNAs, and S RNA species larger than the standard S RNA segment, did occur in the persistently infected cells. The defective viral RNA species were not efficiently encapsidated by N protein. There was a strong selection for encapsidation of full-length S RNAs in the persistently infected cells and the majority of intracellular nucleocapsids contained standard-sized S RNA species.

Stable cell lines were established with cells cloned from the persistently infected C6/36/BUN cell line. The cloned cells were heterogeneous in their ability to produce virus, and their susceptibility to superinfection with wtL₉BUN virus, and in the levels of intracellular viral RNA they contained. Individual clones contained a single prominent defective L RNA species and resistance of the cells to superinfection appeared to correlate more with the levels of full-length viral genomic RNA resident in the cells than with the amount or occurence of defective viral RNAs.

Abbreviations

Viruses

ALMV	alfalfa mosaic virus
BUN	Bunyamwera virus
FMDV	foot and mouth disease virus
HTLV-1	human T-cell lymphotrophic virus
INSV	impatiens necrotic spot virus
LAC	La Crosse virus
MVM	minute virus of mice
NDV	Newcastle disease virus
NEV	Nephropathia Epidemica virus
PIV-1	parainfluenza virus-1
PIV-2	parainfluenza virus-2
PIV-3	parainfluenza virus-3
RSV	respiratory syncytial virus
RVFV	Rift Valley fever virus
SFV	Semliki Forest virus
SSH	snowshoe hare virus
SV-5	Simian virus-5
TMV	tobacco mosaic virus
TSWV	tomato spotted wilt virus
VSV	vesicular stomatitis virus
wtL9BUN	triply plaque-purified stock of Bunyamwera virus
Cell lines	
BHK	baby hamster kidney
C6/36	Aedes albopictus C6/36
C6/36/BUN	Aedes albopictus C6/36 cell line persistently infected with Bunyamwera virus (established during the course of this work)

 \times

Ń

C6/36-PI HI	Aedes albopictus C6/36 cell line persistently infected with Bunyamwera virus (established by Elliott & Wilkie, 1986)
C6/36-PI LO	Aedes albopictus C6/36 cell line persistently infected with Bunyamwera virus (established by Elliott & Wilkie, 1986)
C7/10	Aedes albopictus C7/10
U.4.4	Aedes albopictus U.4.4

Bunyaviridae genome segments

L	large (RNA segment)
М	medium (RNA segment)
S	small (RNA segment)

Bunyaviridae proteins

G1	the larger of the envelope glycoproteins
G2	the smaller of the envelope glycoproteins
L	the viral L protein (viral polymerase)
Ν	nucleocapsid protein
NSm	non-structural protein (encoded in the M RNA)
NSs	non-structural protein (encoded in the S RNA)
Conoral	

General

APS	antiviral polyclonal sera
bp	base pair(s)
BSA	bovine serum albumin
cDNA	complementary deoxyribonucleic acid
Ci	Curie(s)
CIP	calf intestinal phosphatase
CNS	central nervous system
сре	cytopathic effect
CSF	cerebrospinal fluid
ddNTP	2' 3' -dideoxyribonucleoside 5'-triphosphate

DI	defective interfering
DNase	deoxyribonuclease
DTT	dithiothreitol
dNTP	2' -deoxyribonucleoside 5' -triphosphate
E. coli	Escherichia coli
EDTA	sodium ethylenediamine tetra-acetic acid
endo-H	endoglycosidase-H
GMEM	Glasgow modification of Eagle's medium
h	hour(s)
kb	kilobase(s)
kDa	kiloDalton(s)
L-15	Leibovitz' L-15 medium
min	minute(s)
moi	multiplicity of infection
mRNA	messenger ribonucleic acid
NTP	ribonucleoside 5'-triphosphate
O.D.	optical density
ORF	open reading frame
PAGE	polyacrylamide gel electrophoresis
PBS	phosphate buffered saline
RF	replicative form
pfu	plaque forming unit(s)
pi	post infection
RNase	ribonuclease
RNasin	ribonuclease inhibitor
RNP	ribonucleoprotein
rpm	revolutions per minute
S	Svedberg unit (sedimentation coefficient)
SDS	sodium dodecyl sulphate

SSPE	subacute sclerosing panencephalitis
TEMED	N, N, N', N' -tetramethylethylenediamine
Tris	tris (hydroxymethyl) aminomethane
ts	temperature-sensitive
UV	ultraviolet
v	volume
VRCs	vanadylribonucleoside complexes
vRNA	virion RNA
w	weight
w	Watts
wt	wild-type

Amino acids

Symbol	3-letter code	Amino acid	Codons
Α	Ala	Alanine	GCA, GCC, GCG, GCU
С	Cys	Cysteine	UGC, UGU
D	Asp	Aspartate	GAC, GAU
Е	Glu	Glutamate	GAA, GAG
F	Phe	Phenylalanine	UUC, UUU
G	Gly	Glycine	GGA, GGC, GGG, GGU
н	His	Histidine	CAC, CAU
Ι	Ile	Isoleucine	AUA, AUC, AUA
К	Lys	Lysine	AAA, AAG
L	Leu	Leucine	UUG, UUA, CUA, CUC, CUG, CUU
М	Met	Methionine	AUG
Ν	Asn	Asparagine	AAC, AAU
Р	Pro	Proline	CCA, CCC, CCG, CCU
Q	Gln	Glutamine	CAA, CAG
R	Arg	Arginine	CGA, CGC, CGG, CGU, AGA, AGG
S	Ser	Serine	UCA, UCC, UCG, UCU, AGC, AGU
Т	Thr	Threonine	ACA, ACC, ACG, ACU
v	Val	Valine	GUA, GUC, GUG, GUU
W	Тгр	Tryptophan	UGG
Y	Tyr	Tyrosine	UAC, UAU

INTRODUCTION

The work presented in this thesis was aimed at elucidating the molecular mechanisms which enable Bunyamwera virus, the prototype virus of the *Bunyavirus* genus of the family Bunyaviridae, to persist in *Aedes albopictus* C6/36 cells. The literature review is divided into two sections: 'The Bunyaviridae' and 'Persistent viral infections'.

1 The Bunyaviridae

1.1 General biology

1.1.1 Taxonomy and classification of the Bunyaviridae

The Bunyaviridae comprises more than 300 viruses (Karabatsos, 1985). Members are assigned to the family by serological methods such as neutralization, haemagglutination inhibition, complement fixation and enzyme-linked immunosorbent assays (ELISA). In general, viruses within a genus share complement fixation antibodies and are segregated into serogroups on the basis of cross-neutralization and haemagglutination inhibition studies. The neutralization and haemagglutination inhibition tests are thought to be specific for antigenic determinants on the viral glycoproteins, while the complement fixation test is mainly directed against determinants on the nucleocapsid protein. For most member viruses the nucleocapsid protein is more conserved than the glycoproteins and the complementation fixation test will identify antigenic determinants which are shared by different viruses (Shope & Casey, 1962), while the neutralization and haemagglutination inhibition tests will distinguish between closely related viruses, particularly if monoclonal antibodies are employed (Gonzalez-Scarano et al., 1982). Phleboviruses are exceptions in that the complement fixation test is group specific while the haemagglutination inhibition test is cross-reactive (Calisher et al., 1981; Shope, 1985). Recent serological and sequencing data on the S segment of a second tospovirus isolate Impatiens necrotic spot virus (INSV) suggest that the tospoviruses may also be exceptions (Law et al., 1991).

The classification of the Bunyaviridae has recently been reorganised with the merger of the former *Phlebovirus* and *Uukuvirus* genera and the inclusion of a new genus, *Tospovirus*. Hence the Bunyaviridae now consists of five genera: *Bunyavirus; Hantavirus; Phlebovirus; Nairovirus* and *Tospovirus* (Calisher, 1991). The fifth genus was created to include tomato spotted wilt virus (TSWV), an arthropod transmitted virus of plants, as the first plant virus in the family (Milne & Francki, 1984; de Haan *et al.*, 1989a, 1989b, 1990); a second isolate, INSV, has now been identified (Law & Moyer, 1989, 1990). Serological studies, and sequence analysis of the S RNAs of these viruses, support the separation of the two known tospoviruses into distinct serogroups (Law *et al.*, 1991). In addition at least 7 serogroups containing 19 viruses, as well as 22 ungrouped viruses, are considered to be

1

members of the family, but have yet to be assigned to a particular genus (Calisher, 1991). The recent reclassification of the *Phlebovirus* and *Uukuvirus* genera into a single genus arose in the light of sequencing studies, which showed that phleboviruses and uukuviruses express their S RNA via an ambisense coding strategy, distinct from the negative-sense coding strategy employed by the majority of the family. They have identical 5' and 3' terminal nucleotide sequences on their RNA segments, show a high degree of homology in their N proteins and low but significant homology between their glycoproteins. In addition, certain members of each group have been shown to be antigenically related to certain members of the other group (Simons *et al.*, 1990). A brief summary of the taxonomic groups in the family Bunyaviridae is given in Table 1.1. As more data become available and as new viruses are discovered the classification may change.

1.1.2 Characteristics of the Bunyaviridae

Some distinguishing features of the Bunyaviridae are listed below.

(i) The virions appear spherical or pleomorphic under the electron microscope, according to which fixation conditions are used, and are 80 -120 nm in diameter. They are enveloped by a 4-5 nm thick lipid bilayer which is cell-derived and accounts for 20-30% of the weight of the virus particle (Gonzalez-Scarano & Nathanson, 1990; Calisher, 1991).

(ii) Each virion contains four structural proteins, two glycoproteins (G1 and G2), a nucleocapsid protein (N) and the largest of the viral proteins, L, which is believed to be the viral transcriptase (Calisher, 1991). The glycoproteins are embedded in the lipid envelope and project 5-10nm from the virion surface; they possess type-specific antigenic determinants and variations in these glycoproteins contribute unique external features to representatives of different genera (Martin *et al.*, 1985; Schmaljohn & Patterson, 1990).

(iii) Each virus particle contains three molecules (large [L], medium [M] and small [S]) of single-stranded RNA. Each RNA species has conserved complementary 3' and 5' ends which are genus specific (see Table 1.2). The ends of the RNAs are hydrogenbonded to form circular RNA molecules (Calisher, 1991).

(iv) The viral RNAs are present in the virion as nucleocapsids, each consisting of viral nucleoprotein, a unique, single-stranded, non-covalently closed circular genomic RNA segment and a transcriptase enzyme activity. The nucleocapsids are arranged as closed helical circles of ribonucleoprotein and have been designated small, medium and large (S, M and L) according to which of the three viral genome segments is enclosed. The circular nucleocapsids are 2.0-2.5nm in diameter, are sometimes supercoiled and can range in size from 0.2 to 3.0µm in length (Calisher, 1991; Bishop, 1990).

(v) Generally, viral morphogenesis occurs by budding into smooth-surfaced vesicles in or near the Golgi region (Calisher, 1991).

(vi) The viruses can interact genetically with serologically closely related viruses and recombinant viruses may arise by RNA segment reassortment (Bishop, 1990).

BUNYAVIRUS	PHLEBOVIRUS	NAIROVIRUS	HANTAVIRUS	TOSPOVIRUS
(162)	(35)	(33)	(9<)	(2)
Antigenic groups.	Antigenic groups.	Antigenic groups.	Antigenic group.	Antigenic Groups.
Anopheles A(12)	Sandfly fever group	Crimean-Congo	Hantaan (6)	TSWV
Anopheles B (2)	(23)	hemorrhagic fever (3)		NSVI
Bakau (5)	Sandfly fever Naples (4)	Dera Ghazi Khan (6)		
Bunyamwera (32)	Bujaru (2)	Hughes (10)		
Bwamba (2)	Candiru (6)	Qalyub (3)		
C (14)	Chilibre (2)	Sakhalin (7)		
California (13)	Frijoles (2)	Thiafora (2)		
Capim (10)	Punta Toro (2)	Nairobi sheep disease (2)		
Gamboa (8)	Rift Valley fever (3)			
Guama (12)	Salehabad (2)			
Koongol (2)				
Minatitlan (2)	Uukuniemi group.			
Nyando (2)	(Single serogroup			
Olifantsvlei (5)	containing 12 viruses)			
Patois (7)				
Simbu (24)				
Tete (6)				
Turlock (4)				

Table 1.1. Summary of the present classification of the Bunyaviridae. Numbers in brackets indicate the number of viruses in a given genus/antigenic complex (data from Calisher, 1991).

Genus	Sizes of	RNA segments (nuc	leotides)	3' Terminal sequence	5' Terminal sequence
	Γ	W	S		
Bunyavirus	6,875 (1)	4,458 - 4,534 (4)	850 - 984 (6)	UCAUCACAUGA	UCGUGUGAUGA
Phlebovirus	6,423 (1)	3,231 - 4,330 (4)	1,746 - 1,904 (2)	UGUGUUUC	GAACACA
Nairovirus	11,400 - 14,400	4,888 (1)	1,712 (1)	AGAGUUUCU	AGAAACUCU
Hantavirus	6,530 - 6,550 (2)	3,616 -3,682 ⁽⁴⁾	1,675 - 1,785 (4)	AUCAUCAUCUG	AUGAUGAU
Tospovirus	8,897 (1)	4,972 -4,821 (2)	2,916 (1)	UCUCGUUAG	CUAACGAGA

Table 1.2. Sizes and consensus genus specific terminal sequences of the genomic RNAs of the Bunyaviridae. RNA segment sizes are as determined by sequencing of cloned cDNAs; numbers in brackets indicate the number of known sequences. Numbers in italics are size estimates from gel electrophoresis of the RNA where sufficent sequencing data are not available. For references see text.

1.1.3 Host range and transmission

The Bunyaviridae have a wide variety of animal hosts (see Table 1.3). Human to human transmission does not generally occur, and humans are dead end hosts, not producing sufficient viremia to transmit virus to a feeding mosquito (Gonzalez-Scarano & Nathanson, 1990). Crimean-Congo haemorrhagic fever nairovirus is exceptional amongst the Bunyaviridae in that person to person spread can occur and has lead to a number of hospital outbreaks (Gonzalez-Scarano & Nathanson, 1990). TSWV and INSV cause great yield losses worldwide in a large number of economically important crops; for example considerable losses in floral crops throughout North America due to INSV have been reported (Law & Moyer, 1989). The host range of TSWV includes at least 370 plant species in 50 botanical families, including *Nicotiania rustica*, *N. tabacum* and *N. benthamiana* as well as tomato plants (de Haan *et al.*, 1989b; Calisher, 1991; de Oliveira Resende *et al.*, 1991).

Excluding hantaviruses, all members of the family are transmitted by invertebrate vectors, including mosquitoes, ticks, biting midges (*Culicoides* species) and sandflies (*Phlebotomus* species). The Bunyaviridae are maintained and amplified in nature by biological transmission. This involves ingestion of the virus by the vector when feeding upon a viremic vertebrate host; after an extrinsic incubation period (see Section 1.1.5.1) the virus can be transmitted to another vertebrate from the salivary gland secretions of the vector as it seeks another blood meal (Turell, 1988). Another potential means of transmission is mechanical transmission, where the vector feeds upon another host promptly, before the virus becomes inactivated, transmission may occur without a requirement for virus replication in the vector (Beaty & Calisher, 1991). The Bunyaviridae cause a persistent, lifelong infection in their invertebrate vectors.

During major __epizootics of Rift Valley fever, the virus is likely to be amplified by mechanical transmission because of the high level of viremia prevalent in the infected vertebrate hosts (Meegan & Bailey, 1988). Members of the Bunyaviridae can also be maintained within the vector species by transovarial, transtadial and venereal transmission (see Section 1.1.5). TSWV is the only plant virus to date known to be transmitted by thrips (Sakimura, 1962). It has also been reported that TSWV can be vertically transmitted through the seeds of certain plant hosts (Mandahar, 1981). Hantaviruses are primarily rodent-borne and establish persistent infections in these animals..Transmission of hantaviruses is via aerosolized rodent excreta (Lee *et al.*, 1981a, 1981b). In nature individual members of the Bunyaviridae infect a limited number of arthropod and vertebrate hosts (Mclintock, 1978; Murphy *et al.*, 1975). Studies on La Crosse variant and revertant viruses suggest that the specificity is related to the larger of the two viral glycoproteins, G1, probably at the level of viral attachment to susceptible cells (Sundin *et al.*, 1987). The

Genus	Vector species	Natural vertebrate (or plant)	Mechanism of transmission to
		host(s)	humans
Bunyavirus	Mosquitoes (Aedes species), midges	Rodents, lagomorphs and ruminants	Insect bite.
	(Culicoides) and ticks (tete group only).	(acute infection).	
Phlebovirus	Many genera of mosquitoes and	Sheep and other domestic animals;	Insect bite and also through contact with
	Phlebotomus species.	Uukuniemi group viruses infect birds	infected tissues. No human infections
		(acute infections).	with Uukuniemi group viruses have
			been reported.
Nairovirus	Ixodid ticks, Hyalomma species.	Herbivores and lagomorphs (acute	Insect bite and contact with infected
		infection).	tissues. Human to human transmission
			is believed to occur.
Hantavirus	None.	Rodents (persistent infection).	Aerosolized rodent excreta.
Tospovirus	Thrips.	Over 360 plant species belonging to 50	
		families (eg.tomatoes).	

Table 1.3. Host range and transmission of the Bunyaviridae (adapted from Gonzalez-Scarano & Nathanson, 1990 and Calisher, 1991).

molecular basis of the persistent, non-cytopathic infection of the vector species remains to be elucidated.

1.1.4 Disease and geographical distribution

Members of the Bunyaviridae have been isolated from every continent except Antartica, however serotypes tend to have a localised distribution (Bishop & Shope, 1979). A number of viruses belonging to the Bunyaviridae are serious human and veterinary pathogens, for example Crimean-Congo haemorrhagic fever virus (Nairovirus), Rift Valley fever virus (Phlebovirus), the California encephalitis viruses (Bunyavirus) and a number of hantaviruses. Hantaan and related viruses cause a severe haemorrhagic disease (haemorrhagic fever with renal syndrome) with significant mortality throughout Asia, and especially China (Elliott, 1990). Crimean-Congo haemorrhagic fever virus is the etiologic agent of another haemorrhagic fever which results in 10% to 20% mortality over a wide geographic area including Sub-Saharan Africa, eastern Europe, the Middle East and areas in central Asia (Gonzalez-Scarano et al., 1991). An average of 75 cases of California encephalitis are reported annually in the United States; over 80% of patients recover fully after the acute disease, but evidence of residual central nervous system (CNS) infection is apparent in some patients. 10% of affected children develop epilepsy, but fatality is rare (only 0.3%; Kappus et al., 1983). Rift Valley fever virus (RVFV) causes an acute disease in humans with a convalescense period which may be prolonged (Eddy & Peters, 1980). Serious complications do occur but are rare (Laughlin et al., 1979). RVFV causes a more serious disease in domestic animals, sheep are more susceptible than cattle; pregnant ewes normally abort and over 90% mortality amongst infected lambs has been reported (Daubrey & Hudson, 1931; Meegan & Shope, 1981). Outbreaks of fetal abnormalities in Texan sheep have been associated with Cache Valley bunyavirus (Edwards et al., 1989; Chung et al., 1991). Other members of the Bunyaviridae, including Bunyamwera virus itself, cause transient fevers which although self-limiting in nature are important in terms of man-hours lost from work.

1.1.5 Invertebrate infection

Infection of invertebrates can take place via horizontal (oral and venereal transmission) or vertical (transovarial and transtadial transmission) mechanisms.

1.1.5.1 Oral infection

Electron microscopic studies have been used to trace the spread of California serogroup viruses through *Aedes* mosquitoes following a blood meal (Murphy *et al.*,1975). Ingestion of virus is followed by infection of the epithelial cells which line the mosquito mid-gut. An eclipse phase follows during which no infectivity can be detected. Virus first appears in the haemocoel just outside the basal lamina of the mid-gut. From here the virus has access to all body tissues, many of which are permissive. Fluorescent antibody studies

of Aedes triseriatus mosquitoes orally infected with La Crosse virus have demonstrated large amounts of viral antigen in the heart, the neural ganglia, fat body, ovaries and salivary glands (Beaty & Thompson, 1978). The salivary glands are one of the last tissues to become infected, and only a limited number of cells are antigen positive. The infected cells discharge large numbers of virions into the lumen of the gland, from where they can be injected into the vertebrate host during feeding (Murphy *et al.*, 1975). The interval from feeding until the mosquito is infectious, termed the extrinsic incubation period, is 7 to 14 days for La Crosse virus (Tesh & Beaty, 1983). The shorter the extrinsic incubation period, the greater is the likelihood of viral transmission. Typically, vectors with shorter extrinsic incubation periods are considered to be vector competent (Beaty & Calisher, 1991). With most arthropod-borne viruses, the duration of the extrinsic incubation period is a function of temperature; increased temperature reduces the time required (Turell, 1989).

Failure of arboviruses to disseminate from the mid-gut of non-permissive vector species has lead to the concept of a mid-gut barrier. Accumulation of viral antigen in the mid-gut epithelium has been observed in instances when ingestion of virus has not lead to completion of the extrinsic incubation period, and there has been no dissemination of the virus throughout the mosquito (Kramer et al., 1981; Hardy et al., 1983). Intrathoracic injection of the same virus can result in a disseminated infection (Miller, 1983; Hardy et al., 1983). In support of the idea of a midgut barrier, Turell et al. (1984) demonstrated enhanced dissemination and transmission of arboviruses in mosquitoes when an infectious blood meal was contaminated with microfilarae. Microfilarae produce holes in the insect mid-gut through which virus may enter the haemocoel directly. The midgut barrier may also be breached by mechanical means; piercing the midgut with a needle has been shown to permit efficient infection of previously refractory mosquitoes (Hardy et al., 1983). Individual species of invertebrate vectors display differential susceptibility to different viruses, ranging from being totally permissive for a particular virus to completely refractory for that virus (Mitchell, 1983; Hardy, 1988). Typically, there is a threshold level of infectious virus which must be present in an ingested blood meal for midgut infection to occur and with high virus titers most arthropods will become infected. The more susceptible the vector, the greater its competence to transmit virus (Mitchell, 1983). The factors involved in the midgut barrier are not known but a lack of midgut receptors, improper proteolytic processing of viral proteins, proteolytic inactivation of the virus in the non-preferred arthropod host, or an inability of the virus to fuse with the midgut cells are potential mechanisms (Hardy et al., 1983; Turell, 1988; Beaty & Bishop, 1988).

1.1.5.2 Transovarial transmission

Infection of ovaries, including oocytes, occurs as part of the systemic infection and results in transovarial transmission to both male and female offspring (Tesh & Beaty, 1983; Watts *et al.*, 1973). Transovarial transmission has been demonstrated for most members of the California serogroup and mosquitoes infected in this way can transmit virus by bite (Watts et al., 1973; Miller et al., 1977; Tesh, 1980; Tesh & Shroyer, 1980; Le Duc et al., 1975; Le Duc, 1979; Turell et al., 1982b; Patrician et al., 1985). California serogroup viruses have been recovered from naturally infected mosquito larvae suggesting that transovarial transmission occurs in nature (Le Duc, 1979). Transovarial transmission is believed to serve as an overwintering mechanism by which virus is maintained during long periods of vector inactivity and as such is a major survival mechanism for the virus. San Angelo bunyavirus was shown to survive 3 months in dried Aedes albopictus eggs maintained at 28°C without any decrease in the infection rate of F1 offspring (Tesh, 1980), while California encephalitis bunyavirus survived storage in mosquito eggs for up to 19 months and withstood repeated freeze-thawing (Turell et al., 1982c). Stabilized infections with highly efficient transovarial transmission from generation to generation have been reported for laboratory populations of mosquitoes infected with California serogroup viruses (Tesh & Shroyer, 1980; Turell et al., 1982b). This mechanism, combined with venereal transmission of the virus, relieves much of the requirement for amplification through horizontal transmission (Turell & Le Duc, 1983). It has been proposed that virus survival rests mainly on vertical transmission (DeFoliart, 1983).

Transovarially infected mosquitoes have disseminated infections (Beaty & Thompson, 1976). Fluorescent antibody studies on infected larvae detected the highest levels of viral antigen in the alimentary tract, then the ganglia, malpighian tubules, muscles and salivary glands. Antigen was detected in identifiable organs immediately upon emergence from the egg. The same tissues remained infected through each developmental stage, including after metamorphosis. In pupae and adults, antigen was detected at high levels in the foregut, gonadal and associated tissues and in the salivary glands. Infection of gonadal tissue of both sexes provides mechanisms for further vertical transmission of the virus and for venereal spread within the mosquito population (Thompson & Beaty, 1978).

1.1.5.3 Venereal transmission

Venereal transmission of La Crosse virus has been demonstrated in *Aedes triseriatus* mosquitoes (Thompson & Beaty, 1978). Investigations with monoclonal antibodies revealed virus antigen in accessary sex gland fluid, but not within sperm (Thompson & Beaty, 1977). Direct paternal transfer of virus to progeny is therefore possible, since virus present in the spermathecal fluid could attach to and accompany sperm into the egg. Venereally infected females harbour viral antigen in ovaries, heart, gut and salivary glands amongst other organs, and were shown to be capable of transmission of La Crosse virus transovarially and by bite (Thompson & Beaty, 1978).

1.1.5.4 Transtadial transmission

A further mechanism of virus transmission within the vector population involves vertical transmission of the virus through the different stages of arthropod development following an initial horizontal infection of larvae. The initial infection might occur by ingestion of portions of infected dead larvae, ecdysed exoskeletons, or simply by contact with the virus shed from infected larvae. This mode of transmission has been demonstrated with California encephalitis bunyavirus in *Aedes dorsalis* and *Aedes melanimon* mosquitoes as well as with the flavivirus, St. Louis encephalitis virus (Collins, 1962; Turell *et al.*, 1982c).

Transtadially infected mosquitoes can transmit virus to vertebrates by bite. Recently Booth *et al.* (1991) examined tissue tropisms of Dugbe nairovirus in the tick vector *Amblyomma variegatum* by performing immunohistochemical analysis, *in situ* hybridization and infectivity assays on dissected tissue. They determined that the primary site of transtadial persistence of Dugbe virus was the hemocytes and proposed that these motile cells were important in the transmission of persistent virus infection from one cell or organ to another.

1.1.5.5 Effect of infection on the invertebrate host

Arbovirus replication in arthropods is characteristically apathogenic and no major untoward effects have been documented. However, studies carried out on experimentally infected mosquitoes have brought to light subtle effects of bunyavirus infection on the behaviour and development of mosquitoes. Grimstad *et al.* (1980) found that La Crosse virus infected *Aedes triseriatus* mosquitoes were less successful at taking a blood meal than were uninfected siblings. *Aedes albopictus* females infected transovarially with San Angelo virus produced 12% fewer viable eggs than did uninfected controls (Tesh, 1980) and transovarially infected larvae have been shown to take considerably longer to develop than their uninfected counterparts (Tesh, 1980; Turell *et al.*, 1982b). In one study mosquitoes infected with California serogroup viruses, either orally or by intrathoracic injection, were shown to have enhanced sensitivity to carbon dioxide; this sensitivity was inhibited by antiserum to California encephalitis virus (Turell & Hardy, 1980). Mosquitoes infected transovarially with California serogroup viruses do not develop carbon dioxide sensitivity (Turell & Hardy, 1980; Turell *et al.*, 1982a).

1.1.6 Evolution of the Bunyaviridae

Evolution of members of the Bunyaviridae is believed to occur by two major mechanisms, genetic drift and RNA segment reassortment. Until recently, evidence was weighted towards the view that the Bunyaviridae exhibit the genomic plasticity characteristic of RNA genomes (Domingo & Holland, 1988). Bishop and Shope (1979) calculated the spontaneous mutation rate of snowshoe hare virus to be 1 to 2% and, more recently,

Hewlett *et al.* (1992) have reported major sequence differences between two snowshoe hare virus isolates collected from mosquitoes in different areas of Canada in 1975 and 1976. Oligonucleotide fingerprint analysis of 30 field isolates of La Crosse virus yielded 30 non-identical fingerprints (El Said *et al.*, 1979; Klimas *et al.*, 1981), and Tesh & Gubler (1975) demonstrated the presence of small-plaque variants in populations of transovarially passaged La Crosse virus, indicative of a heterogeneous population. In addition, variation in the La Crosse virus genome in response to selective pressure has been demonstrated by the revertion of a monoclonal antibody selected variant, V22, which exhibited decreased oral infectivity for *Aedes triseriatus* mosquitoes amongst other altered biological properties, upon three oral passages in mosquitoes (Gonzalez-Scarano *et al.*, 1983; Sundin *et al.*, 1987).

Genetic drift occurs via the accumulation of point mutations and, less frequently, deletions, duplications and inversions in the genome. Oligonucleotide fingerprinting studies carried out by Bilsel et al. (1988) and Baldridge et al. (1989), suggested that the frequency of such events in persistently infected arthropods may be low. Bilsel et al. (1988) reported that Toscana phlebovirus underwent no detectable genomic change upon serial transovarial transmission through 12 generations of an experimentally infected laboratory colony of its natural sandfly host, Phlebotomus pernicious. Later, Baldridge et al. (1989) demonstrated that the La Crosse virus genome underwent no detectable change during transovarial transmission through two generations of a laboratory colony of Aedes triseriatus mosquitoes, nor during transfer between the insect and vertebrate host. Furthermore, the rate of evolution of the alphavirus Ross River virus during a 10-month outbreak of epidemic polyarthritis in a non-immune island population of humans was determined to be low by direct genomic sequencing of 1.6kb of the virus genome; only one nucleotide substitution was found in the virus isolates examined during the course of the epidemic (Burness et al., 1988). Taken together these observations suggest that arboviruses may exhibit a slower rate of evolution relative to non-arthropod-borne viruses. The molecular constraints imposed on arboviruses by their disparate hosts may act as a strong selective force for genomic stability by limiting the spectrum of viable and competitive variants in a cycling arbovirus population. If the laboratory observations of genomic stability of La Crosse and Toscana viruses during transovarial transmission reflect the situation in nature; and if it is true that certain arboviruses, including La Crosse virus, are principally maintained in nature within the invertebrate host (by transovarial and venereal transmission; Tesh & Shroyer 1980; Turell 1988), then genomic stability during transovarial transmission would confer a low rate of evolution upon these viruses. The genetic variability seen amongst field isolates of La Crosse virus (El Said et al., 1979; Klimas et al., 1981) and snowshoe hare virus (Hewlett et al., 1992) might arise partly as a result of reproductively isolated mosquito populations, each maintaining a population of virus which may differ slightly from that maintained in the other mosquito populations. Conservation of dominant variants in a given arbovirus population is reputed to be aided by physical and biological limitations imposed by small blood meal volumes, small injected saliva volumes and low virus titres during transmission between vertebrate and invertebrate hosts (Burness *et al.*, 1988; Yuill, 1983).

The evolutionary potential of the Bunyaviridae is enhanced by RNA segment reassortment, which has been documented experimentally both *in vitro* and *in vivo*, and in nature (Bishop & Beaty, 1986; Bishop *et al.*, 1987). Since only viruses from a single serogroup will reassort (Bishop, 1985; Pringle *et al.*, 1984) individual serogroups may potentially constitute divergently evolving virus pools. High frequency reassortment between mutant La Crosse viruses or mutant La Crosse and snowshoe hare viruses has been demonstrated in dually infected *Aedes triseriatus* mosquitoes following ingestion of mixtures of these viruses (Beaty *et al.*, 1985). The second infection must occur within one or two days of the initial infected blood meal (Beaty *et al.*, 1985). Interference to superinfection is apparent by three days post infection, once replication of the virus in the midgut has produced sufficient quantities of virus and virus antigen (Sundin & Beaty, 1988). Many mosquitoes exhibit interrupted feeding; if a mosquito is disturbed during engorgement it may finish feeding on an alternate host. Thus superinfection within a time frame which would preclude interference is possible and in the instance of dual infection with viruses from the same serogroup, reassortment could occur.

Transmission of reassortant viruses from mosquitoes to vertebrates was demonstrated by Beaty *et al.* (1985). They isolated reassortant viruses from mice fed upon by dually infected mosquitoes. More recently, Chandler *et al.* (1990) reported high frequency reassortment in *Aedes triseriatus* mosquitoes dually infected with temperaturesensitive variants of La Crosse and snowshoe hare bunyaviruses. Viral replication and genome segment reassortment in the ovaries of the infected mosquitoes were enhanced following ingestion of a blood meal; coinciding with high ovarian metabolical activity. Transovarial transmission of both parental and reassortant viruses was demonstrated; ten percent of infected progeny mosquitoes were infected with reassortant viruses and of these 50% were demonstrated to be able to transmit reassortant viruses to mice.

1.2 Molecular biology of the Bunyaviridae

1.2.1 The virion

Bunyaviridae particles appear spherical or pleomorphic when viewed by electron microscopy. They range in size from 80 to 120 nm in diameter and are usually surrounded by a fringe of glycoprotein spikes (Pettersson *et al.*, 1971; Murphy *et al.*, 1973; Smith & Pifat, 1982; Hung *et al.*, 1983; Pettersson & von Bonsdorff, 1987). Subtle differences have been observed in the surface morphology of different viruses which correlate with their genus (Martin *et al.*, 1985). Cryoelectron microscopy of vitrified-hydrated La Crosse virions has shown these virions to be uniformly spherical, varying in diameter between 75

to 115nm. They are surrounded by a lipid bilayer 4nm thick, in which are embedded 10nm long spike proteins (Talmon *et al.*, 1987). Other enveloped viruses, for example Sendai virus, have been found to be spherical under the neutral pH and 4°C conditions prevailing during cryoelectron microscopy. It is possible therefore, that Bunyaviridae virions are not pleomorphic as suggested by images of negative stained particles, but that the observed pleomorphism is a function of the desiccation which occurs during the specimen preparation required for this technique (Hewlett & Chiu, 1991).

The virus particles consist of four structural proteins: two internal proteins, the L (transcriptase component) and the N (nucleocapsid) proteins and two external glycoproteins, termed G1 and G2, which are inserted into the viral membrane. By convention, the larger molecular weight glycoprotein species is designated G1. Foulke *et al.* (1981) detected three glycoproteins in Hazara nairovirus virions suggesting that nairoviruses may be exceptions to this pattern. The Bunyaviridae do not encode an internal matrix protein. Direct interaction of the internal nucleocapsids with the membrane, or with the cytoplasmic domain of the inserted glycoproteins, may stabilize the virion structure (Pettersson & von Bonsdorff, 1987; Talmon *et al.*, 1987).

The three genome segments of members of the Bunyaviridae, designated small (S), medium (M) and large (L) are present in virions as circular, helical nucleocapsids (Calisher, 1991). In the virion these structures consist of a single segment of negative or ambisense viral RNA which is intimately associated with numerous molecules of the viral N protein and minor amounts of the viral L protein (Obijeski et al., 1976a). Initial studies based on the chemical composition of La Crosse virions and the relative proportions of viral proteins present in La Crosse virions as judged by SDS-PAGE, estimated that each nucleocapsid contained one N protein for every 6 nucleotides of viral RNA (Obijeski et al., 1976a). The RNA within nucleocapsids of members of the Bunyaviridae is inaccessible to reasonable concentrations of nucleases. Pettersson et al. (1971) reported that 1µg/ml of RNase at 37°C did not solubilize the RNA in Uukuniemi virus nucleocapsids and Hacker et al. (1989) demonstrated that La Crosse virus nucleocapsids were resistant to concentrations of RNase A or micrococcal nuclease far in excess of the concentration required to completely digest viral mRNA. The RNA within Bunyavirus nucleocapsids does not however appear to be as tightly complexed as the RNA within, for example, paramyxovirus nucleocapsids. The addition of oligonucleotides complementary to genomic or antigenomic La Crosse virus RNA to infected cell extracts, rendered up to 50% of the nucleocapsid RNA sensitive to RNase H (Bellocq & Kolakofsky, 1987; Kolakofsky et al., 1987); paramyxovirus nucleocapsids were insensitive to this treatment (Kolakofsky & Hacker, 1991).

The precise structural arrangement of bunyavirus nucleocapsids is unknown. In order to explain the ability of the RNA within nucleocapsids to base pair with oligonucleotides it is necessary to assume that the N protein in the vicinity of the base pairing is displaced from the RNA. The accessibility of the RNA to nucleases after the addition of oligonucleotides suggests that there are exposed regions of RNA. A nucleosome-like structure with the RNA arranged on the outside of the protein, or tobacco mosaic virus-like nucleocapsids with the RNA lying within a groove created by the assembled N have been proposed as possibilities (Kolakofsky & Hacker, 1991).

Each genome segment possesses complementary 3' and 5' termini with free 5' (pppA) (Gentsch et al., 1977a; Obijeski et al., 1976b; Pettersson et al., 1977) and 3' ends (hydroxyl) ends (Obijeski et al., 1976b). The terminal sequences are highly conserved (Clerx-van Haaster & Bishop, 1980; Obijeski et al., 1980; Clerx-van Haaster et al., 1982a, 1982b; Schmaljohn & Dalrymple, 1983; de Haan et al., 1989b) and are genus specific (see Table 1.2). Circular and panhandle forms of naked viral RNAs have been observed by electron microscopy (Bouloy, 1973/1974; Samso et al., 1976; Hewlett et al., 1977; Pardigon et al., 1982) and are thought to be a consequence of base pairing of the complementary ends of the RNAs. It is postulated that the complementary sequences may provide signals for recognition by the virus encoded polymerase or are involved in packaging of the viral genome. The terminal 11 bases of Bunyamwera virus L, M and S RNAs are conserved and complementary except for positions 9 and -9, with the following 20 or so bases showing segment specific complementarity (Elliott, 1990). The mismatch at positions 9 and -9 has been found in all bunyavirus RNAs sequenced so far, with the exception of the terminal sequences of the M and S segments of sandfly fever group phleboviruses, the M segments of uukuniemi group phleboviruses and in the hantavirus M and S segments. However, sequencing of the Hantaan virus M segment by Yoo & Kang (1987) revealed a mismatch between positions 12 and -12 in the terminal 18 complementary nucleotides of this RNA (Elliott, 1990). Unmatched and mismatched bases within double stranded regions of RNA are important for recognition by proteins (Wickens & Dahlberg, 1987; Raju & Kolakofsky, 1989). RNAs of positive- and negative-polarity would have significantly different structures due to G-U in genomic RNA being an "allowed" basepair, while C-A in the complementary RNA is not. If a protein (e.g. the N protein) could distinguish these structures this may be important in transcription and the selective packaging of genomic RNA into virions. A similar situation exists with influenza virus where Hsu et al. (1987) have shown that the negative-sense genomic RNA segments of influenza virus are circular. They suggested that the terminal basepaired structures were cis-acting regulatory elements for transcription and possibly the packaging of the influenza virus genome.

Nucleocapsids of members of the *Bunyavirus* genus and the uukuniemi serogroup of the *Phlebovirus* genus have been observed under the electron microscope as circular structures (Pettersson & von Bonsdorff, 1975; Samso *et al.*, 1975; Obijeski *et al.*, 1976b). Psoralen cross-linking studies carried out on La Crosse virus nucleocapsids derived from virions and from infected cells confirmed that base pair interactions occur between the complementary ends of the viral RNA within nucleocapsids (Raju & Kolakofsky, 1989). The observation that viral genomic (-) and full-length complementary (+) viral RNAs are encapsidated, whereas viral mRNAs, which have 5' extensions and are shorter at their 3' ends, are usually not found in nucleocapsids, led to speculation that the conserved terminal sequences of the genome-length viral RNAs might harbour the encapsidation signal for the N protein (Raju & Kolakofsky, 1986). Subsequently, Raju and Kolakofsky (1987b) discovered minor amounts of encapsidated mRNA-like transcripts at late times post infection in La Crosse virus-infected cells along with some full-length (+) molecules with 5' extensions which were not encapsidated. The authors concluded that the nucleocapsid recognition sequence was at the 5' end of the RNA and had to be in the correct context for initiation of encapsidation.

1.2.2 Coding assignments

The trisegmented single-stranded RNA genome of the Bunyaviridae makes up 1% to 2% of the mass of the virion (Obijeski *et al.*, 1976b). The sizes of the RNA segments characteristic of each genus, as determined by sequencing of cloned cDNAs, or, where sequencing data are not available, as estimated by their electrophoretic migration in polyacrylamide or agarose gels, are shown in Table 1.2. The pattern of RNA segment size is distinctive for bunyaviruses, nairoviruses and TSWV, while the RNA profiles of the hantaviruses and phleboviruses are more similar.

For some viruses the mapping of viral proteins to genome segments has been achieved by genetic or biochemical methods (Bishop, 1985). Nucleotide sequence studies, along with N-terminal protein sequence data, have confirmed these results and revealed the coding strategies employed by individual genome segments (see Section 1.3.3.1). The L RNA segment encodes the L protein, the putative virion transcriptase, as verified for Bunyamwera virus by Jin and Elliott (1991). The M RNA segment encodes the two virion glycoproteins, G1 and G2, in the form of a putative precursor polypeptide which is thought to be processed by co-translational cleavage. Bunyavirus and phlebovirus M RNA segments also encode a non-structural protein (NSm) as part of the putative precursor polyprotein. The S RNA segment encodes the nucleocapsid protein (N), and in the case of bunyaviruses, phleboviruses and TSWV a non-structural protein (NSs). For bunyaviruses the N and NSs proteins are translated from overlapping reading frames (ORFs) in a single subgenomic mRNA species. All Bunyaviridae L and M segment RNAs with the exception of the MRNA segment of INS tospovirus, which has an ambisense MRNA segment (Law et al., 1992), are of negative-polarity, as are the S RNA segments of bunyaviruses, hantaviruses and nairoviruses. The infecting genome segments must be transcribed into positive-sense mRNA prior to protein production. The S segments of phleboviruses and TSWV exhibit an ambisense coding strategy. For these viruses the N and NSs protein ORFs are in opposite orientations in the genomic S RNA segment. The N protein is

translated from a viral complementary-sense subgenomic mRNA whereas the NSs protein is translated from a viral-sense subgenomic mRNA. This expression strategy has an intrinsic requirement for a round of genome replication to provide the full-length viralcomplementary template from which the NSs mRNA is transcribed; this may provide a degree of temporal control in the expression of the ambisense S segment specified proteins during infection.

1.2.3 Genetics

RNA segment reassortment occurs between closely related members of the Bunyaviridae. However, cross-genus and cross-serotype reassortment has not been reported and some viruses, even from within the same serogroup, appear to be incompatible with each other (Gentsch & Bishop, 1976; Gentsch et al., 1977b, 1979; Ozden & Hannoun, 1978,1980; Iroegbu & Pringle, 1981; Rozhon et al., 1981; Pringle & Iroegbu, 1982; Elliott et al., 1984; Pringle et al., 1984; Janssen et al., 1986). Further restrictions on reassortment are evident in the co-segregation of the L and S RNA segments of Batai, Bunyamwera and Maguari bunyaviruses in heterologous crosses. This linkage was broken when heterologous reassortants were used as parental viruses (Elliott et al., 1984; Pringle et al., 1984). The alleviation of co-segregation in heterologous crosses suggests that there is no inherent incompatibility between the L, M and S RNAs of these viruses. Restriction may therefore be mediated at the gene product level, possibly involving interaction of the core proteins (N and L) with the RNA template, or proteinprotein interactions between heterologous core proteins. This phenomenon has not been observed in heterologous crosses between California serogroup viruses (Bishop et al., 1981; Gentsch et al., 1977b), possibly due to the closer serological relationship which exists between La Crosse virus and snowshoe hare virus compared to between the Bunyamwera group viruses studied.

Genome segment reassortment has been demonstrated experimentally in dually infected mosquitoes (Beaty *et al.*, 1981, 1982, 1985; Shope *et al.*, 1981) and analysis of field isolates suggests that reassortment can also occur in nature (Klimas *et al.*, 1981; Ushijima *et al.*, 1981). The construction of reassortant viruses provides a powerful strategy for delineating the role of the gene products of individual viral RNA segments in determining the biological character of members of the Bunyaviridae. Laboratory studies with La Crosse, snowshoe hare and La Crosse-snowshoe hare reassortant viruses identified the middle RNA segment of La Crosse virus as the major determinant of infectivity for *Aedes triseriatus* mosquitoes, dissemination within the mosquito and transmission to mice (Beaty *et al.*, 1982; Beaty & Bishop, 1988). Recent studies with reassortant La Crosse-snowshoe hare and La Crosse-Tahyna viruses demonstrated the existence of distinct sites within the bunyavirus M RNA segment carrying genetic determinants for four different biological markers: subcutaneous and intracerebral mouse virulence and oral and intrathoracic infection of mosquitoes (Gonzalez-Scarano et al., 1988).

1.3 Replication of the Bunyaviridae

1.3.1 Attachment and entry

In common with other enveloped viruses, it is the glycoproteins which mediate the attachment of Bunyaviridae virions to host cells. Obijeski *et al.* (1976a) demonstrated a 10^5 -fold reduction in infectivity when La Crosse virus particles were rendered "spikeless" by proteolytic enzyme treatment. Experiments using monoclonal and polyclonal antibodies to block infection or haemagglutinating activity have suggested that G1 is more actively involved in binding to vertebrate host cells than is G2 (Kingsford & Hill, 1981; Gonzalez-Scarano *et al.*, 1982; Grady *et al.*, 1983; Kingsford *et al.*, 1983). In addition, treatment of La Crosse virions with bromelain or pronase, which degrade portions of G1 but leave G2 unchanged, rendered the virus completely non-infectious (Kingsford & Hill, 1981). In contrast, neutralizing and haemagglutination inhibition sites have been found on both the G1 and G2 proteins of Hantaan (Arikawa *et al.*, 1989; Dantas *et al.*, 1986) and Punta Toro viruses (Pifat *et al.*, 1988) suggesting that both proteins may be required for attachment.

Efficient transmission, amplification and dissemination of arboviruses requires replication in both vertebrate and invertebrate hosts. As a result virions are exposed to two significantly different environments. Treatment of La Crosse virus with proteolytic enzymes such as those found in the mosquito midgut, eg. trypsin and pronase, increased virus affinity for mosquito cells (Ludwig et al., 1989). These enzymes cleave the major envelope glycoprotein (G1) exposing the second glycoprotein (G2) which remains intact. Processing of La Crosse virus glycoproteins in the mosquito midgut therefore may be necessary to expose attachment proteins on the virion surface before attachment to and infection of midgut cells can occur. In line with the earlier work by Kingsford & Hill (1981), such enzyme treatment was shown to concurrently cause a significant decrease in viral attachment to Vero cells (Ludwig et al., 1989). More recently, binding studies have been carried out using affinity purified La Crosse virus glycoproteins, labelled with [35S]methionine, to determine which glycoprotein is responsible for attachment to vertebrate and mosquito cell lines, as well as to freshly harvested mosquito midguts. Competition binding studies with radiolabelled glycoproteins suggested that La Crosse virus utilises a receptor mediated attachment mechanism for both vertebrate and invertebrate cells, G1 being important for vertebrate cell attachment and G2 for attachment to mosquito midgut cells (Ludwig et al., 1991b).

Fusion of infected cells at acidic pH values has been reported for viruses in the Bunyaviridae (Gonzalez-Scarano *et al.*, 1984, 1985; Gonzalez-Scarano, 1985; Arikawa *et al.*, 1985). pH-dependent fusion is generally believed to relate to early events in the infection process, particularly the translocation of RNA and proteins into the cell

cytoplasm. Whether one or both of the glycoproteins are required for the fusion process has yet to be elucidated for most members of the family. Work carried out with the V22 mutant of La Crosse virus which exhibits a defective fusion function suggested that G1 mediated fusion (Gonzalez-Scarano et al., 1985). This was supported by results which demonstrated that G1 of La Crosse virus underwent a conformational change at the pH of activation of the fusion function, resulting in both an alteration in the cleavage pattern produced by amino acid specific proteases and in a change in its antigenicity as defined by altered binding of monoclonal antibodies (Gonzalez-Scarano, 1985). In two well studied systems, the Orthomyxoviridae and the Paramyxoviridae, activation of the fusion function is accompanied by a conformational change in the fusion protein (Scheid & Choppin, 1977; Hsu et al., 1981; Skehel et al., 1982). In addition, anti-G1 monoclonal antibodies have been shown to inhibit fusion (Gonzalez-Scarano et al., 1987). However, Pobjecky et al. (1989) reported that liposomes containing G1 were unable to induce fusion, whereas virus particles which had been protease treated so that only intact G2 was present at the virion surface were able to fuse liposomes. The authors suggested that a large hydrophobic domain in G2 is responsible for fusion and that G1 is concerned with interactions with the cell surface receptor rather than with the lipid bilayer directly.

Recently, Ludwig et al. (1991a) produced and characterised the first neutralizing monoclonal antibodies specific for the G2 glycoprotein of La Crosse virus. They also described three neutralizing monoclonal antibodies which bound both to G1 and G2. Amongst these was monoclonal antibody 807-22 which had been used previously by Gonzalez-Scarano et al. (1985) to select the fusion-defective La Crosse variant V22. In the course of their work, Ludwig et al. (1991a) found that monoclonal antibody 807-22 recognised both G1 and G2 of La Crosse virus, raising the possibility that the epitope recognised by this monoclonal antibody on G2 was at least part of the fusion domain previously assigned to G1 (Gonzalez-Scarano et al., 1985). Additionally, Ludwig et al. (1991a) demonstrated that the G2 specific monoclonal antibodies had high neutralizing titres when assayed in mosquito cells but limited ability to neutralize virus in mammalian cells. The G1/G2 directed monoclonal antibodies neutralized virus infectivity both in vertebrate and invertebrate cells. These results support the hypothesis that G2 is involved in the interaction of virus with mosquito cells and suggest that G1 and G2 may share a common structural epitope relevant to their role as attachment proteins in vertebrate and mosquito cells.

Using electroncryomicroscopy, Wang *et al.* (1991) demonstrated that low pH conditions were sufficient to induce membrane fusion events between La Crosse virions. The ability of La Crosse virions to form aggregates at low pH lead these authors to propose that La Crosse virus may be capable of fusing to cell membranes without a requirement for a specific receptor.

1.3.2 RNA synthesis

1.3.2.1 Transcription

Transcriptase activity has been detected in detergent disrupted preparations of Lumbo (Bouloy & Hannoun, 1976), La Crosse (Patterson *et al.*, 1984), Germiston (Gerbaud *et al.*, 1987), Uukuniemi (Ranki & Pettersson, 1975) and Hantaan (Schmaljohn & Dalrymple, 1983) viruses.

(i) Initiation

The mRNAs of both bunyaviruses and phleboviruses are known to possess 5' terminal extensions, 12 to 18 nucleotides long, that are heterogeneous in sequence and are not templated from virion RNA (Bishop *et al.*, 1983; Patterson & Kolakofsky, 1984; Eshita *et al.*, 1985; Collett, 1986; Bouloy *et al.*, 1990; Simons & Pettersson, 1991). Patterson *et al.* (1984) demonstrated that oligonucleotides such as $(A)_nG$, cap analogues such as $m^7GpppAm$ and alfalfa mosaic virus (ALMV) RNA 4, stimulated the transcriptase activity associated with La Crosse virions. Primer extension studies of the polymerase products confirmed that these oligonucleotides were acting as primers (Patterson *et al.*, 1984). This was taken as indirect evidence that the non-templated nucleotides observed at the 5' ends of Bunyaviridae mRNAs were methylated, capped fragments of RNA parasitised from host cell mRNAs, analogous to those used to prime influenza virus mRNA synthesis (Krug *et al.*, 1987). More recently, Hacker *et al.*(1990) confirmed that La Crosse virions which are capped by selecting these mRNAs with anti-cap antibodies. Patterson *et al.* (1984) also detected a methylated cap-dependent endonuclease activity in La Crosse virions which specifically cleaved ALMV RNA 4, approximately 14 nucleotides from its 5' end.

Influenza virus primary transcription takes place in the host cell nucleus due to a requirement for host cell mRNAs synthesised *de novo* as a source of primers for mRNA initiation (Ishihama & Nagatak, 1988). The available evidence suggests that La Crosse virus transcription is quite different in this respect. Rossier *et al.* (1986) pulse-labelled La Crosse virus-infected cells at various times post infection and examined both cytoplasmic and nuclear fractions for labelled La Crosse virus mRNAs. The viral mRNAs were found exclusively in the cell cytoplasm. In addition, unlike influenza virus, La Crosse virus transcription is resistant to drugs such as actinomycin D, which inhibit DNA dependent RNA synthesis by inhibiting host RNA polmerases such as polymerase II (Obijeski & Murphy, 1977). It appears therefore that La Crosse virus acquires its primers from a stable pool of host mRNAs in the cytoplasm rather than from newly synthesised nuclear transcripts, and consequently La Crosse virus mRNA synthesis has no requirement for ongoing host RNA synthesis (Rossier *et al.*, 1986).

Bouloy et al. (1990) characterised the 5' extensions of a number of Germiston bunyavirus mRNAs. cDNA copies of single mRNA molecules were cloned into pBR327 and sequenced. Sequencing revealed that the primers have a high G and C content and invariably possess a C or a U residue adjacent to the viral sequence with U preferred. No two primers were of identical sequence. In several cases the adenosine present at position +1 of the viral sequence was missing. This may be an artifact due to the cloning step and therefore it is difficult to ascertain whether mRNA molecules lacking the ultimate adenosine residue of the viral sequence exist in infected cells. However, similar observations have been reported for influenza virus (Beaton & Krug, 1981), snowshoe hare virus (Bishop *et al.*, 1983) and more recently Uukuniemi virus mRNAs (Simons & Pettersson, 1991).

The initiation of viral mRNA synthesis in cells infected by members of the Bunyaviridae may therefore proceed as follows. The viral polymerase, possibly attached to the 3' end of the genomic RNA, or alternatively free in the cell cytoplasm, binds to the cap group of a host mRNA. The viral exonuclease activity may then cleave the bound host mRNA 10 to 18 nucleotides downstream from the cap structure, exposing a 3' hydroxyl group on the capped fragment. The 3' end of the parasitised RNA may then be aligned at the 3' end of the template RNA and used as a primer by the virion transcriptase to proceed with mRNA synthesis. The first base incorporated by the polymerase would be expected to correspond to the penultimate nucleotide of the genomic template (Kolakofsky & Hacker, 1991).

Hacker *et al.* (1990) reported the existence of a heterogeneous population of 'antimRNAs' in La Crosse virus-infected cells. These subgenomic RNAs were transcribed from antigenomic RNA and possessed non-templated sequences at their 5' end. The ratio of anti-mRNAs to mRNAs in infected cells appeared to be similar to that observed with ambisense mRNAs.

(ii) Elongation

There are conflicting data on the requirement for ongoing host protein synthesis during primary transcription of bunyavirus genomes. A translational requirement has been reported for the primary transcription of a number of bunyaviruses in BHK cells including Bunyamwera (Abraham & Pattnaik, 1983); Akabane (Pattnaik & Abraham, 1983); La Crosse (Raju & Kolakofsky, 1986;1987a) and Germiston (Gerbaud *et al.*, 1987) viruses. In contrast, snowshoe hare bunyavirus has been reported to have no translational requirement for primary transcription in BHK cells (Vezza *et al.*, 1979; Eshita *et al.*, 1985). However, Kolakofsky and Hacker (1991) claimed that in the BHK cell line routinely used in their laboratory, snowshoe hare virus exhibited the same translational requirements as La Crosse virus. Another report from that laboratory claimed that there was only a very slight translational requirement for La Crosse virus mRNA synthesis in *Aedes albopictus* C6/36 cells (Raju *et al.*, 1989). It appears therefore that the translational requirement may be cell-type dependent and may even vary between different lines of BHK cells (Raju *et al.*, 1989).

Further information has been obtained from *in vitro* studies. La Crosse virions were found to produce only incomplete transcripts in an *in vitro* transcription assay unless rabbit reticulocyte lysate was added to the system. In this coupled transcription-translation
system, drugs which inhibited protein synthesis also inhibited full-length mRNA synthesis and resulted in the reappearance of incomplete transcripts (Bellocq et al., 1987). These results suggested that the translational requirement observed *in vitro* and in most cells *in* vivo was not for mRNA initiation, but to prevent premature termination of transcription. The replacement of guanosine with inosine in the nascent mRNA, which weakens RNA-RNA interactions, substituted for the translational requirement (Bellocq & Kolakofsky, 1987). This supported the idea that concurrent translation of the nascent mRNA was required to prevent interactions between the nascent chain and its RNA template which would otherwise lead to premature termination of transcription. Conversely, replacement of uridines in the nascent RNA with bromouridine was found to enhance premature termination of mRNA synthesis (Kolakofsky & Hacker, 1991). Raju et al. (1989) proposed that the translational requirement observed in BHK cells was due to a host factor present in BHK cells but absent from C6/36 cells. The addition of uninfected BHK cell extracts to an in vitro transcription system containing La Crosse virions and La Crosse virus nucleocapsids derived from infected mosquito cells resulted in an increased efficiency of initiation of mRNA synthesis and created a translational requirement for full-length mRNA synthesis. In vitro transcription of mosquito cell-derived La Crosse virus RNA in the absence of concurrent translation, or any additional factors, yielded full-length mRNAs (Raju et al., 1989). Also, the addition of uninfected mosquito cell extracts to an in vitro transcription reaction containing BHK cell-derived La Crosse nucleocapsid RNA failed to relieve the requirement for concurrent translation. Pelleting BHK cell-derived La Crosse virus nucleocapsids through a sucrose cushion reduced the requirement for concurrent translation in vitro. It may be that this procedure removes the putative host component.

(iii) Termination

X

The 3' ends of mRNAs of several members of the Bunyaviridae have been determined. The S segment mRNAs of La Crosse, snowshoe hare, Akabane and Germiston bunyaviruses and Uukuniemi virus have been shown to be truncated at their 3' terminus by approximately 100 nucleotides as compared to virion RNA (vRNA; Cash *et al.*, 1979; Pattnaik & Abraham, 1983; Bouloy *et al.*, 1984; Patterson & Kolakofsky, 1984; Pettersson *et al.*, 1985). Similarly, the M segment mRNAs of snowshoe hare and Germiston bunyaviruses and the phlebovirus Rift Valley fever virus terminate 60, 80 and 112 nucleotides prior to the 3' end of the genomic RNA template, respectively (Eshita *et al.*, 1985; Collett, 1986; Bouloy *et al.*, 1990). Bunyavirus, phlebovirus and uukuvirus mRNAs are probably not polyadenylated, as judged by their inability to bind to oligo(dT)-cellulose (Ulmanen *et al.*, 1981; Abraham & Pattnaik, 1983; Pattnaik & Abraham, 1983; Bouloy *et al.*, 1987), but Cash *et al.* (1979) did claim to select a proportion of snowshoe hare bunyavirus mRNA molecules by this method.

The template sequences corresponding to the termination sites of six bunyavirus mRNAs, Rift Valley fever phlebovirus M mRNAs and Uukuniemi virus S mRNAs have been mapped by nuclease protection assays. The La Crosse virus S (Patterson & Kolakofsky, 1984) and L (Hacker *et al.*, 1990), snowshoe hare S (Eshita *et al.*, 1985) and Germiston M mRNAs (Bouloy *et al.*, 1990) all terminate near the sequence 3' GUUUUU 5', while Germiston S mRNA (Bouloy *et al.*, 1990) ends at a similar template sequence 3' GUUUGU 5'. The snowshoe hare (Eshita *et al.*, 1985) and Rift Valley fever virus (Collett, 1986) M mRNAs, however, end at the sequence 3' ACCCC 5'. While these sequences may signal transcription termination, this has not been proven. Indeed, the Urich template sequences corresponding to the 3' ends of most of the above mentioned mRNAs seem unlikely to be the only termination signal recognised by the transcriptase, as similar U-rich sequences are present upstream of the termination site.

Stable secondary structures, such as hairpins, may be candidates for an additional signal, as has been proposed for Punta Toro phlebovirus ambisense S mRNAs (Emery & Bishop, 1987). These authors mapped the 3' termini of Punta Toro N and NSs mRNAs to a region of the viral S RNA between residues 977 and 1017, by Northern analyses using a series of [³²P]-labelled synthetic oligonucleotide probes. Computer analysis revealed a potential hairpin structure between nucleotides 866-1092 of the S RNA, and transcription of the N and NSs mRNAs appeared to terminate at either side of the top of this hairpin. The intergenic region of the S RNA of Uukuniemi virus has also been shown to contain a short palindromic sequence with the potential to form an A/U rich hairpin structure (Simons et al., 1990), although the existence of secondary structure in this region and any role it might play in transcription termination remain to be determined. Simons and Pettersson (1991) reported that the 3' ends of the Uukuniemi virus S segment mRNAs overlap each other by approximately 100 nucleotides across this palindromic sequence and no other transciption termination signal was identified. Computer analysis of the 3' noncoding region of Germiston and La Crosse virus mRNAs failed to predict stable hairpin structures (Bouloy et al., 1990), and no evidence has been found for potential secondary structure in the intergenic regions of the ambisense S segments of Rift Valley fever, Toscana, and sandfly fever Sicilian phleboviruses (Giorgi et al., 1991). Thus, the precise mechanisms involved in mRNA transcription termination amongst the Bunyaviridae remain to be elucidated.

Little is known about the primary transcriptional properties of viruses in the *Hantavirus*, and *Nairovirus* genera or in the Uukuniemi serogroup of the recently expanded *Phlebovirus* genus. A manganese-dependent, virion-associated polymerase was found to produce Uukuniemi virus transcripts capable of hybridizing to viral RNA templates, although the transcripts appeared to be'small (Ranki & Pettersson, 1975). A polymerase with similar divalent cation requirements was found in association with Hantaan virions (Schmaljohn & Dalrymple, 1983).

1.3.2.2 Genome replication

For members of the Bunyaviridae the initiation of genome replication requires a switch from the synthesis of subgenomic mRNA to the synthesis of full-length, complementary template RNA, and subsequently full-length virion RNA. The processes involved in making that switch have not been defined for any member of the Bunyaviridae. Genome replication and secondary transcription are prevented by translational inhibitors such as cycloheximide, indicating that continuous protein synthesis is required for replication of the genome. It is not known which protein(s) are required, but they are likely to be of viral origin. The virus non-structural proteins (NSs for example) may be potential candidates. The onset of genome replication requires that the mRNA transcription termination signals are suppressed and that the addition of host derived primers to newly synthesised viral-complementary RNAs is prevented. With vesicular stomatitis virus (VSV) the encapsidation of nascent viral RNAs by the viral nucleocapsid protein (N) seems to serve as an antitermination signal, thus allowing full-length genome synthesis (Arnheiter *et al.*, 1985). A role for VSV NS in controlling the availability of the N protein has also been suggested (Howard *et al.*, 1987).

1.3.3 Protein expression strategies and protein processing

Most of the current knowledge regarding the expression of Bunyaviridae proteins has been deduced from nucleotide sequence data. The L, M and S RNA segments of individual viruses employ distinct strategies for protein expression. In addition the gene organisation and expression strategies of each RNA segment, particularly the M and S RNA segments, varies between genera.

1.3.3.1 Expression strategies

(i) L segment proteins

To date, five L gene sequences are available, two from the *Hantavirus* genus and one from each of the *Bunyavirus*, *Phlebovirus* and *Tospovirus* genera (Elliott, 1989a; Schmaljohn, 1990; de Haan *et al.*, 1991; Elliott *et al.*,1992). Each sequence contains a large open reading frame (ORF) in the viral complementary-sense RNA which codes for a primary translation product (L protein) of approximately 2150 amino acids in length in the case of the hantaviruses, and 2238, 2104 and 2875 amino acids in length for Bunyamwera, Uukuniemi and tomato spotted wilt virus, respectively. Additional small ORFs have been identified in the viral-sense L RNAs of both Bunyamwera virus and Hantaan virus which potentially encode proteins of 129 and 221 amino acids in length. The significance of these small open reading frames during a natural infection is questionable since there is no homology between the two putative small polypeptides. In addition, neither the proteins themselves nor their mRNAs have been detected in infected cells and no corresponding ORF has been identified in the viral-sense L RNA of TSWV (de Haan *et al.*, 1991) or Seoul 80-39 hantavirus (Antic *et al.*, 1991b). The viral L proteins are thought to be translated from a mRNA species just shorter than the full-length genomic RNA segment (de Haan *et al.*, 1991), but no information is available regarding processing of the L protein.

(ii) M segment proteins

With the exception of the *Nairovirus* genus, at least one complete M segment sequence is available for each genus of the Bunyaviridae. In general, a single large ORF encoding a polyprotein precursor to the two envelope glycoproteins (and in the case of bunyaviruses and phleboviruses the NSm protein) is contained within the viral complementary-sense M RNA. However, the M segment RNA of INSV has an ambisense gene arrangement (Law *et al.*, 1992).

The M RNA segments of La Crosse (Grady *et al.*, 1987), snowshoe hare (Eshita & Bishop, 1984), Bunyamwera (Lees *et al.*, 1986) and Germiston (Pardigon *et al.*, 1988) bunyaviruses have been sequenced, revealing ORFs which code for polypeptides ranging from 1433 to 1441 amino acids in length. By sequencing the amino and carboxy termini of the G1 and G2 glycoproteins of snowshoe hare virus, and analysing the reactivity of these glycoproteins with antibodies raised against specific peptides, Fazakerley *et al.* (1988) concluded that the order of the snowshoe hare virus M segment encoded proteins was NH₂-G2-NSm-G1-COOH.

Sequencing studies on a number of hantavirus M segments (Schmaljohn *et al.*, 1987; Yoo & Kang, 1987; Giebel *et al.*, 1989; Arikawa *et al.*, 1990; Antic *et al.*, 1991a) revealed that the hantavirus M segment polyproteins are approximately 1135 to 1148 amino acids in length. By direct sequencing of the amino termini of the individual glycoproteins the order of proteins within the precursor polyprotein encoded by Hantaan virus was determined to be NH₂-G1-G2-COOH (Schmaljohn *et al.*, 1987). The G2 protein of Hantaan virus was expressed efficiently by baculovirus and vaccinia virus recombinants which contained cDNA representing the Hantaan virus G2 sequence plus 32 upstream amino acids. The only possible initiation codon was a natural, inframe ATG located 17 amino acids upstream from the amino terminus of G2 (Schmaljohn *et al.*, 1987). Whether this independent initiation of translation of the G2 glycoprotein occurs in authentic viral infections is not known.

M segment sequences have been published for Punta Toro phlebovirus (Ihara *et al.*, 1985b), for three isolates of Rift Valley fever virus (Collett *et al.*, 1985; Takehara *et al.*, 1989) and for Uukuniemi virus (Ronnholm & Pettersson, 1987). The M segments of Punta Toro and Rift Valley fever viruses encode a nonstructural protein in their precursor polyprotein but Uukuniemi virus does not. The order of proteins in the precursor polypeptides has been determined to be NH₂-NSm-G1-G2-COOH for Punta Toro virus, NH₂-NSm-G2-G1-COOH for Rift Valley fever virus and NH₂-G1-G2-COOH for Uukuniemi virus. The apparent difference in the gene order of Punta Toro and Rift Valley fever virus

M segments is misleading and arises due to the convention of defining G1 as the larger of the two glycoproteins and the similarity in the sizes of the two phlebovirus glycoproteins.

Suzich et al. (1990) investigated the expression strategy of the Rift Valley fever virus (RVFV) M segment by performing site-directed mutagenesis on potential translation start sites in RVFV M segment-vaccinia virus recombinants and then expressing these recombinants. The RVFV M segment ORF contains four AUG codons between the AUG initiating the ORF and the amino terminus of mature G2 (Collett et al., 1985). Suzich et al. (1990) revealed that RVFV M segment employs a relatively complex expression strategy which involves the use of at least two AUG codons in the same ORF to produce its full complement of gene products. In addition, the biosynthetic pathways for the two major viral glycoproteins, G2 and G1, appeared to be distinct from each other with a significant proportion of G1 synthesised independently of the other M gene products. Since Northern analysis of RNA isolated from cells infected with the recombinant viruses revealed only a single species of M segment sized mRNA which was found to possess preglycoprotein G1 and G2 coding sequences, the mechanism for independent production of G1 must occur at the post-transcriptional level. Conceivably, a proportion of RVFV G1 may be produced by internal initiation of translation downstream of the preglycoprotein region. Initiation of translation from the first ATG of the ORF appeared to predetermine the utilization of an Nlinked glycosylation site within the preglycoprotein region and also precluded cleavage at the preglycoprotein-G2 junction (Kakach et al., 1989). Conversely, initiation from the second ATG prevented glycosylation at this site but allowed proteolytic cleavage at the amino terminus of G2. Proteolytic cleavage at this junction is known to be independent of glycosylation (Kakach et al., 1989) which suggests that the 37 amino acids between ATG codons 1 and 2 can influence proteolytic processing and glycosylation further along the polyprotein. Thus, a multisite translation initiation strategy may serve as a mechanism for controlling post-translational protein modifications.

In the absence of a complete nairovirus M segment sequence, information regarding the synthesis and processing of nairovirus glycoproteins is limited. The M RNA of nairoviruses generally encodes two glycoproteins, although Hazara nairovirus encodes three glycoproteins (Foulke *et al.*, 1981). Recently the first tospovirus M RNA sequence became available; revealing that the M RNA segment of INSV encodes a polyprotein precursor to glycoproteins G1 and G2 and a non-structural protein (NSm) in an ambisense arrangement (Law *et al.*, 1992). By *in vitro* transcription and translation the order of G1 and G2 in the precursor polyprotein was determined to be NH₂-G2-G1-COOH.

(iii) S segment proteins

The S RNA segments of hantaviruses and nairoviruses code only for the nucleocapsid protein (N), while those of the bunyaviruses, phleboviruses and tospoviruses code for two proteins, the N protein and a non-structural protein (NSs). Bunyavirus S segment proteins are translated from a single bicistronic mRNA species, while viruses of

the *Phlebovirus* and *Tospovirus* genera translate their S segment proteins from two monocistronic mRNAs of opposite polarity, which are transcribed from an ambisense S RNA (see Section 1.2.2). An additional ORF has been identified in the S RNA segments of Germiston and Maguari bunyaviruses. This is situated downstream of and in the same reading frame as the NSs protein (Gerbaud *et al.*, 1987; Elliott & McGregor, 1989). The product of this third ORF has been translated *in vitro* from RNA transcribed from cloned Maguari virus cDNA (Elliott & McGregor, 1989), but neither of these putative proteins have been detected in infected cells (Elliott & McGregor, 1989; Bouloy *et al.*, 1990).

Initiation of translation at the second AUG in the bicistronic bunyavirus S mRNA is thought to occur via "leaky scanning" of ribosomes along the mRNA (Kozak 1986a, 1986b). It has been proposed that a proportion of ribosomes scanning the 5' noncoding region can bypass the first AUG initiation codon and continue scanning until they find the AUG which initiates the second ORF. Kozak (1986b) defined the optimum consensus sequence for the initiation of translation as 5' (A/G)CCAUGG 3'; with a purine (preferentially an A residue) at position -3 and a G residue at position +4 (where the A of the AUG is designated +1) being the most important determinants. The leaky scanning model proposes that when the 5' proximal AUG in a bicistronic mRNA is in a suboptimal sequence context, ribosomal preinitiation complexes can bypass the first AUG and initiate at a downstream AUG which is in a more favourable context.

The NSs ORFs in Maguari and La Crosse bunyaviruses commence with tandem AUG codons (Cabradilla *et al.*, 1983; Elliott and McGregor, 1989). Which of these AUG codons is used for translation initiation remains to be determined.

	- 3	+1	+ 4
Optimal	ACC	AUG	G
N	UCA	AUG	Α
NSs (First)	UCA	AUG	Α
NSs (Second)	AUG	AUG	U

Figure 1.1 Sequence contexts of the AUG codons initiating the N and NSs ORFs in the Maguari virus S RNA; including the optimum sequence context for initiation of translation as defined by Kozak (1986b). Adapted from Elliott and McGregor (1989).

Elliott and McGregor (1989) compared the sequence contexts of the AUG codons initiating the N and NSs ORFs in Maguari virus (see Figure 1.1) and found that the sequence context of the second AUG at the initiation site of NSs was the most favorable. The first AUG at the start of the NSs protein and the AUG initiating the N protein ORF were in identical sequence contexts.

The S RNA segment of Hantaan virus possesses a single ORF in the viral complementary-sense RNA which encodes an N protein 429 amino acids in length (twice as large as the N proteins of the other genera; Schmaljohn *et al.*, 1986b). The other known Hantaan S segment sequences: Sapporo rat virus (Arikawa *et al.*, 1990); Nephropathia Epidemica virus (NEV) strain Hallnas B1 (Stohwasser *et al.*, 1990) and Prospect Hill virus (Parrington & Kang, 1990) display a similar gene organisation. Additional small ORFs have been detected in the viral complementary-sense RNA of Hantaan virus, NEV Hallnas strain B1 and Prospect Hill virus and also in the virion-sense RNA of Sapporo rat virus. However, the products of these ORFs have never been detected in infected cells.

The only sequence information available for the *Nairovirus* genus is the complete sequence of the S RNA of Dugbe virus (Ward *et al.*, 1990). Dugbe virus S segment RNA contains one ORF in the viral complementary-sense RNA which encodes the N protein, and another small ORF in the virion RNA which may not be expressed. Five phlebovirus S segment sequences have been determined; Punta Toro (Ihara *et al.*, 1984), Sandfly fever Sicilian (Marriott *et al.*, 1989); Uukuniemi virus (Simons *et al.*, 1990) and Toscana and Rift Valley fever virus (Giorgi *et al.*, 1991). The N proteins of these viruses exhibit low sequence homology and the NSs proteins are even less well conserved.

De Haan et al. (1990) sequenced the S RNA segment of TSWV. They found two ORFs, one in the genomic-sense RNA, which encoded a 465 amino acid protein and another in the virus complementary-sense RNA, which coded for a smaller protein of 259 amino acids in length. The smaller of these two proteins was identified as the virus nucleocapsid protein by immunoprecipitation with antisera raised against purified TSWV nucleocapsids. Kormelink et al. (1991) made recombinant baculoviruses that expressed high levels of the larger of these two proteins and demonstrated by western blot analysis that this protein was absent from purified TSWV virions. This is the largest non-structural protein so far identified for any member of the Bunyaviridae. Conceivably, the TSWV S segment encoded non-structural protein (NSs) may contain an extra domain which is important for infection processes specific to plants. These authors observed a correlation between the levels of TSWV NSs protein in infected plants and the severity of disease symptoms. The nucleocapsid protein ORF of a second tospovirus isolate, INSV has now been sequenced (Law et al., 1991). This ORF encodes a protein of 262 amino acids in length which is 67% homologous to TSWV at the amino acid level. (N protein)

1.3.3.2 The kinetics of viral protein synthesis in infected cells

Viral protein synthesis in cell cultures infected with members of the Bunyaviridae has been followed most commonly by pulse-labelling infected cells with [³⁵S]-methionine and fractionating radiolabelled cell extracts on SDS-PAGE gels. The viral N protein, being the most abundant viral protein in infected cells, is reliably detected by this procedure. The other viral proteins can be obscured if host cell protein synthesis is not inhibited by the

infection, but this problem can be overcome by immunoprecipitation of the viral proteins prior to electrophoresis (Watret & Elliott, 1985a; Ulmanen *et al.*, 1981). The G2 glycoprotein is not reliably detected on SDS-PAGE gels and this has recently been shown to be due to the heat-labile nature of this protein (R.M. Elliott, personal communication).

The viral glycoproteins are not detected in infected cells until after the onset of synthesis of the other viral proteins. This has been demonstrated with, for example, Bunyamwera virus (Pennington *et al.*, 1977), St. Abb's Head virus (Uukuniemi serogroup of the phleboviruses; Watret & Elliott, 1985a) and Uukuniemi virus (Ulmanen *et al.*, 1981). The rate of viral protein synthesis in cells infected with members of the Bunyaviridae depends to a degree on the moi used, but in general viral messages are transcribed and translated rapidly. High multiplicity infections usually result in maximum levels of viral protein synthesis being observed 3 to 5 h pi, as was observed in BSC-1 and BHK cell cultures infected with 40 pfu/cell of Bunyamwera and 50 pfu/cell La Crosse bunyaviruses respectively (Pennington *et al.*, 1977; Madoff & Lenard, 1982). Parker *et al.* (1984) detected RVFV N protein as early as 2 h pi in Vero cells which had been infected at a high moi.

Watret et al. (1985) determined the protein synthesis profiles for members of the Bunyavirus, Phlebovirus and Nairovirus genera in XTC-2 cells infected at a moi of 1 to 5 pfu/cell. They determined that Bunyamwera virus N protein was synthesised at high levels in XTC-2 cells by 4 h pi and continued to be translated at high levels up to the last time point tested (24h pi); host protein synthesis had been completely abolished by this time. Rates of viral protein synthesis were slower for nairoviruses and phleboviruses; neither Clo Mor nor Dugbe nairoviruses significantly inhibited host cell protein synthesis in XTC-2 cells, and the N protein of Clo Mor virus was not detected in infected XTC-2 cells until 8h pi. Low multiplicity studies (0.1 to 2 pfu/cell) with Dugbe nairovirus in BSC-1 cells also revealed slow kinetics of viral protein synthesis, with maximum levels of viral protein synthesis occurring between 24 and 31 h pi, and no significant inhibition of host protein synthesis (Cash, 1985). Sicilian sandfly fever phlebovirus failed to shut off host cell protein synthesis in XTC-2 cells, and viral protein synthesis was relatively slow (Watret et al., 1985). The exact onset of N protein synthesis was not determined but significant levels of N protein were still being synthesised at 48 h pi. In a parallel infection of BSC-1 cells by Sicilian sandfly fever virus, significant levels of N protein were detected at 24 and 48 h pi and host protein synthesis was shut off. St. Abb's Head virus (Uukuvirus serogroup of the *phlebovirus* genus) did not shut off host cell protein synthesis in XTC-2 cells and also displayed slower kinetics of protein synthesis than bunyaviruses; only low levels of N protein were detected at 4 h pi but levels increased gradually throughout the period studied (up to 16 h pi). Similar slow rates of viral protein synthesis, without host cell protein synthesis shut off, have been observed during Uukuniemi virus infection of chick embryo cells (Ulmanen *et al.*, 1981) and during hantavirus infection of Vero cells (Schmaljohn & Dalrymple, 1984).

1.3.3.3 Protein processing

The L and S segment encoded proteins of the Bunyaviridae do not require posttranslational processing, as they are translated directly from individual mRNA species. Post-translational modification has only been reported for the NSs protein of Rift Valley fever virus which appears to be phosphorylated (Struthers *et al.*, 1984). In contrast, the M segment proteins are both processed and modified. With the exception of INSV, which has an ambisense M RNA segment (Law *et al.*, 1992), each Bunyaviridae M segment RNA encodes a single large ORF in the viral complementary sense from which a mRNA coding for a polyprotein consisting of the viral glycoproteins (and in the case of the bunyaviruses and phleboviruses the NSm protein) is transcribed.

Little is known regarding the processing of this polyprotein to yield the individual viral proteins. However, alanine specific proteases have been implicated in the cleavage events generating the amino termini of the G1 and G2 proteins of Rift Valley fever and Punta Toro phleboviruses (Ihara et al., 1985b), and also in the cleavage events generating the carboxy terminus of G2 and possibly the amino terminus of G1 from bunyavirus precursor polypeptides (Elliott et al., 1991). Precursor polyproteins have only been detected in a few instances, and only in nairovirus infected cells (Clerx & Bishop, 1981; Cash, 1985; Watret & Elliott, 1985b). The difficulty encountered in detecting the precursor polyproteins in infected cells suggests that processing occurs cotranslationally (Pennington et al., 1977; Lees et al., 1986; Schmaljohn et al., 1986a; 1990; Bouloy, 1991). Attempts to translate bunyavirus M segment specific mRNA in vitro have been unsuccessful (Abraham & Pattnaik, 1983; Elliott, 1985). However Ulmanen et al. (1981) produced a 110kDa protein by in vitro translation of Uukuniemi virus M segment specific mRNA, which was cleaved in the presence of microsomal membranes to generate G1 and G2. Independent initiation of translation of the downstream glycoprotein has been observed in vitro with recombinant baculovirus and vaccinia virus constructs expressing M gene sequences from Hantaan and Rift Valley fever viruses (Schmaljohn et al., 1989; Suzich et al., 1990). However, it is not known if independent initiation of translation of viral glycoproteins occurs in vivo.

All of the Bunyaviridae envelope proteins examined to date possess N-linked oligosaccharides. Bunyavirus glycoproteins contain relatively few potential glycosylation sites, with only four potential glycosylation sites evident in the M segment polyprotein encoded by Bunyamwera virus (Vorndam & Trent, 1979; Cash *et al.*, 1980; Lees *et al.*, 1986; Pardigon *et al.*, 1988). Studies with glycosylation inhibitors have confirmed that at least some of these sites are utilised in the mature glycoproteins (Cash *et al.*, 1980). The oligosaccharide side chains on bunyavirus glycoproteins consist mainly of endoglycosidase

H (endo-H) resistant complex glycans (Madoff & Lenard, 1982; Pesonen *et al.*, 1982), although some endo-H sensitive high mannose and some small endo-H-resistant intermediate-type glycans are found. The bunyavirus glycoproteins also contain covalently attached fatty acids (Madoff & Lenard, 1982).

Five N-linked glycosylation sites were identified in the glycoprotein coding sequences of Rift Valley fever and Punta Toro phleboviruses, however, the number of sites utilized in the mature glycoproteins was not defined (Collett et al., 1985; Ihara et al., 1985b). Uukuniemi virus glycoprotein coding sequences contain eight potential N-linked glycosylation sites (Ronnholm & Pettersson, 1987). Pesonen et al. (1982) determined that the G2 protein of Uukuniemi virus possesses mainly high mannose glycans, whereas G1 contains both complex and novel intermediate-type oligosaccharides. Hantaan virus glycoproteins possess a total of seven potential glycosylation sites, five were identified in G1 and two in G2. In this case the glycans were found to be mainly of the high mannose type (Elliot et al., 1984; Schmaljohn et al., 1986a; Schmaljohn et al., 1987). Giebel et al. (1989) sequenced the M segment of Nephropathia Epidemica virus, strain Hallnas B1 (NEV) and found that all but one of the potential glycosylation sites observed in Hantaan virus were found in identical positions in NEV and two potential tyrosine sulphation sites (which are physiologically relevant only to proteins passing through, or located in the Golgi complex) were conserved between the two viruses. Recently, Law et al. (1992) identified nine putative N-linked glycosylation sites in the glycoprotein coding sequences of INSV.

All M segment gene products appear from their cDNA sequences to have a high cysteine content (5 - 7%). In related viruses the positions of these cysteine residues are conserved (Fazakerley *et al.*, 1988; Ihara *et al.*, 1985b; Ronnholm & Pettersson, 1987; Schmaljohn *et al.*, 1988). These findings suggest that there may be extensive disulphide bridge formation within the glycoproteins and the positions of the cysteines may be crucial for the correct folding of the protein. The molecular structure of the glycoprotein complexes at the surface of virions has not been determined. However, despite the abundance of cysteine residues in both G1 and G2, it appears from work with La Crosse virus that these proteins are not linked to each other by disulphide bonds (Gonzalez-Scarano & Nathanson, 1990).

1.3.3.4 Intracellular transport and targetting of the viral glycoproteins

The hydropathy profiles of the polyproteins of La Crosse, snowshoe hare, Bunyamwera and Germiston bunyaviruses are similar. Each begins with a hydrophobic region which is believed to act as a signal peptide for translocation of the polypeptide. Another hydrophobic domain towards the centre of the G2 molecule may serve as a transmembrane domain, while another signal-like sequence lies between NSm and G1. A hydrophobic domain at the carboxy terminus of G1 may function as a membrane anchor sequence. The hydropathy profile of the Hantaan virus M segment polyprotein predicts a short hydrophobic leader sequence prior to the amino terminus of mature G1 and another short hydrophobic region preceding the amino terminus of G2 (Schmaljohn *et al.*, 1987). Giebel *et al.* (1989) sequenced the M segment of Nephropathia Epidemica nairovirus, strain Hallnas B1 (NEV). Comparison of the polyprotein coding sequences of Hantaan and NEV revealed a number of conserved features. The positions of the hydrophobic leader sequences preceeding Hantaan virus G1 and G2 glycoproteins were conserved in NEV, as was the amino acid sequence of a short hydrophilic domain within which cleavage of the hydrophobic leader is thought to occur to yield mature G2.

Chen *et al.* (1991) found that the G2 glycoprotein of Punta Toro phlebovirus did not contain an integral signal for Golgi retention but depended upon an association with G1. Subsequently, Chen and Compans (1991) reported that G1/G2 heterodimers were formed in Punta Toro virus infected cells within three minutes of the onset of viral protein synthesis.

Persson and Pettersson (1991) analysed the heterodimerization and intracellular transport of Uukuniemi virus G1 and G2 glycoproteins. They determined that G1 acquired its mature form within 10 minutes, having completed its disulphide bond formation, while completion of disulphide bond formation in G2 required up to 60 minutes. Pulse-chase experiments in the presence of cycloheximide indicated that the efficient transport of G2 to the Golgi required continuous synthesis of G1. It appeared that G1 and G2 were dependent upon each other for transport to the Golgi from the ER.

Ruusala *et al.* (1992) investigated the intracellular transport and targeting of the Hantaan virus glycoproteins by expressing the glycoproteins either individually or together from recombinant vaccinia viruses in Hela cells. Using immunofluorescence and subcellular fractionation these authors determined that G2 and G1 of Hantaan virus had to interact with each other in the endoplasmic reticulum (ER) in order to be transported to the Golgi complex. When either of the glycoproteins were expressed in the absence of the other they were retained in the ER. Studies with monoclonal antibodies suggested that G1 expressed alone did not acquire its correct conformation.

1.3.4 Assembly and virus release

The Bunyaviridae characteristically mature at Golgi membranes (Lyons & Heyduk, 1973; Murphy *et al.*, 1973; Kuismanen *et al.*, 1982, 1984; Smith & Pifat, 1982), although, Rift Valley fever phlebovirus has been observed to mature at the cell surface of infected rat hepatocytes (Anderson & Smith, 1987). Virus assembly appears to occur via an interaction between viral ribonucleoprotein (RNP) structures, which accumulate on the cytoplasmic side of Golgi membranes, and viral envelope proteins, which are displayed on the luminal side. Such interactions have only been observed across regions of Golgi membrane directly involved in the budding process. RNPs have not been found associated

with Golgi membranes devoid of viral glycoproteins, suggesting that transmembranal recognition between the viral glycoproteins and the N protein is a prerequisite for budding (Smith & Pifat, 1982). Accumulation of the viral glycoproteins in the Golgi complex causes progressive vacuolization (Kuismanen *et al.*, 1984), and this effect has been shown to be mediated by the glycoproteins alone (Gahmberg *et al.*, 1986a). The morphologically altered Golgi complex remains functionally active (Gahmberg *et al.*, 1986b). Expression of hantavirus and phlebovirus glycoproteins in recombinant vaccinia vectors has shown that Golgi targeting is an intrinsic property of the glycoproteins (Matsuoka *et al.*, 1988; Pensiero *et al.*, 1988). However, the signal responsible for Golgi localization has yet to be determined. After the virus particles bud into the Golgi cisternae, it is thought that they are released into the cell cytoplasm in small vesicles, in a manner analagous to secretory granules (Broadwell & Oliver, 1981; Rothman, 1981; Smith & Pifat, 1982). The virus particles are then believed to be released from the cell by exocytosis.

1.3.5 Effects of viral replication on host cells

The cytopathic effects observed in cell cultures infected with members of the Bunyaviridae vary widely, depending both on the virus and the type of host cell studied. With the exception of the *Hantavirus* genus all members of the Bunyaviridae are capable of alternate replication in vertebrate and invertebrate hosts. Generally the Bunyaviridae are cytolytic for their vertebrate hosts but cause little or no pathogenicity in their invertebrate hosts (James & Millican, 1986; Lyons & Heyduk, 1973).

Variation has been observed in the ability of members of the Bunyaviridae to induce host cell protein synthesis shut off in infected vertebrate cells. Bunyaviruses and phleboviruses have been shown to cause a progressive decrease in host protein synthesis during the course of infection in vertebrate cells. In Bunyamwera virus infected BSC-1 cells host cell protein synthesis declined rapidly between 5 and 7 h pi. In the absense of detectable host protein synthesis these cells continued to support virus replication. The synthesis of virus structural proteins was detected up to 22 h pi and infectious virus production continued up to 30 h pi (Pennington et al., 1977). Similar results were obtained with La Crosse virus infected BHK-21 cells (Madoff & Lenard, 1982). Reduced host cell protein synthesis was observed in Rift Valley fever virus infected Vero cells and became gradually more pronounced between 4 to 20 h pi (Parker et al., 1984). However, Sicilian sandfly fever phlebovirus did not significantly inhibit host cell protein synthesis in XTC-2. cells, although this virus efficiently shut off BSC-1 cell protein synthesis (Watret et al., 1985). No reduction in host protein synthesis has been observed in mammalian cells infected with Uukuniemi virus or Clo Mor and Dugbe nairoviruses (Pettersson, 1974; Ulmanen, 1981; Watret et al., 1985; Cash, 1985).

Hantaviruses cause no detectable reduction in host macromolecular synthesis (Elliot et al., 1984; Schmaljohn & Dalrymple, 1984) and routinely establish persistent,

non-cytolytic infections in susceptible mammalian host cells, reflecting their nonpathogenic persistence in their natural rodent hosts (Lee *et al.*, 1981a). Interestingly, Dugbe and Hughes serogroup nairoviruses have been shown to readily establish persistent infections in mammalian cells (BSC-1 and Vero cells respectively; Gould *et al.*, 1983; Cash, 1985). In both virus-cell systems an initial acute phase of infection was superceded by a persistent infection. In the case of Dugbe virus, recovery of the cell monolayer was accompanied by a decline in virus induced cytopathic effect (cpe) and virus protein synthesis (Cash, 1985). Immunofluorescence studies on Vero cells persistently infected with Hughes serogroup nairoviruses demonstrated that 20 to 40% of these cells contained cytoplasmic viral antigen (Gould *et al.*, 1983). The arthropod-borne members of the Bunyaviridae cause no significant cytopathology in mosquito cell cultures and persistent infections are readily established (Newton *et al.*, 1981; Carvalho *et al.*, 1986; Elliott & Wilkie, 1986; Rossier *et al.*, 1988; de Haan *et al.*, 1990).

2 Persistent viral infections

Persistent viral infections can be categorized as slow, latent or chronic. Persistent infections described as slow are associated with the so-called "unconventional viruses" of which scrapie is the prototype. Following incubation periods of months to years these agents give rise to a unique set of degenerative brain diseases, the spongiform encephalopathies. These fatal diseases include scrapie in ruminants and rodents, bovine spongiform encephalopathy in cattle, and Creutzfeld-Jakob disease, Kuru and Gerstmann-Straussler syndrome in humans (Gajdusek, 1990). During a latent infection no infectious virus is present, for example herpes viruses and retroviruses persist latently by integration into the host genome (Weinberg, 1980; Roizman_A 1987; Stevens, 1989). Chronic virus infections are productive persistent infections which can arise after an acute episode or an inapparent sub-clinical infection. Two well studied examples are hepatitis B virus infection of humans (Marion & Robinson, 1983) and lymphocytic choriomeningitis virus infection that the work in this thesis is concerned and any reference to persistent infection in the text refers specifically to chronic infections.

For a virus to persist in the course of a natural infection it must accomplish two things: first it must not be overtly cytolytic (for example most arenaviruses can replicate in host cells without adverse effects; Buchmeier et al., 1980), or it must have available a means of attenuating its lytic potential (for example via restricted gene expression or by the generation of viral variants which are less cytolytic or interfere with the growth of wild-type virus; Chambers & Webster, 1987; Cattaneo et al., 1988a, 1988b); second it has to avoid detection and elimination by the host's immune system. Despite the obvious importance of the immune system in modulating persistent infections in vivo it is likely that mechanisms fundamental to the establishment of persistence are to be found in the molecular interactions between the virus and its host cell. It is reasonable therefore to expect the mechanisms involved in the establishment and maintenance of persistent infections in tissue culture to resemble those underlying persistent infections in vivo. The readiness with which Bunyamwera virus establishes a persistent infection in Aedes albopictus C6/36 cells and the ease with which the persistent state can be maintained (Elliott & Wilkie, 1986) are two factors which make this virus-cell combination a useful system with which to study the molecular mechanisms leading to persistence, in the absence of the complexities of the host's immune response and clearance mechanisms.

As the work presented here was performed solely in tissue culture this review focuses mainly on *in vitro* studies but does include relevant *in vivo* investigations. While I have not attempted to cover the vast literature on the immunology of persistent infections (reviewed by Oldstone, 1989b; 1991), I occasionally refer to immunological aspects, for instance with regard to the mechanisms by which persistent infections might cause disease, or with regard to the evolution of virus variants capable of escaping immune surveillance during the course of a persistent infection.

2.1 Involvement of persistent viral infections in human disease

There has been increasing interest in the possible involvement of persistent viral infections in a variety of human diseases. Chronic infections of the central nervous system (CNS) of man have been reported for measles virus, papovaviruses (JC and SV-40), rubella virus, cytomegalovirus, herpes simplex virus, adenovirus types 7 and 32, Russian spring-summer encephalitis (RSSE) virus, and the human retroviruses human T-cell lymphotrophic virus 1 and human immunodeficiency virus (Gajdusek, 1990).

Subacute sclerosing panencephalitis (SSPE) is a fatal, slowly progressing, inflammatory disease of the CNS. In 1967 Connolly *et al.* demonstrated that SSPE patients had extremely high measles antibody titres in their serum and cerebrospinal fluid (CSF). Subsequent immunofluorescence studies demonstrated that measles virus antigens were present in neurons and glial cells (Freeman *et al.*, 1967; Connolly, 1968). Infectious virus was obtained from these cells by co-cultivation or fusion with continuous cell lines susceptible to measles virus. In this way SSPE virus (a measles virus variant) was isolated (Horta-Barbosa *et al.*, 1969; Barbanti-Brodano *et al.*, 1970). This agent has been repeatedly obtained from both biopsy and autopsy material and from the lymph nodes of SSPE patients (Agnarsdottir, 1977).

Paget's disease is a slowly progressive, painful, deformity-inducing disease of bone. Electron microscopic studies revealed the presence of paramyxovirus nucleocapsids in abnormal osteoclasts taken from patients with Paget's disease (Mills & Singer, 1976; Rebel *et al.*, 1977; Harvey *et al.*, 1982). Both respiratory syncitial virus (RSV) and measles virus have been implicated in the etiology of this disease. Measles virus antigens were detected in bone sections, using immunofluorescence and peroxidase techniques (Rebel *et al.*, 1980; Basle *et al.*, 1985). In later experiments, Basle *et al.* (1986) demonstrated measles virus nucleic acid in both osteoclasts and osteoblasts (bone forming cells) of Pagetic bone, but not in control sections. RSV antigens were detected using fluorescent antibodies in abnormal bone osteoclasts in each of 12 patients with Paget's disease but not in osteoclasts from controls with other bone diseases (Mills *et al.*, 1981). Later, Mills *et al.* (1984) reported the simultaneous detection of both RSV and measles virus antigens in osteoclasts from the same patients. Other paramyxoviruses, simian virus 5 (SV-5) and parainfluenza virus 3 (PIV-3), have also been detected by immunofluorescence in Pagetic bone osteoclasts (Basle *et al.*, 1985).

Many viruses, including paramyxoviruses, have been proposed as etiological agents in multiple sclerosis. Measles virus, parainfluenza virus type-1 (PIV-1) and (SV-5), have all been isolated from tissues taken from multiple sclerosis patients (Russell, 1983).

Antibodies to measles virus, parainfluenza viruses 1, 2 and 3, and SV-5 have been detected in the cerebrospinal fluid of multiple sclerosis patients (Goswami *et al.*, 1987; Russell *et al.*, 1989; Vandvik & Norrby 1989), and levels of humoral antibodies against measles virus were much higher in multiple sclerosis patients than in controls (Adams & Imagawa, 1962; Albrect *et al.*, 1983). More recently, Reddy *et al.* (1989) demonstrated an association between the occurence of HTLV-1 related sequences in peripheral blood monocytes and macrophages from multiple sclerosis patients and the development of disease.

Progressive encephalitides develop in some individuals due to the persistence of certain echoviruses and polioviruses in the CNS (Wyatt, 1973; Davis *et al.*, 1977; Wilfert *et al.*,1977). Dalakas (1986) proposed that a post-poliomyelitis syndrome observed in some individuals was due to the persistence of poliovirus in these individuals. The case for the involvement of a persistent infection has been strengthened by the detection of an intrathecal immune response against poliovirus in patients with post-poliomyelitis syndrome (Sharief *et al.*, 1991). Currently, there is interest in the possible association between persistent enterovirus infection of muscle and post-viral fatigue syndrome (Archard *et al.*, 1988; Cunningham *et al.*, 1990).

2.1.1 Mechanisms by which persistent viral infections could cause disease

There are many possible mechanisms by which persistent viral infections could cause disease. Long term, low level genome replication within certain differentiated cell types or tissues may cause direct cell damage. Evidence is available which suggests that viruses can cause disease by altering the differentiated functions of a cell without affecting its vital functions (Holtzer, 1975; Oldstone, 1984; 1989a). Viral antigens exposed at the surface of cells may trigger immune system damage to the cells, while the interaction of viral proteins with "self" antigens may lead to the generation of "anti-self" responses; for example paramyxoviruses are known to break tolerance to cellular actin (Tyrrell et al., 1987). The interaction of the M protein of Newcastle disease virus (NDV) with actin has been demonstrated to induce a conformational change in these proteins and to enhance the antigenicity of actin (Giuffre et al., 1982; Tyrrell et al., 1987). It is believed that the conformational change induced in actin by interaction with the M protein is sufficient to expose a new antigenic site at the carboxyl end of the actin molecule. It is possible that antigenic mimicry, where an antigenic epitope on a viral protein induces an antibody or Tcell clone which cross reacts with an epitope on a normal host cell protein, may have a role to play. Cross-reacting monoclonal antibodies for measles virus phosphoprotein and the intermediate filaments of neuronal cells (Fu jinami et al., 1983), and measles virus fusion protein and cellular stress protein (Sheshberadoran & Norrby, 1984), have been studied. In addition, persistent shedding of viral antigens could lead to immune complex disease and infection of immunocytes could result directly in depression or hypereactivity of immune responses.

2.2 Characteristics of persistently infected cell cultures

Persistently infected cell cultures continue to shed virus after prolonged passage, although the levels of released virus may be very low. Sometimes infectious virus is not produced because mutations, host cell factors and defective interfering (DI) particles prevent virus maturation. Alternatively low levels of slowly replicating virus variants may be produced. Characteristically, persistently infected cultures harbour intracellular viral genomes, express viral antigens and are resistant to superinfection with homologous virus. The mechanisms involved in viral persistence are poorly understood. In the light of a multitude of studies carried out on a variety of virus-cell systems, persistence has been ascribed to:

(i) The evolution of viral mutants with a range of phenotypes. Temperature-sensitive and small-plaque mutants are commonly observed in association with persistent infections. Mutants defective in RNA synthesis and viruses which have lost their ability to switch off host protein synthesis are also believed to play a role in some persistent infections.

- (ii) The inherent resistance of cells to virus.
- (iii) The co-evolution of cells and virus.
- (iv) The production of interferon or antiviral activity.
- (v) The presence of DI particles.

While each of these factors is known to play a role in the establishment and maintenance of persistent viral infections, it is far from clear how these mechanisms operate, either alone or in combination with each other. The relative importance of each of the factors listed above appears to vary widely in different virus-cell systems and each will be discussed in the following sections.

2.2.1 Evolution of viral variants during persistent infections

Oligonucleotide fingerprinting and sequencing studies have revealed extreme variability amongst RNA genomes. Most natural isolates and clonal populations of RNA viruses consist of mixtures of variant genomes. Upon replication new variants constantly arise and the proportion of each of them increases or decreases in competition with other variants in the population. Thus a given RNA virus population is in a dynamic equilibrium defined by a "consensus" sequence. The composition of this consensus sequence is dependent on prevailing selective pressures (Domingo & Holland, 1988). Persistent infections might be expected to provide an excellent opportunity for the inherent high mutation rate of RNA genomes (due to the lack of proof reading activity associated with their encoded RNA polymerases) to be reflected in high rates of genome change. Genomic

* Also (Burrows, 1966).

r

sequence alterations during RNA virus persistence have been well documented. For instance, virus variants exhibiting an increased resistance to a particular species of DI RNA, relative to the original virus used to establish the persistent infection, have been observed in persistent infections established with rabies virus (Kawai & Matsumoto, 1977), VSV (Horodyski & Holland, 1980), and lymphocytic choriomeningitis virus (Jacobson & Pfau, 1980). Depolo *et al.* (1987) reported the continual, out of phase, co-evolution of DI RNAs and DI resistant variants of VSV, upon undiluted passage of this virus in BHK cells.

The introduction of a number of different variants into a cell which will support long-term replication provides an environment where intergenomic complementation could result in the propagation of even lethal mutations until further mutations suppress their lethality. Rapid changes in phenotypic properties are possible through the selection of variants present or arising in the heterogeneous population. Foot-and-mouth disease virus (FMDV), a picornavirus, shows remarkable antigenic variability. This virus causes an acute disease of domestic and wild cloven-hooved animals, and inapparent persistent infections are common in areas where the disease is enzootic (Burrows, 1966; Auge de Mello et al., 1970). The virus may persist in the oesophageal pharyngeal region of cattle and other ruminants for up to several years without causing signs of disease (Condy et al., 1985). Early studies showed that FMDV isolates from carrier animals differed from FMDV associated with acute disease in being temperature-sensitive and producing small plaques on cell monolayers (Straver & van Bekkum, 1972). More recent studies have indicated that the emergence of FMDV variants stems from the extreme genetic heterogeneity of the virus (Domingo et al., 1980; Domingo et al., 1985; Sobrino et al., 1983), and the selection or random drift of variants continuously arising during viral replication (Rowlands et al., 1983).

Gebauer and coworkers (1988) demonstrated the rapid selection of genetic and antigenic variants of FMDV during persistence in cattle. They established a carrier state experimentally and recovered virus from the oesophageal pharyngeal area of the animals up to 539 days postinfection. Their data, based on the analysis of viral capsid proteins by electrofocusing, and the electrophoretic mobility of the genomic poly(C)-rich tract of FMDV isolates, showed heterogeneity in several isolates and sequential dominance of viral subpopulations. Sequence analysis of the VP1-coding region (which codes for one of the virus structural proteins) of the parental FMDV RNA and of seven isolates from carrier cattle, revealed rates of fixation of mutations of 0.9×10^{-2} to 7.4×10^{-2} substitutions per nucleotide per year; 59% of the base changes lead to amino acid substitutions, some of which were located in a region involved in neutralization of FMDV. FMDV-neutralizing activity was detected in the oesophageal pharyngeal fluid samples, and antigenic variation amongst viral isolates from carrier cattle was demonstrated with monoclonal antibodies raised against the parental virus. Two isolates from the persistent infection displayed a 10^2 and 10^3 fold decreased neutralization with these monoclonal antibodies. Thus persistent infection of ruminants with FMDV appears to promote the rapid selection of antigenically variant foot and mouth disease viruses.

These results were supported by a subsequent study performed in cell culture in the presence of antiviral polyclonal sera (Carrillo *et al.*, 1989). Cloned FMDV was serially passaged on secondary monolayers of bovine foetal kidney cells in the presence of subneutralizing antiviral polyclonal sera (APS). After a limited number of passages the virus population displayed increased resistance to neutralization by APS, altered electrophoretic mobility of VP1, a pronounced reduction in plaque size, temperature sensitivity and decreased pathogenicity for mice. Analysis of cloned virus harvested after 23 passages indicated that the temperature-sensitive character was displayed by the smallplaque variants. The evolution towards attenuated viruses with decreased pathogenicity demonstrated that under the appropriate conditions, less virulent virus variants can prevail over those which destroy the host (Steinhauer & Holland, 1987). Persistent infections were readily established with virus variants isolated from passage 23.

In contrast, isolates from clinical cases of SSPE and acute measles virus infections appeared to be virtually identical by classical serology (Stephenson & ter Meulen, 1982). However, sequencing studies revealed a 1% difference in two overlapping cDNA clones of viral RNA extracted from the brain of an SSPE patient, while the sequenced regions of either clone differed from the corresponding region of the reference Edmonston measles virus RNA in 2.8 and 3.7% of nucleotides. These results suggested that a rapid evolution of measles virus RNA occurs during persistence in the brain and, in conjunction with earlier results from oligonucleotide fingerprint analysis, strongly suggested that the viral RNAs in SSPE consist of related, non-identical nucleotide sequences (Stephenson & ter Meulen, 1982; Cattaneo *et al.*, 1986). Variant genomes with different tropisms and altered replication cycles may arise during the persistent infection and in conjunction with the host's immune response may induce SSPE in rare individuals.

Studies on persistent virus infections of tissue culture cells have provided considerable evidence that viral mutants can modulate the lytic potential of normally cytocidal viruses (Friedman & Ramseur, 1979; Younger & Preble, 1980; Wechsler & Meissner, 1982; Mahy, 1985). The mutants most commonly observed in association with persistent infections have temperature-sensitive and/or small-plaque phenotypes. Temperature-sensitive mutants appear to compete successfully with, and displace, the wild-type viruses in cultures maintained at relatively non-permissive conditions for these mutants (Preble & Youngner, 1975). In most of the studies with arboviruses in invertebrate cell culture, temperature-sensitive and small-plaque mutants accumulated after the onset of the persistent infection, suggesting they were not involved in the establishment of the persistent state (Newton *et al.*, 1981; Verani *et al.*, 1984). Cloned small-plaque variants of Bunyamwera virus released from persistently infected *Aedes albopictus* cells induce cytopathic effects in BHK cells or persistence in mosquito cells identically to wild-type

virus (Newton *et al.*, 1981). However, studies on reovirus persistence have demonstrated the importance of temperature-sensitive variants, and it has been shown that mutations in specific viral genes are crucial for the establishment and maintenance of persistent infection in mouse L-929 cells (Ahmed & Graham, 1977; Ahmed *et al.*, 1980; Ahmed & Fields, 1981, 1982). A genetic approach was used to determine the mutations which had a crucial role in persistence and it became clear that although attenuated viruses could contain multiple defects, the viral components and genetic changes responsible for initiating or maintaining persistence were highly specific (Ahmed & Fields, 1982).

2.2.2 The role of the host cell in the establishment and maintenance of persistent viral infections

Persistent infections in tissue culture may be maintained either at the level of the individual cell or at the population level. The survival of cells following infection is the key to the former mode of persistence. In such "steady-state" infections, 100% of cells express viral antigen and the infected cultures cannot be cured easily with antiviral serum. For persistent infections maintained at the population level a minority of cells are susceptible to infection at any one time, either because a large proportion of cells in the culture is inherently resistant to the virus or because of the action of soluble inhibitors such as interferon. The spread of virus is limited so that cell death is counter-balanced by the production of new cells by cell division. A low and variable percentage of the cells are antigen-positive and such "carrier cultures" can be cured with antiviral serum (Walker, 1964; 1968; Mahy, 1985).

Igarashi *et al.* (1977) studied an *Aedes albopictus* cell line persistently infected with Sindbis alphavirus. They cloned individual cells from the persistently infected cell culture and found that about 50% of the cells released infectious virus. Recloning of the virus-positive clones yielded both infected and uninfected cultures (62% of the recloned cells released infectious virus). Immunofluorescence studies indicated that 40 to 80% of cells in the persistently infected cultures contained viral antigen, a result largely consistent with the cell cloning experiments. The existence of virus negative clones suggested that these cells had a self-curing mechanism. The argument for a self-curing mechanism was strengthened by the observation that growth of the persistently infected cultures for 24 weeks in the presence of antiviral serum (10% v/v) resulted in cultures remaining consistently virus negative through 16 subsequent serial transfers without antisera. Curing in the presence of antiviral serum suggested that extracellular virus was required to maintain the long-term infection and that the infection could not be maintained indefinitely in the culture by vertical transmission from parent cell to daughter cell.

Riedel and Brown (1977) proposed a hypothetical scheme for the maintenance of the persistent state established by Sindbis virus in *Aedes albopictus* cells, based on a number of observations. Differences in the process of virus morphogenesis in vertebrate and invertebrate cultures had previously been demonstrated by Gliedman *et al.* (1975). Using electron microscopy these authors observed membranous cytoplasmic vacuoles in Sindbis virus infected *Aedes albopictus* cells, which were not present in uninfected cells, and which contained Sindbis virions. Viral nucleocapsids at various stages of envelopment were observed in the vesicles, suggesting that envelopment takes place in these internal structures. In addition, virus release appeared to take place via fusion of these vesicles with the plasma membrane. Few free nucleocapsids were seen in the cytoplasm of infected mosquito cells and virions were rarely observed budding from the surface of these cells. The confinement of stages of viral replication to discrete areas of invertebrate cells may afford the cell protection from the toxic effects of virus infection, whereas these effects cause death in infected vertebrate cells.

In their work Riedel and Brown (1977) demonstrated that during the initial acute phase of Sindbis virus infection in *Aedes albopictus* cells, 100% of the cells released infectious virus (resulting in relatively high yields of virus) and expressed viral antigens. No cytopathic effects were observed during the acute phase of infection, yet once persistence was established only a small percentage of the cultured cells contained detectable viral antigen as monitored by immunofluorescence, and less than 1% of the cells released infectious virus as measured by an infectious centres assay. Relatively low levels of virus were released from the persistently infected cells. The culture as a whole however, was as resistant to homologous superinfecting virus as cells in the acute phase of infection. Elimination of the extracellular virus, by growing the cells in the presence of Sindbis virus antiserum, resulted in curing of the culture, such that the culture responded to infection as would an uninfected culture. It was clear therefore that the presence of extracellular virus was important to the maintenance of the persistent state.

Combining these observations it appeared that when mosquito cells were infected with Sindbis virus the cells entered a phase of maximum virus synthesis (acute infection). A large number of virus-producing vesicles were generated in the cell cytoplasm, each being, to an extent, isolated biochemically and genetically from other vesicles and from the host cell. This proposed compartmentalisation of infection was supported by observations made during two earlier studies. Renz and Brown (1976) had observed that many of the temperature-sensitive mutants of Sindbis virus which complement each other readily in vertebrate cells failed to do so in cultured mosquito cells, and Igarashi and Stollar (1976) had demonstrated that DI particles of Sindbis virus could neither be propagated, nor exert their inhibitory action, in mosquito cells. These findings were interpreted as reflecting the isolation of the replication of individual infecting virions in the cytoplasm of infected cells, resulting in an inability of normal particles to provide the necessary helper functions for defective virion synthesis in the mosquito cells. The large amounts of virus released during the first 48 hours post infection from Sindbis virus infected mosquito cells suggested that many of the internal vesicle structures were lost by this time. If the virus induced vesicles were released into the surrounding medium the cells would have had less virus synthesizing capability and contained less viral antigen; the presence of a single vesicle might not have been detectable by immunofluorescence or infectious centres assay but might have been adequate to maintain resistance to superinfection.

The progressive loss of virus-induced vesicles from the infected cells coupled with normal cell division would result in the appearance of uninfected cells in the culture; these cells would be susceptible to reinfection by virions in the medium and the cycle from the acute to the persistent phase of infection would be repeated for these particular cells. From the observations of Riedel & Brown (1977) it appeared that at any given time a very small percentage of the cells in the persistently infected Sindbis virus/*Aedes albopictus* culture were in the acute phase of infection. Using immunofluorescence these authors detected a decrease in the rate of synthesis of virus components in the persistently infected cells, which suggested that the persistently infected cells might possess some mechanism for specifically retarding virus production without interfering with normal cell function. The persistently infected cells did however support the replication of West Nile virus, a non-homologous superinfecting virus, indicating that the observed suppression of virus synthesis was not the result of a classical interferon-like agent (see Section 2.3.2).

Other morphological profiles of alphavirus growth in *Aedes albopictus* cells have been presented. Stollar *et al.* (1979) observed Sindbis virus nucleocapsids in the cytoplasm of infected *Aedes albopictus* cells and saw virions budding at the surface of these cells; suggestive of a mode of replication more akin to that observed in Sindbis virus infected vertebrate cells. Simizu and Maeda (1981) examined the morphogenesis of western equine encephalitis alphavirus in the C6/36 clone of *Aedes albopictus* cells. These authors found cells containing virus inclusions similar to those described by Gliedman *et al.* (1975) in addition to virus budding from the cell surface. The different profiles of virus development observed for alphaviruses in *Aedes albopictus* cells may be due to differences between the subclones of *Aedes albopictus* cells used (see Section 2.3). It is possible that both modes of virus development co-exist in the mosquito. The maturation pathway described by Gliedman *et al.* (1975) may be important in the establishment of inapparent persistent infections, essential for maintenance of the life-long infection of the mosquito, whereas the pathway described by Stollar *et al.* (1979) may be important for the production of large amounts of extracellular virus in tissues such as the salivary gland of the mosquito.

2.2.3 Inherent resistance of cells to virus

Vogt & Dulbecco (1958) studied the properties of a cloned HeLa cell line (S3) which exhibited an increased resistance to poliovirus type 3. Although no evidence was obtained which suggested that this culture could become persistently infected, this study revealed the existence of individual cells within the population which displayed different

degrees of resistance to the infecting virus. Repeated exposure of the standard culture to virus resulted in a cell line with increased resistance to poliovirus infection. The "resistant" culture contained a higher proportion of cells displaying resistance to poliovirus than did the original culture and was proposed to arise by the selection of pre-existing "resistant" variants in the S3 population. More recently, Kaplan *et al.* (1989) isolated and characterised HeLa cell lines blocked at different stages in the poliovirus life-cycle. They detected lower levels of expression of the poliovirus receptor on HeLa cell clones resistant to poliovirus infection.

Takemoto & Habel (1959) established a carrier-culture of HeLa cells infected with Coxsackie A9 virus which was passaged for many cell generations with consistent production of virus. Clonal analysis of the persistently infected culture revealed that only a small proportion of the cells were infected. Prolonged cultivation of the carrier-culture selected for cells with increased resistance to Coxsackie A9 virus and also generated virus with altered antigenic composition.

Mizzen *et al.* (1983) proposed that persistence of mouse hepatitis virus (coronavirus) in mouse LM cells was maintained by subsets of host cells which could resist infection and withstand cytopathic effects, i.e. cell fusion. Persistence was proposed to be a feature of the entire culture, not of individual cells, dependent upon a dynamic balance between infected and uninfected cells.

Lymphotrophic minute virus of mice (a parvovirus) establishes a persistent infection in mouse L cells which is believed to be maintained by co-evolution of cells and virus in the carrier culture (see Section 2.2.4). Spontaneous curing of the culture occurs after 20 to 30 passages (Ron *et al.*, 1984; Ron & Tal, 1985). Ron and Tal (1986) demonstrated that curing occurred via the selection of virus-resistant cells in the culture. Resistance to virus replication was shown to be due to an intracellular block, and the data presented suggested that the replication of minute virus of mice required host-coded helper functions for at least two stages of its replication cycle. These authors hypothesized that during the persistent infection, either the cell-coded helper functions were modified or their expression was reduced.

2.2.4 Co-evolution of cells and virus

Ahmed and coworkers (1981) performed a detailed study of reovirus persistence in mouse L cells. Their persistently infected culture was established by infecting a monolayer of mouse L cells with a serially passaged stock of reovirus tsC(447) (Ahmed & Graham, 1977). Cell clones free of virus (LR cells) were obtained by growing the persistently infected L cells in the presence of anti-reovirus serum for 30 days and isolating cell clones from the cured culture. Infection of the LR cells with reovirus type-3, or with reovirus isolated from the carrier culture, readily gave rise to a persistent infection. In contrast, infection of L cells with reovirus type-3 is known to result in inhibition of host cell macromolecular synthesis and complete cell destruction (Zweerink & Joklik, 1970). Infection of L cells with reovirus isolated from persistently infected cultures also resulted in a lytic infection with no surviving cells. Ahmed *et al.* (1981) argued that the mutant cells evolved during the course of persistence, rather than having been selected from a sub-population of cells, on the basis that such a sub-population of cells should always survive and become persistently infected.

The LR cells were partially resistant to infection with reovirus type-3 resulting in a low level persistent infection with inefficient virus replication. The replication of reovirus type-3 in the LR cells appeared to be restricted at an early step such as absorption, penetration or uncoating. Most of the LR cells were morphologically distinct from the parental L cells, generally appearing larger and less fibroblast-like. Preliminary observations suggested that LR cells had an altered cytoskeleton and contained a large number of lysosome-like structures in the cytoplasm. The LR cells were also resistant to vesicular stomatitis virus which is extremely cytopathic for L cells. The evolution of virus was evident in so far as virus released from later passages of the persistently infected cells grew better in LR cells than reovirus type-1 and type-2 and displayed complete dominance in co-infection experiments. By infecting mutant cells with mutant virus it was possible to construct the original carrier culture. Carrier cultures established with LR cells and wildtype reovirus produced less virus and contained fewer infected cells than the cultures created with mutant cells and mutant virus.

Co-evolution of cells and virus has also been proposed as the mechanism whereby lymphotrophic minute virus of mice (MVM(i); parvovirus) persists in L cells (Ron & Tal, 1985) and FMDV persists in BHK-21 cells (de la Torre *et al.*, 1988). In the former study, infection of L cells with MVM(i) resulted in the emergence of host-range mutant viruses. Parental MVM(i) grew well in T-lymphocyte derived cell lines but its replication in L cells was restricted. Mutant viruses which arose after infection of L cells were capable of a lytic infection which destroyed most of the initially restrictive parental L cells. However the culture was not completely lysed and a persistent infection was established. Throughout the persistent infection extensive changes occurred in both the tissue tropism of the progeny virus and in the phenotypic properties of the cells. Mutant cells were selected which were increasingly restrictive to the replication of the resident virus, but concomitant changes in the virus enabled it to replicate in a subpopulation of these cells. Infection of mutant cells with mutant virus nor infection of mutant cells with parental virus led to persistence.

De la Torre *et al.* (1988) studied the evolution of cells and virus during serial passage of a cloned BHK-21 cell line persistently infected with FMDV C-S8c1. The persistently infected cells termed C1-BHK-Rc1 (de la Torre *et al.*, 1985) became constitutively resistant to the parental FMDV C-S8c1. Treatment of persistently infected

cells from late passages of the carrier culture with ribavirin cured the cells of virus (de la Torre *et al.*, 1987) but did not restore sensitivity to FMDV C-S8c1. The resistance of C1 BHK-Rc1 cells to FMDV C-S8c1 was due to an intracellular block which resulted in a 100-fold decrease in the amount of FMDV RNA in the infected cells. Virus isolated from late-passage carrier cells (FMDV R59) partly overcame this cellular block and was also more cytolytic than the parental FMDV for BHK-21 cells. Sequencing of the VP1 gene from nine viral clones arising from C1-BHK-21 cells revealed genetic heterogeneity between these isolates.

C1-BHK-Rc1 cells had a characteristic round cell morphology, and grew faster than BHK-21 cells in liquid culture. They were less subject to contact inhibition of growth, and had an increased ability to form colonies in semi-solid agar. The persistent state was readily reconstituted with late-passage C1-BHK-Rc1 cells and FMDV C-S8c1 or FMDV R59. Clonal analysis of the cells in the C1-BHK-Rc1 carrier culture revealed a range of cell phenotypes coexisting in the cell population (de la Torre et al., 1989b). A total of 248 stable cloned cell lines were derived from 294 individual cells isolated from the original carrier culture. None of the stable cloned cell lines tested released infectious FMDV, or contained levels of viral RNA detectable by dot blot hybridization using cDNA probes. FMDV was present in the culture medium of 8 of 177 single cells isolated from pass 19 of the carrier culture, none of which gave rise to stable cell clones, suggesting that a small proportion of cells in the carrier culture (approximately 4%) were lytically infected. In fact the proportion of lytically infected cells in the persistently infected culture probably fluctuated between 4%-20% based on the above observation and the fact that 80% of cells isolated from the carrier culture gave rise to stable cell clones. At least six distinct cell phenotypes with regard to cell morphology, resistance to FMDV strain C-S8c1 and cell growth characteristics, were present in the cloned cell cultures. The increased resistance to FMDV C-S8c1 and enhanced transformation phenotype exhibited by late passages of C1-BHK-Rc1 cells (de la Torre et al., 1988) would appear therefore to be the additive effect of a range of cell responses within the population. A number of cell clones were demonstrated to retain their distinct phenotype after 20 serial passages, subcloning and further propagation. Overall the results suggested that the altered phenotypes of individual cell clones were the result of inheritable cell modifications. De la Torre et al. (1989b) suggested that the origin of this cell diversity may lie in the selection of increasingly transformed cells in the culture or may be due to DNA hypermutability which has been described in association with other systems: for example shuttle plasmid vectors during their replication in mammalian cells (Razzaque et al., 1983) and immunoglobulin gene segments (O'Brien et al., 1987). The FMDV carrier culture (C1-BHK-Rc1) appears to exist as a dynamic interaction between an evolving heterogeneous population of FMDV (Steinhauer & Holland, 1987; de la Torre et al., 1988; Domingo & Holland, 1988) and multiple cell variants. It is likely that cell heterogeneity confers a selective advantage for long-term virus

and cell survival by providing the cell population with a range of responses towards FMDV.

In later experiments de la Torre and co-workers (1989a) transformed two cell clones from pass 74 of the C1-BHK-Rc1 culture, which were highly resistant to FMDV C-S8c1, to puromycin resistance (Pur^r) and fused these cells to BHK-21 cells transformed to neomycin resistance (Neo^r). The hybrid Neo^r Pur^r cells displayed the specific resistance to FMDV C-S8c1 characteristic of the original clones. The results therefore suggested that resistance to FMDV C-S8c1 was mediated by dominant *trans*-acting cellular products.

2.2.5 Interferon

Interferon may be involved in the maintenance of a number of persistent infections of vertebrate cell lines. For example, interferon has been detected in cultures of mouse L cells persistently infected with Semliki Forest virus (Meinkoth & Kennedy, 1980); VSV (Ramseur & Friedman, 1977; Nishiyama *et al.*, 1978) and Sindbis virus (Inglot *et al.*, 1973). However, it is likely that interferon plays a subsiduary rather than a principal role in these persistent infections (Friedman & Ramseur, 1979). Interferon has never been detected in persistently infected invertebrate cell cultures, although antiviral activity, distinct from interferon in the specificity of its action, has been detected in mosquito cell cultures persistently infected with Sindbis and Semliki Forest alphaviruses (see Section 2.3.2).

2.2.6 Defective interfering particles

Defective interfering (DI) particles were first observed by von Magnus in 1947 as "incomplete" virus particles generated on passaging influenza virus serially undiluted in embryonated chicken eggs. They have subsequently been observed in association with most well studied viruses. The best characterised DI particles are those of the negativestrand RNA viruses vesicular stomatitis virus (VSV; rhabdovirus), influenza virus (orthomyxovirus) and Sendai virus (paramyxovirus). DI particles of positive-strand RNA viruses such as poliovirus and FMDV (Picornaviridae) and of reoviruses, with their double-stranded RNA genome, have also been well studied. In the case of icosahedral viruses such as the Picornaviridae, the structural similarity between standard virus and DI particles extends to overall shape and size; in most cases the defective viruses contain less total RNA than the parent virus and can be separated by density gradient centrifugation. Defective forms of rhabdoviruses such as VSV can be separated from standard virus relatively easily on the basis of size difference since the length of the viral RNA governs the overall size of these bullet-shaped viruses and defective viruses contain shorter RNAs. In more pleomorphic viruses, such as influenza and measles viruses, DI virus particles tend to be smaller than the standard particle, but the variability in size of the virion population as a whole makes it difficult to achieve clean separation (Huang & Baltimore, 1977). Because

of the relative ease of isolation of VSV DI particles most of the early work on the general properties of DI particles was carried out with this virus.

Huang and Baltimore (1970) coined the term "DI particle" having recognised the ubiquitous nature of DI viruses. They defined these entities as defective viruses which lacked an essential portion of the viral genome, required homologous parental virus as helper for replication, contained virus structural proteins and antigens; and exhibited the capacity to interfere specifically with homologous helper virus, by replicating at its expense in doubly infected cells. In the case of non-segmented viruses such as VSV defective RNAs are the basis of defective interfering virus particles. The DI particles of segmented RNA viruses such as influenza virus generally contain an increased proportion of lower molecular weight RNAs and fewer high molecular weight RNAs (DI RNAs) have been observed in influenza virus DI preparations (Nayak *et al.*, 1989).

The majority of DI RNAs originate from the polymerase genes of the parental virus. For instance, 88% of influenza virus DI RNAs originate from the three polmerase genes (Nayak *et al.*, 1989). The same progenitor gene can give rise to multiple DI RNAs of varying lengths, and the smaller DI RNAs are not always a subset of the larger DI RNAs. It is worth noting that defective RNAs which are not interfering, but which are subgenomic RNAs arising from polymerase errors, have also been recognised. Comparative sequence analysis of DI RNAs and subgenomic RNAs has not revealed any distinguishing features between the two species (Nayak *et al.*, 1989).

2.2.6.1 Generation of DI RNAs

Sequencing studies have revealed four major types of DI RNAs of negative-strand RNA viruses: internally deleted RNAs, which retain their parental 3'and 5' ends but display one or more internal deletions; stem or panhandle RNAs, where the parental 5' terminus is retained but the 3' terminus is complementary to the 5' terminus; snapback or hairpin RNAs, where the 3' half of the RNA is complementary to the 5' half; and mosaic RNAs, where the defective RNAs can possess either of the terminal arrangements detailed above, but due to numerous intermolecular or intramolecular recombinations have relatively complex sequence alterations. DI RNAs are believed to arise as a consequence of polymerase errors, as originally suggested by Huang (1977) and Leppert et al. (1977). These authors suggested that DI genomes are created by genome rearrangements or recombinations resulting from "jumping" of the viral replicase from one viral RNA template to another, or from one segment of a template to another. The RNA replicase would carry the incomplete nascent strand to the new template and then use this RNA as a primer for resumption of chain elongation. This "polymerase copy-choice" mechanism for DI RNA generation was generalised by Lazzarini et al. (1981) and Perrault (1981) to explain the generation of all known forms of DI particles, and was subsequently verified experimentally for polioviruses by Kirkegaard and Baltimore (1986) and for coronaviruses by Lai (1990). The incorporation of cellular transfer RNA into the 5' termini of a class of Sindbis virus DI particles has been observed (Schlesinger, 1988). The ease with which DI RNAs are generated via the polymerase copy-choice mechanism is dependent upon the ability of the replicase, carrying the nascent RNA strand, to jump between templates or between different sites on the same template. The efficiency of DI generation must therefore be inversely proportional to the processivity of the polymerase, that is its tendency to continue transciption on its original template (Roux *et al.*, 1991). Variant viruses which generate VSV DI RNAs with increased efficiency have been reported (Depolo & Holland, 1986). These variants might be expected to possess a decreased processivity and/or an increased ability to bind to a new template.

The Wa-182 strain of influenza virus, a single-gene reassortant containing the NS gene from the A/Aichi/2/68 (H3N2) wild type (wt) strain of influenza virus against an A/WSN/33 (H1N1) wt background, and which harbours three point mutations in its NS2 gene compared to the wild-type A/WSN/33 (H1N1), is readily converted to a DI form by a single high-multiplicity infection (Odagiri & Tobita, 1990a). In contrast, wild-type A/WSN/33 (H1N1) virus requires at least three consecutive high-multiplicity passages to generate a comparable amount of DI particles (Janda et al., 1979). This reassortant was generated by Odagiri and Tobita (1990a) who demonstrated that the three point mutations in the NS2 gene of Wa-182 were responsible for it's enhanced efficiency of DI generation. The DI RNAs of Wa-182 were formed by deletion of the PA gene of the virus. Sequence analysis revealed no mutations in the PA gene of Wa-182 when compared to the progenitor gene of A/WSN/33 (H1N1) wt (Odagiri & Tobita, 1990b), suggesting that a mutation or unique region of the nucleotide sequence, such as a uracil-rich region on the polmerase genes (Fields & Winter, 1982), is not a prerequisite for DI RNA generation. This is consistent with the study carried out by Jennings et al. (1983) in which no consensus sequences were found which might favor the generation of DI RNAs. In addition, determination of the sequences flanking the deleted regions of the progenitor genes failed to reveal significant homology with consensus cellular RNA splicing sequences (Fields & Winter, 1982; Nayak et al., 1982; Sivasubramanian & Nayak, 1983). Odagiri and Tobita (1990a) postulated that the non-structural influenza virus protein NS2 plays an important role in the synthesis of intact viral genomic RNAs. It may be a component of the replicase, determining the balance between the processivity of the polymerase and its ability to continue chain elongation on a new template.

2.2.6.2 Interference

DI particles amplify their genomes at the expense of homologous helper virus, which encodes the replication and encapsidation proteins essential for the replication of both. The interfering effects of DI particles can reduce the yield of virus to zero in doubly infected cells (Sekellick & Marcus, 1980). The helper virus must compete against the DI particles for its own gene products and consequently is hampered in its replication. Often the genomes of DI particles are rearranged in a manner which enhances their ability to replicate and to compete for replication and encapsidation proteins. This results in dominant replication of the DI particle and concomitant interference with the replication of the infectious helper virus. For example, VSV DI particles usually contain rearranged termini which favour replication and preclude transcription. Therefore, while helper virus genomes are engaged mainly in transcription DI genomes, which are devoted solely to replication, become dominant in the viral RNA population. Re and Kingsbury (1986, 1988) investigated the factors influencing the replication "fitness" of Sendai virus DI RNA species. They determined that there was a minimum size of about 450 bases for a defective RNA to be viable. However, any DI RNAs smaller than 1600 nucleotides (10% of the genome size) were enveloped with decreasing efficiency as their size decreased. This was thought to be due to limited membrane flexibility: as the limit of curvature is approached the efficiency of particle budding would be expected to decrease and smaller nucleocapsids would be discriminated against. For RNAs larger than 1600 nucleotides, the data were consistent with smaller RNAs being replicated more quickly. Additionally, superimposed on the size constraints, were the findings that copyback 3' termini were more competitive for replication than genomic 3' termini and non-transcribing RNAs were replicated more efficiently than transcribing RNAs.

DI genomes which are transcriptionally and/or translationally active can interfere at a level other than replication. For instance, relatively stable defective polypeptides of PB2 gene origin have been detected in influenza virus infected cells (Akkina *et al.*, 1984a, 1984b; Chambers *et al.*, 1984; Penn & Mahy, 1985; Akkina & Nayak, 1987). These polypeptides can bind to RNP complexes and by competing with functionally active PB2 render the complexes transcriptionally or replicationally defective. Despite being made in large amounts these polypeptides are not assembled into mature virions, and neither are the RNP complexes bound to defective PB2. Interference may therefore occur at the level of assembly leading to decreased virus production (Akkina & Nayak, 1987). DI particles are also capable of interfering indirectly with virus, for example by interferon induction (Johnson, 1981), modulation of immune responses (Dimmock *et al.*, 1986), or restriction of cell surface expression of virus proteins. In addition, the presence or absence of encapsidation or packaging signals in the DI RNAs determines their relative efficiency of maturation.

2.2.6.3 Influence of host cell type

The variability of DI particle effects within different cell types has been reported. DI particles which interfere strongly in one cell type may interfere only weakly, or not at all, in another. In some cases this is due to poor replication and amplification of the DI particles in particular cell types, but in some cell types the DI particles may replicate efficiently, but exert weak interference with helper virus replication (Choppin, 1969; Huang, 1988; Brinton *et al.*, 1984; Cave *et al.*, 1985; Kang *et al.*, 1981; Holland, 1987; Gillies & Stollar, 1980; Barrett & Dimmock, 1986).

2.2.6.4 Cell protection, disease attenuation and persistence

In addition to indirect cell-sparing effects such as interferon induction, alteration of immune responses and reduction of early virus yields, it has been proposed that DI particles exhert a direct cell-sparing effect (Huang & Baltimore, 1970). To date, numerous DI particles have been demonstrated to facilitate the establishment and maintenance of persistent infections by a wide variety of animal viruses in tissue culture (Huang & Baltimore, 1977; Holland *et al.*, 1980; Barrett & Dimmock, 1986; Holland, 1987; Schlessinger, 1988). In a number of systems an initial high multiplicity infection or undiluted passage of a normally cytolytic virus will result in a persistent infection. High multiplicity infection or undiluted passage of virus increases the probability of introducing DI viruses into the host cell or animal, as it is very difficult to prepare virus totally free of DI particles (Holland, 1987). Also, co-infection of DIs and a homologous cytolytic virus can convert a normally lytic infection to a persistent one, as reported for Sendai virus in BHK cells (Roux & Holland, 1979).

In vivo studies with influenza virus have demonstrated that influenza A/chicken/ Pennsylvania/1/83 (H5N2), a natural isolate consisting primarily of DI particles (Chambers & Webster, 1987), protects chickens from the virulent strain of the virus (influenza A/ chicken/Pennsylvania/1370/83) upon co-inoculation. Chambers and Webster (1991) investigated the biological basis of this effect and found that protection correlated with a complete inhibition of virulent virus spread to the brains of infected chickens. It appeared that the protective effect was afforded by DI particle-mediated interference with virulent virus multiplication within the respiratory tract of infected chickens. The defective viral isolate had to be administered within 20 hours of the virulent inoculum for protection to be achieved.

2.3 Persistent infections established by arboviruses

Arboviruses characteristically replicate in both vertebrate and invertebrate hosts, displaying very different outcomes of infection. Usually an inapparent persistent infection is established upon infection of an invertebrate vector, but on transmission to the vertebrate host an acute cytocidal infection ensues. These disparate outcomes of infection can be reproduced in tissue culture. In particular the replication of members of the Bunyaviridae and Alphaviridae in mosquito cell cultures has been extensively studied.

Persistent infections have been established in mosquito cell cultures with Marituba, Germiston, La Crosse and Bunyamwera bunyaviruses (Carvalho et al., 1986; Delord *et al.*, 1989; Rossier *et al.*, 1988; Newton *et al.*, 1981; Elliott & Wilkie, 1986) and with Toscana phlebovirus (Nicoletti & Verani, 1985). In addition, mosquito cell cultures persistently infected with Sindbis and Semliki forest alphaviruses have been well studied (Davey & Dalgarno, 1974; Shenk *et al.*, 1974; Gliedman *et al.*, 1975; Igarashi *et al.*, 1977; Riedel & Brown, 1977; Eaton, 1981; Stalder *et al.*, 1983; Brown & Condreay, 1986). In each case, an initial acute phase of infection, on average of about 3-4 days duration, was followed by a long-term persistent infection.

Invertebrate cell cultures first became available in 1967 when Singh derived Aedes albopictus cells from the macerated tissues of whole mosquito larvae (Singh, 1967). While infection of cultured Aedes albopictus cells with arboviruses has consistently been reported to lead to the establishment of persistent infections, different laboratories have reported varying degrees of cytopathic effect during the acute phase of infection. Tooker and Kennedy (1981) cloned subpopulations of cells from an Aedes albopictus cell line. Of 115 clones, they found that 70% produced poor yields of Semliki Forest virus and exhibited no cpe while 30% produced high virus yields and exhibited extensive cpe. In contrast, Brown and coworkers (Gliedman et al., 1975; Riedel & Brown, 1977,1979) did not observe cpe at any time after infection of their Aedes albopictus cells with Sindbis virus; repeated attempts to subclone cells which demonstrated cpe, from this particular cell line, were unsuccessful (Brown & Condreay, 1986).

All Aedes albopictus cell lines currently in use originated from the parental Singh line (Singh, 1967). This original culture probably contained a diverse population of partially differentiated cells destined for development into particular tissues in the mature mosquito. Variation between Aedes albopictus cell lines in different laboratories may have arisen as a consequence of subtle differences in tissue culture materials and cell-handling techniques, imposing slightly different selection pressures on the cell lines. The individual tissues of the mature mosquito are known to be differentially susceptible to arbovirus infection. While the mosquito as a whole shows no deleterious effects of virus infection and transovarial transmission to offspring has been reported (for both bunyaviruses and alphaviruses) without any indication of virus in the abdomen or thorax (Leake, 1984), the salivary glands produce large amounts of mature virus throughout the life time of the mosquito.

Studies with cell lines subcloned from Singh's original Aedes albopictus cultures (u4.4 cells, C6/36 cells and C7/10 cells) have demonstrated heterogeneity in the response of individual cell clones to arbovirus infection. Each of these cell lines produce different amounts of virus and display correspondingly different degrees of cytopathology following infection with Sindbis virus (Miller & Brown, 1992). In addition, these cell lines respond differently to homologous interference (Condreay & Brown, 1986) and actinomycin D treatment (Condreay *et al.*, 1988; see Section 2.3.2), and vary in their ability to produce and respond to an antiviral factor (Condreay & Brown, 1988; see Section 2.3.2).

There is evidence that host cell functions participate in the replication of Sindbis virus in cultured invertebrate cells. Studies with cloned *Aedes albopictus* cell lines, including treatment with actinomycin D (u.4.4 and C6/36 cells), and enucleation (u.4.4 cells) have been shown to interfere with Sindbis virus production (Scheefers-Borchel *et al.*, 1981; Erwin & Brown, 1983). Differences in the participation of the host cell in virus replication in each of these cell types could be responsible for the variability in virus production exhibited by these cells as well as the differences seen in the establishment of homologous interference and superinfection exclusion. By studying the effects of actinomycin D and cycloheximide on the synthesis of viral RNA and the production of mature virus in u.4.4, C6/36 and C7/10 cells, Condreay *et al.* (1988) demonstrated that host components were involved in at least two distinct steps in the production of Sindbis virus by mosquito cells: in the production of viral RNA and in the synthesis of viral structural polypeptides.

Vertebrate cell cultures which are permissive for bunyavirus replication usually perish within 72 hours of infection (Bishop & Shope, 1979). Uukuniemi virus (Pettersson, 1974) and Dugbe virus (David-West & Porterfield, 1974) have been shown to be exceptions. Uukuniemi virus did not affect host macromolecular synthesis in either resting BHK-21 or chick embryo cells, while Dugbe virus produced only very slight cpe in three continuous vertebrate cell lines (a pig kidney cell line, African green monkey cells and Rhesus monkey kidney cells). Viruses which are normally cytocidal for a particular cell type can be made to establish persistent infections under certain conditions. An example from the Bunyaviridae is the persistent infection established by Toscana phlebovirus upon undiluted passage in Vero cells (Verani *et al.*, 1984).

Studies on arbovirus persistence in invertebrate cell cultures have suggested that DI particles and antiviral proteins produced by the host cell may promote the persistent state. Evidence for the involvement of these factors in arbovirus persistence will be discussed in the following sections.

2.3.1 Defective interfering particles associated with arboviruses 2.3.1.1 The Bunyaviridae

Kascsak and Lyons (1978) demonstrated the generation of DI virus particles upon high multiplicity passage of Bunyamwera virus in BHK cells. Two major classes of virus particle were detected: standard virus which banded during isopycnic centrifugation at a buoyant density of 1.19 g/cm³ and non infectious particles with a buoyant density of 1.17 g/cm³. The latter particles contained only S segment sized RNA and interfered with the replication of standard Bunyamwera virus. Infection of BHK cells with a mixture of standard and defective Bunyamwera virus resulted in an alteration in the intracellular profile of viral RNA. Incorporation of radiolabelled precursors into the L and M RNAs was greatly reduced whereas labelling of the S RNA species was unaffected. The interference observed in this virus-cell system appeared therefore to stem from an inhibitory effect exerted at the level of viral genome replication. Defective viral stocks only inhibited the replication of homologous standard virus and this interference was host cell dependent. Studies on autointerference (defined as the inhibition of infectious virus yield at a high moi) in MDBK, *Aedes albopictus*, BHK and Vero cells, identified BHK and Vero cells as "high interference" cell lines, while MDBK and *Aedes albopictus* cells were designated "low interference" cell lines. The defective virus used in these studies was grown in BHK cells from virus which had been plaque-purified on Vero cell monolayers.

There are two reports to date of studies on the establishment and maintenance of persistent infection by Bunyamwera virus in *Aedes albopictus* mosquito cell lines. In the earlier paper Newton *et al.* (1981) used an uncloned *Aedes albopictus* cell line and no evidence was presented for the presence of defective virus. However, in a more recent study using the cloned mosquito cell line *Aedes albopictus* C6/36, Elliott and Wilkie (1986) described interfering particles which were released from the persistently infected cells and which contained full-length S RNA segments but no L or M segment RNA. In addition, viral S RNA was the major viral RNA species detected in the persistently infected cells. Some subgenomic RNAs were observed which originated from the L RNA segment.

More recently in a study carried out by Cunningham and Szilagyi (1987) on the replication of Germiston bunyavirus in BHK cells, discrete truncated RNAs (both genomic and antigenomic and of the same mobility on agarose-urea gels) were detected and shown to be derived from the L RNA. The genomic form was packaged into virions and released from infected cells. Defective virus particles were therefore expected to be present but interfering activity was not tested for. These RNAs represented 20% of the large genomic RNA and first became detectable in infected cells four hours after infection with 50 pfu/cell of Germiston virus. When two strains of Germiston virus were compared each gave rise to a unique size of subgenomic L RNA, suggesting that the defective replication event giving rise to these RNAs was a non-specific event. The positive-sense subgenomic L RNAs (which were not packaged into virions) were most likely to be replicative intermediates for replication of the defective L RNAs.

Recently, de Oliveira Resende *et al.* (1991) investigated the effect of repeated mechanical transfer of TSWV between plants on the genome constitution of the virus. Hybridization with an L RNA specific probe revealed subgenomic L RNA species after 20 or so mechanical passages. By 6 days post infection with the defective isolates, more defective L RNAs than full-length L RNAs were detected in the plants, suggesting that replication of the defective forms interfered with standard genome replication. While both negative- and positive-sense full-length and truncated L RNAs were detected in intracellular nucleocapsids, only negative-sense RNA species, both full-length and truncated, were detected in virus particles. It appeared therefore that negative-sense full-length and truncated L RNAs were sense full-length and truncated therefore that negative-sense full-length and truncated therefore that negative-sense full-length and truncated L RNAs were sense full-length and truncated L RNAs were sense full-length and truncated therefore that negative-sense full-length and truncated truncated therefore that negative-sense full-length and truncate

of defective L RNAs was associated consistently with symptom attenuation. Differential hybridization suggested that deletions may have occured in different regions of the L RNA as has already been described for other viruses (Holland, 1986).

In addition to the defective L RNA containing isolates, another distinct class of mutants was generated which were morphologically deficient in that they were incapable of producing enveloped particles and lacked the ability to synthesise the membrane glycoproteins. RNA analysis failed to reveal any large deletions in the three genomic RNA species of this isolate. Indeed the M RNA segment appeared to be full-length. The morphological defect may therefore be due to point mutations or very small deletions in the M RNA which do not affect the size of this molecule. Truncated L RNAs were not detected in the morphologically defective isolate.

Biological evidence has been presented for the involvement of DI particles in persistent infections established by Dugbe nairovirus in pig kidney cells (David-West & Porterfield, 1974) and Toscana phlebovirus in Vero cells (Verani *et al.*,1984), but no direct biochemical proof was obtained. Carvalho *et al.* (1986) examined Marituba bunyavirus replication in cultured *Aedes albopictus* cells and in a mouse fibroblast cell line (L-A9) but found no evidence for autointerference during primary infection of the invertebrate cells. However, studies carried out by Volkmer *et al.* (1983) revealed that Marituba virus replication in L-A9 cells generated DI particles. Marituba virus DI particles interfered with the replication of homologous but not heterologous virus.

2.3.1.2 The Alphaviridae

DI particles of alphaviruses have been generated in both vertebrate (Johnson *et al.*, 1975; Guild & Stollar, 1977; Stark & Kennedy, 1978; Barrett *et al.*, 1984) and invertebrate cell lines (Brown & Gliedman, 1973; Logan, 1979; King *et al.*, 1979; Tooker & Kennedy, 1981; Stalder *et al.*, 1983). Serial undiluted passage of alphaviruses in vertebrate cells has been demonstrated to generate DI particles with a host cell-restricted ability to interfere with standard virus. While these DI particles interfere with standard virus replication in vertebrate cells, they do not interfere with the replication of standard virus in invertebrate cell lines (Eaton, 1975; Igarashi & Stollar, 1976; King *et al.*, 1979). Co-inoculation of standard and defective alphaviruses into vertebrate cells has been demonstrated to generate alphavirus into vertebrate cells has been demonstrated to generate be particles interfere with standard virus in invertebrate cells, they do not interfere with the replication of standard virus in invertebrate cells interfere with standard virus and defective alphaviruses into vertebrate cells has been demonstrated to result in a persistent infection (Stollar, 1980).

Igarashi *et al.* (1977) and Eaton (1981) studied *Aedes albopictus* cell lines persistently infected with Sindbis virus. These authors reported the generation of Sindbis DI particles 7 to 10 weeks after the initiation of the persistent infection. The relatively late appearance of these DI particles suggested that they did not have an essential role to play in the establishment of the persistent infection. Subsequently, Tooker and Kennedy (1981) and Stalder *et al.* (1983) detected DI RNAs in cloned *Aedes albopictus* cell lines persistently infected with Semliki Forest virus. In both studies DI RNAs were observed within 24 h of

the initial infection. The cell lines underwent cytopathic crises during the acute phase of infection and 10% of the cells survived to establish the persistently infected cultures. It is possible that the rapid generation of DI RNAs prevented cell death in a subpopulation of the cells and so enabled transition to the persistent infection. Stalder *et al.* (1983) reported that the DI RNAs generated in their *Aedes albopictus* C6/36 cell line were not efficiently packaged into virions.

2.3.2 Antiviral activity associated with persistent arbovirus infections of invertebrate cell lines

Newton *et al.* (1981) could not detect any interferon-like activity in their persistently infected *Aedes albopictus* cell line. However an antiviral activity, distinct from interferon in the specificity of its action, has been shown to be produced by *Aedes albopictus* mosquito cells infected with Sindbis and Semliki Forest alphaviruses and also Banzi flavivirus (Riedel & Brown, 1979; Newton & Dalgarno, 1983; Condreay & Brown, 1986; 1988; Lee & Schloemer 1981a,b). Riedel and Brown (1979) described a low molecular weight antiviral protein in the culture medium of *Aedes albopictus* cells infected with Sindbis virus infected *Aedes albopictus* cells, prior to infection with Sindbis virus, reduced the virus yields and gave rise to levels of virus production and percentages of infectious centres characteristic of persistently infected cell populations. The maximal effect was achieved by incubating the cells in medium containing the antiviral activity 48 hours prior to infection. The antiviral activity was insensitive to antisera against BHK cell derived Sindbis virus and was rapidly inactivated at 56°C. The factor was sensitive to proteinase K but insensitive to α - and β -galactosidases, phospholipase C and a range of DNases and RNases.

The most striking observation was the specificity of action of this antiviral protein. The antiviral activity produced by Sindbis virus-infected mosquito cells did not interfere with Sindbis virus production in BHK-21 cells, nor did it interfere with the replication of Semliki Forest virus or West Nile virus in mosquito cells. Newton and Dalgarno (1983) identified an antiviral activity produced by Semliki Forest virus infected mosquito cells. This antiviral activity was also cell and virus specific, sensitive to proteinase K and heat sensitive. They demonstrated that no analogous antiviral activity was produced by *Aedes albopictus* cells persistently infected with Kunjin flavivirus or Bunyamwera bunyavirus, which suggested that the activity might be specific to alphaviruses with little significance in relation to infections of arthropod cells by other arboviruses. However, another group described an antiviral activity specific for Banzi flavivirus produced by infected mosquito cells (Lee & Schloemer 1981 a,b).

Based on experiments where RNA synthesis was measured as incorporation of [³H]-uridine into cells treated with actinomycin D, Newton and Dalgarno (1983) concluded that the antiviral activity released from Semliki Forest virus infected *Aedes albopictus* cells
blocked viral RNA synthesis, and that host transcription was required to establish the antiviral state. The validity of these results was brought into question when it was realised that actinomycin D itself efficiently arrests virus protein and RNA synthesis in cultured insect cells (Scheefers-Borchel, 1981; Condreay & Brown, 1988).

Since the observed repression of [³H]-uridine incorporation into RNA could have resulted from the presence of actinomycin D, Condreay and Brown (1988) studied the effect of the antiviral protein produced in Sindbis virus infected mosquito cells on virus specific RNA synthesis using a procedure not requiring actinomycin D. They studied the sensitivity to and the production of the antiviral protein in three independently subcloned Aedes albopictus cell lines, u.4.4, C6/36 and C7/10. The activity produced by infected u.4.4 cells was demonstrated to inhibit total viral RNA synthesis in these cells and in C6/36 cells, in a time dependent manner. Its effect was maximally realised when the cells were treated with the activity 48 hours before infection. Interestingly, Sindbis virus-infected C7/10 cells did not produce an antiviral protein and were not sensitive to the antiviral factor produced by Sindbis virus-infected u.4.4 cells. The data were consistent with an antiviral state induced by the antiviral protein which prevented the formation or efficient functioning of viral RNA-synthesising complexes in treated cells. Northern blot analysis showed that both 49S and 26S RNA species were reduced in Sindbis virus infected u.4.4 cells treated with the antiviral activity. It was not determined whether the activity affected the stability or the synthesis of the viral RNA. Experiments with a mutant of Sindbis virus which was known to be able to overcome an actinomycin D induced block in RNA synthesis in vertebrate cells (Baric et al., 1983), demonstrated that the block to viral replication induced by the antiviral factor was distinct from that induced by actinomycin D. This mutant was shown to bypass the actinomycin D induced block of viral RNA synthesis in invertebrate cells (Condreay et al., 1988) but did not overcome the block induced by the antiviral agent (Condreay & Brown, 1988).

2.3.3 The value of studying arbovirus persistence

A multitute of studies on a variety of virus-cell (or indeed virus-host) systems have created a wealth of published observations on persistent viral infections, but no real understanding of the mechanisms underlying the establishment and maintenance of the persistent state has been achieved. Persistent viral infections are difficult to study particularly *in vivo*, where the role of the virus, the infected tissue/cell and the host's immune response must be evaluated. *In vitro* models are therefore important tools with which to unravel the complexities of viral persistence. A number of tissue culture systems have been developed in which persistence is established by introducing interference at an early stage in infection. This can be achieved by using a high moi or serially passaged virus preparations (both of which contain a mixture of standard and defective virus), or by co-inoculation of a standard virus preparation and homologous defective interfering particles. Viral variants generated during persistent infections established by such manipulations are then often capable of establishing persistence by themselves. In arbovirus-invertebrate cell systems persistence is the natural course of infection. Since no manipulation of the system is required to establish persistence this is an ideal system with which to investigate the molecular mechanisms fundamental to the establishment of persistence.

2.4 Aims of this study

A detailed study of Bunyamwera virus persistence in C6/36 cells was carried out to gain some insight into the mechanisms underlying the establishment and maintenance of the persistent infection. The major questions addressed were as follows:

- What early molecular events limit the acute phase of Bunyamwera virus infection in C6/36 cells?
- How do levels of extracellular virus and intracellular viral RNAs vary during the course of persistence?
- Are defective viral RNAs generated, and if so are they fundamental to the maintenance of the persistent state?
- Is there heterogeneity in the response of individual cells within the culture to the persistent infection?

* Then subsequently from Dr Paul Young (London School of Hygiene and Tropical Medicine, London, U.K.).

.

•

3 Materials and Methods

3.1 Cells and virus

3.1.1 Cells

Aedes albopictus C6/36 cells (Igarashi, 1978) were obtained originally by Dr. R. M. Elliott from Dr. D. L. Knudson (Yale Arbovirus Research Unit, New Haven, Connecticut, USA). *

BHK-21 cells (MacPherson & Stoker, 1962) were obtained from the Cytology department of this Institute.

3.1.2 Virus

Bunyamwera virus (Watret et al., 1985) was three times plaque-purified in BHK cells from a laboratory stock originally obtained from Dr. N. Karabatsos, Vector-borne Diseases Laboratory, Center for Disease Control, Fort Collins, Co., USA. Working stocks of virus were only two passages in BHK cells away from the triple plaque-purified stock.

3.2 Tissue culture materials

Materials marked with an asterisk were obtained from the media department of this Institute (details are given in Section 3.2.2). "HSA" grade agarose, which was used for the plaque-assay overlays, was obtained from Park Scientific Ltd., Northampton, U.K., and Cidex fixative (activated glutaraldehyde solution) from Johnson & Johnson Co. The remaining tissue culture materials were obtained from Life Technologies, Paisley.

3.2.1 Tissue culture media

Glasgow Minimal Essential medium (GMEM) containing 10% newborn calf serum and 10% tryptose phosphate broth.

Leibovitz L-15 medium (Leibovitz, 1963) supplemented with 5% heatinactivated foetal calf serum and 10% tryptose phosphate broth.

3.2.2 Tissue culture solutions

All solutions were obtained from the Institute's in-house media department. **Amphotericin B:** 0.025% (w/v) amphotericin B in distilled water. **Eagles A:** 1.6mM CaCl₂.2H₂O; 1mM MgSO₄.7H₂O; 0.001% (v/v) HCl.

Eagles B (without phenol red): Solution of salts, amino acids and vitamins in distilled water, with final concentrations as follows: 0.88M NaCl; 43mM KCl; 6mM NaH₂PO₄.2H₂O; 0.2M glucose; 0.0001% ferric nitrate; 0.26M NaHCO₃; 0.02% (w/v) Lglutamine; 8x10⁵ Units of penicillin; 0.08% (w/v) streptomycin; 0.0001% (w/v) antimycotic; 0.033% (w/v) L-arginine mono HCl; 0.019% (w/v) L-cysteine; 0.015% (w/v) L-histidine mono HCl; 0.041% (w/v) Isoleucine; 0.041 (w/v) L-leucine; 0.056% (w/v) L- lysine mono HCl; 0.026% (w/v) L-phenylalanine; 0.038% (w/v) L-threonine; 0.006% (w/v) L-tryptophan; 0.028% (w/v) L-tyrosine; 0.037% (w/v) L-valine; 0.012% (w/v) L-methionine; 0.003% (w/v) inositol; 0.002% (w/v) each choline chloride, folic acid, nicotinamide; DL pantothenic acid (calcium salt); pyridoxal HCl; thiamine HCl and 0.0002% riboflavin. Carbon dioxide was bubbled through the solution until the pH reached 6.5 and the solution was filter sterilised.

Giemsa stain: 1.5% suspension of Giemsa stain (BDH) in glycerol, heated at 56°C for 2 hours, diluted with an equal volume of methanol, and diluted 1:10 in buffered distilled water before use.

Neutral red: 0.4% (w/v) solution of neutral red chlorine in distilled water.

PBS working solution: 137mM NaCl; 2.7mM KCl; 4.3mM

Na₂HPO₄.7H₂O; 1.4mM KH₂PO₄, pH 7.3.

Trypsin: 0.25% (w/v) Difco 1: 250 trypsin in Tris-saline (136mM NaCl; 5.4mM KCl; 0.4mM Na₂HPO₄; 5.6mM dextrose [D-glucose]; 24.8mM Tris (hydroxymethyl) aminomethane). HCl was added to bring the pH to 7.7 and 0.002% (w/v) phenol red was added along with 100,000 Units of penicillin and 0.01% streptomycin. The trypsin was added last, prior to filter sterilization.

Tryptose phosphate broth: 3% (w/v) Difco Bacto tryptose phosphate in distilled water.

Versene: 0.5mM EDTA; 137mM NaCl; 2.6mM KCl; 4.3mM Na₂HPO₄.7H₂0; 2.4mM KH₂PO₄ (pH 7.12 to 7.3) plus 0.002% (w/v) phenol red.

3.2.3 Overlay for plaque titrations

- 310ml Eagles A*
 - 65ml Eagles B (without phenol red)*
- 20ml Newborn calf serum
- 125ml Agarose (2.4% in Eagles A w/o phenol red)*

3.2.4 Neutral red overlay

- 310ml Eagles A*
 - 65ml Eagles B (without phenol red)*
 - 7ml Neutral red (0.4%)*
- 125ml Agarose (2.4% in Eagles A w/o phenol red)*

3.2.5 Tissue culture plasticware

Tissue culture plasticware was obtained either from Life Technologies Ltd. (flasks, Limbro well plates and 35mm diameter Petri dishes) or Sterilin Ltd. (50mm and 100mm diameter Petri dishes).

3.3 Bacteria

Escherichia coli strain JM101 (supE44, thi, Δ (lac-proAB) / F', [tra $\Delta 36$, proAB⁺, lac I^q lacZ $\Delta M15$]; Messing, 1979) was used throughout the course of this work.

3.4 Bacterial culture materials

Escherichia coli strain JM101 was grown in 2YT broth. 2YT broth: 85mM NaCl, 10g/l Difco Bactotryptone, 10g/l yeast extract. L-broth: 117mM NaCl, 10g/l Difco Bactopeptone, 5g/l yeast extract (pH 7.5 prior to sterilization).

Top agar: 1% (w/v) agar in water.

Agar plates contained 1.5% w/v agar in L-broth.

Petri dishes were obtained from Bibby Sterilin Ltd., U.K.

3.5 Plasmids and vectors

3.5.1 Plasmids

pMG1: Mouse cytoskeletal γ -actin cDNA cloned in pBR322 (Peter et al., 1988); kindly provided by Jasmine Daksis. The [³²P]-labelled actin probe was prepared from *Pst* I-digested pMG1 DNA.

pBUN127: The Bunyamwera virus M gene sequences inserted into M13mp18 bacteriophage during the construction of templates for the synthesis of labelled cDNA probes were obtained from pBUN127 (Lees et al., 1986). This recombinant plasmid contained a G/C tailed cDNA of approximately 3600 nucleotides derived from the Bunyamwera virus M segment cloned into the *Pst* I site in the ampicillin resistance gene of pBR322.

pTZ18R-BUN L: positive-sense (L RNA specific) riboprobes were synthesised from a full-length L segment cDNA cloned into the *Sma* I site of pTZ18R, under the control of the bacteriophage T7 promoter (Jin & Elliott, 1991; unpublished data).

pT7T3 M: positive-sense (M RNA specific) riboprobes were transcribed from a full-length M segment cDNA cloned into the *Pst* I site of pT7T319U, under the control of the T3 promoter, obtained from George Nakitare.

pUC9 BUNS 378/79: positive-sense (S RNA specific) riboprobes were transcribed from a full-length S segment cDNA inserted into pUC9, under the control of the T7 promoter, obtained from Alastair McGregor.

3.5.2 Cloning Vectors

M13mp18 bacteriophage vectors (Messing & Vieira, 1982) were used for the production of single-stranded DNA. *Hinc* II fragments from pBUN127 (see later) were cloned into the double-stranded replicative form of M13mp18.

3.5.3 Recombinant M13mp18 bacteriophage

Recombinant M13mp18 bacteriophage, containing fragments of cDNA to Bunyamwera virus S (Elliott, 1989b), M (Lees et al., 1986) or L (Elliott, 1989a) RNA inserted into the lacZ coding sequence of the vector, were used as templates from which [³²P]-labelled strand specific cDNAs were synthesised for use as hybridization probes. For the L and S segment specific probes, single-stranded template DNA was prepared from selected laboratory stocks of recombinant bacteriophage which had been established during the sequencing of the Bunyamwera virus L and S segments (Elliott, 1989a, 1989b). M13mp18 recombinants containing fragments of the Bunyamwera virus M RNA were constructed from the cDNA insert from pBUN127 (Lees et al., 1986; see Section 3.12.1). Details of the recombinant bacteriophage DNAs used as templates and the probes synthesised from them are given in Table 3.1.

Template DNA	Strand detected	Bases covered	Size	Probe
			(nuc)	name
pBUN357.A8	Genomic L	997-1436	439	L5
pBUN363.86	Antigenomic L	5664-5864	200	L7
pBUN308.6	Genomic S	10-400	390	S8
pBUN308.15	Antigenomic S	550-944	394	S9
pBUN.M1	Genomic M	2233-2517	285	M1
pBUN.M6	Antigenomic M	2517-2233	285	M6

Table 3.1Details of the recombinant bacteriophage DNAs used as
templates to synthesise radiolabelled cDNA probes

3.6 Chemicals, radiochemicals, miscellaneous reagents and enzymes

3.6.1 Chemicals

Chemicals were obtained from either BDH Ltd. or Sigma Chemical Co. Ltd., with the following exceptions: Seakem GTG Agarose (FMC Bioproducts); ammonium persulphate and TEMED [N,N,N'N' tetramethylethylenediamine] (Bio-Rad Laboratories Ltd.); caesium chloride [99.9%] (Melford Laboratories Ltd.); methylmercury (II) hydroxide [1M in water] (ALFA Products); En³hance (NEN Research Products); formamide (Fluka Chemicals Ltd.); Sephadex G-50, Fine [20-80µm] (Pharmacia Fine Chemicals); IPTG and X-gal (BRL, Life Technologies Inc.).

3.6.2 Radiochemicals

Radioisotopes were supplied by Amersham International plc. at the following specific activities:

[³⁵ S]-L-methionine	800mCi/mmol	(approx.)
5' α-[³² P] dNTPs	3000 Ci/mmol	(10µCi/µl)
5' α-[³² P] CTP	400 Ci/mmol	(10µCi/µl)
γ-[³² P] dATP	5000 Ci/mmol	(10µCi/µl)

3.6.3 Miscellaneous materials

Dialysis membrane (A & J Beverage Ltd); nitrocellulose membranes [0.2 or 0.45 µm pore size] (Schleicher & Schuell, Inc.); RNasin (Promega); Millipore disposable filters (Nalge Company); Spin-X centrifuge filter units (Costar); Falcon tubes (15 and 50 ml) (Becton Dickinson Labware); rabbit reticulocyte lysate (New England Nuclear); Multiprime DNA labelling system (Amersham International plc.); and Sequenase DNA sequencing kit (United States Biochemicals).

3.6.4 Enzymes

Restriction enzymes and buffers and calf intestinal phosphatase (CIP) were supplied by Boehringer-Mannheim. Lysozyme was obtained from Sigma; AMV reverse transcriptase from Life Sciences; T7 RNA polymerase and RQ1 RNase-free DNase from Promega; T4 polynucleotide kinase from Northumbrian Biologicals Ltd. (NBL); T7 DNA polymerase from Pharmacia LKB Ltd., while large fragment DNA polymerase I (Klenow enzyme), DNA polymerase I, and T4 DNA ligase were supplied by Bethesda Research Laboratories (BRL).

3.7 Commonly used solutions

3.7.1 Reaction buffers

In vitro transcription buffer (5x): 200mM Tris.HCl (pH 7.5), 30mM MgCl₂, 10mM spermidine, 50mM NaCl.

Kinase buffer (5x): 250mM Tris.HCl (pH 7.6), 50mM MgCl₂, 25mM DTT, 0.5mM spermidine.

Ligase buffer (BRL; 5x): 0.25M Tris.HCl (pH 7.6); 50mM MgCl₂; 5mM ATP; 5mM DTT; 25% (w/v) polyethylene glycol-8,000.

NT buffer (10x): 0.5M Tris.HCl (pH 7.5), 0.1M MgSO₄, 10mM DTT, 0.5mg/ml BSA.

NT buffer (10x) & dNTP working solution: 500μ l of 10x NT buffer; 5μ l of 4x 5mM dNTPs.

NTE buffer: 100mM NaCl, 10mM Tris, 1mM EDTA.

TA buffer (10x): 330mM tris acetate pH 7.9, 660mM potassium acetate, 100mM magnesium acetate, 1mg/ml nuclease free bovine serum albumin, 5mM DTT.

TE buffer: 10mM Tris.HCl pH 8.0, 1mM EDTA pH 8.0. **TM buffer (10x):** 100mM Tris.HCl pH 8.0, 100mM MgCl₂.

3.7.2 Buffers for RNA extractions

Guanidinium thiocyanate (GSCN) buffer: 4M guanidinium thiocyanate; 25mM sodium citrate, pH 7.0; 0.5% (w/v) sarcosyl; 0.1M β -mercaptoethanol; 0.33% (v/v) Antifoam A. The guanidinium thiocyanate, sodium citrate, sarcosyl and Antifoam A were dissolved in 40ml of sterile distilled water, the volume was adjusted to 100ml and the solution was filtered through a disposable millipore filter. Finally the β -mercaptoethanol was added.

Lysis buffer (to disrupt virions): 50mM Tris.HCl pH 8.0; 10mM NaCl; 0.6% NP40.

Lysis buffer (to disrupt infected cells): 50mM Tris.HCl pH 7.5; 150mM NaCl; 0.6% NP40.

3.7.3 Buffers for direct RNA sequencing

5x Transcription buffer: 120mM Tris.HCl pH 8.3; 80mM MgCl₂; 40mM DTT; 2mM dATP, dCTP and dTTP; 4mM dGTP; 500µg/ml actinomycin D.
5x Annealing buffer: 50mM Tris.HCl pH 8.3; 5mM EDTA; 1.25M KCl. (Both were stored frozen at -20°C).

3.7.4 Sample buffers (containing dyes to mark progress of

electrophoresis)

Formamide dyes: 100% formamide; 0.3% bromophenol blue; 0.3% xylene cyanol F.F.

Protein Dissociation buffer: 4% (w/v) SDS; 1.15M β -mercaptoethanol; 0.16M Tris.HCl pH 6.8; 25% glycerol and 0.05% bromophenol blue.

Sample Buffer (2x) for methylmercury gels: 20% glycerol; 0.05% bromophenol blue; 2x EM buffer.

3.7.5 Electrophoresis buffers

EM buffer (10x): 0.05M sodium borate; 0.1M sodium sulphate; 0.01M EDTA; 0.15M boric acid.

TBE buffer (10x): 0.89M Tris; 0.89M boric acid; 0.002M EDTA.

Tris-glycine (10x): 1.9M Glycine; 0.25M Tris.

(SDS-PAGE running buffer consisted of 1x Tris-glycine made 0.1% with respect to SDS)

3.7.6 Solutions for hybridization/Northern blotting

Denhardt's (100x): 2% (w/v) Ficoll 400; 2% (w/v) polyvinylpyrrolidone; 20mg/ml bovine serum albumin (Pentex fraction V, Miles Laboratories). Dissolved in sterile distilled water, filtered and stored at -20°C (Adams et al., 1990).

SSC (20x): 3M NaCl; 0.3M sodium citrate, pH adjusted to 7.0 with 1M HCl. SSPE (20x): 3.6M NaCl; 200mM NaH₂PO₄ (pH 7.4); 20mM EDTA (pH 7.4).

3.7.7 Other solutions

Gel fix: 50% (v/v) methanol; 10% (v/v) acetic acid; 40% water.

3.8 Synthetic oligonucleotides

Oligonucleotides were synthesised by Dr. J. McLauchlan using a Biosearch DNA synthesiser, model 8600. Six 20mer oligonucleotides were used during the course of this work, details of which are given in Table 3.2. Tubes containing freshly synthesised oligonucleotides were heated at 55°C for 5 hours and opened on ice in a fume hood to release ammonia vapours. The oligonucleotides were transferred to labelled 1.5ml reaction vials and lyophilised for a minimum of 5 hours in a vacuum concentrator.

Oligo No.	Annealing temp. (T ^o C)	Sequence
1	51	5' -CAACAGAAGGTCATTAAAGG- 3'
2	51	5' -TTCTACATTAAAGGACGCGA- 3'
3	61	5' -CTCCACCGCCTCAGTGGATG- 3'
4	55	5' -CTCTCTTTCTTCCCAGGATC- 3'
5	55	5' -GACAGAAGTTCAGTCTGCAC- 3'
6	53	5' -AGGGTTGTTTCAACCCACAA- 3'

Table 3.2Sequences and annealing temperatures $[T (^{o}C) = 4 (G+C) + 2 (A+T) - 5^{o}C]$ of the oligonucleotides used for direct RNA sequencing.

3.9 Cell manipulations

3.9.1 Maintenance of cells

Aedes albopictus C6/36 cells were maintained at 28°C in L-15 medium (Leibovitz, 1963) supplemented with 5% heat inactivated foetal calf serum and 10% tryptose phosphate broth. The cells were subcultured at a 1/20 split at approximately 6 day intervals and were regularly tested for mycoplasma contamination.

BHK-21 cells were maintained at 37°C in Glasgow minimal essential medium (GMEM) supplemented with 10% newborn calf serum and 10% tryptose phosphate broth

(Stoker & MacPherson, 1961). The BHK cells were not routinely subcultured in our laboratory but were obtained regularly from the cytology department of this Institute.

3.9.2 Establishment of an *Aedes albopictus* C6/36 cell line (C6/36/BUN) persistently infected with Bunyamwera virus

Aedes albopictus C6/36 cells were grown to 90% confluency and infected with 5 pfu/cell of Bunyamwera virus. After four days the cells were subcultured at a 1/10 split and thereafter at a 1/20 split at approximately 6 day intervals. The cells were maintained at 28°C in supplemented L-15 medium.

3.9.3 Cloning of cells from the C6/36/BUN cell line

Persistently infected Aedes albopictus C6/36 cells were cloned from a given pass of the C6/36/BUN cell line, essentially as described by Igarashi et al. (1977). The cells were dispersed into a single-cell suspension in L-15 medium and aliquots containing approximately 20 single cells were seeded onto 100mm diameter tissue culture plates. The absence of aggregates of cells was confirmed by examination of an aliquot of the suspension under the light microscope. The cells were maintained at 28°C in L-15 medium and within two weeks colonies of cells were visible to the eye. The medium was carefully removed from the plates and isolated colonies were circled on the underside of the plates using a marker pen. Small (2mm²) pieces of sterile filter paper were immersed in a 1:4 mixture of trypsin:versene. Circled colonies were overlaid with a piece of saturated filter paper for a few minutes while the trypsin dislodged the cells from the plate. The cells were transferred attached to the filter paper, using a sterile syringe needle, into 1ml of L-15 medium contained in individual wells (15mm diameter) of a 24 well plate. The 24 well plate was then incubated at 31°C until confluent monolayers were observed for each clone. Transfer of confluent monolayers from multiwell plates to small (25cm³) tissue culture flasks was carried out by washing the cells off the well surface using a sterile Pasteur pipette and L-15 medium. Transfer of cells to medium (80cm³) and large (175cm³) tissue culture flasks was achieved by shaking the cells into fresh medium. These cloned cell lines were maintained in an identical manner to the uninfected Aedes albopictus C6/36 cell line.

3.9.4 Cryopreservation of cell cultures

Liquid-nitrogen stocks of cells from each pass of the C6/36/BUN cell line and of cloned cell lines derived from this cell line were established. The cells were frozen at a concentration of approximately 1x10⁷ cells/ml in Leibovitz's L-15 medium, containing 10% dimethyl sulphoxide (DMSO) which had been filter sterilized. The cell suspension was transferred to 2ml plastic cryotubes and stored at -70°C overnight before being immersed in liquid-nitrogen for long-term storage.

3.9.5 Revival of liquid-nitrogen cell stocks

Ampules of cells were thawed at 37°C and decanted into small (25cm³) tissue culture flasks along with 5ml of Leibovitz's L-15 medium. The cells were left to settle for 4 h at 28°C after which time the culture fluid was replaced with fresh Leibovitz's L-15 medium.

3.10 Virus manipulations

3.10.1 Plaque-purification of virus

Fifty millimetre diameter tissue culture dishes were seeded with $2x10^6$ BHK cells/dish in 4ml of GMEM and incubated overnight at 37°C. Serial ten fold dilutions of virus, in GMEM, were prepared to 10^{-6} dilution and 100μ l of each dilution was used to infect individual monolayers. Each dilution was plated out in duplicate. The virus was left to adsorb for one hour at 31°C. The inoculum was then removed and replaced with 4ml of agarose overlay (see Section 3.2.3). After four days at 31°C, a second agarose overlay containing 0.01% neutral red (see Section 3.2.4) was applied to the plates and the plates were returned to the 31°C incubator for a further 24 hours. At the end of the incubation period plaques were visible as clear round areas against a red background. Well separated plaques were picked into 0.5ml of medium using a sterile Pasteur pipette and stored frozen at -20°C. The second and third successive plaque picks were carried out as above, with the inoculum comprising ten-fold dilutions (to 10^{-3}) of 30μ l of the preceeding plaque isolate.

3.10.2 Assay of viral infectivity by plaque-titration

Virus titres were determined by plaque-titration on BHK cell monolayers of 90% confluency. Fifty millimetre diameter tissue culture dishes were seeded with 2x10⁶ BHK cells/dish in 4ml of GMEM and incubated overnight at 37°C. Serial ten fold dilutions of virus in GMEM were prepared to 10⁻⁸ dilution. 100µl of each dilution was used to infect a single monolayer taking care to distribute the inoculum over the entire monolayer. Each dilution was assayed in duplicate. The virus was left to adsorb for 1 hour at 31°C after which time the inoculum was removed and replaced with 4ml of agarose overlay (see Section 3.2.3). The overlaid, infected cell monolayers were incubated at 31°C for 4 to 5 days and then fixed with activated Cidex (1% glutaraldehyde) for 4 hours. The monolayers were stained with Giemsa stain for 10 minutes, excess stain was washed off and the plates were left to air dry prior to counting the plaques.

3.10.3 Preparation of elite stocks of virus

Small (25cm^2) tissue culture flasks were seeded with $2x10^6$ BHK cells in 15 ml of GMEM and incubated overnight at 37°C. The next day the medium was removed from the flasks and the cells were infected at a multiplicity of 0.001 to 0.01 pfu/cell with triple plaque-purified virus (200µl out of the 500µl total plaque isolate). The flasks were shaken gently to disperse the virus over the cells and incubated at 31°C for one hour while virus

adsorbed to the cells. The inoculum was then removed and 5ml of GMEM was added to each flask. The infection was allowed to proceed at 31°C until cpe was evident (usually 48 to 72 hours). The medium was then decanted from each flask into a 15 ml Falcon tube and clarified by centrifugation in an MSE Coolspin centrifuge at 3,000 rpm for 8 minutes at 4°C. The clarified virus suspension was divided into 2.5ml aliquots and stored frozen at -70°C. Each elite stock was titrated in BHK cells.

3.10.4 Preparation of virus working stocks

Large (175cm²) tissue culture flasks were seeded with 1×10^7 BHK cells/flask and incubated overnight at 37°C. The resulting 90% confluent cell monolayers were infected with 0.001 to 0.01 pfu/cell of Bunyamwera virus elite stock and the virus was adsorbed for one hour at 31°C. Fifteen millilitres of GMEM was added to each flask and the infection was allowed to proceed at 31°C for 4 to 5 days, by which time extensive cpe was evident. The culture fluid was harvested and clarified by centrifugation in an MSE Coolspin centrifuge at 3,000 rpm for 8 minutes at 4°C. The virus suspension was divided into 5ml aliquots and stored frozen at -70°C. Each working stock was titrated in BHK cells.

3.10.5 Assay for resistance to superinfection

Cells from a chosen pass of the persistently infected C6/36/BUN cell line, or from cloned cell lines derived from this culture, were harvested into L-15 medium and dispersed into a single-cell suspension. Duplicate 35mm diameter tissue culture plates were seeded with 1.5x10⁶ cells/plate and the cells were left at 28°C overnight to grow to 90% confluency. The cell monolayers were either infected with 5 pfu/cell of wt Bunyamwera virus or mock infected with 250µl of L-15 medium, and incubated for 1 hour at 28°C. After 1 hour the inoculum was removed, the monolayers were washed 2x with PBS and 2.5ml of fresh L-15 medium was added to each plate. The plates were maintained at 28°C for 48 hours. The supernatant was then removed for plaque-titration in BHK cells, as described previously.

3.10.6 Infectious centres assay

Cells from passage 6 of the persistently infected C6/36/BUN cell line were washed twice in PBS, and harvested as a single-cell suspension in 10ml of Leibovitz L-15 medium. The concentration of the cell suspension was calculated by counting the number of cells in an aliquot, using a haemocytometer. Ten-fold dilutions of the cell suspension (down to 10^{-6}) were prepared in supplemented GMEM medium and 200µl aliquots of each dilution were plated onto 90% confluent BHK cell monolayers in 50mm diameter tissue culture dishes. The C6/36 cells were left to adsorb to the BHK cell monolayers at 31°C for one hour. The plates were then overlaid with 4ml of agarose overlay (see Section 3.2.3) and incubated at 31°C for 4 to 5 days. The cell monolayers were subsequently fixed and stained as detailed in Section 3.10.2. Plaques were counted and the proportion of C6/36 cells releasing infectious virus was calculated.

3.11 Bacterial manipulations

3.11.1 Preparation of competent bacteria

A single colony of *E. coli* JM101 was picked off an agar plate (which had been stored sealed at 4°C) and was inoculated into 10ml of 2YT broth. The culture was shaken overnight at 37°C. 0.5ml of the overnight culture was inoculated into a conical flask containing 50ml of 2YT broth and incubated at 37°C with shaking for 2.5 to 3 hours, until the optical density reading at 590nm was 0.3. The flask was then placed on ice for a few minutes. The bacterial culture was transferred to a 50ml Falcon tube and the bacteria were pelleted by centrifugation at 3,000 rpm for 8 minutes in a Wifug 500E bench centrifuge. The supernatant was discarded and the bacterial pellet was resuspended in 20ml of cold 0.1M CaCl₂. The bacterial suspension was left on ice for 20 minutes to 2 hours before being pelleted as before. The bacterial pellet was resuspended in 3ml of cold 0.1M CaCl₂ and stored on ice until required.

3.11.2 Transformation of E. coli strain JM101

Selected stocks of M13mp18 replicative form (RF) DNA were diluted in 2YT medium and 50 to 500 ng of DNA was used to transform *E. coli* JM101 bacteria. Competent cells and DNA were mixed as follows:

300µl Competent cells

1.0µl DNA (0.05 to 0.5 mg/ml)

The mixture was incubated on ice for one hour, and then heat-shocked at 42°C for 40 seconds. Meanwhile 1ml of an *E. coli* JM101 overnight culture was added to 10ml of 2YT broth and shaken at 37°C for one hour. A bottle of top agar was melted and equilibrated at 42°C and the required number of LB agar plates were dried at 37°C. Each transformation mix was added to a prewarmed tube (42°C) containing the following:

- 250µl Freshly grown JM101
- 25µl IPTG (100mM)
- 25µl X-gal (20mg/ml in dimethylformamide)
- 3ml Top Agar

The contents of the glass tubes were mixed and poured immediately onto dry, prewarmed LB agar plates. The plates were tipped gently from side to side to ensure an even distribution of bacteria and top agar and left at room temperature for 5 minutes to set. The plates were incubated inverted at 37°C overnight. Clear plaques, indicative of bacteriophage containing insertions into the vector cloning sites within lacZ coding sequences, were visible the next day and isolated plaques were picked.

3.12 DNA manipulations

3.12.1 Synthesis of recombinant M13mp18 containing cDNAs to the Bunyamwera virus M segment

The 3.6 kb Bunyamwera virus M segment cDNA from pBUN127 was excised with *Pst* I and digested for two hours at 37°C with 20 Units of *Hinc* II (see Section 3.12.6). After phenol/chloroform extraction, the DNA fragments were ethanol precipitated and resuspended in 20 μ l of water. The fragments were then ligated into M13mp18 (previously linearised with *Sma* I and dephoshorylated using calf intestinal phosphatase) by incubation with T4 DNA ligase at 15°C overnight. Three reactions were set up as shown in Table 3.3. After the overnight incubation, the recombinant phage were extracted once with phenol and once with chloroform, ethanol precipitated (see Section 3.12.5) and dissolved in 20 μ l of water.

	1	2	3
Hinc II digest.	1µ1	5µl	10µ1
5x Ligase buffer	4µl	4µl	4μl
T4 DNA ligase	0.5U	0.5U	0.5U
Sma/cap M13mp18	1µ1	1µl	1µ1
Water	To make	total volun	ne to 20µl

Table 3.3 Ligation reactions set up to generate Bunyamwera virusM segment cDNA/M13mp18 recombinants.

3.12.2 Preparation of single-stranded template DNA

A 100ml bottle of 2YT broth was prewarmed to 37° C and 1ml of a fresh overnight culture of *E. coli* JM101 bacteria was added. Single, isolated plaques, freshly picked with a sterile cocktail stick, were added to 1.5ml aliquots of this suspension in sterile universals (the cocktail stick was left in the universal) and the cultures were incubated in an orbital shaker, set at 100-200 rpm and 37° C, for 5-6 hours. One ml of the final infected culture was transferred to a 1.5ml Eppendorf tube and the bacteria were pelleted by centrifugation at 13,000 rpm for 5 minutes in a microcentrifuge. The supernatants were decanted into a fresh Eppendorf tube (taking care not to transfer any of the pelleted bacteria) and 150µl of 20% polyethylene glycol in 3.5M potassium acetate (or 2.5 M NaCl) was added. The samples were mixed well by several vigorous inversions and left at room temperature for at least 30 minutes, or at 4°C overnight. The bacteriophage were pelleted at 13,000 rpm for 5 minutes in a microcentrifuge and the supernatant fluid was removed using a sterile Pasteur pipette attached to a suction pump. The bacteriophage pellets were forced to the other side of the Eppendorf tubes by centrifugation at 13,000 rpm for 30 seconds, and the last traces of supernatant were removed as before. The

bacteriophage were resuspended in 100 μ l of TE and 50 μ l of TE-equilibrated phenol was added. The samples were vortexed for 10 seconds, left on the bench for 5 minutes and vortexed again briefly before centrifugation at 13,000 rpm for 10 minutes. The upper aqueous phase containing the phage DNA was transferred into a fresh tube and the DNA was precipitated by the addition of 10 μ l 3M sodium acetate pH 5.5 and 250 μ l of ethanol. The samples were vortexed briefly and left overnight at -20°C or on dry ice for 30 minutes. The DNA was pelleted by centrifugation at 13,000 rpm for 5 minutes. The supernatant was removed using a Pasteur pipette and the pellet of single-stranded bacteriophage DNA was washed with 70% ethanol. The DNA was lyophilised and dissolved in 30 μ l of sterile distilled water. The templates were stored at -20°C.

3.12.3 Quantitation of nucleic acid

The concentration of RNA and DNA preparations was determined from optical density measurements at 260nm (OD_{260}), measured using Beckman microcell 8mm high quartz cuvettes in a Beckman DU-62 spectrophotometer. It was assumed that an OD_{260} of one was equivalent to a concentration of 50µg/ml for double-stranded DNA, 40µg/ml for single-stranded DNA and RNA, and 20µg/ml for single-stranded oligonucleotides.

3.12.4 Assessment of the purity of nucleic acid preparations

Two optical density measurements were taken for each preparation, one at 260nm and another at 280nm. Calculation of the OD_{260}/OD_{280} ratio provided an assessment of the purity of a given preparation. Ratios of 1.8 and 2.0 are indicative of relatively pure DNA and RNA preparations respectively. Contamination with phenol or protein lowers the ratio.

3.12.5 Ethanol precipitation

Nucleic acids were precipitated with one tenth volume of 3M NaOAc pH 5.5 and 2.5 volumes of ice-cold ethanol. The samples were mixed well and frozen at -20 or -70° C for at least one hour. The precipitate was pelleted by centrifugation at 12,000 x g for 20 minutes and the supernatant was removed. The pellet was washed with 70% ethanol and recentrifuged briefly. Finally the supernatant was removed, the pellet was dried in a vacuum concentrator and resuspended in the appropriate buffer.

3.12.6 Restriction enzyme digestion

Restriction digests were set up using 2 units of enzyme per μ g of double-stranded DNA (BRL enzymes and buffers were used). A typical reaction would be:

- 10µg double-stranded DNA
- 6μl 10x restriction enzyme buffer
- $2\mu l$ (20 U) restriction enzyme H_2O to $60\mu l$

Digestion usually required a 2-5 hour incubation at the optimum temperature for the restriction enzyme. The extent of digestion was assessed by running $1/10^{\text{th}}$ of the sample on an agarose minigel with ethidium bromide in the running buffer, and visualizing the digestion pattern with a long wave UV source. The remainder of the sample was phenol extracted, ethanol precipitated and taken up in 10µl of H₂O.

3.12.7 Gel purification of oligonucleotides

A 16% denaturing polyacrylamide/urea gel was prepared as follows:

- 25ml 16% sequencing gel mix (see Section 3.18.4)
- 0.25ml 10% ammonium persulphate
 - 25µl TEMED

The gel was poured to 0.3mm thickness. Formamide dyes were added to each sample (1/10th of the sample volume) and the gel was loaded. Electrophoresis was carried out at 37W for approximately 3 hours or until the bromophenol blue had migrated 3/4 of the way down the gel. The glass plates were removed from the gel which was then covered with "Clingfilm". The gel was placed face upward on a fluorescent thin-layer chromatography (TLC) plate and the oligonucleotides were visualized by UV shadowing. Oligonucleotides have a UV absorption peak at 260nm; when the electrophoresed gel was placed on the fluorescent TLC plate and illuminated from above by short wave UV light the oligonucleotides were visible as dark bands against the light surface. The oligonucleotides were excised from the gel using a clean scalpel and transferred to a 15ml Falcon tube. The gel slices were covered with water and the oligonucleotides were eluted into the water by shaking at 37°C overnight. The water containing the oligonucleotides was then spun through a Spin-X centrifuge filter unit to remove gel pieces and the oligonucleotide solutions were lyophilised. Finally each oligonucleotide was dissolved in 50µl of water and a 1µl aliquot was removed to determine the concentration and purity of the preparation by spectroscopy.

3.13 DNA sequencing

3.13.1 Sequencing and size determination of recombinant M13mp18 cDNAs

All single-stranded templates were sequenced using Sequenase. The relative sizes of the cDNA inserts were checked by generating prime-cut probes and running them side by side on a 6% denaturing polyacrylamide/urea gel (see Section 3.18.4). Details of the recombinant bacteriophage selected from laboratory stocks, or constructed from pBUN.127, and the probes synthesised from them are given in Table 3.1.

3.13.2 Single-stranded DNA sequencing (using Sequenase)

(i) Annealing reaction

For each set of four sequencing lanes one annealing reaction was set up as follows:

1µl Primer (provided in kit)

2µl Sequencing buffer (provided in kit)

 $2\mu g$ DNA (reaction volume made up to 10 μ l with H₂0)

Samples were heated to 65°C for two minutes and allowed to cool slowly to room temperature over a period of 30 minutes (using a beaker of water heated initially to 65°C).

(ii) Labelling reaction

Each labelling reaction was set up as follows:

10µl Template-primer

1µl 0.1M DTT

 2μ l Labelling mix (diluted five-fold with distilled H₂O)

0.5 μ l α -[³²P] dATP

 2μ l Sequenase enzyme (diluted 1:8 with ice cold TE)

The samples were mixed thoroughly and incubated at room temperature for 5 to 10 minutes.

(iii) Termination reaction

For each sequence determination four Eppendorf tubes were labelled G, A, T or C and 2.5 μ l of ddGTP, ddATP, ddTTP or ddCTP was added to the appropriately labelled tube. The chain terminating nucleotides were prewarmed at 37°C for at least 1 minute and 3.5 μ l of the above labelling incubation was transferred to each tube. The samples were mixed, spun down briefly in a microcentrifuge and incubated for a further 5 minutes at 37°C. Finally, 4 μ l of stop solution was added to each tube, the samples were mixed thoroughly and stored on ice until they were loaded onto the gel. Just prior to loading the samples were heated to 80°C for 2 minutes and 2 to 3 μ l of each sample was loaded onto an 8% sequencing gel (see Section 3.18.4).

3.14 Radiolabelling procedures

3.14.1 Prime cut probes

M13 Universal primer was annealed to recombinant M13 DNA in TM buffer for one hour at 60°C. One microlitre of a 0.5mM solution of dCTP, dTTP and dGTP was added along with 10 μ Ci of α -[³²P]-dATP (10mCi/ml) and 1 unit of large fragment DNA polymerase (Klenow); the sample was incubated at room temperature for 10 minutes. A chase mix consisting of 1 μ l of 4 x 0.5mM dNTP was added to the reaction and the sample was incubated at room temperature for 5 minutes. The large fragment DNA polymerase was then inactivated by heating at 70°C for 10 minutes. The appropriate restriction enzyme was added to the reaction mixture and the newly generated double-stranded DNA was digested with 10U of restriction enzyme for 90 minutes at the appropriate temperature. Labelled DNA fragments were purified by electrophoresis through a thin (0.3mm) denaturing 6% polyacrylamide/urea gel (see Section 3.18.4). Prior to loading the gel, 0.75µl of 0.5M EDTA and 25µl of formamide dyes were added to the DNA and each sample was boiled for 2 minutes. Each sample was loaded onto the gel and electrophoresed at 37 W for 4 hours or until the bromophenol blue band was 3/4 of the way down the gel. The probe was located by autoradiography, excised from the gel and the gel slice was broken into small pieces by centrifugation into a 1.5ml Eppendorf tube through a hole in the bottom of a 0.5ml Eppendorf tube. The gel pieces were transferred to a 15mlFalcon tube in 1ml of water and the probe was eluted into the water by shaking overnight at 37°C. The gel pieces were removed by centrifugation through a Spin-X centrifuge filter unit.

3.14.2 Nick translation

The labelling reaction, in a total volume of 50μ l, was set up using 100-200ng of template DNA to which the following were added:

- 5µl 10x NT buffer & dNTP
- 4µl 4 x [³²P]-dNTPs.
- 2μ l 10⁻⁴ dilution of DNase 1 (stock solution concentration = 1,000 U/ml).
- 1µl DNA polymerase 1 (10 U).

The reaction was incubated at room temperature for 30 minutes. Labelled DNA was separated from unincorporated nucleotides by chromatography on a 1ml column of Sephadex G-50. Ten microlitres of Orange-G was added to the sample prior to loading onto the column; this dye chromatographed with the unincorporated dNTPs. The sample was applied to the top of the column and the labelled DNA was eluted with water until the orange-G had almost reached the bottom of the column. The probe was denatured prior to use by boiling for 5 minutes.

3.14.3 Generation of single-stranded α -[³²P]-labelled cDNAs (using the Amersham 'rapid multiprime DNA labelling system')

Recombinant M13mp18 bacteriophage containing fragments of cDNA to the Bunyamwera virus S, M or L segment RNAs were used as templates to make strand specific probes. Single-stranded template DNA was stored at -20°C at a concentration of 25 μ g/ml in sterile distilled water. For each probe, 25ng of template DNA was denatured by heating to 95 to 100°C for two minutes followed by incubation on ice. The labelling reaction was set up on ice as follows:

- $4\mu l$ Each unlabelled dNTP (omitting ones to be used as label)
- 5µl Buffer provided in kit
- 5µl Primer provided in kit
- 5µl Combined volume of α -[³²P]-dNTPs (3000 Ci/mmol; 10mCi/mmol)
- 2µl Enzyme
- 15μl H₂O

The reaction mix was vortexed briefly and spun down in a microfuge for a few seconds before incubation at 37°C for 30 minutes. Unincorporated nucleotides were removed by gel filtration through a Sephadex G-50 mini column as described in the preceeding section on nick translation. Before use, the labelled DNA was denatured at 95 to 100°C for two minutes and chilled on ice. The probe can be stored at -20°C after the addition of EDTA to 20mM.

3.14.4 End-labelling of oligonucleotides with γ -[³²P]-dATP

The reaction was set up as follows:

- $1\mu l$ 0.1 $\mu g/\mu l$ oligonucleotide
- 4µl 5x kinase buffer
- 10µl γ -[³²P]-dATP (100µCi)
- 1.4µl T4 polynucleotide kinase (7U)
- 3.6 μ l H₂O

Each sample was incubated at 37° C for 30 minutes, followed by a 5 minute incubation at 65°C to inactivate the T4 polynucleotide kinase. The radiolabelled oligonucleotides were stored frozen at -20°C; 1µl (5ng of labelled oligonucleotide) was sufficient for one primer extension reaction.

3.14.5 Synthesis of Riboprobes

Riboprobes were synthesised from full-length L, M and S segment cDNAs cloned under the control of the bacteriophage T7 promoter (Jin and Elliott, 1991 and unpublished data), using T7 RNA polymerase (Promega) and 50μ Ci α -[³²P]-CTP in an *in vitro* transcription reaction as detailed below.

(i) Standard transcription protocol

The template DNA was linearised and the reaction was set up as follows:

- 1.0µg Linearised template DNA
- 4.0µl 5x transcription buffer
- 2.0µl 100mM DTT
- 20 U RNasin
- 1.0μl 10mM ATP
- 1.0µl 10mM GTP
- 1.0µl 10mM UTP

2.4µl 100µM CTP

5.0µl α -[³²P]-CTP (50µCi at 10mCi/ml)

1.0µl T7 RNA polymerase (at 15-20 Units/µl)

The reaction volume was adjusted to 20μ l with water and the samples were incubated at 37 to 40° C for 60 to 120 minutes.

(ii) Digestion of the DNA template and removal of unincorporated nucleotides following transcription.

After the transcription reaction, RQ1 RNase-free DNAse was added to a concentration of 1 unit/ μ g of template DNA and the sample was incubated for 15 minutes at 37°C. The radiolabelled RNA was phenol/chloroform extracted and recovered by ethanol precipitation in the presence of 1M NH₄OAc. For the precipitation step each sample was mixed thoroughly with 1/4 volume of 5M NH₄OAc and 3 volumes of ethanol and placed at -20°C for 30 minutes. The RNA was pelleted by centrifugation at 12,000 x g for 30 minutes and the supernatant was decanted. The RNA was resuspended in 50 μ l of water which was then made 1M with respect to NH₄OAc and a second precipitation step was carried out as described above. The pellet was dried under vacuum and resuspended in TE buffer for storage at -70°C.

3.14.6 Pulse-labelling with [³⁵S]-methionine

Protein synthesis in infected cells was monitored by labelling the cells with [³⁵S]methionine during a one hour pulse. Confluent monolayers of *Aedes albopictus* C6/36 or BHK cells, in 35mm diameter tissue culture plates, were infected at a moi of 5 pfu/cell with Bunyamwera virus and incubated at 28°C and 31°C respectively. Uninfected controls were included for both cell types. At designated time points post infection the culture medium from infected cell monolayers was replaced with 500µl of PBS containing 50µCi of [³⁵S]methionine and incubation was continued for one hour. The labelling solution was removed and the monolayers were washed twice with PBS. The cells were scraped into 80µl of protein dissociation buffer and stored at -20°C prior to SDS-PAGE. The samples were boiled for 2 minutes and spun briefly in a microcentrifuge before loading 1/3rd of each sample onto a polyacrylamide gel.

3.15 RNA isolation

3.15.1 Preparation of virion RNA

Culture fluid from Bunyamwera virus infected BHK cells was harvested and clarified by centrifugation at 3,000 rpm, for 10 minutes, in an MSE Coolspin centrifuge, maintained at 4°C. The clarified supernatant was centrifuged at 35,000 rpm for 3 hours in a Sorvall TST41 rotor, again at 4°C. Pelleted virus was resuspended in lysis buffer comprising 50mM Tris HCl, pH8, 10mM NaCl and 0.6% NP40, and left for one hour at 4°C. The lysed virus suspension was centrifuged at 6,500 rpm in a Sorvall SS34 rotor, the

supernatant was harvested and EDTA was added to a concentration of 6mM. This suspension was layered onto a preformed CsCl gradient overlaid with 2.5ml of 5% sucrose in 50mM NaCl, 25mM Tris and 2mM EDTA. The CsCl gradient consisted of 6ml of 20-40% (w/w) CsCl in 50mM NaCl, 25mM Tris and 2mm EDTA. The loaded gradients were centrifuged at 35,000 rpm at 10°C in a Sorvall TST-41 rotor for 15 hours. The viral nucleocapsids formed a sharp band in the middle of the CsCl gradient and were harvested through the side of the centrifuge tube using a sterile syringe and needle. The nucleocapsid material was diluted in 1x NTE and concentrated by centrifugation at 39,000 rpm for 2 hours at 4° C in a Sorvall TST-41 rotor. RNA was extracted from the pelleted nucleocapsids using the acid-phenol-guanidinium thiocyanate protocol of Chomczynski & Sacchi (1987), as outlined below.

3.15.2 Preparation of fractionated infected cell RNA

Viral nucleocapsids were separated from the remainder of the infected cell RNA as described by Leppert et al., (1979) and Rossier et al., (1988). Infected cell monolayers were washed twice with PBS and scraped into PBS using a rubber policeman. The cells were pelleted by centrifugation at 3,000 rpm in an MSE Coolspin centrifuge, resuspended in PBS and repelleted. The cells were lysed on ice for 15 minutes in buffer containing 150mM NaCl, 50mM Tris.HCl pH 7.5 and 0.6% NP40. The cells were further disrupted by vortexing for 2 minutes. Nuclei and cell debris (membranes) were removed from the suspension by centrifugation at 8,000 rpm for 5 minutes at 4°C in a Sorvall SS34 rotor. The cytoplasmic extract was made 6 mM with respect to EDTA and centrifuged through a preformed 20-40% (w/w) CsCl gradient overlaid with 5% sucrose, as described in section 3.15.1. Bunyamwera virus nucleocapsids appeared as a sharp band in the centre of the gradient and the remainder of the infected cell RNA pelleted. The pelleted RNA was used as the source of viral mRNA after a further extraction was performed using the acid-phenolguanidinium thiocyanate protocol of Chomczynski and Sacchi (1987). The nucleocapsids were harvested using a syringe, diluted with 1x NTE buffer (100mM NaCl, 10mM Tris.HCl (pH 7.5), 1mM EDTA) and concentrated by centrifugation at 39,000 rpm for 2 hours at 4°C in a Sorvall TST-41 rotor. RNA was extracted from the purified nucleocapsids using the acid-phenol-guanidinium thiocyanate protocol of Chomczynski and Sacchi (1987).

3.15.3 Preparation of total cell RNA

Total cellular RNA was extracted by the acid-phenol-guanidinium thiocyanate protocol of Chomczynski and Sacchi (1987). The reagent volumes specified below were sufficient to harvest RNA from 1 x 10^6 cells and were scaled up as appropriate. Infected cell monolayers were washed twice with PBS before lysis with 100μ l of GSCN buffer (see Section 3.7.2). The lysed cells were transferred to an Eppendorf tube to which 10μ l of 2M

NaOAc pH 4.0, 100µl of water-equilibrated phenol, and 20µl of a 49:1 mixture of chloroform/amyl alcohol were added with thorough mixing after the addition of each reagent. Each sample was then vortexed for 10 seconds and left on ice for 15 minutes. The aqueous and organic phases were separated by centrifugation at $10,000 \times g$ for for 15 minutes at 4°C. The upper aqueous phase was harvested and mixed with 100µl of isopropanol and the samples were incubated at -20°C for one hour. The RNA was pelleted by centrifugation at 10,000 x g for 15 minutes at 4°C. The pellet was dissolved in 300µl of GSCN buffer and mixed with 300µl of isopropanol. The samples were incubated at -20°C for one hour and centrifuged at 10,000 x g for 15 minutes. The pelleted RNA was dissolved in water (brief heating to 60°C was sometimes necessary to ensure that the RNA had dissolved completely). An equal volume of a 4:1 mixture of chloroform:butan-1-ol was added and the sample was vortexed to an emulsion and then centrifuged briefly at 10,000 x g to separate the phases. The upper aqueous phase was harvested and the organic phase was back extracted with an equal volume of water. Finally the RNA was recovered from the pooled aqueous phases by precipitation with 1/10 volume of 3M NaOAc pH 5.5 and 2.5 volumes of ice-cold ethanol (see Section 3.12.5).

3.16 RNA manipulations

3.16.1 Slot blot analysis

Each RNA sample was mixed with 100µl of 20x SSC and applied under vacuum to nitrocellulose using a Schleicher and Schuell Minifold II. The nitrocellulose (Schleicher and Schuell) and two pieces of filter paper were pretreated by emersion first in sterile distilled water for 5 minutes and then in 10x SSC for 5 minutes. The manifold was assembled as per the manufacturer's instructions. After application of the RNA, the nitrocellulose was baked at 80°C under vacuum for two hours.

3.16.2 Northern blot analysis

Electrophoresis was performed in 1.0, 1.5, or 1.8% agarose gels containing 5mM methylmercury hydroxide (see Section 3.18.3). Non-submarine gels were electrophoresed in 1x EM buffer, for five hours at 75V, with recirculation of the tank buffer (Perbal, 1988). Transfer of RNA to nitrocellulose was by capillary transfer in 20x SSC, without pretreatment of the gel. Prior to transfer, the nitrocellulose (which had been cut so that it was 1mm longer and wider than the gel) was soaked in sterile distilled water for 5 minutes and then in 10x SSC for 5 minutes. The gel was placed on a sheet of 3MM paper which acted as a buffer bridge (having been wet through with 20x SSC and arranged on a glass plate support spanning a tray containing 20x SSC). The ends of the 3MM paper were immersed in the 20x SSC. The nitrocellulose was placed on top of the gel so that it just overlapped the sides of the gel but was not in danger of contacting the filter paper underneath (which would have short circuited the flow of buffer through the gel). Care

was taken to avoid air bubbles either underneath the 3MM paper or between the paper and the gel, or the gel and the nitrocellulose; any air bubbles were removed by rolling a 10ml pipette over the surface, applying gentle pressure. Two pieces of filter paper which had been cut just smaller than the nitrocellulose and had been presoaked as per the nitrocellulose were placed on top, followed by an 8cm stack of absorbant paper towels. A 500g weight was placed on top of the blot to aid capillary transfer and the blot was left overnight. Finally the nitrocellulose was baked at 80°C in a vacuum oven for two hours.

3.16.3 Hybridization conditions

Prehybridization and hybridization were carried out at 42°C for 4 and 16 hours respectively. The buffer conditions used with particular probes are set out in Table 3.4.

Buffer constituents	Riboprobes		Prime-cut or rapid multiprime generated DNA probes		
	Prehybridization	Hybridization	Prehybridization	Hybridization	
SSPE	6х	бх	5x	5x	
Formamide	50%	50%	50%	50%	
Low fat milk			0.5%	0.1%	
Carrier DNA	0.05 mg/ml	0.05mg/ml	0.1 mg/ml	0.1 mg/ml	
Yeast tRNA	0.05 mg/ml	0.05 mg/ml			
SDS	0.1%	0.1%			
Denhardts	5x	1x			

Table 3.4 Recipes for prehybridization and hybridization buffers.

3.16.4 Post hybridization washes

After hybridization the filters were washed several times at room temperature. Filters which had been probed for viral RNA were washed twice in 2x SSC, 0.5% SDS and twice at higher stringency (0.1x SSC, 0.1% SDS). Filters probed with the actin specific probe were washed twice at the lower stringency only.

3.16.5 In vitro translation

In vitro translation reactions were carried out with New England Nuclear (NEN) rabbit reticulocyte lysate. Each reaction contained 2µg of total infected cell RNA or the cell equivalent of nucleocapsid derived RNA, adjusted to 2µl total volume. For each set of reactions a premix was prepared:-

Per reaction:

5 μ l [³⁵S]-methionine.

- 5.5µl translation cocktail
- 2.0µl 1M potassium acetate
- 0.5µl 32.5mM magnesium acetate

Separate reactions were then set up as follows:

- 10µl lysate
- 13µl premix
- 2µl RNA/H₂O

The samples were incubated at 37° C for 90 minutes and then 25μ l of protein dissociation buffer was added to each. The *in vitro* translation products were analysed by SDS-PAGE.

3.17 Direct RNA sequencing

Direct RNA sequencing was performed as described by Geliebter et al. (1986). This method is based on enzymatic extension of $[^{32}P]$ - end-labelled oligonucleotide primers by reverse transcriptase, in the presence of dideoxynucleoside triphosphates, which function as chain terminators during the reverse transcription step. Direct RNA sequencing yields the consensus sequence of the RNA population under study. Both virion RNA and total infected cell RNA were used as templates for sequencing.

(i) Annealing reaction:

Each annealing reaction was set up:-

1µl (5ng) [³²P]-labelled oligonucleotide primer

10µg total infected cell RNA or 1µg of virion RNA

2μl 5x annealing buffer

The reaction volume was adjusted to 10µl with water and the nucleic acid was denatured by heating to 80°C for 3 minutes. Radiolabelled oligonucleotides were annealed to the template RNA by incubation for one hour at the annealing temperature (T°C), defined as five degrees below the denaturation temperature for the oligonucleotide. Denaturation temperature is determined by nucleotide composition and T°C was calculated as shown below:

 $T(^{o}C) = 4(G+C) + 2(A+T) - 5^{o}C$

(ii) Primer extension reaction

Primer extension reactions were set up in quadruplicate (one reaction for each chain terminator) as follows:-

- 2.0µl annealed RNA-Primer
- 1.0µl 5x transcription buffer
- 3 U reverse transcriptase
- 0.5µl ddNTP (either 1.5mM ddATP, ddCTP or ddGTP or 3.0mM ddTTP)

The samples were incubated at 50°C for one hour, then 2μ l of formamide dyes were added to each. The samples were boiled for 3 minutes and half of each sample was loaded onto an 8% sequencing gel (see Section 3.18.4).

3.18 Electrophoresis 3.18.1 SDS-PAGE

	15% Resolving Gel	3% Stacking Gel
Acrylamide: Bis (75:1) [37.5 and 0.5% (w/v) stock]	24.9ml	1.0ml
1M Tris HCl pH 8.8	24.0ml	
1M Tris HCl pH 6.8		625µl
H ₂ O	14.7ml	8.0ml
10% SDS	650µl	100µl
10% Ammonium persulphate	650µl	100µl
TEMED	60µl	5µl

Table 3.5Recipe for 15% resolving and 3% stacking portions
of SDS-PAGE gels

SDS-PAGE gels were poured between two clean 20 x 22cm glass plates separated by 1mm thick spacers on either side. The gel plates were sealed with Kapton tape. The 15% resolving gel mix was prepared as shown in Table 3.5, adding the ammonium persulphate and TEMED last. Using a 10ml glass pipette the resolving gel was poured to a height approximately two sample depths below the bottom of the sample wells, taking care to avoid air bubbles. A small volume of butan-1-ol was applied to the top of the gel to ensure a level interface with the stacking gel and to aid polymerization by excluding air. The gel was left to set for 15 minutes. Meanwhile the 3% stacking gel mix was prepared as indicated in Table 3.5, the butan-1-ol was poured off the set stacking gel and the top of the gel was rinsed with water. The stacking gel was poured to the top of the gel plates and the gel comb was inserted to form the sample wells. The gel was wrapped in damp tissues, covered in cling film and left to set a few degrees from the horizontal. Best results were obtained if the gels were left at 4°C overnight prior to electrophoresis. Electrophoresis was performed in 1x Tris/glycine buffer containing 0.1% SDS. The gel comb was removed, the sample wells were rinsed with electrophoresis buffer and the gel was preelectrophoresed for 30 minutes at a constant current of 60mA. Samples were boiled for two minutes and centrifuged briefly in a microcentrifuge prior to loading onto the gel.

Electrophoresis was performed at 4°C at a constant current of 60mA for approximately 3.5 hours (until the bromophenol blue in the sample buffer had migrated to between 2 and 3cm from the bottom of the gel).

3.18.2 Agarose gels for electrophoresis of DNA

A molten solution of 1% agarose in 1x TBE (25ml/gel) was left to cool slightly and was poured to form a horizontal slab gel in a BRL Horizon-58 electrophoresis tank, the gel comb was put in place and the gel was left to set for 5 minutes. The gel was emersed in 1x TBE running buffer containing 0.5μ g/ml ethidium bromide to enable visualisation of the DNA under UV illumination after electrophoresis. Samples containing 1/10th volume of glycerol dyes were loaded onto the gel and electrophoresis was carried out at 80 to 100V for 30 minutes, or until the bromophenol blue was three quaters of the way down the gel.

	Gel Concentration		
	1%	1.5%	1.8%
Agarose	0.8g	1.2g	1.44g
1x EM buffer	80m1	80ml	80ml
1M methylmercury hydroxide	400µ1	400µl	400µl

3.18.3 Agarose gels containing 5mM methylmercury hydroxide

Table 3.6 Composition of the agarose gelscontaining methylmercury hydroxide,used for electrophoresis of viral RNAs.

The appropriate quantity of agarose was melted in 1x EM buffer as indicated in Table 3.6. The molten agarose was cooled to 60-62°C before the addition of 400µl of 1M methylmercury hydroxide; every step after and including the addition of methylmercury hydroxide was performed in a fumehood. Horizontal gels were poured in a BRL Horizon 11.14 electrophoresis tank. Each sample was mixed with an equal volume of 2x sample buffer and 1µl of 0.1M methylmercury hydroxide, vortexed briefly and centrifuged for 1 second at 12,000 rpm in a microfuge. The samples were loaded quickly onto the gel to avoid diffusion and nonsubmarine gels were run in 1x EM buffer at 75 V. A peristaltic pump was used to recirculate the buffer between the two reservoirs of the electrophoresis tank.

	Ge	Gel Concentration	
	6%	8%	16%
Urea	288g	288g	288g
Acrylamide	34.2g	45.6g	91.2g
Bisacrylamide	1.8g	2.4g	4.8g

3.18.4 Polyacrylamide/urea denaturing gels

Table 3.7 Sequencing gel mixes

Each of the above was dissolved in sterile distilled water to a total volume of 500ml. 20g of Amberlite was added and the solution was stirred for 30 minutes. The gel solution was filtered, 60ml of 10x TBE was added and the volume was adjusted to 600ml. The sequencing gel mixes were stored at 4°C.

These gel mixes were used in the preparation of 0.3mm thick denaturing polyacrylamide/urea gels required for sequence determinations (8% mix); gel purification of 'prime-cut' probes (6% mix) and gel purification of oligonucleotides (16% mix). In each case the gels were prepared with the same relative proportions of sequencing mix, 10% ammonium persulphate and TEMED (100:1:0.1). The total volume of gel required was generally 60ml for full-size sequencing gels while 25ml was sufficent to prepare gels for the purification of oligonucleotides and probes. Pre-electrophoresis of the gels was carried out at 37W for 30 minutes. The sample wells were washed out with electrophoresis buffer (1x TBE) to remove any local concentration of urea which would prevent the samples from settling in the wells. Formamide dyes were added to each sample (1/10th of the sample volume) and the sequencing and 'prime-cut' probe samples were boiled for 2 minutes prior to loading onto the gel. Electrophoresis was carried out at 60W for sequencing gels and at 37W for the smaller denaturing gels for 2-3 hours, or until the bromophenol blue had travelled most of the way down the gel. For direct RNA sequence analysis longer electrophoretic runs were regularly performed inorder to obtain the maximum amount of sequence information from a given primer.

3.18.5 Treatment of gels after electrophoresis

SDS-PAGE gels of [³⁵S]-methionine labelled proteins were fixed overnight. For gels of proteins translated *in vitro* the gel fix solution was changed several times during the first hour to reduce background due to residual free [³⁵S]-methionine in the gel. After fixation, all [³⁵S]-methionine labelled gels were soaked in En³Hance at room temperature with shaking for 50 minutes, and then washed in water for a further 30 minutes. SDS-PAGE gels were dried under vacuum in a Bio-Rad gel drier set at 80°C for 2.5 hours.

Polyacrylamide/urea gels were transferred directly to 3MM paper. Immediately after electrophoresis one of the glass plates was removed and a single sheet of 3MM paper was pressed firmly onto the exposed gel. The paper was slowly peeled away from the plate with the gel attached. The gel was dried onto the 3MM paper under vacuum in a Bio-Rad gel drier set at 80°C for 1 hour.

3.19 Autoradiography and fluorography

Gels of [³²P]-labelled material were exposed against Kodak X-Omat film, at room temperature or at -70°C if an intensifying screen was used to enhance the signal. [³⁵S]-methionine labelled gels which had been treated with En³Hance (fluorography) were exposed against Kodak X-Omat film at -70°C.

3.20 Densitometric analysis

Densitometric analysis of autoradiographic images was carried out using a GS 300 Transmittance/Reflectance scanning densitometer (Hoefer Scientific Instruments, San Francisco). The relative intensities of individual autoradiographic signals were calculated using the GS 360 data system (IBM).

4 Bunyamwera virus (wtL9BUN): plaquepurification and determination of the consensus population sequence of the genomic S RNA segment

4.1 Production of a triply plaque-purified elite stock of Bunyamwera virus

A triply plaque-purified stock of Bunyamwera virus was prepared from a laboratory stock of virus to provide a pedigreed stock for all future experiments. The laboratory stock was originally plaque-purified in BSC-1 cells from a suckling mouse brain homogenate, and subsequently passaged a number of times in BHK cells (Watret *et al.*, 1985; see Section 3.1.2). Some heterogeneity in plaque size was evident within the laboratory stock of Bunyamwera virus prior to plaque-purification and one of the larger plaques (designated wtL9BUN) was selected. After three successive rounds of plaque-purification, an elite stock of wtL9BUN virus was prepared and titrated. BHK cell monolayers were used for the plaque-purification steps, for growing the elite stock of virus and for plaque-titration of the elite stock. The titre of the elite stock was 2×10^7 pfu/ml and working stocks of wtL9BUN virus were prepared from the elite stock by a single low multiplicity passage in BHK cells (see Section 3.10.4).

4.2 Direct RNA sequence determination of the small (S) genomic RNA segment of wtL9BUN virus

To determine how representative the cDNA derived sequence for the small (S) genomic RNA segment of Bunyamwera virus (Elliott, 1989b) was of the consensus population sequence, the genomic S RNA segment of wtLoBUN virus was sequenced directly, by primer extension of γ -[³²P]-labelled oligonucleotides in the presence of dideoxynucleoside triphosphates (see Section 3.17). On the basis of the cDNA derived sequence, six 20mer oligonucleotides complementary to regions of Bunyamwera virus genomic S RNA were synthesised for use as primers (see Figure 4.1). The priming sites were spaced at intervals of between 130 and 150 nucleotides along the S RNA, and up to 200 bases of sequence could be read from each primer. Nucleotide positions are numbered with respect to the positive-sense RNA, thus corresponding directly to the numbering of the cDNA (Elliott, 1989b). The consensus population sequence agreed with the cDNA derived sequence although the detection of adenosine alongside uridine at position 923 within the 3' non-coding region of the positive-sense S RNA suggested that there might be some heterogeneity within the virus preparation (see Figure 4.2). Since this nucleotide substitution was within the 3' non-coding region and outside the conserved complementary terminal sequences of Bunyamwera virus S RNA (Elliott, 1990) it was unlikely to be of consequence. Its presence does however illustrate the difficulty in obtaining completely

1	AGTAGTGTACTCCACACTACAAACTTGCTATTGTTGAAAATCGCT	45
46	GTGCTATTAAATCCCAACAGAAGGTCATTAAAGGCTCTTTAATGAT	90
91	TGAGTTGGAATTTCATGATGTCGCTGCTAACACCAGCAGTACTTT	135
136	TGACCCAGAGGTCGCATACGCTAACTTTAAGCGTGTCCACACCAC	180
181	TGGGCTTAGTTATGACCACATACGAATC <u>TTCTACATTAAAGGACG</u>	225
226	CGA GATTAAAACTAGTCTCGCAAAAAGAAGTGAATGGGAAGTTAC	270
271	ACTTAACCTTGGGGGGCTGGAAGATTACTGTATATAATACGAATTT	315
316	${\tt TCCTGGCAACCGGAACAACCCAGTTCCTGACGATGGTCTTACC{\tt CT}$	360
361	CCACCGCCTCAGTGGATTCCTTGCCAGGTACCTACTTGAGAAGAT	405
406	GCTGAAAGTCAGTGAACCAGAGAAATTGATTATTAAATCAAAAAT	450
451	AATCAACCCTTTGGCTGAAAAGAATGGGATCACTTGGAATGATGG	495
496	AGAGGAAGTTTAT <u>CTCTCTTTCCCAGGATC</u> AGAGATGTTCTT	540
541	AGGAACTTTCAGATTCTACCCCTTAGCAATCGGGATCTACAAAGT	585
586	TCAGCGCAAGGAAATGGAACCAAAATACCTTGAGAAAACAATGCG	630
631	GCAGAGGTACATGGGACTAGAAGCAGCAACTTGGACTGTTAGTAA	675
676	ATT GACAGAAGTTCAGTCTGCAC TGACAGTTGTCTCTAGCTTAGG	720
721	TTGGAAGAAAACCAATGTTAGTGCAGCTGCCAGGGACTTCCTTGC	765
766	TAAATTCGGAATCAACATGTAAGCAGGGATGCATTTTTAATCGGG	810
811	CTAAAGTCATCTGTTTTAATTTGGCTAAA AGGGTTGTTTCAACCC	855
856	ACAA	900
901	GCGGGCTAAATCAACATTATATTGTTAATGGTATTTTAAGTTTTA	945
946	GGTGGAGCACACTACT	961

Figure 4.1. The cDNA sequence of the S RNA segment of Bunyamwera virus presented as the complementary positive RNA strand, written as DNA (from Elliott, 1989b). The positions of the oligonucleotides used for direct RNA sequencing are shown in **bold** and <u>underlined</u>.



Figure 4.2. Direct RNA sequencing of the genomic S RNA segment of wtL9BUN virus by primer extension from oligonucleotide number 6. The sequence shown is from nucleotide 908 to 939. The sequence from nucleotide 917 to 929 is highlighed, with the nucleotide heterogeneity at position 923 shown in **bold**.

homogeneous preparations of RNA viruses due to the inherent high mutation rate associated with their encoded RNA polymerases (see Section 2.2.1).

In the course of determining the cDNA sequence of Bunyamwera virus genomic S RNA, discrepancies were encountered between the sequence of the largest cDNA clone (pBUN S14) and three other cDNA clones (pBUN308, pBUN309 and pBUN3/59) at nucleotide positions 378 and 379 (Elliott, 1989b; see Figure 4.3). The pBUNS14 cDNA contained guanosine and thymidine at nucleotide positions 378 and 379, which created codons for cysteine at amino acid position 98 of the N protein and valine at amino acid position 92 of the NSs protein respectively. Earlier radiolabelling studies on Bunyamwera virus-infected cells had indicated that the N protein was deficient in cysteine residues (Elliott, 1985), and four overlapping cDNAs (pBUN308, pBUN309 and pBUN3/59) contained thymidine at nucleotide position 378 and cytidine at nucleotide position 379, coding for phenylalanine and serine at amino acid positions 98 and 92 in Bunyamwera virus N and NSs proteins respectively. Derivation of the consensus population sequence of wtL9BUN virus S RNA, by primer extension sequencing on virion RNA, confirmed the presence of uridine (thymidine in the cDNA) and cytidine at nucleotide positions 378 and 379, respectively (see Figure 4.4).

V	D	S	L	P	NS
S	G	F	L	A	N
AGT	GGA	TTC	CTT	GCC	
371-				>385	
AGT	GGA	TGT	CTT	GCC	
S	G	С	L	A	N
v	D	v	L	P	NS

Figure 4.3 Differences between the sequences of the cDNA clone pBUN S14 and the cDNA clones pBUN308, pBUN309 and pBUN3/59 used to sequence Bunyamwera virus S segment RNA (Elliott, 1989b). -----> represents positive-sense Bunyamwera virus S RNA, nucleotides 371-385; the upper half of the figure shows the nucleotide sequence and the deduced amino acid sequences obtained from clones pBUN3/59, pBUN308 and pBUN309; the lower half of the figure shows the corresponding sequences obtained from the cDNA clone pBUN S14 (figure adapted from Elliott, 1989).



Figure 4.4. The consensus population sequence of the genomic S RNA segment of wtL9BUN virus, from nucleotide 367 to 393. The sequence between nucleotide 374 and 385 is highlighted with the T at position 378 and C at position 379 shown in **bold**.

4.3 Summary

Determination of the consensus sequence of the genomic S RNA segment of the wtL9BUN virus preparation confirmed that this triply plaque-purified stock of Bunyamwera virus was homogeneous. The only sequence heterogeneity detected was at nucleotide position 923 within the 3' non-coding region of the viral complementary-sense S RNA. The consensus sequence of the genomic S RNA segment of the wtL9BUN virus preparation was otherwise identical to the published cDNA derived sequence (Elliott, 1989b). A discrepancy between the sequence of the cDNA clone pBUNS14 and three other cDNA clones from which the Bunyamwera virus S sequence was derived, was resolved by primer extension sequencing on virion RNA. It appeared that pBUNS14 may have represented a minor variant in the RNA population, or may have arisen due to a reverse transcriptase error during cDNA synthesis.

5

Comparative study of macromolecular syntheses in Bunyamwera virus infected *Aedes albopictus* C6/36 and BHK cells

The patterns of viral protein and RNA syntheses leading to cell death in Bunyamwera virus-infected BHK cells, and to inapparent, persistent infection in *Aedes albopictus* C6/36 (C6/36) cells were compared by SDS-PAGE analysis of [³⁵S]-methionine pulse-labelled infected cell extracts and Northern analysis of infected cell RNA respectively. Infections were initiated with 5 pfu/cell, using the same stock of virus (wtL9BUN virus), and viral protein synthesis and RNA accumulation were followed for 72 hours post infection (h pi).

5.1 Protein synthesis in Bunyamwera virus-infected BHK and C6/36 cells

Infected BHK and C6/36 cells were pulse-labelled for one hour with [³⁵S]methionine at intervals over 72 h post infection. The pattern of protein synthesis observed in each infected cell type was strikingly different (Figure 5.1). In BHK cells host protein synthesis diminished drastically between 12 and 16 h pi such that, by 20 h pi the viral L, G1, N and NSs proteins (first detected 6 to 8 h pi) were virtually the only proteins detected. Between 16 and 24 h pi viral protein synthesis continued at high levels in the absence of significant host protein synthesis. By 36 h pi only very low levels of viral protein synthesis were detectable during a one hour pulse with [³⁵S]-methionine. The G2 and NSm proteins, which are not reproducibly detected by SDS-PAGE (Elliott, 1985), were not identified in this experiment. [It has subsequently been discovered that G2 is heat-labile and can be detected by not boiling the Bunyamwera virus-infected cell extracts prior to loading SDS-PAGE gels (R.M. Elliott, personal communication)].

In C6/36 cells there was little decrease in host protein synthesis, and only G1, N and NSs could be identified above host background. Overall, Bunyamwera virus infection proceeded more slowly in C6/36 cells. G1 was detected between 16 and 36 h pi, N between 8 and 24 h pi (weakly at 36 h pi) and NSs between 12 and 24 h pi. While G1 and N were detectable at comparable levels in both cell types, the level of NSs was much lower in C6/36 cells relative to BHK cells. No viral proteins were detected after 36 h pi in the C6/36 cells and host protein synthesis continued at a slightly reduced rate. Thus, against a background of continuing host protein synthesis, Bunyamwera virus protein synthesis was down-regulated. Although comigrating host bands made it difficult to determine the precise onset of synthesis of L and G1, the synthesis of N and NSs did appear to preceed that of G1, particularly in C6/36 cells. This illustrates a degree of temporal control of bunyavirus induced polypeptide synthesis as first reported by Pennington *et al.* (1977).


Figure 5.1. Protein synthesis in Bunyamwera virus-infected (a) BHK and (b) C6/36 cells. Cells were infected with 5 pfu/cell of Bunyamwera virus, and the infected cultures were pulse-labelled for 1 h with [35S]-methionine at the times indicated (h pi). MI indicates mock infected cell lysate. Cell lysates were fractionated on 15% SDS-PAGE gels. The positions of the viral proteins L, G1, N and NSs are indicated.

(a)

5.2 S segment mRNA levels in infected C6/36 cells

Slot-blot hybridization was employed to measure the steady-state levels of Bunyamwera virus S mRNA (BUN S mRNA) in infected C6/36 cells, over 72 h pi, to determine if the decline in viral protein synthesis observed in these cells after 24 h pi was due to altered levels of viral mRNAs. Bunyavirus-infected cells contain both encapsidated and unencapsidated viral RNA species. The former are the viral nucleocapsids in which, usually, full-length genomic and antigenomic RNAs are complexed with the viral N protein, while the latter consist of the viral mRNAs. Because viral mRNA and antigenomic RNA are the same sense, they must be separated before mRNA levels can be determined. These populations are readily separated (fractionated) on the basis of their different buoyant densities in CsCl gradients (Leppert *et al.*, 1979; Rossier *et al.*, 1988; see Section 3.15.2). Encapsidated viral RNA species form a sharp band in the middle of the gradient while the remainder of the infected-cell RNA including the viral mRNAs pellet.

Slot blot analysis of the pellet RNA fraction from Bunyamwera virus-infected C6/36 cells (Figure 5.2) revealed that significant levels of Bunyamwera virus S mRNA were present in the infected C6/36 cells beyond 36 h pi, when N protein synthesis could no longer be detected by [35 S]-methionine pulse-labelling *in vivo* (Figure 5.1). Maximum levels of S mRNA were detected 24 to 48 h pi in the C6/36 cells by slot-blot analysis and significant levels were detected up to the last time point tested (72 h pi; Figure 5.2). The presence of Aintact S mRNA, up to 72 h pi in infected C6/36 cells, was confirmed by Northern blot analysis of pellet RNA extracted during a separate experiment (Figure 5.3).

In order to preserve the nucleocapsid structures so that encapsidated and unencapsidated viral RNA species could be isolated separately, cytoplasmic extracts were prepared using a relatively mild procedure based on detergent disruption of infected cells. Examination of a number of Northern blots of pelleted RNAs from both infected cell types revealed that the viral mRNA species, particularly the infected C6/36 cell derived viral mRNAs were susceptible to nuclease degradation during this extraction procedure. The greater susceptibility of C6/36 cell derived viral mRNAs to degradation suggested that C6/36 cells may be richer in ribonucleases, or that the viral mRNAs synthesised in the two cell types differ in their susceptibility to nuclease degradation, possible due to the nature of their host derived primers. The relatively poor detection of C6/36 cell derived BUN S mRNA by Northern analysis, compared to the strong positive signals obtained by slot-blot hybridization (using the same probe; see Figures 5.2 and 5.3), may be explained, at least partly, by the fact that slot-blot hybridization does not rely on the complete integrity of the target RNA for its detection. This may also have contributed to the different patterns of RNA accumulation observed by the slot blot and Northern blot analyses detailed above (Figures 5.2 and 5.3 respectively); although some variation between two separate timecourse RNA extractions might be expected.



Figure 5.2. Slot-blot analysis of the pellet RNA fraction from Bunyamwera virus infected C6/36 cells. $12\mu g$ of CsCl pellet RNA, extracted at the indicated h pi, was immobilized onto nitrocellulose and hybridized with a [³²P]-labelled cDNA probe (S9) specific for positive-sense Bunyamwera virus S RNA.





Figure 5.3. Northern blot analysis of the pellet RNA fraction from Bunyamwera virus-infected C6/36 cells. $20\mu g$ of CsCl pellet RNA, extracted at the indicated h pi, was fractionated in a 1% agarose gel containing 5mM methylmercury hydroxide and blotted to a nitrocellulose filter. The filter was hybridized with a [³²P]-labelled cDNA probe (S9) specific for positive-sense Bunyamwera virus S RNA.

Assuming that a similar ratio of antigenomic S RNA to S mRNA exists in Bunyamwera virus infected BHK cells.

.

:

i.

It was clear, from the slot blot and Northern blot analyses, that the decline in virus protein synthesis observed between 24 and 36 h pi in the C6/36 cells was not paralleled by a decline in S mRNA levels. Interestingly, these observations are similar to those reported by Rossier *et al.* (1988) who detected maximum levels of La Crosse bunyavirus S mRNA in infected C6/36 cells at 24 h pi; although levels of S mRNA remained relatively high up to 80 h pi, N protein synthesis could not be detected after 24 h pi by pulse-labelling of infected cells (Rossier *et al.*, 1988).

5.3 In vitro translation

Total cell RNA, prepared from Bunyamwera virus-infected BHK and C6/36 cultures at time points up to 72 h pi, was used to programme *in vitro* translation reactions (see Section 3.16.5 and Figure 5.4). Maximum levels of translation were achieved with RNA extracted from BHK cells at 12 h pi and from C6/36 cells at 24 h pi, reflecting the slower progress of Bunyamwera virus infection in C6/36 cells. Virtually no host cell protein translation was obtained with RNA samples prepared from infected BHK cells after 12 h pi. The decrease in host cell protein translation *in vitro* paralleled the decline in host cell protein synthesis observed *in vivo* (see Figure 5.1a) and can be attributed to virus-induced mRNA instability in these cells (Raju & Kolakofsky, 1988).

Fractionated RNA preparations from Bunyamwera virus-infected BHK cells were also used to programme in vitro translation reactions (Figure 5.5). High levels of translation were achieved in vitro with CsCl pellet RNA extracted from BHK-21 cells up to 24 h pi, but no translatable free S mRNA was evident in the 48 and 72 h pi samples (Figure 5.5a). At all time points tested low levels of translation were observed from nucleocapsid derived RNA from Bunyamwera virus-infected BHK cells (Figure 5.5b). Raju & Kolakofsky (1987b) reported that approximately 2% of La Crosse bunyavirus mRNAs were encapsidated in infected BHK cells, and Bunyamwera virus S mRNA could be detected as early as 12 h pi amongst the encapsidated RNA population in infected BHK cells by Northern blot analysis (Figure 5.6). The ratio of La Crosse virus S mRNA to antigenomic S RNA in infected BHK cells can be estimated from available data to be 35:1 (Rossier et al., 1986; 1988). *Comparison of the levels of antigenomic S RNA and S mRNA in the banded fractions displayed in Figure 5.6 suggests that a similar proportion of Bunyamwera virus mRNAs was encapsidated during infection of BHK cells. The ratio of S mRNA to antigenomic S RNA in the encapsidated RNA fraction from Bunyamwera virus infected BHK cells remained constant up to 72 h pi. It is probable therefore that the low levels of translation observed in vitro with nucleocapsid-derived RNA extracted from Bunyamwera virus BHK cells (Figure 5.5b) resulted from translation of the small proportion of viral messages which were encapsidated in these cells.

A slot blot hybridization was performed to estimate the levels of positive-sense S RNA in each of the infected-BHK cell samples used in the *in vitro* translation reactions



Figure 5.4. 2µg of total cell RNA, extracted from Bunyamwera virus-infected C6/36 and BHK cells at the indicated h pi, was translated *in vitro* using the NEN rabbit reticulocyte lysate. BUN, marker track of [³⁵S]-methionine-labelled, infected BHK cell extract, prepared at 18 h pi; -, control reaction without added RNA; MI, mock-infected cell RNA. The viral N protein is indicated; in this gel NSs was obscured by the globin band.



Figure 5.5. (a) 1.5µg of CsCl pellet RNA and (b) 1.5µg of nucleocapsidderived RNA, extracted from Bunyamwera virus-infected BHK cells at the indicated h pi, were translated *in vitro* using the NEN rabbit reticulocyte lysate. BUN, marker track of $[^{35}S]$ -methionine-labelled, infected BHK cell extract, prepared at 18 h pi; MI, mock-infected cell RNA. The viral N protein is indicated; in these gels NSs was obscured by the globin band.



Figure 5.6. Northern blot of nucleocapsid-derived RNA extracted from Bunyamwera virus-infected BHK cells at the indicated h pi. $10\mu g$ of RNA was fractionated on a 1.8% agarose gel, containing 5mM methylmercury hydroxide, and transferred to a nitrocellulose filter. The filter was probed for positive-sense Bunyamwera virus S RNA using the [³²P]-labelled cDNA probe S9. (data not shown)

described above, The 48 and 72 h pi banded RNA samples contained four-fold more positive-sense S RNA than the earlier time point samples yet, while N protein was translated from RNA samples extracted up to 72 h pi, the levels translated from RNA extracted from 48 h onwards were 50% lower than at the 12 and 24 h pi time points. It was also clear that the 12 h pi pelleted RNA sample from BHK cells and the 24 h pi banded RNA sample from the same cells contained the same amount of positive-sense S RNA, yet more N protein (greater than ten-fold more) was translated from the pelleted RNA than from the banded RNAs. It appeared therefore that infected-cell derived S antigenomic RNA was translated poorly (if at all) *in vitro* compared to S mRNA. In contrast, *in vitro* translation from the equivalent of full-length positive-sense Bunyamwera virus S RNA, transcribed from a full-length cDNA clone, has been achieved in our laboratory (R. M. Elliott, personal communication).

The levels of host protein translation observed in vitro with total cell RNA prepared at 12, 48 and 72 h pi from Bunyamwera virus-infected C6/36 cells were comparable to the levels seen with RNA extracted from mock infected C6/36 cells (Figure 5.4), and were similar to the levels of host protein synthesis observed by pulse-labelling in vivo at the same time points. However, host protein translation in vitro diminished drastically between 12 and 24 h pi and remained low at 36 h pi. This was in marked contrast to the *in vivo* situation, where no significant decrease in host cell protein synthesis was observed during the course of infection (see Figure 5.1a). It is conceivable that the relatively high ratio of viral to host protein translation observed in vitro with the 24 h pi RNA sample represents the true ratio of viral to host mRNAs in Bunyamwera virus infected C6/36 cells at this time. Competition between viral mRNAs (at their maximum level) and host mRNAs for the limited components of the *in vitro* translation reaction could result in the low levels of host protein translation observed. The lower ratio of viral to host proteins synthesised in vivo at 24 h pi in infected C6/36 cells may be a consequence of a translational block imposed selectively on viral mRNAs. The absence of this block in the in vitro translation reactions suggests that it is removed by the RNA extraction procedure and may be proteinaceous. The possibility that a transient mRNA instability was induced in these cells by the virus infection was not supported by the in vivo pulse-labelling data (see Figure 5.1).

Translatable S mRNA was present in infected C6/36 cells up to the last time point tested (72 h pi; see Figure 5.4) providing further evidence that the decline in viral N protein synthesis observed after 24 h pi *in vivo* (Figure 5.1) could not be explained solely by the absence of translatable S mRNA. However, at later time points (48 and 72 h pi), the levels of N protein synthesised *in vitro* were not representative of the levels of S mRNA estimated to be present by slot blot and Northern blot analyses; Figures 5.2 and 5.3). Judging from the ratio of viral to host proteins translated *in vitro*, the virus specific mRNAs did not appear to compete as successfully with the host cell mRNAs for translation as they had

earlier in infection. This suggested that there may also be some alteration in the viral mRNA molecules synthesised later in infection in Bunyamwera virus-infected C6/36 cells which affects their translatability.

Attempts were made to prepare intact fractionated RNA samples from Bunyamwera virus-infected C6/36 cells which could be used to programme *in vitro* translation reactions. Vanadylribonucleoside complexes (VRCs; transition state analogues that bind to and inhibit many RNases; Berger & Birkenmeier, 1979) were added to the detergent-based lysis buffer, and were maintained at a concentration of 10mM through all stages of extraction and purification of the RNAs. Five μg (12% of the total yield) of the pellet RNA fraction and an equal proportion of the total yield of nucleocapsid-derived RNA, extracted from Bunyamwera virus-infected C6/36 cells at 48 h pi, were translated *in vitro* using the NEN rabbit reticulocyte lysate. The results suggested that by 48 h pi a significant amount of translatable Bunyamwera virus S mRNA was encapsidated in Bunyamwera virus-infected C6/36 cells (Figure 5.7).

5.4 Accumulation of viral RNA species in Bunyamwera virus infected *Aedes albopictus* C6/36 and BHK cells

The accumulation of positive-sense viral RNA species in Bunyamwera virusinfected C6/36 and BHK cells was compared by Northern analysis of total cell RNA (see Section 3.16.2) and of fractionated RNA samples (see Section 3.15.2) prepared from infected cultures at intervals up to 72 h pi. Hybridization was carried out with [³²P]labelled cDNA probes (L7, M6 and S9) specific for positive-sense Bunyamwera virus L, M and S RNA species, respectively. The size differences between full-length L and M RNA species and their respective subgenomic mRNAs were not sufficient to resolve these RNAs by electrophoresis, but Bunyamwera virus S mRNA could be separated from antigenomic S RNA by electrophoresis in 1.5% denaturing agarose gels (see later). Each set of filters containing C6/36 and BHK cell derived RNA samples from parallel time courses were hybridized at the same time, using the same probes so that direct comparisons of the relative levels of each viral RNA species at a given time point during the lytic and persistent infections could be made. However, the specific activities of the segment specific probes used were not determined and therefore interpretations regarding the absolute ratio of the individual RNA segments within any one sample could not be made.

Northern analysis of total cell RNA revealed different patterns of viral RNA accumulation in the two cell types (Figure 5.8). In infected BHK cells, positive-sense Bunyamwera virus L, M and S RNA species were detected at 12 h pi and accumulated, in parallel, up to 48 h pi. Between 48 and 72 h pi levels of each of the positive-sense RNA species declined slightly. In infected C6/36 cells, positive-sense viral RNA species were not detected until 24 h pi. The levels of each of the positive-sense viral RNA species detected at 24 h pi represented the maximum levels reached, since no further accumulation



Figure 5.8. Positive-sense viral RNA synthesis in Bunyamwera virus-infected C6/36 and BHK cells: (a) 10µg of total cell RNA, extracted from specific for Bunyamwera virus positive-sense L, M and S RNA species, respectively; (b) Graphs showing the pattern of accumulation of each of the positive-sense viral RNA species in Bunyamwera virus infected C6/36 and BHK cells as determined by densitometric analysis of the Bunyamwera virus-infected C6/36 and BHK cells at the indicated h pi was fractionated on a 1.0% agarose gel containing 5mM methylmercury hydroxide and transferred to a nitrocellulose filter. The filter was hybridized with a mixture of [³²P]-labelled cDNA probes, L7, M6 and S9, Northern blot shown in (a). of viral RNAs was observed in the C6/36 cells. The ratio of Bunyamwera virus L, M and S RNA segments synthesised in the two cell types differed; relatively more L and M RNA compared to S RNA was synthesised in BHK cells. Notably, the absence of viral RNA accumulation in the infected C6/36 cells, beyond 24 h pi, coincided with the downregulation of viral protein synthesis observed in these cells (see Section 5.1).

Northern analysis of fractionated RNA samples from both cell types facilitated a more detailed comparison of RNA accumulation in the two systems. Separate analysis of encapsidated and unencapsidated viral RNA species enabled viral RNA replication and mRNA transcription to be considered independently. Despite the problems encountered with ribonuclease digestion of the pellet RNA fraction from infected C6/36 cells (see Section 5.2), all three size classes of Bunyamwera virus mRNA were detected in the CsCl pellet RNA fraction from these cells up to late times post infection (albeit at low levels). Figure 5.3 illustrates that S mRNA was present in these cells up to 72 h pi, while Figure 5.9 shows the upper portion of another Northern blot demonstrating that L and M segment mRNAs were also present in these cells up to 72 h pi.

All three mRNA species were detected in the CsCl pellet RNA fraction from infected BHK cells (Figure 5.10). Maximum levels of S mRNA were present at 12 h pi and declined thereafter as would be expected as a result of the general mRNA instability induced in these cells after bunyavirus infection (Raju & Kolakofsky, 1988). Accumulation of the L and M mRNA species lagged behind accumulation of the S mRNA; L and M mRNAs were first detected at 12 h pi, whereas S mRNA first became detectable at 6 h pi. After maximum levels of each of the mRNA species had been reached (12 h pi for Bunyamwera virus S mRNA and 24 h pi for the L and M mRNAs), S mRNA levels declined rapidly, while levels of the L and M mRNAs decreased more slowly. These results are similar to those reported by Rossier et al. (1988) regarding La Crosse virus replication in BHK cells. Although, because La Crosse virus replicates more rapidly than Bunyamwera virus, viral mRNAs were detected earlier during the La Crosse virus infection, and a time interval between the attainment of maximum levels of S mRNA and L and M mRNAs was not observed. The higher multiplicity of infection used by Rossier and co-workers (20 to 50 pfu/cell compared to the 5 pfu/cell used in the work presented here) may have contributed to the differences observed. Figure 5.11 shows a Northern blot of the CsCl pellet RNA fraction from Bunyamwera virus-infected BHK cells, probed for positive-sense Bunyamwera virus S RNA species only. It is interesting to note that at 48 and 72 h pi, when S mRNA levels were low, the slower migrating, full-length antigenomic S RNA was detected in the pellet RNA fraction from infected BHK cells. There is some evidence to suggest that encapsidation of antigenomic RNA occurs concurrently with its synthesis (Raju & Kolakofsky, 1987a; Hacker et al., 1989). If this is the case, the presence of antigenomic S RNA in the pellet RNA fraction suggests that insufficient N



Figure 5.9. Unencapsidated positive-sense L and M RNA species in Bunyamwera virus-infected C6/36 cells up to 72 h pi. 40µg of CsCl pellet RNA, prepared from Bunyamwera virus-infected C6/36 cells at the indicated h pi, was fractionated on a 1.8% agarose gel containing 5mM methylmercury hydroxide and blotted to a nitrocellulose filter. Bunyamwera virus positive-sense L and M RNA species were detected using the [³²P]-labelled cDNA probes, L7 and M6 respectively.



Figure 5.10. Unencapsidated positive-sense viral RNA species in Bunyamwera virus-infected BHK cells up to 72 h pi. $40\mu g$ of CsCl pellet RNA, prepared from Bunyamwera virus-infected BHK cells at the indicated h pi, was fractionated on a 1.8% agarose gel containing 5mM methylmercury hydroxide and blotted to a nitrocellulose filter. The filter was hybridized with a mixture of [³²P]-labelled cDNA probes, L7, M6 and S9, specific for Bunyamwera virus positive-sense L, M and S RNA species, respectively.



Figure 5.11. Northern blot of CsCl pellet RNA prepared from Bunyamwera virus-infected BHK cells at the indicated h pi. 10µg of RNA was fractionated on a 1.8% agarose gel, containing 5mM methylmercury hydroxide, and transferred to a nitrocellulose filter. The filter was probed for positive-sense Bunyamwera virus S RNA using the [³²P]-labelled cDNA probe S9. protein is present in Bunyamwera virus-infected BHK cells at 48 h pi to fully encapsidate all of the antigenomic S RNA synthesised at that time.

Nucleocapsid-derived RNAs from both infected cell types remained intact throughout the isolation procedure (see Figure 5.12). The encapsidation of bunyavirus RNAs by N protein has been demonstrated to protect the RNA against ribonuclease digestion. Hacker et al. (1989) demonstrated that bunyavirus nucleocapsids are resistant to concentrations of RNase A far in excess of those required to completely digest viral mRNAs. During lytic infection in BHK cells, Bunyamwera virus RNA replication continued up to 48 h pi. After 48 h pi, levels of positive-sense L and M RNA declined slowly. However, positive-sense Bunyamwera virus S RNA continued to be encapsidated in the BHK cells between 48 and 72 h pi (see Figure 5.12). By 48 h pi, host protein synthesis had been completely abolished in these cells, viral protein synthesis was barely detectable (see Figure 5.1) and the cells had begun to display cpe. The appearance of antigenomic S RNA in the pellet RNA fraction from infected BHK cells at 48 h pi (see Figure 5.11) suggested that levels of unassembled N protein may be low in these cells by 48 h pi. This was to be expected given that levels of S mRNA were low in the BHK cells at this time. Under conditions where levels of N protein are limiting, the L, M and S segment RNAs would have to compete for the available unassembled N protein. Smaller RNAs would be encapsidated more rapidly, and hence the S segment RNA would be expected to be preferentially assembled into stable nucleocapsid structures later in infection in BHK cells. In contrast, viral RNA replication declined relatively early during Bunyamwera virus infection of C6/36 cells (Figure 5.12). After 24 h pi levels of encapsidated positive-sense L and M RNA species declined to low levels, but encapsidated positive-sense S RNA continued to accumulate up to 72 h pi. When the same infected C6/36 cell derived RNA samples were fractionated on a 1.5% denaturing agarose gel, prior to Northern analysis, the S RNA specific signal at 48 and 72 h pi was resolved into two bands corresponding to antigenomic S and S mRNA sized RNA species (Figure 5.13). It appeared that beyond 24 h pi, after genome replication had declined in Bunyamwera virusinfected C6/36 cells, S mRNA species were encapsidated by the viral N protein. The encapsidation of S mRNA may be responsible for the down-regulation of N protein synthesis observed in these cells beyond 24 h pi.

5.5 Summary

Bunyamwera virus infection induced shut off of host cell protein synthesis in BHK cells but not in C6/36 cells. The overall rate of viral protein synthesis appeared to be slower in C6/36 cells and NSs appeared to be synthesised at lower levels in these cells. Against a background of continuing host protein synthesis, levels of viral protein synthesis decreased after 24 h pi in the infected C6/36 cells; no viral protein synthesis was detected in these cells beyond 36 h pi. The decline in viral N protein synthesis was not paralleled by a



M6 and S9, specific for Bunyamwera virus positive-sense L, M and S RNA species, respectively; (b) Graphs showing the pattern of accumulation of encapsidated positive-sense viral RNA species in Bunyamwera virus-infected C6/36 and BHK cells as determined by Figure 5.12. Patterns of accumulation of encapsidated positive-sense viral RNA species in Bunyamwera virus-infected C6/36 and BHK cells: (a) 0.3µg of RNA extracted from CsCl-banded nucleocapsids at the indicated h pi was fractionated on a 1.8% agarose gel containing 5mM methylmercury hydroxide and transferred to nitrocellulose filters. The filters were hybridized with a mixture of [³²P]-labelled cDNA probes, L9, densitometric analysis of the Northern blot shown in (a).

B



Figure 5.13. Northern blot analysis of nucleocapsid-derived RNA from Bunyamwera virus-infected C6/36 cells. 1µg of RNA, extracted from CsClbanded nucleocapsids at the indicated h pi, was fractionated on a 1.5% agarose gel containing 5mM methylmercury hydroxide. The RNA was transferred to a nitrocellulose filter which was hybridized with a mixture of $[^{32}P]$ -labelled cDNA probes, L9, M6 and S9, specific for Bunyamwera virus positive-sense L, M and S RNA species respectively. decline in the levels of translatable SmRNA. Northern blot analysis of encapsidated viral RNA species extracted from infected C6/36 cells revealed the presence of S mRNA-sized RNA in viral nucleocapsid structures at 48 and 72 h pi whereas only full-length viral RNA species were encapsidated up to 24 h pi. Encapsidation of S mRNA in infected C6/36 cells may promote the establishment of persistent infection by limiting the translation of viral proteins.

Different patterns of viral RNA accumulation were observed in the two cell types following infection with Bunyamwera virus. Viral RNA synthesis proceeded more slowly in C6/36 cells, and L and M RNAs were less abundant relative to S RNA in these cells compared to BHK cells. Northern blot analysis of encapsidated viral RNA species isolated from infected C6/36 cells suggested that levels of viral genome replication were severely reduced after 24 h pi. Levels of encapsidated positive-sense L and M RNA declined to low levels in these cells between 24 and 48 h pi, however, positive-sense S RNA continued to accumulate in nucleocapsid structures up to the last time point tested (72 h pi). The encapsidated positive-sense S RNA species in the C6/36 cells.

6 Studies on *Aedes albopictus* C6/36 cell lines persistently infected with Bunyamwera virus

6.1 Screening of viruses released from two persistently infected cell lines established by Elliott and Wilkie (1986)

Elliott and Wilkie (1986) established two lines of mosquito cells persistently infected with Bunyamwera virus by infecting cultures of *Aedes albopictus* C6/36 cells with 5 or 0.5 pfu/cell of Bunyamwera virus. These cell lines, termed C6/36-PI HI and C6/36-PI LO respectively, were passaged at approximately weekly intervals through more than 40 passages and at no time was cpe observed. Supernatant culture fluid was harvested from each pass immediately before subculture and stocks of virus released at different passage levels of both persistently infected cell lines were stored at -20°C. Initial studies on these cell lines had shown that S segment specific RNAs were dominant both in the persistently infected cells and in virus released from these cells, almost to the exclusion of the L and M segment specific RNAs. It was suggested that the persistent virus released from both cell lines consisted mainly of variants possessing only the genomic S segment RNA. These studies also revealed that the nucleocapsid protein (N) encoded by virus released from one of the later passages of the C6/36-PI LO cell line (only pass 17 was tested) had a slower electrophoretic mobility on SDS-PAGE gels than the N protein of wild-type Bunyamwera virus (Elliott & Wilkie, 1986).

To investigate further the role of the S segment RNA and the evolution of virus variants in the maintenance of the persistent state, all the laboratory stocks of supernatant tissue culture fluid from the C6/36-PI HI and PI LO cell lines were screened for viruses producing variant N proteins. Initially it was important to determine the passage level at which the variants first appeared and to assess the extent of their occurence. Culture fluid from each passage level of both PI cell lines was used to infect BHK cell monolayers, which were pulse-labelled with [³⁵S]-methionine for 1 h at 24 h pi. SDS-PAGE analysis of the radiolabelled cell extracts revealed the emergence of three distinct N protein phenotypes during the course of the persistent infection (Figures 6.1 and 6.2). Variation was apparent in the levels of viral protein synthesis and in the extent of host cell protein synthesis shut-off observed with individual samples. This variation paralleled the fluctuating titres of extracellular virus recorded for the different passage levels of the persistently infected cell lines by Elliott & Wilkie (1986) and therefore probably reflected the faster progress of infections initiated at higher multiplicities. In relatively advanced infections, where host protein synthesis had been severely reduced, the viral L, G1, N and NSs proteins were easily identified. In the instances where host protein synthesis was not yet shut off, only the viral N protein could be identified against a background of comigrating host proteins. Both cell lines yielded viruses encoding N proteins with wild-



altered electrophoretic mobility. BHK cell monolayers were infected with culture fluid from each passage level (1-31) of Figure 6.1 Screening of the virus released from the C6/36-PI HI cell line for viral variants encoding N proteins with extracts were analysed by SDS-PAGE. wt: marker track of [35S]-methionine labelled cell extract from wt Bunyamwera the C6/36-PI HI cell line. At 24 h pi the monolayers were pulse-labelled for 1 h with [35S]-methionine. Labelled cell virus infected BHK cells.

Passage levels of the C6/36-PI HI cell line

Passage levels of the C6/36-PI LO cell line



with altered electrophoretic mobility. BHK cell monolayers were infected with culture fluid from each passage ³⁵S]-methionine. Labelled cell extracts were analysed by SDS-PAGE. wt: marker track of [³⁵S]-methionine Figure 6.2 Screening of the virus released from the C6/36-PI LO cell line for viral variants encoding N proteins level (1-31 and 36) of the C6/36-PI LO cell line. At 24 h pi the monolayers were pulse-labelled for 1 h with abelled cell extract from wt Bunyamwera virus infected BHK cells. MI: labelled cell extracts from uninfected BHK cells. type mobility (N), as well as variants encoding N proteins with intermediate mobility (NINT), and slow mobility (NLO), on SDS-PAGE gels (Figures 6.1 and 6.2). Variant N protein containing viruses first appeared at passages 8 and 9 of the C6/36-PI HI and C6/36-PI LO cell lines respectively. At passage 8 of the C6/36-PI HI cell line the viral N protein synthesised appeared to be of the same mobility as the NLO species which had been detected at passage 17 of the C6/36-PI LO cell line by Elliott and Wilkie (1986). By passage 9 another N protein phenotype was evident which displayed a mobility on SDS-PAGE gels between that of wild-type Bunyamwera virus N protein and NLO (NINT). In addition, at passage 9 two G1 protein species were evident one with wild-type mobility and a slower migrating species. At later passages NLO was most prevalent, appearing at significant levels at passages 10, 21, 23, 26 and 29 to 31 (Figure 6.1). In the C6/36-PI LO cell line, N proteins of all three phenotypes (N, NINT and NLO) were detected concurrently at passage 9; in addition, two G1 species were synthesised at passage 9, as observed for the same passage level of the C6/36-PI HI cell line. At passages 10 and 12 both NLO and N^{INT} were evident. At later passages the dominant N protein phenotype varied: at passage 14 relatively low levels of N^{INT} were observed; at passage 17 only N^{LO} was observed, in agreement with previous observations (Elliott & Wilkie, 1986); at passage 18, NLO, an N species of intermediate mobility (possibly distinct from the other NINT species) and traces of wild-type N protein were synthesised. Then at passage 20 only NLO was evident; through passages 22 to 24 and also at passage 29, N protein of wild-type mobility was synthesised; at passage 31, N^{LO} was the dominant species but at passage 36 wild-type N protein was once again the only species detected (Figure 6.2). At passages 9 and 10 of the C6/36-PI HI and C6/36-PI LO cell lines respectively, a protein corresponding in size to NSm was observed. Thus variant viruses were detected relatively early during the persistent infection and appeared to co-exist with wild-type virus in the persistently infected cells. The relative level of each viral species fluctuated during the course of persistence.

In order to further characterise the variants which emerged during the persistent infection, virus released at passages 8, 18, 20, 29 and 30 of the C6/36-PI LO cell line were selected for further investigation. Supernatant tissue culture fluid from each of these passage levels was titrated on BHK cell monolayers and a minimum of twelve isolated plaques were picked from each passage. The protein synthesis profile produced by each isolate upon infection of BHK cells was determined. All 12 plaque isolates from passage 18, 11 possessed wild-type mobility. Of 12 isolates plaque-purified from passage 18, 11 possessed wild-type N proteins, while 1 isolate encoded an N protein of N^{LO} phenotype. The ratio of the different virus phenotypes selected from this population of persistent viruses did not reflect the relative proportion of each N protein phenotype synthesised during the initial screen (Figure 6.2). No viruses encoding N^{INT} proteins were recovered from passage 18 despite their presence at levels significantly higher than the wild-type phenotype as determined by the initial screen. In addition the initial screen

suggested that N^{LO} encoding variants were more abundant relative to viruses encoding wild-type N proteins than was apparent upon screening the plaque isolates. Sixteen plaque isolates were picked from passage 20, and of these 11 displayed N^{LO} phenotype and 5 possessed wild-type mobility N proteins. Of 14 isolates plaque-purified from passages 29 and 30, 7 and 9 respectively exhibited N^{LO} phenotype. The phenotype(s) of the other isolates fom these passage levels could not be determined due to low levels of viral protein synthesis against a background of high levels of host protein synthesis at the time of pulselabelling. However, once again, the abundance of N^{LO} encoding viruses recovered from passage 29 was not in line with predictions based upon the initial screen (Figure 6.2) during which only wild-type N protein encoding viruses were detected at this passage level.

On the basis of this second screen, two isolates from each passage level were selected for a further two rounds of plaque-purification. Each plaque isolate was identified by passage and plaque number. For instance the ninth plaque isolated from passage 18 was labelled BUN18.9. Elite stocks of triple-plaque purified viruses encompassing the N and N^{LO} phenotypes were then grown on BHK cell monolayers to provide a source of biologically cloned virus which could be characterised further, and from which virion RNA could be prepared for molecular analysis (see Section 6.3). Figure 6.3 shows the protein synthesis profiles of the elite stocks of virus established from the C6/36-PI LO cell line.

6.2 Determination of the temperature-sensitivity of variants arising in the C6/36-PI LO cell line

Elliott and Wilkie (1986) reported the appearance of temperature-sensitive (ts) virus during the establishment of persistence in the C6/36-PI HI and C6/36-PI LO cell lines. By passage 8 the virus released from both persistently infected cell lines displayed temperature-sensitivity. As detailed above, the viruses encoding variant N proteins first appeared in both persistently infected cell lines around this time (see Figures 6.1 and 6.2). To determine if the ts phenotype could be correlated with the altered N protein mobility on SDS-PAGE gels, the temperature-sensitivity of selected elite stocks of virus derived from the C6/36-PI LO cell line was assayed by plaque titration on BHK cell monolayers. Plaque assays were performed at 31, 37 and 38.5°C. Viruses with variant N proteins grew less well (BUN18.9) or not at all (BUN20.2, BUN20.4, BUN29.9 and BUN29.10) at 38.5°C compared to 31 and 37°C, while viruses with wild-type-like N proteins grew equally well at the three temperatures tested (Table 6.1). It appeared that the viruses with variant N proteins also exhibited a ts phenotype.



Figure 6.3 Protein profiles of BHK cells infected with each of the elite stocks of triple plaque-purified virus isolated from the C6/36-PI LO cell line. BHK cell monolayers were infected with the elite stocks of virus at a moi of 10 pfu/cell. At 48 h pi the infected monolayers were pulse-labelled for 1 h with [35S]-methionine and the labelled cell extracts were analysed by SDS-PAGE. wt, [35S]-methionine labelled cell extract from wild-type Bunyamwera virus infected cells; MI, [35S]methionine labelled cell extract from uninfected BHK cells. The elite stocks are identified by passage number and plaque number from the first round of plaque-purification.

Virus isolate		Virus titre (pfu/ml)			ts phenotype	N protein phenotype	
Passage	Plaque	31°C	37°C	38.5°C			
8	4	1.2 x 10 ⁷	1.1 x 10 ⁷	4.5 x 10 ⁷	no	wt	
	6	2.1 x 10 ⁷	2.8 x 10 ⁷	2.9 x 10 ⁷	no	wt	
18	9	2.0 x 10 ⁶	1.2 x 10 ⁶	2.5 x 10 ⁴	yes	NLO	
	10	3.2 x 10 ⁷	3.9 x 10 ⁷	3.8 x 10 ⁷	no	wt	
20	2	2.9 x 10 ⁷	4.0 x 10 ⁷	No plaques	yes	NLO	
	4	3.4 x 10 ⁷	1.8 x 10 ⁷	No plaques	yes	NLO	
29	9	1.1 x 10 ⁶	1.1 x 10 ⁶	No plaques	yes	NLO	
	10	2.2 x 10 ⁷	2.2 x 10 ⁷	No plaques	yes	NLO	
wt	Lg	4.1 x 10 ⁷	6.0 x 10 ⁷	1.9 x 10 ⁷	no	wt	

Table 6.1 Results of the temperature-sensitivity assay performed on BHK cell monolayers with viruses plaque-purified from the supernatant tissue culture fluid of *Aedes albopictus* C6/36 cells persistently infected with Bunyamwera virus.

6.3 Sequence analysis of the genomic S RNA segments of viruses released from the C6/36-PI LO cell line

To determine the molecular basis for the altered electrophoretic mobility of the N proteins encoded by the variant viruses arising during the persistent infection, and to investigate the overall rate of nucleotide change in the Bunyamwera virus S segment RNA during the persistent infection, attempts were made to sequence the S segments of isolates plaque-purified from passages 8, 18, 20 and 29 of the C6/36-PI LO cell line. Virion RNA was prepared (see Section 3.15.1) and sequenced by primer extension from γ -[³²P]-labelled oligonucleotides. The oligonucleotides used were those used to sequence the S RNA segment of wtL₉BUN virus (see Figure 4.1 and section 3.17).

Sequence analysis of the S segment RNA of viruses BUN8.6 and BUN18.10, which produced wild-type mobility N proteins, demonstrated that the S RNA of the BUN8.6 isolate was identical in sequence to the S RNA of wild-type Bunyamwera virus, while the S RNA segment of the BUN18.10 isolate had undergone a single nucleotide substitution (see Figure 6.4). The S RNA segment of BUN18.10 had a guanosine to adenosine transition at nucleotide position 255 in the positive-sense S RNA. This nucleotide change resulted in amino acid substitutions in the N and NSs proteins encoded by this virus. Asparagine replaced serine at amino acid position 57 of the N protein and



Figure 6.4. Direct RNA sequence determination of the genomic S RNA segment of the persistent viral isolate BUN 18.10, by primer extension sequencing from oligonucleotide number 2. The sequence shown covers nucleotides 243 to 263 inclusive, and is written (together with the corresponding Bunyamwera virus S segment sequence; Elliott, 1989b) as the cDNA, alongside. The guanosine to adenosine transition at nucleotide position 255 is indicated in **bold**.

On the left is shown the sequence of the corresponding region of the wtL9BUN virus S RNA. Note the A/T ambiguity at position 256, which is a T residue in the cDNA derived sequence. methionine replaced value at amino acid position 51 of the NSs protein. Both were relatively conservative amino acid changes and neither of these changes occured in conserved regions of the N or NSs proteins as defined by a consensus amino acid sequence composed by alignment of the six known bunyavirus S gene sequences (Figure 6.5).

Problems were encountered during attempts to sequence the S RNA segment of each of the variant viruses, so a primer extension reaction in the absence of dideoxy nucleotides was performed. This too failed to work suggesting that the problem lay in the ability of the oligonucleotides to prime the extension reaction. Eventually, primer extension from oligonucleotide number 2, annealed to BUN18.9 RNA, yielded a stretch of sequence identified as Germiston virus S RNA bases 233 to 433 (of the positive-sense full-length S RNA). Comparison of the published sequences for the S RNAs of Germiston and Bunyamwera bunyaviruses (Gerbaud et al., 1987; Elliott, 1989) revealed a 20 nucleotide stretch of Germiston virus S RNA, between bases 211 and 230, which was identical to the stretch of sequence from bases 209 to 228 inclusive of Bunyamwera virus genomic S RNA to which oligonucleotide 2 was the exact complement. This was confirmed by using the FIND program (Devereux et al., 1984) to scan the Germiston virus S gene sequence for potential priming sites. The perfect match between oligonucleotide 2 and the afore mentioned sequence was detected while none of the remaining 5 oligonucleotides had complementary sites with fewer than 3 mismatched bases within the Germiston virus S gene sequence.

Figure 6.6 shows a portion of the sequencing data obtained by priming with oligonucleotide number 2 on BUN18.9 virion RNA and wtLoBUN virion RNA. As priming was not achieved with oligonucleotides numbers 2, 3 or 4 on BUN20.2 or BUN29.10 virion RNA, whereas parallel reactions, under identical conditions, yielded sequencing data from BUN8.6 and BUN18.10, it appeared that the BUN20.2 and BUN29.10 isolates were also contaminating viruses in the persistently infected C6/36 cell line. It appeared therefore that the variant N protein producing viruses, observed in association with the persistently infected Aedes albopictus C6/36 cell line established and passaged by Elliott and Wilkie (1986), originated by contamination of the cell line with temperature-sensitive viruses which were in use in the laboratory at that time. It is interesting that Bunyamwera virus persistently infected C6/36 cells were susceptible to infection by other bunyaviruses. Verani et al. (1984) reported that Toscana virus persistently infected cells were resistant to infection with other phleboviruses but remained susceptible to viruses within other genera of the Bunyaviridae. However, Newton et al. (1981) reported that Barmah Forest bunyavirus did replicate in their Bunyamwera virus persistently infected uncloned Aedes albopictus cells.

From the sequencing data on the BUN8.6 and BUN18.10 virus isolates, both of which encoded wild-type mobility N proteins, the rate of nucleotide change in the viral genome during the course of the persistent infection appeared to be slow. No nucleotide

(a)

(b)

Bunyamwera .mieLeFhDVaantsstFdPEvaYanFkrvhttglsYdhIRiFYikgreiKtsLakrsew Maguari .mieLeFnDVaantsstFdPEiaYvnFkrihttglsYdhIRvLYikgreiKtsLkkrsew Germiston .mleLeFeDVpnnigstFdPEsgYtnFqrnylpgvtLdqIRiFYikgreiKtsLkkrsew La Crosse .msdLvFyDVastgangFdPDagYmdFcvknaesInLaaVRiFFInaakaKaaLsrkper snowshoe hare .msdLvFyDVastgangFdPDagYmaFcvkyaesvnLaaVRiFFInaakaKaaLsrkper Aino manqFiFqDVpqrnlatFnPEvgYvaFiakhgaqlnFdtVRfFFlnqkkaKmvLsktaqp Consensus L-F-DVF-PEYFLR-FKL 61 120 Bunyamwera evtlpLGgwkitvvNtpFPgnrnpNVpddgLTlHElSGELArYllekm.lkvsepeklij
61 Bunyamwera evtlnLGgwkityyNtnFPgnrnnnVnddgLTlHRlSGFLArYllekm.lkyseneklii
01 Bunyamwera evtlnLGgwkityyNtnFPgnrnnpVnddgLTlHRlSGFLArYllekm,lkyseneklii
MaguarievtlnLGgwkvavfNtnFPgnrnspVpddgLTlHRlSGFLArYlleki.lkvsdpekliiGermistonevtlnLGgwkvavfNtnFPgnrnnaVpdygLTfHRlSGFLArYlleki.lkvsdpekliiLa CrossekanpkFGewqveviNnhFPgnrnnpIgnndLTiHRlSGYLArWvldqynenddesqhelisnowshoe harekanpkFGewqvevvNnhFPgnrnnpInsddLTiHRlSGYLArWvldqykenedesrreliAinosvdltFGgikftlvNnhFPqytanpVpdtaLTlHRlSGYLAkWvadqcktnqiklaeam.ConsensusGNFPVLT-HR-SGYLA
121 180
Bunyamwera kskIInPlAEknGitWndGeevYLsFfPGsEMFLgTFrFYPLaIgiykVqrkeMEpkYLe
Maguari kskIInPlAEknGitWadGeevYLsFfPGsEMFLgTFkFYPLaIgiykVqkkeMEpkYLe
Germiston rtkIVnPlAEknGitWesGpevYLsFfPGaEMFLgTFrFYPLaIgiykVqrkeMDpkFLe
La Crosse rttIInPiAEsnGvgWdsGpeiYLsFfPGtEMFLeTFkFYPLtIgihrVkqgmMDpqYLk
snowshoe hare kttlInPiAEsnGvrWdsGaeiYLsFfPGtEMFLeTFkFYPLtIgiyrVkqgmMDpqYLk
Aino .ekIVmPlAEvkGctWteGltmYLgFaPGaEMFLeTFeFYPLvIdmhrVlkdgMDvnFMr
ConsensusII-P-AEGWGYL-F-PG-EMFL-TF-FYPL-IVMDYL-
181 237
Runvamwera KtMRORYmg]eAatWtysK]teVgsaLtyVssLgWkktnySaaArdFLakFGInm
Maguari KtMRORYmgleAatWtvsKvneVgaaLtvVsgLdWkktnvSaaAreFLakFGInm
Germiston KtMRORYlgidAgtWtttKlgeVeaaLkvVsgLgWkktnvSsaAreFLskFGIrm
La Crosse KalRORYgtltAdkWmsgKvaalaksLkdVegLkWgkgglSdtAktFLgkFGIrlp
snowshoe hare KaLRQRYgsltAdkWmsgKvtaIaksLkeVegLkWgrgglSdtArsFLgkFGIrlp
Aino KvLRQRYgtltAeqWmtqKidaVraaFnaVgqLsWaksgfSpaAraFLaqFGIni
Consensus KRQRYAWKVL-WSAFLFGI
1 * 50

	1	*		60			
Bunyamwera	mMslltpavlLtqrshtltlsvstplglvmttYeSStlkdaRlklvSo	akevno	klhI	LtLg			
Maguari	mMslltpavlLtqrlhtltlsvstplglvmttFeSStlkdaRlklvSo	akevsç	rlrI	LtLg			
Germiston	. Mslits gvlLtqsqdtltfsvttcqglrltkFaSStlkdaRlkivScherkerkerkerkerkerkerkerkerkerkerkerkerke	ikevno	klrI	Ltg			
La Crosse	Crosse mMshqqvqmdLilmqgiwtsvlkmqnhstllqLgSSssmlqRprllSrvsqrgrltLnL						
snowshoe hare	$\tt mMshqqvqmdLilmqgiwhsvlnmqnqsillqLgSSssmpqRprllSrvsqrgrqiLnLe$						
Aino	$. {\tt MflngislrLtrrsgmwhlllnmgpnsisipLdSSssirrRprwySvrrhnqvliLhLv}$						
Consensus	-ML-SSRS		I	-L-			
	61	111					
Bunyamwera	agrllyiIrIflaTgttqfltmvlpStasvdslpgtylrrc						
Maguari	agrllylIqIflaTgtvqfqtmvlpStdsvdslpgtylrkf						
Germiston agrylysIrIsleTgtmqclttvlpStvsvdtlpgtylestlqrqnqkss							
La Crosse sgrwrlsIiIfleTgttqlvttilpStdylgi							
snowshoe hare	sgrwrlsIiIfleTgtiqltatilpStdcqdi						
Aino	asslhwlItIfpnTqqilcqtlpslSivsqdi						
Consensus	S						
in the second second second	and the second of the second	1.1.1.1	1.1.1	1.11			

Figure 6.5. Six-way alignment of (a) the nucleocapsid (N) proteins and (b) the non-structural (NSs) proteins of six different bunyaviruses. The alignments were generated with the GCG programs PILEUP and PRETTY. The consensus sequence shows residues completely conserved in all six sequences. * indicates the position of the amino acid substitutions found in the N and NSs proteins of the BUN 18.10 isolate.



Figure 6.6. Direct RNA sequence determination of the genomic S RNA segments of (a) the persistent viral isolate BUN18.9 and (b) wtL₉BUN virus, by primer extension sequencing from oligonucleotide number 2. The sequences shown are from nucleotide 290 to 351 of BUN18.9 S RNA (identified as Germiston virus S RNA) and nucleotide 288 to 360 of wtL₉BUN virus S RNA. The sequence from nucleotide 322 to 351 of both RNAs is written as cDNA alongside.

changes were observed in the consensus S segment sequence for isolate BUN8.6 which had been plaque-purified from the persistently infected culture after it had been passaged for over two months. The BUN18.10 isolate was plaque-purified from the supernatant tissue culture fluid of the persistently infected culture after it had been passaged for over four months and only one nucleotide substitution, resulting in conservative amino acid substitutions in N and NSs, was observed. The constraints on the evolution of the Bunyavirus S segment RNA may be greater than for the other genome segments due to the overlapping reading frames encoded in the S RNA. Hence the rate of change of the Bunyamwera virus genome as a whole, during the persistent infection, may have been greater than is suggested by the relative stability of the S segment RNA.

6.4 Establishment of an *Aedes albopictus* C6/36 cell line persistently infected with wtL₉BUN virus

Since the available persistently infected cell lines were apparently contaminated further studies on the molecular basis of with at least one other bunyavirus, Bunyamwera virus persistence in C6/36 cells required that another persistently infected cell line be established. Aedes albopictus C6/36 cells were infected with wtL₂BUN virus at a moi of 5 pfu/cell. The cells were subcultured at a 1 in 10 split four days post infection and thereafter at a 1 in 20 split every 6 days. The resultant persistently infected culture (C6/36/BUN) grew at a similar rate to both uninfected C6/36 cells and to the C6/36-PI HI and C6/36-PI LO cell lines. Like the former persistently infected cell lines, the C6/36/BUN cell line was passaged for over a year without observable cytopathic effects. Extracellular virus titres were determined and the plaque-phenotypes and protein synthesis profiles, produced in BHK cells, by virus released at different passage levels of the C6/36/BUN cell line were monitored. In addition, total cell RNA and fractionated RNA samples (see Section 3.15) were prepared from the first 23 passages of the C6/36/BUN cell line. These samples were probed with Bunyamwera virus segment specific riboprobes to follow the synthesis and encapsidation of viral RNA species throughout the persistent infection.

6.5 Virus release from the C6/36/BUN cell line

Supernatant culture fluid from each of the first 30 passages of the C6/36/BUN culture was routinely sampled immediately before subculture and titrated on BHK cell monolayers. During the persistent infection the titre of released virus fluctuated between 10^2 and 10^7 pfu/ml but no regular periodicity was observed (see Figure 6.7). There was a trend towards lower levels of extracellular virus with increasing passage level of the culture and at passages 24, 25 and 28 levels of extracellular virus fell below the threshold of detection of the plaque assay (10 pfu/ml). Superimposed on this general trend towards decreased virus production, occasional increases in levels of released virus were observed;



line.

for instance, marked increases in the levels of extracellular virus were detected at passages 13 and 21 (reaching titres of 3×10^6 and 4×10^5 pfu/ml respectively) and virus production returned to detectable levels at passages 26 and 29 after declining to levels too low to be detected by plaque assay on BHK cells. There was also a marked change in the plaque morphology (in BHK cells) of extracellular virus with increasing passage level; the original well defined, round, lytic-plaque morphology was gradually replaced by a small, cloudy-plaque phenotype. The cloudy-plaque variants arose before passage 5 and became dominant by passage 15 (see Figure 6.7). The decreasing plaque-size made counting of plaques difficult and may have contributed to the low titres recorded for later passages.

An infectious centres assay carried out on cells from passage 6 of the C6/36/BUN cell line revealed that 2% of the cells shed infectious virus. Infectious centres and immunofluorescence studies carried out on an *Aedes albopictus* C6/36 cell culture persistently infected with Semliki Forest virus (Davey and Dalgarno, 1974) also revealed that only 2% of the cells within the persistently infected culture released infectious virus, while 100% of the cells appeared to be productively infected during the acute phase of infection. Immunofluorescence studies carried out with the C6/36/BUN cell line yielded inconclusive results.

6.6 Protein profiles in BHK cells infected with culture fluid from the persistently infected C6/36/BUN cell line

Supernatant culture fluid from the first 30 passages of the C6/36/BUN cell line was used to infect BHK cell monolayers. Initially each monolayer was infected with 400µl of supernatant culture fluid, after 30 minutes at 31°C the culture fluid was replaced with another 400µl of culture fluid from the same passage of the persistently infected cell line. The inoculum was removed after a further 30 minute absorption period and the infected cells were maintained in supplemented GMEM (see Section 3.2.1). At 48 h pi the cells were pulse-labelled for 1 h with [35S]-methionine and labelled cell extracts were analysed by SDS-PAGE (Figure 6.8). No variant N protein-containing viruses were detected, but novel polypeptides, potentially of viral L segment origin, were observed in infections initiated with virus released at passage levels 18, 21, 23, 26 and 29 of the C6/36/BUN culture (see Section 8.4). The levels of viral protein synthesis and host cell shut-off observed for each passage correlated with fluctuations in the levels of extracellular virus in the tissue culture supernatant of the C6/36/BUN culture (see Figure 6.7). It was interesting to note that while the infections initiated with culture fluid from passages 24, 25 and 28 of the C6/36/BUN cell line (which contained levels of virus below the threshold of detection by plaque assay on BHK cell monolayers) were slow, the synthesis of viral N protein was observed in the infected BHK cells, demonstrating that infectious virus was released at these passage levels.





Figure 6.8 Protein profiles of BHK cells infected with supernatant tissue culture fluid from the first 30 passages of pulse-labelled with [35S]-methionine and the labelled cell extracts were analysed by SDS-PAGE. MI, mock infected the C6/36/BUN cell line. BHK cell monolayers were infected with culture fluid from each passage level of the persistently infected cell line, and were maintained in supplemented GMEM at 31°C for 48 h. The cells were then BHK cell extract.

6.7 Levels of intracellular viral RNAs in the persistently infected cell line

Total cell RNA was extracted at every passage level (up to passage 23) of the C6/36/BUN cell line. Each sample was probed for Bunyamwera virus genomic L, M and S RNAs and for Bunyamwera virus complementary-sense S RNA by slot blot analysis using segment- and sense-specific riboprobes (as detailed in Section 3.16.1). An additional filter was hybridized with a [³²P]-labelled cDNA probe specific for mouse actin mRNA to monitor the amount of cellular RNA applied with each sample. The films were scanned by densitometry and each virus-specific signal was normalised against the corresponding actin signal. On the assumption that the actin mRNA levels remained relatively constant at each passage level, fluctuations in the levels of each of the viral RNA species relative to actin mRNA could be expressed as changes in the abundance of each of the viral RNA species per cell in the culture (see Figure 6.10). Levels of each of the viral RNA species varied between passage levels without regular periodicity and no simple correlation between the fluctuating levels of each of the viral RNA species and the titres of virus released from these cells was evident (Figure 6.9). Furthermore, the level of each viral RNA species appeared to vary independently (Figure 6.10). For example, at passage 2 the signal for genomic S RNA was stong relative to the signals for the genomic L and M RNAs; by passage 4 the level of M RNA had increased 3-fold, levels of L RNA had doubled and S RNA levels remained approximately the same. By passage 6, levels of genomic L RNA had increased considerably but levels of M and S genomic RNA remained relatively low. At passage 13, both the L and S genomic RNA segments gave considerably stronger signals than the M genomic RNA segment. The relatively poor signals obtained while probing for Bunyamwera virus genomic M RNA appeared to be due, at least in part, to problems with the probe as a control sample consisting of nucleocapsid derived RNA extracted from Bunyamwera virus-infected C6/36 cells at 24 h pi (which was known to contain significant quantities of M RNA) also gave a weak signal with this M segment specific riboprobe. Viral complementary-sense S RNA was detected at all passage levels (Figure 6.9). Levels fluctuated irregularly between passages and maximum levels were observed at passages 8, 9 and 18. Levels of viral complementary-sense S RNA appeared to vary independently of the levels of genomic S RNA (Figure 6.11).

A more detailed study of intracellular viral RNAs in the C6/36/BUN cell line was carried out by Northern analysis. Initially, total cell RNA from passages 4 and 13 of the persistently infected cell line (chosen because at these passage levels titres of released virus and levels of intracellular viral RNA were relatively high) was analysed using partial cDNA probes (L5, L7, S8, S9, M1 and M6) synthesised from the template cDNAs detailed in Section 3.5.3. In agreement with the slot blot data, genomic M RNA was more abundant relative to genomic L and S segment RNAs at passage 4 than at passage 13 (Figure 6.12a).






Figure 6.10. Levels of Bunyamwera virus genomic L, M and S RNA, relative to actin mRNA levels, resident in the C6/36/BUN cell line during the first 23 passages as determined by densitometric analysis of the slot blot data presented in Figure 6.9.



Figure 6.11. Levels of Bunyamwera virus genomic and complementary-sense S RNA, relative to actin mRNA levels, resident in the C6/36/BUN cell line over the first 23 passages as determined by densitometric analysis of the slot blot data presented in Figure 6.9.



3 of the C6/36/BUN cell line was probed with a [32P]-labelled cDNA probe (L5) specific for Bunyamwera virus genomic Figure 6.12 Northern blot analysis of total infected cell RNA extracted from passage levels 4 and 13 of the C6/36/BUN cell L RNA; (c) 20µg of RNA from passage levels 4 and 13 was probed with [32P]-labelled cDNA probes (L7, M6 and S9) ine. Each RNA sample was fractionated by electrophoresis through a 1.5% agarose gel containing 5mM methylmercury nydroxide. Following capillary transfer to nitrocellulose, the RNA was hybridized with Bunyamwera virus specific [32P]and S8) specific for Bunyamwera virus genomic L, M and S RNA species; (b) 20µg of RNA extracted from passage level abelled cDNA probes. (a) 20µg of RNA from both passage levels was probed with [32P]-labelled cDNA probes (L5, M1 specific for positive-sense Bunyamwera virus L, M and S RNA species. In addition to the full-length viral RNA species, subgenomic viral RNAs were observed at passage 13. These subgenomic (defective) RNAs were identified as L segment derived by hybridization with a $[^{32}P]$ -labelled cDNA probe (L5) specific for Bunyamwera virus genomic L RNA (Figure 6.12b). A different spectrum of defective RNAs was observed when the filter was hybridized with a mixture of $[^{32}P]$ -labelled cDNA probes (L7, M6 and S9) specific for positive-sense L, M and S RNAs (Figure 6.12c). Since the L segment specific partial cDNA probes (L5 and L7) were synthesised from non overlapping clones (see Section 3.5.3) they may have detected different subpopulations of defective RNAs. S mRNA sized RNA was detected at both passage levels by Northern analysis, coinciding with the detection of significant levels of viral positive-sense S RNA by slot blot hybridization (see Figure 6.11).

6.8 Defective viral RNAs in the persistently infected cells

The occurence of defective viral RNAs in the persistently infected cells was investigated further. Total cell RNA extracted from passages 2, 6, 18, 21 and 58 of the persistently infected culture was fractionated in a denaturing agarose gel and transferred to a nitrocellulose filter. The filter was hybridized sequentially with first an L segment specific riboprobe and then an S segment specific riboprobe, in both cases to detect genomic-sense RNA (Figure 6.13). Subgenomic L segment derived RNAs were just detectable at passages 2 and 6, and were abundant at passage 21 of the persistently infected cell line. The ratio of defective L segment RNA to full-length L RNA varied between different passage levels, and at passage 58 defective L RNAs were much more abundant than the full-length L RNA. The pattern of defective L RNAs changed with increasing passage level. Most, but not all, of the size classes of defective RNAs appeared to be stably maintained in the cell line and new species arose with increasing passage level. No subgenomic S segment RNAs were observed, but interestingly, at passage 21, a more slowly migrating S segment RNA was detected. Figure 6.14 compares the Northern profiles of total cell RNA from passages 21 and 58 of the persistently infected cell line, probed for genomic-sense S RNA only. At passage 21 the defective S RNA was far more abundant than the authenic genomic S RNA, but at passage 58 the normal sized genomic S RNA was the only negative-sense S RNA species detected. Slower migrating genomicsense S RNA species were also detected at passages 19 and 22. The defective S RNA species present at passage 19 was intermediate in mobility between authentic sized genomic S RNA and the larger S RNA species observed at passages 21 and 22 (Figure 6.15). Clearly, the population of defective viral RNAs resident in the persistently infected C6/36/BUN cell line altered as the culture matured.



Figure 6.13. Northern blot analysis of total cell RNA extracted from passage levels 2, 6, 18, 21 and 58 of the C6/36/BUN cell line. $20\mu g$ of RNA from each passage level was fractionated on a 1.5% agarose gel containing 5mM methylmercury hydroxide and transferred to a nitrocellulose filter. The filter was hybridized sequentially, first with a genomic L segment specific [³²P]-labelled riboprobe, and then with a genomic S segment specific [³²P]-labelled riboprobe. \diamond indicates S RNA species migrating more slowly.



Figure 6.14. Northern blot analysis of total cell RNA from passage levels 21 and 58 of the C6/36/BUN cell line. $20\mu g$ of RNA from each of these passage levels was fractionated on a 1.5% agarose gel containing 5mM methylmercury hydroxide. The RNA was transferred to a nitrocellulose filter which was hybridized with a [³²P]-labelled riboprobe specific for Bunyamwera virus genomic S RNA.



Figure 6.15. Northern blot analysis of total cell RNA extracted from passage levels 18, 19, 21 and 22 of the C6/36/BUN cell line. $15\mu g$ of RNA from each passage level was fractionated on a 1.5% agarose gel containing 5mM methylmercury hydroxide. The RNA was transferred to a nitrocellulose filter which was hybridized with a [³²P]-labelled riboprobe specific for Bunyamwera virus genomic S RNA.

6.9 Encapsidation of viral RNA species in the C6/36/BUN cell line

In addition to preparing total cell RNA from each passage level of the C6/36/BUN cell line, encapsidated viral RNA was isolated by banding intracellular viral nucleocapsids on CsCl gradients and subsequent extraction using the acid-phenol-guanidinium-thiocyanate protocol of Chomczynski and Sacchi (1987). Encapsidation of viral RNAs in the persistently infected cells was investigated by comparing Northern blots of total cell RNA and nucleocapsid derived viral RNA from the same passage level of the C6/36/BUN cell line.

Analysis of nucleocapsid derived RNA from passage level 18 revealed that the only positive-sense viral RNA species detected in intracellular nucleocapsids was positivesense S RNA (Figure 6.16a). Similarly analysis of nucleocapsid derived RNA from passage levels 18 and 21 revealed that genomic S RNA was preferentially encapsidated over the other negative-sense viral RNA segments (Figure 6.16b). Interestingly, there appeared to be a strong selection for packaging of authentic sized genomic S RNA in the persistently infected cells; at passage 21, when full-length and defective L RNA species and an aberrant S RNA species were abundant (see Figure 6.13), only authentic sized S RNA, which was far less abundant than either of the former RNA species, was encapsidated (Figure 6.17). It appeared therefore that the defective L RNAs derived from the L RNA segment were not efficiently assembled into nucleocapsids. To investigate this further, total cell RNA and nucleocapsid derived RNA samples from passages 18, 19, 21 and 22 of the C6/36/BUN cell line were fractionated on a denaturing agarose gel and transferred to a nitrocellulose filter which was probed sequentially, first with a genomic L segment specific riboprobe, and then a genomic S segment specific riboprobe (Figure 6.18). The result further emphasised the exclusive encapsidation of full-length genomic S RNA at passage level 21, despite its low abundance, while the defective viral RNAs prevalent at this passage level and full-length genomic L RNA remained uncomplexed with N protein.

In this analysis full-length genomic L RNA and some defective L segment derived RNAs were detected amongst the encapsidated RNA species extracted from passage level 18. Indeed when nucleocapsid derived RNA from passage 18 was probed solely for genomic L RNA sequences a range of size classes of defective L RNAs were detected, including a defective RNA not much larger than 1kb in length, and possibly an L RNA species larger than full-length Bunyamwera virus genomic L RNA (Figure 6.19). The encapsidated L segment specific RNAs observed at passage 18 were however present in low abundance relative to encapsidated genomic S RNA (Figure 6.18). Clearly there was a strong selection for encapsidation of standard viral RNA species in the persistently infected cells. In addition full-length S RNA species predominated over full-length L and M RNA species in intracellular nucleocapsids.



Figure 6.16 Northern blot analysis of encapsidated viral RNA species in the C6/36/BUN cell line. Samples of intracellular-nucleocapsid derived RNA were fractionated on a 1.5% agarose gel containing 5mM methyl mercury hydroxide. The RNA was transferred to a nitrocellulose filter which was hybridized with segment and sense-specific [³²P]-labelled probes: (a) One tenth of the total yield of encapsidated RNA from passage level 18 was probed with [³²P]-labelled cDNA probes (L7, M6 and S9) specific for positive-sense Bunyamwera virus L, M and S RNA species; (b) One tenth of the total yield of encapsidated RNA from passage levels 18 and 21 was probed with [³²P]-labelled riboprobes specific for Bunyamwera virus genomic L, M and S RNA species.



D

Genomic S RNA

Figure 6.17. Northern blot analysis of nucleocapsid-derived RNA from passage levels 18 and 21 of the C6/36/BUN cell line. One tenth of the total yield of nucleocapsid-derived RNA from each passage level was fractionated on a 1.5% agarose gel containing 5mM methylmercury hydroxide. The RNA was transferred to a nitrocellulose filter which was hybridized with a [³²P]-labelled riboprobe specific for Bunyamwera virus genomic S RNA.



Figure 6.18. Northern blot analysis of (a) total cell RNA and (b) nucleocapsid-derived RNA from passage levels 18, 19, 21 and 22 of the C6/36/BUN cell line. $12\mu g$ of total cell RNA (one twenty-fifth of the total yield), and one tenth of the total yield of nucleocapsid-derived RNA, from each passage level were fractionated on a 1.5% agarose gel containing 5mM methylmercury hydroxide. The RNA was transferred to a nitrocellulose filter which was hybridized sequentially first with a Bunyamwera virus genomic L segment specific riboprobe and then a genomic S segment specific riboprobe. Both riboprobes were labelled with [³²P].



Figure 6.19. Northern blot analysis of nucleocapsid derived RNA from passage level 18 of the C6/36/BUN cell line. One tenth of the total yield of intracellular-nucleocapsid derived RNA from passage level 18 was fractionated on a 1.5% agarose gel containing 5mM methylmercury hydroxide. The RNA was transferred to a nitrocellulose filter which was hybridized with a [³²P]-labelled riboprobe specific for Bunyamwera virus genomic L RNA.

The detection of encapsidated full-length and defective L RNA species at passage level 18 (albeit at low levels relative to the levels of encapsidated genomic S RNA), coincided with relatively high levels of positive-sense S RNA in the C6/36/BUN cell line (see Figure 6.11). High levels of positive-sense S RNA probably represent high levels of S mRNA, and if viral mRNAs were efficiently translated in the persistently infected cells at this time, relatively high levels of N protein would be present in the cells at passage level 18. This would relieve any competition for encapsidation which might exist in the persistently infected cells when N protein availability is limited. Elliott and Wilkie (1986) reported that levels of Bunyamwera virus N protein did fluctuate between passage levels of their C6/36-PI HI and C6/36-PI LO cell lines. Thus the changing profiles of encapsidated RNA species in the persistently infected cells may to some extent reflect the relative abundance of the viral N protein at each passage level.

6.10 Summary

Direct RNA sequence analysis of the S RNA segment of viruses plaque-purified from culture fluid of the C6/36-PI LO cell line (Elliott & Wilkie, 1986) revealed a low level of genetic drift in the S RNA of Bunyamwera virus during persistent infection of C6/36 cells. The variant N protein-containing viruses reported to arise during propagation of the persistently infected C6/36-PI LO cell line (Elliott & Wilkie, 1986) were determined to be contaminating temperature-sensitive bunyaviruses, not endogenous to the persistently infected cell line.

A new C6/36 cell line persistently infected with Bunyamwera virus (C6/36/BUN) was established, and passaged for over a year without observable cpe. Once the persistent infection was established, levels of released virus fluctuated between 10^2 and 10^7 pfu/ml. Small, cloudy plaque-phenotype variants were detected in the culture fluid of the C6/36/BUN cell line before passage 5 and predominated by passage 15. An infectious centres assay carried out at passage 6 indicated that 2% of the cells in the persistently infected culture shed infectious virus.

Levels of intracellular viral genomic RNA varied between passage levels. Each species of viral RNA varied independently and no simple correlation between the fluctuating levels of viral RNA and titres of released virus was evident. Levels of positivesense S RNA varied independently of levels of genomic S RNA. Positive-sense S RNA was detected at all passage levels with maximum levels at passages 8, 9 and 18.

Subgenomic L segment derived RNAs were detected in the C6/36/BUN cell line. These defective RNAs were present at low levels at passage 2 and were dominant over fulllength viral RNA species by passage 58. Aberrant S RNA species which were larger than the standard genomic RNAs were detected at passages 19, 21 and 22. Full-length S RNA species (both positive- and negative-sense) were preferentially encapsidated over full-length L and M segment RNAs. However the aberrant S RNAs detected at passage levels 19, 21 and 22 were never observed in nucleocapsid structures.

The defective L RNAs were inefficiently encapsidated by the viral N protein, and were observed in nucleocapsid structures only at passage level 18, coinciding with high levels of positive-sense S RNA. It is possible that high levels of positive-sense S RNA represent elevated levels of S mRNA and thus increased levels of N protein, which may allow encapsidation of the defective L RNAs in addition to the more efficiently encapsidated viral RNA species (in particular full-length S RNA). Therefore the availability of N protein, rather than the absence of a specific encapsidation signal, may determine whether or not defective viral RNAs are encapsidated and possibly packaged into defective virions. The observation that full-length Bunyamwera virus S RNA is the most efficiently encapsidated viral RNA species in persistently infected C6/36 cells agrees with the observation that virions, containing predominantly S RNA, are released from persistently infected C6/36 cells (Elliott & Wilkie, 1986).

7

Analysis of cell lines cloned from the C6/36/BUN culture

The C6/36 cell line used in the course of this work was cloned by Igarashi (1978) from an *Aedes albopictus* cell line derived from the macerated tissues of mosquito larvae (Singh, 1967). While the original uncloned Singh cell line was likely to have consisted of a mixed cell population containing partially differentiated cells which were destined for particular tissues of the mature mosquito, the cloned C6/36 cell line was expected to consist of a homogeneous cell population. However, microscopic examination of our uninfected (C6/36) and persistently infected (C6/36/BUN) mosquito cell cultures revealed variation in cell morphology within both cultures. This variation may have been due to individual cells being at different stages in the cell cycle, but may also have reflected cell heterogeneity within the cultures. Variations within other clonal cell lines have been reported (Varma et al., 1979).

Cell heterogeneity can arise during the course of a persistent infection, as exemplified by the rapid generation of cell heterogeneity in cultures of cloned BHK-21c1 cells persistently infected with FMDV (de la Torre et al., 1985). To determine if there was heterogeneity in the response of individual cells within the C6/36/BUN culture to persistent Bunyamwera virus infection, cells were cloned from four different passages of the C6/36/BUN cell line: pass 12 (p12), pass 20 (p20), pass 30 (p30) and pass 58 (p58). Those cells which gave rise to stable cloned cell lines were tested for their ability to produce infectious virus and to resist superinfection by wtL₉BUN virus. In addition the levels of viral RNA resident in each of the cloned cell lines derived from p20 and p58 of the C6/36/BUN culture were determined.

7.1 The ability of cells cloned from p12, p20, p30 and p58 of the C6/36/BUN culture to give rise to stable cell lines

Cells were cloned directly from the ongoing pass 58 of the C6/36/BUN cell line, while cells from the earlier passes (p12, p20 and p30) had to be revived from liquid nitrogen stocks. The cells recovered well from storage and confluent (80cm²) monolayers were grown within one week. The resultant cell lines, p12', p20' and p30' continued to grow well and were subcultured at approximately weekly intervals; at no time was generalized cpe observed. The release of small, cloudy plaque-phenotype virus from the cells and the detection of Bunyamwera virus genomic RNA in these cells indicated that the cultures remained persistently infected (Figure 7.1).

Initial attempts to clone individual cells from p58 by seeding single cells into individual wells of a 96 well plate were unsuccessful. Numerous wells were successfully seeded with single cells, but only wells seeded with two or more cells gave rise to monolayers. Isolated single cells were prone to cytolysis. It would be interesting to



Figure 7.1. Levels of Bunyamwera virus genomic RNA resident in cells revived from frozen stocks (originating from passes 12, 20 and 30 of the C6/36/BUN cell line). Total cell RNA was prepared from each of the revived cell lines and aliquots of RNA were applied in duplicate to nitrocellulose using a vacuum manifold. Replicate strips of nitrocellulose were then hybridized either with a mixture of [³²P]-labelled cDNA probes specific for each Bunyamwera virus genomic RNA segment (i.e. to detect negative-sense L, M and S RNA) or with a [³²P]-labelled actin gene cDNA probe to estimate the relative amount of RNA applied with each sample.

determine if cell death in these instances was due to virus induced lysis of isolated cells, but this question was not persued during the course of this work.

Clonal cultures were obtained by seeding 100mm diameter tissue culture plates with at least 20 cells from a single-cell suspension (see section 3.9.3). The cells were left at 28°C for two weeks after which time colonies of cells were visible to the eye. The colonies were then transferred to individual wells of a 24 well plate. After transfer most of the clones, but not all, grew well giving rise to stable cloned cell lines. Some variation was observed in the rate of growth of individual clonal cell lines, for instance the clones derived from p20' grew more slowly (at least initially) than clones from p12', p30' and p58. A small minority of the cloned cell lines exhibited cpe. One of the p58 clones (p58c9) displayed extensive cpe after transfer to the multiwell plate, but recovered over the course of 10 days and gave rise to a stable cell line. Two out of 18 clones derived from p20 failed to establish stable cell lines but exhibited extensive cpe leading to total cell destruction after transfer to the multiwell plate. 100% of the cloned cells from p12' and p30' gave rise to stable cell lines without any signs of generalised cpe.

7.2 Virus release from the cloned cell lines

Samples of supernatant culture fluid from the revived parental cell lines p12', p20'and p30', from pass 66 of the C6/36/BUN culture and from each of the cloned cell lines were assayed for infectious virus. The culture fluid was sampled 48 h after subculture of the cell lines and was titrated on BHK cell monolayers (see Table 7.1). Only small, cloudy plaque-phenotype virus was detected. Overall, the proportion of the cloned cell lines which released detectable levels of virus decreased with increasing passage level of the parental cell lines. p12' and p30' released similar levels of virus (<1 x 10^2 pfu/ml) however, 57% of p12' derived clones released detectable levels of virus compared to only 6% of p30' derived clones. At p66 no virus was detected in the culture fluid of the C6/36/BUN culture by plaque assay on BHK cell monolayers. However, supernatant culture fluid from p55 of C6/36/BUN cell line did produce cpe in BHK cell monolayers. All of the cloned cell lines derived from p20' released relatively high levels of virus (on average 2 x 10^5 pfu/ml). Given the irregular fluctuations in the titre of virus released over the first 30 passages of the C6/36/BUN culture (see Figure 6.7) it is not suprising that variation was also observed in the levels of virus released from each set of cloned lines.

7.3 Determination of the susceptibility of the cloned cell lines to superinfection with wtL₉BUN virus

Each of the cloned cell lines as well as the parental cell lines p12', p20', p30' and the ongoing pass of the C6/36/BUN cell line (p66) were tested for their ability to support the replication of wtL₉BUN virus. 48 h after superinfection, the supernatant culture fluid was harvested and titrated on BHK cell monolayers, along with control culture fluid from

	Virus release from the cloned cell lines (pfu/ml)					
	p12'	p20'	p30'	p58		
Clone 1	1 x 10 ²	2 x 10 ³	•			
Clone 2		3 x 10 ⁵				
Clone 3	1 x 10 ²	1 x 10 ⁵		· · · · · ·		
Clone 4		3 x 10 ⁵				
Clone 5		6 x 10 ⁵				
Clone 6		1 x 10 ⁵				
Clone 7	< 1 x 10 ²	2 x 10 ⁵	<u> </u>			
Clone 8		3 x 10 ⁵				
Clone 9		2 x 10 ⁵				
Clone 10	< 1 x 10 ²	2 x 10 ⁴				
Clone 11	$< 1 \ge 10^{2}$	2 x 10 ⁵		10		
Clone 12	< 1 x 10 ²	1 x 10 ⁵		nih		
Clone 13	< 1 x 10 ²	4 x 10 ⁵		N/A		
Clone 14	$< 1 \ge 10^{2}$	10	10	N/A		
Clone 15	N/A	8 x 10 ⁵		N/A		
Clone 16	N/A	1 x 10 ⁵		N/A		
Average titres of		philip de tra	240-1-2005			
virus released from	$< 1 \times 10^{2}$	2 x 10 ⁵				
the clones						
% of cloned cell		12-14-14-15	Alar Astrony			
lines releasing	57%	100%	6%	8%		
virus			A. S.			
Virus release from	(p12')	(p20')	(p30')	(p66)		
parental cell lines	$< 1 \ge 10^2$	6 x 10 ⁴	$< 1 \times 10^{2}$	<u> </u>		

Table 7.1. Virus titres in the supernatant culture fluid of each of the cloned cell lines 48 h after subculture. The indistinct plaque-phenotype of the virus produced by the persistently infected cells made accurate counting difficult and therefore the titres given are approximate. The limit of detection of the plaque assay was 10 pfu/ml (----- denotes levels of virus below the limit of detection; N/A, not applicable).

non-superinfected cultures. Cell-released wtL₉BUN virus was easy to distinguish from background virus released from the persistently infected cultures due the marked contrast in plaque-phenotype. The results of the assay are displayed in Table 7.2.

The parental cell lines p12' and p30' displayed significant resistance to superinfection with wtL₉BUN virus and produced 9 x 10^2 and 8 x 10^2 pfu/ml of wtL₉BUN virus respectively, compared to 1.4 x 10⁹ pfu/ml produced by uninfected C6/36 cells. However, the stable cloned cell lines derived from p12' and p30' did not display this degree of resistance; the most marked reduction in virus yield compared to uninfected cells was a 2 Log reduction exhibited by two p12' clones, p12'c11 and p12'c12. The parental cell line p20' also displayed resistance to superinfection although not to the same extent as p12' and p30'; superinfected p20' cells produced 1.1 x 10⁵ pfu/ml of wtL₀BUN virus, 4 Logs less than uninfected C6/36 cells, but 3 Logs above the levels of virus produced by the p12' and p30'cell lines. However, all of the cloned cell lines derived from p20' were resistant to superinfection, although again to varying degrees. By pass 66 of the C6/36/BUN cell line the persistently infected culture no longer displayed resistance to superinfection with homologous virus but supported the replication of wtL_oBUN virus, producing relatively high yields $(1.5 \times 10^8 \text{ pfu/ml})$. The levels of the wild-type virus produced by p66 of the C6/36/BUN cell line were one Log lower than the yields from the cloned cell lines derived from p58.

7.4 Presence/absence of Bunyamwera virus RNA in the cloned cell lines

After 10 passages of the cloned cell lines, total cell RNA was extracted from each of the lines derived from p20' and p58 using the method of Chomczynski and Sacchi (1987). The levels of viral genomic RNA/cell in each cell line were compared by performing parallel hybridizations with virus and cell specific RNA probes. Duplicate samples of RNA (of approximately 8μ g) were immobilised onto nitrocellulose using a Schleicher and Schuell Minifold II slot blot apparatus and were hybridized either with a mixture of [³²P]-labelled cDNA probes specific for Bunyamwera virus L, M and S genomic RNA or with a [³²P]-labelled probe specific for actin mRNA. Determination of the relative levels of actin mRNA in each sample (assuming that actin mRNA levels remain relatively constant) enabled variations in the quantity of total cell RNA applied to the nitrocellulose to be monitored and corrected for. After autoradiography the slot blots were scanned by densitometry and variations in the levels of viral genomic RNA were normalized against actin mRNA levels. The results are presented graphically in Figure 7.2.

Variation was observed in the levels of viral RNA resident in individual clones derived from the same passage level of the C6/36/BUN cell line. Seven of the sixteen p20' clones hybridized strongly for viral genomic RNA, while only one of the fifteen p58 derived cloned cell lines appeared to contain relatively high levels of viral RNA. In

	Yields of wtL9BUN virus (pfu/ml)					
	p12'	p20'	p30'	p58		
Clone 1	3 x 10 ⁸	2 x 10 ⁵	1 x 10 ⁹	4 x 10 ⁹		
Clone 2	2 x 10 ⁸	3 x 10 ⁴ *	2 x 10 ⁹	2 x 10 ⁹		
Clone 3	7 x 10 ⁸	4 x 10 ⁷ *	2 x 10 ⁹	3 x 10 ⁹		
Clone 4	3 x 10 ⁸	6 x 10 ⁴	2 x 10 ⁹	3 x 10 ⁹		
Clone 5	1 x 10 ⁹	3 x 10 ⁴	4 x 10 ⁸	2 x 10 ⁹		
Clone 6	3 x 10 ⁸	3 x 10 ⁴	2 x 10 ⁹	2 x 10 ⁹		
Clone 7	1 x 10 ⁹	6 x 10 ⁴	7 x 10 ⁸	3 x 10 ⁹		
Clone 8	9 x 10 ⁸	1 x 10 ⁴	7 x 10 ⁸	4 x 10 ⁹		
Clone 9	5 x 10 ⁸ *	1 x 10 ⁷ *	9 x 10 ⁸	4 x 10 ⁹		
Clone 10	2 x 10 ⁹ *	2 x 10 ⁴	ND	2 x 10 ⁹		
Clone 11	3 x 10 ⁷ *	8 x 10 ⁴	3 x 10 ⁹	5 x 10 ⁹		
Clone 12	7 x 10 ⁷ *	4 x 10 ⁴	5 x 10 ⁸	1 x 10 ⁹		
Clone 13	8 x 10 ⁸ *	2 x 10 ⁴	7 x 10 ⁸	N/A		
Clone 14	1 x 10 ⁹ *	3 x 10 ⁴	3 x 10 ⁹	N/A		
Clone 15	N/A	4 x 10 ⁴	1 x 10 ⁸	N/A		
Clone 16	N/A	5 x 10 ⁴	1 x 10 ⁹	N/A		
Parental	(p12')	(p20')	(p30')	(p66)		
cell lines	9 x 10 ²	1 x 10 ⁵	8 x 10 ²	2 x 10 ⁸		
Yield from uninfected Aedes albopictus C6/36 cells = 1.4 x 10 ⁹ pfu/ml						

Table 7.2. Susceptibility of the parental and cloned cell lines to superinfection with wtL₉BUN virus, expressed as extracellular levels of wild-type Bunyamwera virus in the supernatant culture fluid (pfu/ml). Bold type denotes a greater than 4 log reduction in virus yield relative to uninfected C6/36 cells; ND (no data); N/A (not applicable). The cloned and the parental cell lines were passaged a number of times prior to the superinfection assay: p12' derived clones c1 to c8 and all of the p20' derived clones apart from c2, c3 and c9 were passaged 3 times; p12' derived clones c9 to c14 and p20' c2, c3 and c9 were passaged 7 times (*), while p30' and p58 derived clones went through 9 and 12 passages respectively. The revived parental cell lines: p12', p20'and p30' were passaged 12 times prior to the superinfection assay. The ongoing pass of the C6/36/BUN cell line (p66) was also tested for resistance to superinfection.



Figure 7.2. Relative levels of Bunyamwera virus genomic RNA, relative to actin mRNA, resident in each of the cloned cell lines derived from pass 20' and pass 58 of the C6/36/BUN culture. RNA levels were determined by densitometric analysis of slot blot data. The slot blot analysis was carried out on duplicate samples of total cell RNA (of approximately 8µg) which were immobilised onto nitrocellulose using a vacuum manifold. Replicate strips of nitrocellulose were then hybridized with a mixture of [32P]-labelled cDNA probes specific for Bunyamwera virus genomic RNA, or with a [³²P]-labelled actin gene cDNA to estimate the relative amount of RNA applied in each sample. general, the levels of viral RNA observed in the p20' clones were significantly higher than for the p58 clones and while most of the p20' derived cloned cell lines clearly harboured viral RNA sequences some of the cloned cell lines derived from passage 58 gave weak hybridization signals which were not significantly above background levels (see Figure 7.2).

Four of the p20' cloned cell lines (c1, c3, c5 and c8), which represented cultures containing high (c1 and c8) and low (c3 and c5) levels of Bunyamwera virus genomic RNA, were selected for Northern analysis to determine the relative levels and the nature of the genomic L and S RNA species resident in each of these cell lines (Figure 7.3). Full-length Bunyamwera virus genomic L RNA was present at significant levels in p20' c5 and p20' c8 and at very low levels relative to the defective L RNA in p20' c1 while the levels of full-length genomic L RNA in p20' c3 were too low to be detected by Northern analysis. However, each of the four cloned lines contained a single species of defective L RNA. For three out of the four clones the defective L RNA predominated over full-length genomic L RNA but in p20' c8 similar levels of both L RNA species were present. Three distinct size classes of subgenomic L RNA were observed amongst the four cloned cell lines.

The levels of actin mRNA contained in each of the RNA samples varied (Figure 7.3c), indicating that more total cell RNA was examined from p20' c3 and p20' c5 than from p20' c1 and p20' c8. Consequently the viral RNA species in p20' c3 and p20' c5 were over represented on the Northern blots relative to p20' c1 and p20' c8. Taking this into account the relative levels of Bunyavirus genomic L and S RNAs observed in the cloned cell lines by Northern analysis were in overall agreement with the data from the slot blot hybridizations carried out with partial [³²P]-labelled cDNA probes (Figure 7.2). There may be some discrepancy between the results for p20' c5 which could be a consequence of the different probes used, since it is possible that the small defective L RNA in p20'c5 might not have been detected by the [³²P]-labelled cDNA probe used for the initial slot blot analysis. The cDNA probe was specific for sequences between nucleotides 997 and 1436 of Bunyamwera virus genomic L RNA (see Section 3.5.3) which may be deleted from the defective L RNA in the p20' c5 cell line. It would be interesting to hybridize a Northern blot of p20' c5 RNA with the partial probe to determine if this region is deleted from the defective L RNA.

Full-length Bunyamwera virus genomic S RNA was detected in the p20' c5 and p20' c8 cell lines but not in the other two cloned cell lines (Figure 7.3b). The presence of detectable levels of full-length genomic S RNA coincided with the highest levels of full-length genomic L RNA, although the ratio of Bunyamwera virus negative-sense L RNA species to negative-sense S RNA species varied between the clones. In p20' c8, genomic S RNA was more abundant relative to both of the L RNA species compared to the RNA profile for p20' c5.



to a nitrocellulose filter using a vacuum manifold. This filter was hybridized with a [32P]-labelled actin gene cDNA probe p20'c3, p20'c5 and p20'c8; derived originally from passage 20 of the C6/36/BUN cell line. Duplicate samples of total cell hydroxide. The RNA was then transferred to nitrocellulose filters which were hybridized with [32P]-labelled riboprobes specific for Bunyamwera virus genomic L RNA (a) or genomic S RNA (b). An equal volume of each sample was applied to estimate the total amount of RNA applied in each sample (c). Total cell RNA from passage 21 was included in each blot RNA from each of these cloned cell lines were fractionated on 1.5% agarose gels containing 5mM methylmercury Figure 7.3 Northern analysis of Bunyamwera virus genomic L and S RNA species resident in the cloned cell lines: p20'c1. to function as a positive control and a marker track Superinfection experiments (see Section 7.3) showed that clones p20' c5 and p20' c8 yielded less virus $(3x10^4 \text{ pfu/ml} \text{ and } 1x10^4 \text{ pfu/ml} \text{ respectively})$ than p20' c1 $(2x10^5 \text{ pfu/ml})$ and p20' c3 $(4x10^7 \text{ pfu/ml})$; the yield from normal C6/36 cells was $1x10^9 \text{ pfu/ml}$. Hence resistance of the persistently infected cells to superinfection with Bunymawera virus appeared to correlate with levels of full-length viral RNA present in the cells and not with the amount or occurence of defective L RNAs.

7.5 Summary

Stable cell lines were established from cells cloned from passages 12, 20 and 30 of the persistently infected C6/36/BUN cell line, which had been stored frozen. The revived cell lines (p12', p20' and p30') remained persistently infected as judged by their resistance to superinfection with wtL₉BUN virus, the presence of intracellular viral RNA sequences and the release of small, cloudy plaque-phenotype virus. All of the cell lines were passaged a number of times before they were analysed and consequently no longer represented passages 12, 20 and 30 of the original C6/36/BUN cell line.

Levels of virus released from the p12', p20' and p30' parental cell lines varied from below 1×10^2 pfu/ml for the p12' and p30' cell lines, to 6×10^4 pfu/ml for the p20' cell line. The percentage of virus producing clones derived from each parental cell line also varied; p12' (57%), p20' (100%), p30' (6%) and p58 (8%).

The ability of the parental and cloned cell lines to support the replication of wtL₉BUN virus was tested in superinfection experiments. wtL₉BUN virus replicated to high titres in cells from passage 66 of the C6/36/BUN culture and in cloned cell lines derived from passage 58 of the C6/36/BUN cell line, suggesting that the persistently infected culture had lost its ability to exclude superinfecting homologous virus at these higher passage levels. However the p12', p20' and p30' cell lines, and the cloned cell lines derived from them, displayed a range of magnitude of resistance to superinfection. The results suggested that (at least at relatively early passage levels) resistance to superinfection varied just as levels of released virus and levels of intracellular viral RNA varied in the persistently infected culture (see Chapter 6). No simple correlation between levels of production of the persistent, cloudy plaque-phenotype virus and superinfectability of the cell cultures was evident.

The level of viral genomic RNA resident in each of the p20' and p58 cloned cell lines was measured by slot blot analysis. In general, higher levels of viral RNA were detected in the p20' clones. Seven out of sixteen p20' clones contained relatively high levels of viral genomic RNA compared to one out of fifteen of the p58 clones. Northern blot analysis of RNA extracted from four of the p20' clones (p20' c1, p20' c3, p20' c5 and p20' c8) revealed that each clone harboured a single prominent species of defective L RNA. However, resistance to superinfection appeared to correlate more with the level of fulllength viral RNAs resident in the cloned cell lines than with the amount or occurence of defective L RNAs.

.

8 Discussion

During the course of this work, the molecular mechanisms leading to the establishment of persistent Bunyamwera virus infection of *Aedes albopictus* C6/36 cells, and the factors responsible for the maintenance of the persistent infection, were investigated. In this section, the results presented in the preceeding chapters will be discussed in the light of the findings of other research groups, particularly during the period since the commencement of this work. Possible directions for future research will be outlined.

8.1 Establishment versus maintenance of persistent viral infections

For some time it has been recognised that the establishment and maintenance of persistent viral infections should be considered independently. Recently, Borzakian *et al.* (1992) demonstrated that the establishment and maintenance of poliovirus persistence in a non-neural human cell line (HEp-2c) involve distinct mechanisms. Wild-type as well as attenuated strains of poliovirus can establish persistent infection in human neuroblastoma cells, and poliovirus mutants with altered biological properties can be isolated from these cells (Colbere-Garapin *et al.*, 1989). These variants have a highly mutated genome and an altered cell specificity, and can induce a secondary persistent infection in cells of non-neural origin (Pelletier *et al.*, 1991). Borzakian *et al.* (1992) used poliovirus mutants, derived from wild-type and attenuated strains of poliovirus by selection in neuroblastoma cells, to establish persistent infections in HEp-2c cells. They determined that while the maintenance of the persistent infection was dependent upon the evolution of mutant host cells within the persistently infected culture in which poliovirus multiplication was restricted.

In the above study, viral variants were used to establish the persistent infection. Other popular means of establishing persistent viral infections with RNA viruses are to coinfect cells with defective interfering particles and homologous standard virus preparations, or to use serially passaged stocks of virus and/or a high moi to initiate the infection. Each of these procedures increases the probability of the occurence of interference between standard and defective virus in the infected cells, which could then lead to persistence (see Section 2.2.6.4 and references therein). The widely documented use of defective interfering virus particles to establish persistent viral infections (e.g.Roux & Holland, 1979; Meinkoth & Kennedy, 1980; Weiss *et al.*, 1980; Weiss *et al.*, 1983; Cave *et al.*, 1985; Atkinson *et al.*, 1986; Poidinger *et al.*, 1991) and reports on the generation of defective interfering (DI) particles during the course of persistence in a number of *in vivo* and *in vitro* systems (Nuesch *et al.*, 1988; Nuesch *et al.*, 1989; Moscana, 1991) suggest that DI particles may have a role to play in many persistent viral infections. However, arboviruses establish persistent infections in invertebrate cell cultures without coinoculation of standard and defective viruses. When defective viral RNAs or DI particles have been detected during these infections, the defective species were usually not evident until well after the establishment of the persistent infection (Igarashi *et al.*, 1977; Eaton, 1981; Kascsak & Lyons, 1978; Elliott & Wilkie, 1986). There have been reports of the detection of Semliki Forest virus (SFV) DI RNAs within 24 h of infection of *Aedes albopictus* C6/36 cells (Tooker & Kennedy, 1981; Stalder *et al.*, 1983). However, since cloned virus stocks were not used in these studies, the possibility exists that the defective SFV RNAs were introduced with the virus inoculum.

In the system described in this thesis, working stocks of biologically cloned Bunyamwera virus readily established persistent infection in Aedes albopictus C6/36 cells. The newly initiated persistent infection mimicked that reported previously (Elliott & Wilkie, 1986), demonstrating the reproducibility of the bunyavirus system. Working stocks of Bunyamwera virus were prepared by a total of two low multiplicity passages, in BHK cells, of a triply-plaque purified viral isolate (wtL₂BUN). The low moi used, and the limited passage history of the wtL₉BUN virus preparation, were designed to limit the possibility of introducing defective virus into the stocks. In addition, defective viral RNAs could not be detected in the initial virus stocks by Northern blot analysis of virion RNA. The consensus sequence of the genomic S segment RNA of this virus preparation (which was obtained by primer extension sequencing using virion RNA as template and consequently represented virus which had undergone a total of three low multiplicity passages in BHK cells) revealed that the virus preparation remained homogeneous after these low multiplicity passages. The readiness with which this biologically cloned Bunyamwera virus preparation established a persistent infection of Aedes albopictus C6/36 cells, at a relatively low moi (5 pfu/cell), suggests that neither defective interfering virus nor viral variants are essential to the initiation of this persistent infection. Similar conclusions have been reached for the establishment of persistent La Crosse virus infection of Aedes albopictus C6/36 cells (Rossier et al., 1988; Hacker et al., 1989). However, certain characteristics of the persistently infected C6/36/BUN cell culture described herein, and of mosquito cell cultures persistently infected with other bunyaviruses, such as fluctuating titres of released virus and resistance to superinfection with homologous virus, suggest that DI genomes may arise once the persistent infection has been established (Newton et al., 1981; Elliott & Wilkie, 1986; Holland, 1990; see Section 8.4).

8.2 Molecular events underlying the establishment of persistent viral infections

One of the difficulties encountered in studying persistent viral infections is distinguishing mechanisms fundamental to the maintenance of the persistent state from other phenomena which arise due to the long-term residence of virus in the host cell, but which are not essential to the maintenance of the persistent infection. It therefore seemed pertinent to consider the early molecular events underlying the establishment of Bunyamwera virus persistence in Aedes albopictus C6/36 cells. Arboviruses are particularly valuable in this respect, as comparative analyses of the replication of these viruses in vertebrate and invertebrate cell cultures facilitates a direct comparison of the molecular mechanisms leading to the expression of the lytic and persistent phenotypes (respectively) of the same virus. The Alphaviridae and Bunyaviridae together account for most of the described arboviruses. More extensive molecular studies have been carried out on alphavirus persistence in invertebrate cell culture than on bunyavirus persistence, but in the past five years the balance has been redressed to some extent by a number of publications on La Crosse bunyavirus persistent infection of Aedes albopictus C6/36 cells (Rossier et al., 1988; Hacker et al., 1989). Besides the work presented here, only one other study has been undertaken to directly compare Bunyamwera virus infection of vertebrate and invertebrate cell lines. Newton et al. (1981) investigated Bunyamwera virus infection of BHK cells and an uncloned mosquito cell line (Aedes albopictus cells) at 37°C and 26°C respectively and followed gene expression, in both infected cell types, by metabolic labelling.

As reported in Chapter 5, detailed examination of viral and host protein synthesis, and of viral RNA synthesis, in Bunyamwera virus-infected BHK and C6/36 cells over 72 h pi, revealed a number of differences underlying the divergent, host-dependent, outcomes of infection. Following infection with Bunyamwera virus, host protein synthesis was inhibited in BHK cells but not in C6/36 cells, in agreement with the observations of Newton et al. (1981). Between 48 and 72 h pi the BHK cell cultures exhibited extensive cytopathic effects. However, the infected mosquito cell cultures continued to grow without any obvious cytopathic effects. The protein synthesis profile obtained for Bunyamwera virus-infected BHK cells was similar to the pattern of protein synthesis observed in La Crosse virus infected BHK cells by Raju & Kolakofsky (1988). In agreement with this report, the onset of the decline in host protein synthesis in Bunyamwera virus-infected BHK cells coincided with maximum levels of viral gene expression. However, in the Bunyamwera virus system, the rapid decrease in host protein synthesis observed between 10 and 16 h pi was not paralleled by an equivalent decline in viral protein synthesis. Viral proteins continued to be synthesised at high levels for a further 12 h (up to 24 h pi). In contrast, Raju & Kolakofsky (1988) reported that viral and cellular protein synthesis declined in parallel following La Crosse virus infection of BHK cells. They determined the steady-state levels of viral and host mRNAs in La Crosse virus-infected BHK cells and concluded that the decline in viral and cellular protein synthesis was due to a general mRNA instability induced in these cells following infection with La Crosse virus.

La Crosse virus infection of BHK cells (Raju & Kolakofsky, 1988) proceeds more rapidly than Bunyamwera virus infection, and the higher multiplicity of infection used by Raju & Kolakofsky (20 to 50 pfu/cell) may also have contributed to the different rates of viral replication observed in the two systems. The slower progress of the Bunyamwera virus infection (initiated at 5 pfu/cell) enabled the decay in viral and cellular protein synthesis to be examined in more detail. In our hands, viral protein synthesis did not decay with the same kinetics as cellular protein synthesis in Bunyamwera virus-infected BHK cells. Whether or not this reflected a greater stability of viral mRNAs in Bunyamwera virus-infected BHK cells relative to host cell mRNAs is open to question. Raju & Kolakofsky (1988) demonstrated that viral mRNA synthesis continued at significant levels in La Crosse virus-infected BHK cells beyond 12 h pi (well after the decline in viral protein synthesis was observed). These authors suggested that, in these cells, the decrease in viral protein synthesis arose due to a net decrease in the rate of viral mRNA synthesis relative to mRNA degradation with time.

Raju & Kolakofsky (1988) presented evidence that the mRNA instability induced in BHK cells following La Crosse virus infection was not mediated via an interferonactivated latent cellular RNase. They demonstrated that the activity reponsible for the degradation was operative only when viral replication was at near maximal levels (in this case at 6 h pi) and was specific for mRNAs. They proposed that the endonuclease activity detected in La Crosse virions (Patterson *et al.*, 1984) might be responsible. The viral endonuclease activity is presumed to be a function of the viral polymerase (L protein) and the fact that the induction of mRNA instability paralleled viral gene expression supports a role for this enzyme in the induction of mRNA instability.

It is interesting, therefore, to speculate on the apparent lack of mRNA instability induction following bunyavirus infection of C6/36 cells. These cells support high levels of viral gene expression during the acute phase of infection (up to 24 h after infection with 5 pfu/cell of wtL₉BUN virus) without any overt detrimental effects upon the host cell. C6/36 cell derived bunyavirus mRNAs have heterogeneous non-templated primers at their 5' end (Hacker *et al.*, 1989), which are presumably derived from host cell mRNAs, as in infected BHK cells. The virion associated endonuclease activity is believed to perform the 'capsnatching' activity responsible for generating primers from host mRNAs (Patterson *et al.*, 1984; Patterson & Kolakofsky, 1984). Therefore the virion associated endonuclease must function in infected C6/36 cells. If this activity is responsible for inducing mRNA instability in bunyavirus-infected mammalian cells what modulates this activity in C6/36 cells ?

The analysis of positive-sense viral RNA species in Bunyamwera virus-infected C6/36 cells and BHK cells (presented in Chapter 5) revealed a relatively low abundance of positive-sense L RNA species in infected C6/36 cells compared to BHK cells. This probably reflects lower levels of the L mRNA in Bunyamwera virus-infected C6/36 cells relative to BHK cells. Lower levels of La Crosse virus L mRNA were also detected in infected C6/36 cells relative to BHK cells relative to BHK cells infected under the same conditions (Rossier *et*

al., 1988). Lower levels of expression of the viral L protein in C6/36 cells would be consistent with mRNA instability in bunyavirus-infected mammalian cells being induced by the viral endonuclease activity. The destruction of host mRNAs would be expected to result ultimately in cell death, as is observed following bunyavirus infection of BHK cells.

Viral RNA replication and gene expression in Bunyamwera virus-infected C6/36 cells declined after 24 h pi, while levels of host protein synthesis remained relatively This observation agreed with that of Newton et al. (1981) regarding constant. Bunyamwera virus infection of uncloned Aedes albopictus cells. Measurement of S mRNA levels, by slot blot and Northern blot analyses, suggested that the levels of viral transcription were not dramatically altered at this stage in the infection. Rossier et al. (1988) reported similar findings for La Crosse virus infection of C6/36 cells. These authors judged that the transition from the acute to the persistent phase of La Crosse virus infection of C6/36 cells took place at 24 h pi. At this time, the synthesis of La Crosse virus genomic RNA and N protein was severely reduced, although S mRNA levels remained high. By pulse-labelling infected cultures with [³H]-uridine, Rossier et al. (1988) determined that S mRNA synthesis declined relatively gradually during the initial days of the persistent infection and was still at 50% of its maximum level at 5 days pi. In contrast, genome and antigenome synthesis declined rapidly and became undetectable by 2 days pi. Levels of La Crosse virus S mRNA were still approximately 65% of the maximum levels (observed on day 1) on day 3 pi. However, despite significant levels of S mRNA synthesis, S mRNA levels fell to low levels, between 3 and 5 days post infection, in the infected C6/36 cells. Presumably the fall in S mRNA levels represented an increased rate of turnover with time.

Northern blot analysis of encapsidated RNA species in Bunyamwera virusinfected C6/36 cells revealed a dramatic decrease in the levels of encapsidated, positivesense L and M RNAs between 24 and 48 h pi. However, encapsidated positive-sense S RNA continued to accumulate throughout the three days monitored (Figure 5.12). Northern blot analysis and *in vitro* translation of CsCl fractionated RNA samples from Bunyamwera virus-infected C6/36 cells revealed that by 48 h pi significant levels of Bunyamwera virus S mRNA were encapsidated. This may account, at least partly, for the continuing accumulation of encapsidated positive-sense S RNA species in the infected C6/36 cells.

Hacker *et al.* (1989) similarly found that within 24 to 48 h of La Crosse virus infection of C6/36 cells, 75% of the S mRNA was encapsidated. The encapsidated S mRNA could not be translated *in vitro* unless the N protein was first dissociated from the RNA, but, once free of N protein, the S mRNA was translatable. These findings are in line with the *in vitro* translation data shown in Figure 5.4; total infected-cell RNA samples were used. The total cell RNA samples were prepared using the acid-phenol-guanidinium-thiocyanate protocol of Chomczynski and Sacchi (1987) which will dissociate RNA-protein

interactions. The high ratio of viral to host protein translation observed *in vitro* compared to *in vivo* using the 24 and 36 h pi RNA samples from Bunyamwera virus-infected C6/36 cells may reflect the release of translatable S mRNA from nucleocapsid structures.

Hacker et al. (1989) examined the S mRNA-N protein complexes in some detail. They determined that the binding of N protein to its own mRNA was highly specific (no host cell mRNAs appeared to be associated with the viral N protein). N protein assembly occurred on preformed mRNAs, rather than concurrently with their synthesis, and in some cases the primers on the mRNAs were also encapsidated. By 16 h pi in La Crosse virusinfected C6/36 cells, significant levels of viral genome replication continued in the absence of new protein synthesis, yet the genomes which were made were all encapsidated. The continued availability of N protein in the absence of detectable levels of N protein synthesis suggested that a pool of N protein had accumulated in La Crosse virus-infected C6/36 cells by 16 h pi. It may be the existence of this N protein pool which facilitates the encapsidation of S mRNA. With time, the S mRNA appeared to become the preferred target for encapsidation. Potentially, some alteration in the S mRNA molecules which are transcribed later in infection may allow them to compete efficiently with the full-length viral RNA species for encapsidation. Assuming that bunyavirus genome replication and encapsidation are coupled, as suggested by studies on La Crosse virus infection of BHK cells (Patterson & Kolakofsky, 1984; Raju and Kolakofsky, 1987a), depletion of the N protein pool might then result in a decline in genome replication. However, the affinity of the N protein for S mRNA does not appear to be particularly strong, since the mRNA remains unencapsidated for up to 4 h after its synthesis (Hacker et al., 1989). It seems more likely that the decline in genome replication is the initial event. The lack of genome synthesis might then lead to accumulation of a pool of unassembled N protein which can interact with the S mRNA. An early event in the establishment of persistent infection of C6/36 cells would therefore appear to be the restriction of genome replication.

Hypothetically, before the viral polymerase (L protein) can synthesise full-length viral RNAs, some factor (viral or cellular) may be required to promote the association of the viral L and N proteins so that genome replication and encapsidation are coupled. Depletion or exhaustion of this putative factor in the infected C6/36 cells may release the L protein from its replicase activity so that it functions solely as a transcriptase. In this situation viral RNA species may still be encapsidated, but not concurrently with their synthesis. It is also possible that a factor involved in the initiation of genome replication (i.e. initiation with ATP rather than with a primer) which modifies the activity of the L protein from a transcriptase to a replicase may be present in limiting amounts in infected C6/36 cells. In this respect, it is interesting to note the relatively low levels of expression of the NSs protein in infected C6/36 cells, compared to infected BHK cells. The function of the NSs protein has not yet been elucidated, but a role in modulating the activity of the

viral polymerase is possible. Alternatively, a factor which inhibits genome replication may accumulate in bunyavirus-infected C6/36 cells.

Encapsidation of the S mRNA may act as a secondary level of control in the persistent infection. By blocking translation of the S mRNA, this association of N protein with S mRNA may prevent cytotoxic effects due to the accumulation of viral N protein, which might begin to interact non-specifically with host mRNAs or become insoluble.

Before the molecular events underlying bunyavirus persistence in C6/36 cells can be understood, we need to know more about the molecular biology of genome replication and transcription of viral mRNAs in this system. In particular, a detailed understanding of the function of the viral polymerase and the role of any accessory factors (viral or cellular) will pave the way towards elucidating the mechanisms fundamental to the establishment of persistence. The sequence of the Bunyamwera virus L segment is available (Elliott, 1989a), and functional Bunyamwera virus L protein has been expressed from recombinant vaccinia viruses (Jin & Elliott, 1991). More recently mutagenesis studies, using Bunyamwera virus L gene-vaccinia virus recombinants, have highlighted regions of the L protein which are functionally important (Jin & Elliott, 1992). The Bunyamwera virus M segment proteins (G1, G2 and NSm) and S segment proteins (N and NSs) have also been cloned and expressed (Nakitare, 1992 (PhD thesis); R. M. Elliott, personal communication). Thus, the individual constructs required to develop a system in which Bunyamwera virus RNA synthesis is driven by plasmid generated viral proteins are available and could facilitate the elucidation a^{ot} the role of each of the Bunyamwera virus proteins in RNA synthesis. The development of an infectious Bunyamwera virus cDNA system is also underway.

Relatively recently, Curran *et al.* (1991) proposed a model for Sendai virus RNA replication, based on deletion studies using an *in vivo* transcription system. Within this model there is potential for negative-regulation of genome replication, which could lead to persistent infection. These authors determined that the Sendai virus P protein is a modular protein which is essential for genome replication. The N-terminal domain appears to bind the viral polymerase (L) protein, whereas the C-terminal domain binds the viral nucleoprotein (NP). Earlier studies revealed that the P gene of Sendai virus expresses up to eight proteins; two of these proteins (V and W) are translated from edited mRNAs but have N-terminal domains identical to the N-terminal domain of the P protein (Pelet *et al.*, 1991). The V protein had been suspected to play a role in RNA synthesis since it contains the N-terminal half of P and is strongly conserved across the majority of paramyxoviruses. Contrary to expectations that V might enhance RNA synthesis, genome replication was inhibited in a dose dependent manner when V was co-expressed with NP, P and L (the later three being essential for paramyxovirus genome replication). It appeared that both V and W could bind to nucleocapsids and inhibit the functional interaction between P protein and

nucleocapsids. It is possible that V and/or W are components of a negative-regulatory mechanism which restricts viral replication during the establishment of persistent infection.

Paramyxovirus genome replication is thought to differ from transcription in its requirement for unassembled NP protein. During genome replication, NP protein initiates nucleocapsid assembly on the nascent leader RNA. Concurrent encapsidation of the growing nascent RNA suppresses polymerase termination signals at the gene junctions, so that full-length viral RNAs are synthesised as opposed to mRNAs. From studies with vesicular stomatitis virus, N is thought to function in genome replication as a complex with P (Peluso & Moyer, 1988; Howard & Wertz, 1989).

Curran *et al.* (1991) proposed that for Sendai virus RNA synthesis, P must contact both the NP of the template and the L protein. Their model makes two assumptions: firstly one L protein molecule must interact with two molecules of P and, secondly, NP protein must be associated with the P protein before it will assemble nascent RNA. When the levels of unassembled NP are low, both P proteins (associated with the L protein) would contact the assembled NP of the genome template as RNA synthesis is initiated. In this situation the polymerase (L) would recognise the gene termination signals, on the nascent RNA, yielding leader and mRNAs. When the level of unassembled NP protein rises, P-NP complexes can form off the template. The availability of these 'free' P-NP complexes means that the nascent RNA can be encapsidated concurrently with its synthesis and consequently full-length viral RNA molecules will be synthesised. V and W could potentially interfere with the P-L complex by sequestering L in a form which is nonfunctional for genome replication. Thus the relative levels of the Sendai virus V and P proteins might control genome replication. Hypothetically, these levels may be controlled by the frequency of RNA editing.

The relative ratios of paramyxovirus mRNAs can also be influenced by the host cell. Paramyxoviruses display a gradient of decreasing mRNA abundance with increased distance of the gene from the 3' end of the genome template. The steepness of this gradient differs in different host backgrounds (Cattaneo *et al.*, 1987; Schneider-Schaulies *et al.*, 1989). The gradient is relatively shallow during acute infection of epithelial cells and fibroblasts but steep in cells such as neurons, in which infection is slow and persistent. Since the L and P genes are distantly spaced on the paramyxovirus transcriptional map, their ratios will be strongly affected by changes in this gradient. The ability of V to inhibit RNA synthesis is dependent upon limiting levels of L. It is possible, therefore, that V exerts a strong negative effect on RNA replication only when L levels are relatively low, such as in neurons. This fits with the pattern of persistence observed with paramyxoviruses: acute infection of epithelial cells and persistent infection of neurons.

8.3 Analysis of *Aedes albopictus* C6/36 cells persistently infected with Bunyamwera virus

Preliminary sequencing studies revealed that Bunyamwera virus undergoes minimal genetic drift during persistent infection of C6/36 cells. The S RNA segments of two viral isolates (8.6 and 18.10), which were plaque-purified from passage levels 8 and 18, respectively, of the C6/36-PI LO cell line, established by Elliott & Wilkie (1986), were sequenced by primer extension sequencing of virion RNA. No nucleotide changes were evident in the 8.6 isolate and a single nucleotide substitution was detected in the 18.10 isolate. The mutation in the 18.10 isolate resulted in conservative amino acid changes in both the N and NSs proteins of this virus. Neither of these amino acid substitutions occurred in conserved regions of the N or NSs proteins (see Section 6.3). It is possible that the constraints on the evolution of the S RNA segment are greater than on the other genome segments, due to the overlapping reading frames encoded in the S RNA. The rate of evolution of the Bunyamwera virus genome as a whole may therefore be greater than is suggested by the relative stability of the S RNA. It has, however, been suggested that arboviruses exhibit a slower rate of evolution relative to non-arthropod-borne RNA viruses (Bilsel et al., 1988; Baldridge et al., 1989), and Poidinger et al. (1991) reported that Murray Valley encephalitis flavivirus underwent few genomic changes during 300 days of persistent infection in Vero cells. These results are in contrast to the rapid rates of virus evolution observed in cell cultures persistently infected with FMDV, VSV and influenza virus (de la Torre et al., 1988; Depolo et al., 1987; Frielle et al., 1984).

Sequencing studies also revealed that the variant-N protein containing viruses reported to be present in *Aedes albopictus* C6/36 cells persistently infected with Bunyamwera virus (Elliott & Wilkie, 1986) were not endogenous to the persistently infected culture, but represented contaminating temperature-sensitive bunyaviruses.

As mentioned above, cloned stocks of Bunyamwera virus (wtL₉BUN virus) readily established persistent infection in *Aedes albopictus* C6/36 cells. The persistently infected cells were indistinguishable from uninfected C6/36 cells in terms of cell morphology and growth rate, and were passaged for over a year without observable cytopathic effect. The newly established persistent infection mimicked that reported previously in C6/36 cells (Elliott & Wilkie, 1986) and also the persistent Bunyamwera virus infection established in an uncloned *Aedes albopictus* cell line by Newton *et al.* (1981).

Levels of virus released from the persistently infected culture fluctuated between 10² and 10⁷ pfu/ml. The persistently infected culture also displayed resistance to superinfection with wtL₉BUN virus, which varied in magnitude at different passage levels. There was a marked change in the plaque-morphology of released virus with increasing passage level: well defined lytic plaques were gradually replaced by small, cloudy plaques.

Each of the above characteristics were also observed by Newton *et al.* (1981) and Elliott & Wilkie (1986), and are generally associated with the presence of defective interfering particles in the culture (Holland, 1990).

The relative levels of each of the viral genomic RNAs, and of positive-sense S RNA, at different passage levels of the C6/36/BUN cell line were determined. The levels of each of the viral RNA species varied between passage levels, and each of the viral RNAs appeared to vary independently. No simple correlation between the fluctuating levels of each of the viral RNAs and the titres of virus released from the C6/36 cells was evident. More detailed examination of the viral RNAs resident in the C6/36/BUN culture, by Northern blotting, revealed the presence of defective viral RNAs.

8.4 Bunyaviridae DI particles

Prior to the study presented in this thesis, there had only been two reports on Bunyaviridae DI particles. Kascsak & Lyons (1978) reported the generation of defective Bunyamwera virus particles upon high multiplicity passage of Bunyamwera virus in BHK cells. These particles contained predominantly S segment-sized RNA and interfered with the multiplication of homologous virus. Elliott & Wilkie (1986) observed the generation of defective Bunyamwera virus particles, which contained only S segment sequences, during persistent Bunyamwera virus infection of Aedes albopictus C6/36 cells. These particles also interfered with the replication of homologous virus. In neither case were novel viral RNA species (DI RNAs or defective RNAs without interfering activity) observed in the defective interfering virus preparations. More recently, subgenomic RNAs derived from the L RNA segment of Germiston bunyavirus and tomato spotted wilt tospovirus (TSWV) have been detected (Cunningham & Szilagyi, 1987; de Oliveira Resende et al., 1991; see Section 2.3.1.1). Interfering activity was inferred, but not proven, in the former case. TSWV defective L RNAs out-compete full-length L RNAs in infected Nicotiana rustica plants and are associated with disease attenuation, suggesting that these DI RNAs possess interfering activity.

A range of size classes of subgenomic RNAs derived from the Bunyamwera virus L RNA segment were observed in the persistently infected C6/36/BUN cell line. Low levels of the defective RNAs were first detected (by Northern blot analysis) at passage level two of the persistently infected culture and they were clearly detectable by passage level 21. By passage level 58 a wider spectrum of defective L RNAs were seen and defective RNA was much more abundant than the full-length L RNA segment. No subgenomic M segment RNA or S segment RNA was observed, but interestingly at passages 19, 21 and 22 more slowly migrating S segment RNA species were detected.

When passage levels 18, 19, 21 and 22 of the C6/36/BUN cell line were probed for encapsidated Bunyamwera virus genomic RNAs, only passages 18 and 21 contained detectable levels of encapsidated viral RNAs. Defective L RNAs were observed in
nucleocapsid structures, at passage 18 only, while no encapsidated viral RNAs were detected at passages 19 and 22. It is interesting to compare the levels of positive-sense S RNA in the persistently infected C6/36/BUN cell culture at these passage levels. At passage 18, relatively high levels of intracellular positive-sense S RNA coincided with significant levels of encapsidated defective L RNAs. At passage 21, the levels of positive-sense S RNA were approximately 40% of the levels observed at passage 18, and only full-length genomic S RNA was observed in viral nucleocapsids, despite an abundance of defective L RNAs being present in these cells. At passages 19 and 22, the levels of positive-sense S RNA were low and no encapsidated viral RNAs were detected in these cells.

High levels of positive-sense S RNA most probably represent high levels of S mRNA and, in the absence of a translational block, relatively high levels of N protein. Encapsidation of the defective L RNAs, generated in the C6/36/BUN cell line, may therefore be dependent upon the levels of N protein in the persistently infected cells at a particular passage level, rather than the absence of encapsidation signals in the subgenomic RNAs. However, the larger S RNA species detected at passages 19, 21 and 22 may lack encapsidation signals, since these RNAs appeared not to be encapsidated. Stalder et al. (1983) reported the generation of SFV DI RNAs, which were inefficiently encapsidated, during persistent infection of C6/36 cells. Sequence comparisons between the defective S RNAs and standard Bunyamwera virus genomic S RNA may identify genomic signals required for efficient encapsidation and packaging of the viral genome. The defective S RNA at passage level 21 was particularly abundant relative to the standard genomic S RNA segment, but only the authentic S RNA was encapsidated (Figure 6.18). There does appear to be a strong selection for encapsidation of full-length S RNA species in the persistently infected cells, and full-length S RNA species predominate over full-length L and M RNA species in intracellular nucleocapsids. This may be due to the greater efficiency with which smaller RNA molecules can be encapsidated and is in line with early reports of Bunyamwera virus DI particles containing only S RNA (Kascsak & Lyons, 1978; Elliott & Wilkie, 1986). Rossier et al. (1988) failed to detect defective viral RNAs in their culture of Aedes albopictus C6/36 cells persistently infected with La Crosse bunyavirus; this may reflect the fact that they examined relatively early passage levels (4, 6 and 8), when the levels of defective viral RNAs are low.

Recently, there have been two reports of the characterisation of Bunyaviridae DI RNAs (Patel & Elliott, 1992; de Oliveira Resende *et al.*, 1992). In both instances the DI RNAs were subgenomic species, derived from the L RNA segment. Patel & Elliott (1992) characterised Bunyamwera virus DI RNAs generated in mouse L cells. These defective L RNAs were efficiently packaged into nucleocapsids. In addition, the DI particle preparations produced novel polypeptides of various sizes in infected BHK and mouse L cells. These novel polypeptides could be precipitated with antisera raised against either the

N or C terminus, or both, of the Bunyamwera virus L protein. Nucleotide sequence analysis of cloned cDNAs to the most abundant DI RNAs revealed that in each case a single internal deletion of the L segment had taken place, removing between 72% and 77% of the L RNA. The 5' and 3' terminal sequences were retained.

De Oliveira Resende et al. (1992) analysed four TSWV DI L RNAs by hybridization studies, PCR cloning and sequence determination. They determined that, like the Bunyamwera virus DI L RNAs characterised by Patel & Elliott (1992), these RNAs belonged to the 5'-3' class of DI RNAs (Lazzarini et al., 1981). The TSWV DI L RNAs retained both genomic termini and displayed a single extensive internal deletion of 60% to 80% of the standard genomic L RNA. All four TSWV DI L RNAs examined contained ORFs. In each case the encoded protein retained the original C-terminal amino acid sequence of the viral polymerase (L). De Oliveira Resende et al. (1992) also determined the junction sites within each DI L RNA. Analysis of the nucleotide sequences surrounding the recombination sites of the TSWV DI L RNAs revealed the involvement of identical sequences (such as UA, UAG or CCACU) at the release and reinitiation sites of individual DI L RNAs. It is thought that the repeat sequences may enable the viral polymerase to jump from one site to another on the L RNA molecule during virus replication. This hypothesis fits with the copy-choice mechanism of DI RNA generation (Lazzarini et al., 1981; Perrault, 1981; Re et al., 1985; Cascone et al., 1990). Meier et al. (1984) proposed that repeat sequences in the vesicular stomatitis virus genome were involved in defining junction sites during the generation of vesicular stomatitis virus DI RNAs. However, in this case, the repeated sequences are not immediately adjacent to the junction site, as in **TSWV DI L RNAs.**

Comparison of the nucleotide sequences immediately preceeding and at the 3' end of the deleted region of the Bunyamwera virus DI L RNAs characterised by Patel and Elliott (1992), revealed the occurence of similar repeat sequences. In two of the DI RNAs (DI-M5 and DI-M7) the sequences AUA and AA, respectively, were found immediately preceeding the polymerase dissociation and reattachment sites. In the third DI RNA (DI-M3) the sequence GGA was repeated prior to the termination and reinitiation sites, however, in this case the sequence was close to, but not immediately adjacent to the recombination site. Engelhorn *et al.* (1993) recently characterised a Sendai virus internal deletion defective RNA (E307). Examination of the sequences preceeding the deletion and recombination sites required to generate this defective RNA, revealed the sequence UAC close to the polymerase dissociation site and immediately preceeding the reattachment site.

These short sequences are likely to occur at other sites in the viral genome. This was found to be the case with the short sequences repeated at the junction sites of TSWV DI L RNAs. It is possible therefore that, in addition to the repeat sequences, secondary structure(s) in the L RNA, or higher order structures in the ribonucleoprotein, cause the viral polymerase to "jump" from one site to the other. Computer models predict hairpin

structures at the release and reinitiation sites on the TSWV genome. The first of the pair of repeat sequences is predicted to occur in the end loop of the first hairpin, while the repeat sequence at the release site is located in unpaired sequences in the stem of another hairpin structure. These conformations, within the higher order structure of the ribonucleoprotein, may expose the repeat sequences at the recombination sites, allowing the polmerase to jump from one to the other. In all TSWV DI L RNAs characterised so far, and in the Bunyamwera virus, VSV and Sendai virus examples cited above, one of the repeat sequences defining the release and reinitiation sites is lost during the formation of the DI RNA. This is consistent with a model in which sequences in the nascent RNA which are complementary to sequences at the release site in the L RNA template, anneal to the repeat sequence at the reinitiation site. Upon reinitiation of RNA synthesis, the second sequence will be lost. The existence of repeat sequences in the viral genome is likely to increase the probability of the viral polymerase jumping between between these sites and deleting the intervening sequences.

Analysis of the defective Bunyamwera virus L RNAs produced in the C6/36/BUN cell line suggested that these defective RNAs were inefficiently encapsidated. However, when BHK cells were infected with virus released from each of the first thirty passages of the C6/36/BUN cell line, novel polypeptides were observed in infections initiated with virus released at passage levels 18, 21, 23, 26 and 29 (Figure 6.8), when host protein synthesis was efficiently inhibited. Immunoprecipitation studies are required to determine if these polypeptides are virus-specified and akin to those observed by Patel & Elliott (1992). If these proteins are virus specific, a small proportion of the defective L RNAs in the C6/36/BUN cell line may be encapsidated and packaged into virions. As discussed earlier, the number of defective virions released from the C6/36/BUN culture might vary according to the level of unassembled N protein in the cells at a particular passage level.

8.5 Cell heterogeneity within persistently infected cell cultures

Cells were cloned from passages 12, 20, 30 and 58 of the persistently infected C6/36/BUN cell line. When the cell clones, which gave rise to stable cell lines, were examined for levels of intracellular viral RNA, levels of released virus, and their ability to resist superinfection with wild-type Bunyamwera virus, it became clear that there was heterogeneity within the persistently infected cell culture. Individual cells harboured different defective L RNAs and responded differently to superinfection with homologous virus. In addition, no Bunyamwera virus RNA was detected in some clones, and a minority of cells cloned from the C6/36/BUN cell line displayed cpe and failed to establish stable cell lines.

Similar observations have been made with other persistently infected cell cultures, for example BHK cells persistently infected with FMDV (de la Torre et al., 1989b). Clonal analysis of this persistently infected cell line revealed that none of the stable cloned cell lines, derived from the original persistently infected culture, released infectious FMDV, or contained levels of viral RNA detectable by dot blot hybridization with cDNA probes. Eight out of a total of 177 single cells isolated from the original persistently infected culture (termed C1-BHK-Rc1), released FMDV, but none of these cells gave rise to stable cell clones. Presumably these cells were lytically infected. On the basis of their studies, de la Torre et al. (1989b) estimated that between 4 to 20% of the cells in the C1-BHK-Rc1 culture were lytically infected. Heterogeneity was evident in the morphology of individual cells within the culture, in the resistance of individual cells to superinfection with homologous virus, and in the growth characteristics of individual cells. Upon serial passage of the persistently infected culture, cells with increased resistance to FMDV were selected. When cloned cells derived from this culture were infected with the original FMDV, the virus yields were 10^3 to 10^5 fold lower than those obtained with BHK-21 cells (de la Torre et al., 1989b). There was no evidence for DI involvement (de la Torre et al., 1988; 1989b). The resistance was specific for FMDV and appeared to be due to an intracellular block which resulted in a 100-fold decrease in the levels of FMDV RNA relative to infected BHK-21 cells. As stated earlier, no viral RNA was detected in the resistant cells, and the resistance appeared to be mediated by trans-acting cellular factors (de la Torre et al., 1989a).

Two, out of a total of 58, single cells isolated from the persistently infected C6/36/BUN cell line, exhibited cpe and failed to establish stable cell clones. Since it was not ascertained if infectious virus was released from these cells, we cannot be certain that cell death was due to virus-induced lysis of the cells. In contrast to the C1-BHK-Rc1 persistently infected culture, most of the single cells isolated from the C6/36/BUN cell line gave rise to stable cell clones which harboured intracellular viral RNA. The majority of the cloned cell lines derived from passages 12 and 20 released detectable levels of infectious virus, while 2 out of 28 cell clones derived from passages 30 and 58 of the C6/36/BUN culture, released low, but detectable, levels of infectious virus. Thus, by passage 12 of the persistently infected C6/36/BUN cell line, the majority of individual cells within the culture appeared to able to withstand the cytopathic effects of Bunyamwera virus infection. The virus released from each of the cloned cell lines, and from later passage levels of the parental C6/36/BUN cell line had a cloudy plaque morphology, suggestive of interfering activity (Welsh & Pfau, 1972).

In contrast to the C1-BHK-Rc1 culture, the C6/36/BUN cell line no longer exhibited resistance to infection with wild-type Bunyamwera virus by passage level 58. Northern blot analysis of total cellular RNA from passage level 58 of the C6/36/BUN cell line revealed mainly defective L RNAs in these cells. In addition, resistance to superinfection appeared to vary at different passage levels of the persistently infected culture, and between individual cloned cell lines derived from a given passage level of the culture. Northern blot analysis of total cellular RNA from a selection of cloned cell lines derived from passage 20 of the C6/36/BUN culture, showed that individual cells harboured different defective L RNAs. All of the clones examined contained a single prominent defective L RNA species. The resistance of the cloned cells to superinfection with Bunyamwera virus appeared to correlate with the amount of full-length L and S RNA present, and not with the amount or occurence of defective L RNA.

8.6 Directions for future work

The mechanisms by which persistent viral infections are established and maintained are still far from clear. With the bunyavirus system a more detailed understanding of the molecular biology of infection, and in particular the functioning of the viral polymerase and the function/role of the viral non-structural proteins must precede advances in our understanding of the establishment of persistent bunyavirus infections.

The role of the host cell in persistent viral infections must be addressed. The expression of the lytic and persistent phenotypes of Bunyamwera virus, upon infection of BHK and C6/36, respectively, may well depend upon host cell factor(s) present in one cell type but not the other. Recently, Levine et al. (1993) demonstrated that Sindbis alphavirus infection of a number of vertebrate cell lines (BHK, mouse neuroblastoma (N18) and rat prostatic adenocarcinoma (AT-3) cells) induces an active cell suicide process (apoptosis). In contrast to the lytic replication cycle observed in these vertebrate cell lines, Sindbis virus establishes a persistent infection in differentiated primary cultured rat neurons (Levine et al., 1991). However, Sindbis virus infection of undifferentiated primary cultures of embryonic rat neurons results in cell death. This differential susceptibility to virus induced cell death was associated with levels of expression of the proto-oncogene bcl-2 which is known to block apoptosis. Levine et al (1993) also generated stable transfectant AT-3 cell clones expressing the human bcl-2 protein (AT3Bcl2). They demonstrated that expression of the bcl-2 protein protected the AT-3Bcl2 cells from Sindbis virus-induced cell death and lead to the establishment of a persistent infection. It would be interesting to investigate whether or not Bunyamwera virus infection kills vertebrate cells via apoptosis. Since Sindbis and Bunyamwera virus readily establish persistent infections in mosquito cell cultures (Newton et al., 1981; Stalder et al., 1983; Elliott & Wilkie, 1986), it would be interesting to determine if mosquito cell cultures express bcl-2, or an equivalent protein which promotes cell survival.

A role for DI particles in maintaining bunyavirus persistent infections seems likely, but we are still some way from understanding how DI particles exert their effects. In the meantime DI genomes generated during persistent infections may serve as tools with which to identify functionally important regions of the virus genome. For instance, characterisation of the larger S RNA species detected in the C6/36/BUN culture, which appeared not to possess the signals required for encapsidation, may help identify RNA encapsidation signals. In addition, sequence analysis of the defective L RNA species generated in the C6/36/BUN cell line may increase our understanding of the mechanisms of DI RNA generation.

An understanding of invertebrate cell tolerance to arbovirus infection may lead to the development of novel antiviral therapies. In particular, the purification and characterisation of the antiviral proteins produced by alphavirus-infected invertebrate cells (see Section 2.3.2) could lead to the development of therapeutic agents for the control of arbovirus-induced disease.

REFERENCES

- Abraham, G. & Pattnaik, A. K. (1983). Early RNA synthesis in Bunyamwera virusinfected cells. *Journal of General Virology* 64, 1277-1290.
- Adams, J. M. & Imagawa, D. T. (1962). Measles antibodies in multiple sclerosis. Proceedings of the Society of Experimental Biology and Medicine 111, 562-566.
- Adams, L. D., Albright, L. M., Ausubel, F. M., Besmond, C., Bloch, K. & et al (1990). Appendices. *Current protocols in molecular biology*. New York, John Wiley and sons. p. A.2.1.
- Agnarsdottir, G. (1977). Subacute sclerosing panencephalitis. *Recent Advances in Clinical Virology*. Edinburgh, Churchill Livingstone. p.21-49.
- Ahmed, R., Canning, W. M., Kauffman, R. S., Sharpe, A. H., Hallum, J. V. & Fields, B. N. (1981). Role of the host cell in persistent viral infection: coevolution of L cells and reovirus during persistent infection. *Cell* 25, 325-332.
- Ahmed, R., Chakraborty, P. R. & Fields, B. N. (1980). Genetic variation during lytic virus infection: High passage stocks of wild-type reovirus contain temperature-sensitive mutants. *Journal of Virology* 34, 285-287.
- Ahmed, R. & Fields, B. N. (1981). Reassortment of genome segments between reovirus defective interfering particles and infectious virus: construction of temperature-sensitive and attenuated viruses by rescue of mutations from DI particles. Virology 111, 351-363.
- Ahmed, R. & Fields, B. N. (1982). Role of the S4 gene in the establishment of persistent reovirus infection in L cells. *Cell* 28, 605-612.
- Ahmed, R. & Graham, A. F. (1977). Persistent infections in L cells with temperaturesensitive mutants of Reovirus. *Journal of Virology* 23, 250-262.
- Akkina, R. K., Chambers, T. M. & Nayak, D. P. (1984). Expression of defectiveinterfering influenza virus-specific transcripts and polypeptides in infected cells. *Journal* of Virology 51, 395-403.
- Akkina, R. K., Chambers, T. M. & Nayak, D. P. (1984). Mechanism of interference by defective-interfering particles of influenza virus : Differential reduction of intracellular synthesis of specific polymerase proteins. *Virus Research* 1, 687-702.
- Akkina, R. K. & Nayak, D. P. (1987). Inteference by defective interfering influenza virus: role of defective viral polypeptides in infected cells. *The Biology of Negative Strand Viruses*. New York, Elsevier Science Publishers. p.183-187.
- Albrecht, P., Tourtellotte, W. W., Hicks, J. T., Sato, H., Boone, E. J. & Potvin, A. R. (1983). Intra-blood-brain barrier measles virus antibody synthesis in multiple sclerosis patients. *Neurology* 33, 45-50.
- Anderson, G. J. & Smith, J. F. (1987). Immunoelectron microscopy of Rift Valley fever viral morphogenesis in primary rat hepatocytes. *Virology* **161**, 91-100.

- Antic, D., Lim, B. U. & Kang, C. Y. (1991). Molecular characterization of the M genomic segment of the Seoul 80-39 virus; nucleotide and amino acid sequence comparisons with other hantaviruses reveal the evolutionary pathway. *Virus Research* 19, 47-58.
- Antic, D., Lim, B. U. & Kang, C. Y. (1991). Nucleotide sequence and coding capacity of the large (L) genomic RNA segment of Seoul 80-39 virus, a member of the hantavirus genus. Virus Research 19, 59-65.
- Archard, L. C., Bowles, N. E., Behan, P. O., Bell, E. J. & Doyle, D. (1988). Postviral fatigue syndrome: persistence of enterovirus RNA in muscle and elevated creatine kinase. *Journal of the Royal Society of Medicine* 81, 326-329.
- Arikawa, J., Lapenotiere, H. F., Iacono, C. L., Wang, M. L. & Schmaljohn, C. S. (1990). Coding properties of the S and the M genome segments of Sapporo rat virus: comparison to other causative agents of hemorrhagic fever with renal syndrome. Virology 176, 114-25.
- Arikawa, J., Schmaljohn, A. L., Dalrymple, J. M. & Schmaljohn, C. S. (1989). Characterization of Hantaan virus envelope glycoprotein antigenic determinants defined by monoclonal antibodies. *Journal of General Virology* 70, 615-624.
- Arikawa, J., Takashima, I. & Hashimoto, N. (1985). Cell fusion by hemorrhagic fever with renal syndrome (HFRS) viruses and its application for titration of virus infectivity and neutralizing antibody. *Archives of Virology* **86**, 303-313.
- Arnheiter, H., Davis, N. L., Wertz, G., Schubert, M. & Lazzarini, R. A. (1985). Role of the nucleocapsid protein in regulating vesicular stomatitis virus RNA synthesis. *Cell* 41, 259-267.
- Atkinson, T., Barrett, A. D., Mackenzie, A. & Dimmock, N. J. (1986). Persistence of virulent Semliki Forest virus in mouse brain following co-inoculation with defective interfering particles. *Journal of General Virology* 67, 1189-1194.
- Augé de Mello, P., Honigman, M. H. & Fernández, M. V. (1970). Further information on the survival of modified foot-and-mouth disease virus in cattle. Bulletin Off. Int. Epizoot. 73, 489-505.
- Baldridge, G. D., Beaty, B. J. & Hewlett, M. J. (1989). Genomic stability of La Crosse virus during vertical and horizontal transmission. Archives of Virology 108, 89-99.
- Barbanti-Brodona, G., Oyanagi, S., Katz, M. & et al (1970). Presence of two different viral agents in brain cells of patients with subacute sclerosing panencephalitis. *Proceedings of the Society of Experimental Biology and Medicine* **134**, 230-236.
- Baric, R. S., Lineberger, D. W. & Johnston, R. E. (1983). Reduced synthesis of Sindbis virus negative-strand RNA in cultures treated with host transcription inhibitors. *Journal* of Virology 47, 46-54.
- Barrett, A. D. & Dimmock, N. J. (1986). Defective interfering viruses and infections of animals. Current Topics in Microbiology and Immunology 128, 55-84.
- Barrett, A. D. T., Crouch, C. F. & Dimmock, N. J. (1984). Defective interfering Semliki Forest virus populations are biologically and physically homogeneous. *Journal of General Virology* 65, 1273-1283.

- Basle, M. F., Fournier, J. G., Rozenblatt, S., Rebel, A. & Bouteille, M. (1986). Measles virus RNA detected in Paget's disease bone tissue by *in situ* hybridization. *Journal of General Virology* 67, 2103-2110.
- Basle, M. F., Russell, W. C., Goswami, K. K., Rebel, A., Giraudon, P., Wild, F. & Filmon, R. (1985). Paramyxovirus antigens in osteoclasts from Paget's bone tissue detected by monoclonal antibodies. *Journal of General Virology* 66, 907-913.
- Beaton, A. R. & Krug, R. M. (1981). Selected host cell capped RNA fragments prime influenza viral RNA transcription *in vivo*. *Nucleic Acids Research* 9, 4423-4436.
- Beaty, B. J. & Bishop, D. H. L. (1988). Bunyavirus-vector interaction. Virus Research 10, 289-302.
- Beaty, B. J. & Calisher, C. H. (1991). Bunyaviridae-Natural History. Current Topics in Microbiology and Immunology 169 Berlin, Springer-Verlag. p.27-78.
- Beaty, B. J., Miller, B. R., Shope, R. E., Rozhon, E. J. & Bishop, D. H. L. (1982). Molecular basis of bunyavirus per os infection of mosquitoes: role of the middle-sized RNA segment. *Proceedings of the National Academy of Sciences*, U.S.A. 79, 1295-1297.
- Beaty, B. J., Rozhon, E. J., Gensemer, P. & Bishop, D. H. (1981). Formation of reassortant bunyaviruses in dually-infected mosquitoes. Virology 111, 662-665.
- Beaty, B. J., Sundin, D. R., Chandler, L. R. & Bishop, D. H. L. (1985). Evolution of bunyaviruses by genome reassortment in dually infected mosquitoes (*Aedes triseriatus*). *Science* 230, 548-550.
- Beaty, B. J. & Thompson, W. H. (1976). Deliniation of La Crosse virus in developmental stages of transovarially infected Aedes triseriatus. American Journal of Tropical Medicine and Hygiene 25, 505-515.
- Beaty, B. J. & Thompson, W. H. (1978). Tropisms of La Crosse virus in Aedes triseriatus (Diptera: Culicidae) following infective blood meals. Journal of Medical Entomology 14, 499.
- Bellocq, C. & Kolakofsky, D. (1987). Translational requirement for La Crosse virus SmRNA synthesis: a possible mechanism. *Journal of Virology* 61, 3960-3967.
- Bellocq, C., Raju, R., Patterson, J. & Kolakofsky, D. (1987). Translational requirement of La Crosse virus S-mRNA synthesis: *in vitro* studies. *Journal of Virology* 61, 87-95.
- Berger, S. L. & Birkenmeir, C. S. (1979). Inhibition of intractable nucleases with ribonucleoside-vanadyl complexes: Isolation of messenger ribonucleic acid from resting lymphocytes. *Biochemistry* 18, 5143-5149.
- Bilsel, P. A., Tesh, R. B. & Nichol, S. T. (1988). RNA genome stability of Toscana virus during serial transovarial transmission in the sandfly *Phlebotomus perniciosus*. Virus Research 11, 87-94.
- Bishop, D. H. L. (1985). Replication of arenaviruses and bunyaviruses. *Fields Virology*. New York, Raven Press. First, ed. p.1083-1119.

- Bishop, D. H. L. (1990). Bunyaviridae and their replication. Part I: Structure of bunyaviridae. *Fields Virology*. New York, Raven Press. Second, ed. p.1155-1173.
- Bishop, D. H. L. & Beaty, B. J. (1986). Interference-immunity of mosquitoes to bunyavirus superinfection. Symposium of the Zoological Society London 56, 95-115.
- Bishop, D. H. L., Clerx, J. P. M., Clerx-van Haaster, C. M., Robeson, G., Rozhon, E. J., Ushijima, H. & Veerisetty, V. (1981). Molecular and genetic properties of the Bunyaviridae. *The replication of negative strand viruses*. North Holland, New York, Amsterdam, Elsevier. p.135-146.
- Bishop, D. H. L., Gay, M. E. & Matsuoko, Y. (1983). Non-viral heterogeneous sequences are present at the 5' ends of one species of snowshoe hare bunyavirus S complementary DNA. *Nucleic Acids Research* 11, 6409-6418.
- Bishop, D. H. L. & Shope, R. E. (1979). Bunyaviridae. Newly characterized vertebrate viruses New York, Plenum Press. p.1-156.
- Bishop, D. H. L., Shope, R. E. & Beaty, B. J. (1987). Molecular mechanisms in Arbovirus disease.40, Cambridge University Press. p.135-166.
- Booth, T. F., Steele, G. M., Marriott, A. C. & Nuttall, P. A. (1991). Dissemination, replication, and trans-stadial persistence of Dugbe virus (Nairovirus, Bunyaviridae) in the tick vector Amblyomma variegatum. American Journal of Tropical Medicine and Hygiene 45, 146-157.
- Borzakian, S., Couderc, T., Barbier, Y., Attal, G., Pelletier, I. & Colbere, G. F. (1992). Persistent poliovirus infection: establishment and maintenance involve distinct mechanisms. *Virology* **186**, 398-408.
- Bouloy, M. (1973/1974). Three segment RNA genome of Lumbo virus (Bunyavirus). Intervirology 2, 173-180.
- Bouloy, M. (1991). Bunyaviridae: Genome organization and replication strategies. Advances in Virus Research 40, London, Academic Press, Inc. p.235-275.
- Bouloy, M. & Hannoun, C. (1976). Studies on Lumbo virus replication I. RNAdependent RNA polymerase associated with virions. *Virology* 69, 258-264.
- Bouloy, M., Pardigon, N., Vialat, P., Gerbaud, S. & Girard, M. (1990). Characterization of the 5' and 3' ends of viral messenger RNAs isolated from BHK-21 cells infected with Germiston virus (Bunyavirus). Virology 175, 50-58.
- Bouloy, M., Vialat, P., Givard, M. & Pardigon, N. (1984). A transcript from the S segment of Germiston bunyavirus is uncapped and codes for the nucleoprotein and a non-structural protein. *Journal of Virology* **49**, 717-723.
- Brinton, M. A., Blank, K. J. & Nathanson, N. (1984). Host genes that influence susceptibility to viral diseases. *Concepts in viral pathogenesis*. New York, Springer-Verlag. p.71-78.
- Broadwell, R. & Oliver, C. (1981). Golgi apparatus, GER L, and secretory granule formation within neurons of the hypothalamo-neuro-hypophysial system of control and hyperosmotically stressed mice. *Journal of Cell Biology* **90**, 474-484.

- Brown, D. T. & Condreay, L. D. (1986). Replication of alphaviruses in mosquito cells. *The Togaviridae and Flaviviridae*. Schlesinger and Schlesinger (eds.) New York and London, Plenum Press. p. 171-207.
- Brown, D. T. & Gliedman, J. B. (1973). Morphological variants of Sindbis virus obtained from infected mosquito tissue culture cells. *Journal of Virology* 12, 1535-1539.
- Buchmeier, M. J., Welsh, R. M., Dutho, F. J. & Oldstone, M. B. A. (1980). The virology and immunology of lymphocytic choriomeningitis virus infection. Advances in Immunology 30, 275-331.
- Burness, A. T. H., Pardoe, I., Faragher, S. G., Vrati, S. & Dalgarno, L. (1988). Genetic stability of Ross River virus during epidemic spread in nonimmune humans. Virology 167, 639-643.
- Burrows, R. (1966). Studies on the carrier state of cattle exposed to foot-and-mouth disease virus. *Journal of Hygiene* 64, 81-90.

Cabradilla, C. D., Holloway, B. P. & Obijeski, J. F. (1983). Molecular cloning and sequencing of the La Crosse virus S RNA. Virology 128, 463-468.

Calisher, C. H. (1991) Bunyaviridae. Archives of Virology Supplementum 2, 273-283

- Calisher, C. H., Monath, T. P., Karabatsos, N. & Trent, D. W. (1981). Arbovirus subtyping: applications to epidemiological studies, availability of reagents, and testing services. *American Journal of Epidemiology* **114**, 619-630.
- Carrillo, E. C., Rojas, E. R., Cavallaro, L., Schiappacassi, M. & Campos, R. (1989). Modification of foot-and-mouth disease virus after serial passages in the presence of antiviral polyclonal sera. Virology 171, 599-601.
- Carvalho, M. G., Frugulhetti, I. C. & Rebello, M. A. (1986). Marituba (Bunyaviridae) virus replication in cultured *Aedes albopictus* cells and in L-A9 cells. *Archives of Virology* **90**, 325-335.
- Cascone, P. J., Carpenter, C. D., Li, X. H. & Simon, A. E. (1990). Recombination between satellite RNAs of turnip crinkle virus. *EMBO Journal* 9, 1709-1715.
- Cash, P. (1985). Polypeptide synthesis of Dugbe virus, a member of the Nairovirus genus of the Bunyaviridae. *Journal of General Virology* 66, 141-148.
- Cash, P., Hendershot, L. & Bishop, D. H. (1980). The effect of glycosylation inhibitors on the maturation and intracellular polypeptide synthesis induced by snowshoe hare virus. Virology 103, 235-240.
- Cash, P., Vezza, A. C., Gentsch, J. R. & Bishop, D. H. L. (1979). Genome complexities of the three mRNA species of snowshoe hare bunyavirus and *in vitro* translation of SmRNA to viral N polypeptide. *Journal of Virology* 31, 685-694.
- Cattaneo, R., Rebmann, G., Baczko, K., ter, Meulen, V. & Billeter, M. A. (1987). Altered ratios of measles virus transcripts in diseased human brains. Virology 160, 523-526.

- Cattaneo, R., Schmid, A., Billiter, M. A., Sheppard, R. D. & Udem, S. A. (1988). Multiple virus mutations rather than host factors cause defective measles virus gene expression in a subacute sclerosing panencephalitis cell line. *Journal of Virology* 62, 1388-1397.
- Cattaneo, R., Schmid, A., Eschle, D., Baczko, K., ter Meulen, V. & Billeter, M. A. (1988). Biased hypermutation and other genetic changes in defective measles viruses in human brain infections. *Cell* 55, 255-265.
- Cattaneo, R., Schmid, A., Rebmann, G., Baczko, K., ter, Meulen, V., Bellini, W. J., Rozenblatt, S. & Billeter, M. A. (1986). Accumulated measles virus mutations in a case of subacute sclerosing panencephalitis: interrupted matrix protein reading frame and transcription alteration. Virology 154, 97-107.
- Cave, D. R., Hendrickson, F. M. & Huang, A. S. (1985). Defective interfering virus particles modulate virulence. *Journal of Virology* 55, 366-373.
- Chambers, T. M., Akkina, R. K. & Nayak, D. P. (1984). In vivo transcription and translation of defective interfering particle-specific RNAs of influenza virus. Segmented Negative Strand viruses. Orlando, Academic. p. 85-91.
- Chambers, T. M. & Webster, R. G. (1987). Defective interfering virus associated with A/Chicken/Pennsylvania/83 influenza virus. Journal of Virology 61, 1517-1523.
- Chambers, T. M. & Webster, R. G. (1991). Protection of chickens from lethal influenza virus infection by influenza A/chicken/Pennsylvania/1/83 virus: characterization of the protective effect. *Virology* 183, 427-432.
- Chandler, L. J., Beaty, B. J., Baldridge, G. D., Bishop, D. H. L. & Hewlett, M. J. (1990). Heterologous reassortment of bunyaviruses in *Aedes triseriatus* mosquitoes and transovarial and oral transmission of newly evolved genotypes. *Journal of General Virology* 71, 1045-1050.
- Chen, S. Y. & Compans, R. W. (1991). Oligomerization, transport, and Golgi retention of Punta Toro virus glycoproteins. *Journal of Virology* 65, 5902-5909.
- Chen, S. Y., Matsuoka, Y. & Compans, R. W. (1991). Golgi complex localization of the Punta Toro virus G2 protein requires its association with the G1 protein. Virology 183, 351-365.
- Chomczynski, P. & Sacchi, N. (1987). Single-step method of RNA isolation by acid guanidinium thiocyanate-phenol-chloroform extraction. *Analytical Biochemistry* 162, 156-159.
- Choppin, P. W. (1969). Expression of DI particles appears to be host controlled. Virology 39, 130-134.
- Chung, S. I., Livingston, C. J., Jones, C. W. & Collisson, E. W. (1991). Cache Valley virus infection in Texas sheep flocks. *Journal of the American Vetinary Medicial Association* 199, 337-340.
- Clerx, J. P. & Bishop, D. H. (1981). Qalyub virus, a member of the newly proposed Nairovirus genus (Bunyaviridae). Virology 108, 361-372.

- Clerx-van Haaster, C. & Bishop, D. H. L. (1980). Analysis of the 3' terminal sequences of snowshoe hare and La Crosse bunyaviruses. Virology 105, 564-574.
- Clerx-van Haaster, C. M., Akashi, H., Auperin, D. D. & Bishop, D. H. L. (1982). Nucleotide sequence analyses and predicted coding of bunyavirus genome RNA species. Journal of Virology 41, 119-128.
- Clerx-van Haaster, C. M., Clerx, J. P. M., Ushijima, H., Akashi, H., Fuller, F. & Bishop, D. H. L. (1982). The 3' terminal sequences of Bunyaviruses and Nairoviruses (Bunyaviridae): evidence of end sequence generic differences within the virus family. *Journal of General Virology* **61**, 289-292.
- Colbere-Garapin, F., Christodoulou, C., Crainic, R. & Pelletier, I. (1989). Persistent poliovirus infection of human neuroblastoma cells. *Proceedings of the National Academy of Sciences*, U.S.A. 86, 7590-7594.
- Collett, M. S. (1986). Messenger RNA of the M segment of Rift Valley fever virus. Virology 151, 151-156.
- Collett, M. S., Purchio, A. F., Keegan, K., Frazier, S., Hays, W., Anderson, D. K., Parker, M. D., Schmaljohn, C., Schmidt, J. & Dalrymple, J. M. (1985). Complete nucleotide sequence of the M RNA segment of Rift Valley fever virus. Virology 144, 228-245.
- Collins, W. E. (1962). Trans-stadial passage of St. Louis encephalitis virus in Aedes aegypti mosquitoes. American Journal of Tropical Medicine and Hygiene 11, 535-538.
- Condreay, L. D., Adams, R. H., Edwards, J. & Brown, D. T. (1988). Effect of actinomycin D and cycloheximide on replication of Sindbis virus in Aedes albopictus (mosquito) cells. Journal of Virology 62, 2629-2635.
- Condreay, L. D. & Brown, D. T. (1986). Exclusion of superinfecting homologous virus by Sindbis virus-infected Aedes albopictus (mosquito) cells. Journal of Virology 58, 81-86.
- Condreay, L. D. & Brown, D. T. (1988). Suppression of RNA synthesis by a specific antiviral activity in Sindbis virus-infected Aedes albopictus cells. Journal of Virology 62, 346-348.
- Condy, J. B., Hedger, R. S., Hamblin, C. & Barnett, I. T. (1985). The duration of the foot-and-mouth disease virus carrier state in African buffalo (i) in the individual animal and (ii) in a free-living herd. *Comparative Immunology, Microbiology and Infectious Diseases* 8, 259-265.
- Connolly, J. H. (1968). Additional data on measles virus antibody and antigen in subacute sclerosing panencephalitis. *Neurology* 18, 87-90.
- Connolly, J. H., Allen, I. V., Hurwitz, L. J. & et al. (1967). Measles virus antibody and antigen in subacute sclerosing panencephalitis. *Lancet* 1, 542-544.
- Cunningham, C. & Szilagyi, J. F. (1987). Viral RNAs synthesized in cells infected with Germiston Bunyavirus. Virology 157, 431-439.

- Cunningham, L., Bowles, N. E., Lane, R. J., Dubowitz, V. & Archard, L. C. (1990). Persistence of enteroviral RNA in chronic fatigue syndrome is associated with the abnormal production of equal amounts of positive and negative strands of enteroviral RNA. Journal of General Virology 71, 1399-1402.
- Curran, J., Boeck, R. & Kolakofsky, D. (1991). The Sendai virus P gene expresses both an essential protein and an inhibitor of RNA synthesis by shuffling modules via mRNA editing. *EMBO Journal* **10**, 3079-3085.
- Dalakas, M. C., Elder, G., Hallett, M., Ravits, J., Baker, M., Papadopoulos, N., Albrecht, P. & Sever, J. (1986). A long-term follow-up study of patients with postpoliomyelitis neuromuscular symptoms. New England Journal of Medicine 314, 959-963.
- Dantas, J. R., Okuno, Y., Asada, H. & et al. (1986). Characterization of glycoproteins of viruses causing hemorrhagic fever with renal syndrome (HFRS) with monoclonal antibodies. *Virology* 151, 379-384.
- Daubney, R. & Hudson, J. R. (1931). Enzootic hepatitis or Rift Valley fever: an undescribed virus disease of sheep, cattle and man, from East Africa. Journal of Pathology and Bacteriology 34, 545-557.
- Davey, M. W. & Dalgarno, L. (1974). Semliki Forest virus replication in cultured Aedes albopictus cells: studies on the establishment of persistence. Journal of General Virology 24, 453-463.
- David-West, T. S. & Porterfield, J. S. (1974). Dugbe virus: a tick-borne arbovirus from Nigeria. Journal of General Virology 23, 297-307.
- Davis, L. E., Bodian, D., Price, D. & et al. (1977). Chronic progressive poliomyelitis secondary to vaccination of an immunodeficient child. New England Journal of Medicine 297, 241-245.
- de Haan, P., Kormelink, R., de Oliveira Resende R., van Poelwijk F., Peters, D. & Goldbach, R. (1991). Tomato spotted wilt virus L RNA encodes a putative RNA polymerase. *Journal of General Virology* **72**, 2207-2216.
- de Haan, P., Wagemakers, L., Goldbach, R. & Peters, D. (1989). Tomato spotted wilt virus, a new member of the Bunyaviridae? *Genetics and Pathogenicity of Negative strand viruses*. Amsterdam, Elsevier Science. p. 287-290.
- de Haan, P., Wagemakers, L., Peters, D. & Goldbach, R. (1989). Molecular cloning and terminal sequence determination of the S and M RNAs of tomato spotted wilt virus. *Journal of General Virology* **70**, 3469-3473.
- de Haan, P., Wagemakers, L., Peters, D. & Goldbach, R. (1990). The S RNA segment of tomato spotted wilt virus has an ambisense character. *Journal of General Virology* 71, 1001-1007.
- de la Torre, J., Davila, M., Sobrino, F., Ortin, J. & Domingo, E. (1985). Establishment of cell lines persistently infected with foot-and-mouth disease virus. Virology 145, 24-35.

- de la Torre, J. C., Alarcon, B., Martinez, S. E., Carrasco, L. & Domingo, E. (1987). Ribavirin cures cells of a persistent infection with foot-and-mouth disease virus in vitro. Journal of Virology 61, 233-235.
- de la Torre, J. C., de la Luna, S., Diez, J. & Domingo, E. (1989). Resistance to foot-andmouth disease virus mediated by trans-acting cellular products. *Journal of Virology* 63, 2385-2387.
- de la Torre, J. C., Martinez, S. E., Diez, J. & Domingo, E. (1989). Extensive cell heterogeneity during persistent infection with foot-and-mouth disease virus. *Journal of Virology* 63, 59-63.
- de la Torre, J. C., Martinez, S. E., Diez, J., Villaverde, A., Gebauer, F., Rocha, E., Davila, M. & Domingo, E. (1988). Coevolution of cells and viruses in a persistent infection of foot-and-mouth disease virus in cell culture. *Journal of Virology* 62, 2050-2058.
- de Oliveira Resende, R., de Haan, P., de Avila, A., Kitajima, E. W., Kormelink, R., Goldbach, R. & Peters, D. (1991). Generation of envelope and defective interfering RNA mutants of tomato spotted wilt virus by mechanical passage. *Journal of General Virology* 72, 2375-2383.
- de Oliveira Resende, R., de Haan, P., van de Vossen, E., de Avila, A. C., Goldbach, R. & Peters, D. (1992). Defective interfering L RNA segments of tomato spotted wilt virus retain both virus genome termini and have extensive internal deletions. *Journal of General Virology* 73, 2509-2516.
- DeFoliart, G. R. (1983). Aedes triseriatus : vector biology in relationship to the persistence of La Crosse virus in endemic foci. California Serogroup Viruses. New York, Alan R. Liss Incorporated. p. 89-105.
- Delord, B., Poveda, J. D., Astier, G. T., Gerbaud, S. & Fleury, H. J. (1989). Detection of the bunyavirus Germiston in Vero and Aedes albopictus C6/36 cells by in situ hybridization using cDNA and asymmetric RNA probes. Journal of Virological Methods 24, 253-264.
- DePolo, N. J., Giachetti, C. & Holland, J. J. (1987). Continuing coevolution of virus and defective interfering particles and of viral genome sequences during undiluted passages: virus mutants exhibiting nearly complete resistance to formerly dominant defective interfering particles. *Journal of Virology* **61**, 454-464.
- DePolo, N. J. & Holland, J. J. (1986). Very rapid generation/amplification of defective interfering particles by vesicular stomatitis virus variants isolated from persistent infection. *Journal of General Virology* 67, 1195-1198.
- Devereux, J., Haeberli, P. & Smithies, O. (1984). A comprehensive set of sequence programs for the VAX. Nucleic Acids Research 11, 387-395.
- Dimmock, N. J., Beck, S. & McLain, L. (1986). Protection of mice from lethal influenza: evidence that defective interfering virus modulates the immune response and not virus multiplication. *Journal of General Virology* 67, 839-850.
- Domingo, E., Davila, M. & Ortin, J. (1980). Nucleotide sequence heterogeneity of the RNA from a natural population of foot-and-mouth disease virus. *Gene* **11**, 333-346.

- Domingo, E. & Holland, J. J. (1988). High error rates, population equilibrium and evolution of RNA replication systems. E. Domingo, J. J. Holland, and P. Ahlquist (Eds.). RNA genetics. Boca Raton, Florida, CRC Press. p. 3-36.
- Domingo, E., Martinez-Salas, E., Sobrino, F., de la Torre, J. C., Portela, A., Ortin, J., Lopez-Galindez, C., Perez-Brena, P., Villanneva, N., Najera, R. & et al. (1985). The quasispecies (extremely heterogeneous) nature of viral RNA genome populations: biological relevance-a review. *Gene* 40, 1-8.
- Eaton, B. T. (1975). Defective interfering particles of Semliki Forest virus generated in BHK cells do not interfere with viral RNA synthesis in *Aedes albopictus* cells. *Virology* 68, 534-538.
- Eaton, B. T. (1981). Viral interference and persistence in Sindbis virus infected Aedes albopictus cells. Canadian Journal of Microbiology 27, 563-567.
- Eddy, G. A. & Peters, C. J. (1980). The extended horizons of Rift Valley fever: current and projected immunogens. *New developments with human and veterinary vaccines*. New York, Liss, Alan R. p. 179-191.
- Edwards, J. F., Livingston, C. W., Chung, S. I. & Collisson, E. C. (1989). Ovine arthrogryposis and central nervous system malformations associated with *in utero* Cache Valley virus infection: spontaneous disease. *Veterinary Pathology* **26**, 33-39.
- El Said, L. H., Vorndam, V., Gentsch, J. R., Clewley, J. P., Calisher, C. H., Klimas, R. A., Thompson, W. H., Grayson, M., Trent, D. W. & Bishop, D. H. L. (1979). A comparison of La Crosse virus isolates obtained from different ecological niches and an analysis of the structural components of California encephalitis serogroup viruses and other bunyaviruses. *American Journal of Tropical Medicine and Hygiene* 28, 364-386.
- Elliot, L. H., Kiley, M. P. & McCormick, J. B. (1984). Hantaan virus: identification of virion proteins. *Journal of General Virology* 65, 1285-1293.
- Elliott, R. M. (1985). Identification of nonstructural proteins encoded by viruses of the Bunyamwera serogroup (family Bunyaviridae). Virology 143, 119-126.
- Elliott, R. M. (1989a). Nucleotide sequence analysis of the large (L) genomic RNA segment of Bunyamwera virus, the prototype of the family Bunyaviridae. Virology 173, 426-436.
- Elliott, R. M. (1989b). Nucleotide sequence analysis of the small (S) RNA segment of Bunyamwera virus, the prototype of the family Bunyaviridae. *Journal of General Virology* **70**, 1291-1285.
- Elliott, R. M. (1990). Molecular biology of the Bunyaviridae. Journal of General Virology 71, 501-522.
- Elliott, R. M., Dunn, E., Simons, J. F. & Pettersson, R. F. (1992). Nucleotide sequence and coding strategy of the Uukuniemi virus L RNA segment. *Journal of General* Virolgy 73, 1745-1752.
- Elliott, R. M., Lees, J. F., Watret, G. E., Clark, W. & Pringle, C. R. (1984). Genetic diversity of bunyaviruses and mechanisms of genetic variation. A. Kohn and P.

Fuchs (Eds.). Mechanisms of Viral Pathogenesis: From Gene to Pathogen. Boston, Martius Nijhoff. 61-76.

- Elliott, R. M. & McGregor, A. (1989). Nucleotide sequence and expression of the small (S) RNA segment of Maguari bunyavirus. Virology 171, 516-524.
- Elliott, R. M., Schmaljohn, C. S. & Collett, M. S. (1991). Bunyaviridae Genome Structure and Gene Expression. *Current Topics in Microbiology and Immunology* 169 Berlin, Springer-Verlag. p. 91-141.
- Elliott, R. M. & Wilkie, M. L. (1986). Persistent infection of Aedes albopictus C6/36 cells by Bunyamwera virus. Virology 150, 21-32.
- Emery, V. C. & Bishop, D. H. L. (1987). Characterization of Punta Toro S mRNA species and identification of an inverted complementary sequence in the intergenic region of Punta Tovo phlebovirus ambisense S RNA that is involved in mRNA transcription termination. Virology 156, 1-11.
- Engelhorn, M., Stricker, R. & Roux, L. (1993). Molecular cloning and characterization of a Sendai virus internal deletion defective RNA. *Journal of General Virology* 74, 137-141.
- Erwin, C. & Brown, D. T. (1983). Requirement of cell nucleus for Sindbis virus replication in cultured *Aedes albopictus* (mosquito) cells. *Journal of Virology* **45**, 792-799.
- Eshita, Y. & Bishop, D. H. L. (1984). The complete sequence of the M RNA of snowshoe hare bunyavirus reveals the presence of internal hydrophobic domains in the viral glycoprotein. *Virology* 137, 227-240.
- Eshita, Y., Ericson, B., Romanowski, V. & Bishop, D. H. L. (1985). Analyses of the mRNA transcription processes of snowshoe hare bunyavirus S and M RNA species. *Journal of Virology* 55, 681-689.
- Fazakerley, J. K., Gonzalez, S. F., Strickler, J., Dietzschold, B., Karush, F. & Nathanson, N. (1988). Organization of the middle RNA segment of snowshoe hare Bunyavirus. Virology 167, 422-432.
- Fields, S. & Winter, G. (1982). Nucleotide sequences of influenza virus segments 1 and 3 reveal mosaic structure of a small viral RNA segment. *Cell* 28, 303-313.
- Foulke, R. S., Rosato, P. R. & French, G. R. (1981). Structural polypeptides of Hazara virus. Journal of General Virology 53, 169-172.
- Freeman, J. M., Magoffin, R. L., Lennette, E. H. et al. (1967). Additional evidence of the relation between subacute inclusion-body encephalitis and measles virus. *Lancet* 2, 129-131.
- Friedman, R. M. & Ramseur, J. M. (1979). Mechanisms of persistent infections by cytopathic viruses in tissue culture. Archives of Virology 60, 83-103.
- Frielle, D. W., Huang, D. D. & Younger, J. S. (1984). Persistent infection with influenza virus: evolution of virus mutants. *Virology* 138, 103-117.

- Fujiami, R. S., Oldstine, M. B. A., Wroblewska, Z., Frankel, M. E. & Koprowski, H. (1983). Molecular mimicry in virus infection: crossreaction of measles virus phosphoprotein of of herpes simplex virus protein with human intermediate filaments. *Proceedings of the National Academy of Sciences, U.S.A.* 80, 2346-2350.
- Gahmberg, N., Kuismanen, E., Keranen, S. & Pettersson, R. F. (1986). Uukuniemi virus glycoproteins accumulate in and cause morphological changes of the Golgi complex in the absence of virus maturation. *Journal of Virology* 57, 899-906.
- Gahmberg, N., Pettersson, R. F. & Kaariainen, L. (1986). Efficient transport of Semliki Forest virus glycoproteins through a Golgi complex morphologically altered by Uukuniemi virus glycoproteins. *EMBO Journal* 5, 3111-3118.
- Gajdusek, D. C., Ed. (1990). Subacute Spongiform Encephalopathies: Transmissible Cerebral Amyloidoses caused by Unconventional Viruses. Fields Virology. New York, Raven Press. p. 2289-2324.
- Gebauer, F., de, I. T. J. C., Gomes, I., Mateu, M. G., Barahona, H., Tiraboschi, B., Bergmann, I., de, M. P. A. & Domingo, E. (1988). Rapid selection of genetic and antigenic variants of foot-and-mouth disease virus during persistence in cattle. *Journal* of Virology 62, 2041-2049.
- Geliebter, J., Zeff, R. A., Melvold, R. W. & Nathenson, S. G. (1986). Mitotic recombination in germ cells generated two major histocompatibility complex mutant genes shown to be identical by RNA sequence analysis: K^{bm9} and K^{bm6}. Proceedings of the National Academy of Sciences, U.S.A. 83, 3371-3375.
- Gentsch, J. R. & Bishop, D. H. L. (1976). Recombination and complementation between temperature-sensitive mutants of the bunyavirus snowshoe hare virus. Journal of Virology 20, 351-354.
- Gentsch, J. R., Bishop, D. H. L. & Obijeski, J. F. (1977). The virus particle nucleic acids and proteins of four bunyaviruses. *Journal of General Virology* 34, 257-268.
- Gentsch, J. R., Robeson, G. & Bishop, D. H. L. (1979). Recombination between snowshoe hare and La Crosse bunyaviruses. *Journal of Virology* 31, 707-717.
- Gentsch, J. R., Wynne, L. R., Clewley, J. P., Shope, R. E. & Bishop, D. H. L. (1977). Formation of recombinants between snowshoe hare and La Crosse bunyaviruses. Journal of Virology 24, 893-902.
- Gerbaud, S., Vialat, P., Pardigon, N., Wychowski, C., Girard, M. & Bouloy, M. (1987). The S segment of the Germiston virus RNA genome can code for three proteins. *Virus Research* 8, 1-13.
- Giebel, L. B., Stohwasser, R., Zoller, L., Bautz, E. K. & Darai, G. (1989). Determination of the coding capacity of the M genome segment of nephropathia epidemica virus strain Hallnas B1 by molecular cloning and nucleotide sequence analysis. *Virology* 172, 498-505.
- Gillies, S. & Stollar, V. (1980). Generation of defective interfering particles of vesicular stomatitis virus in *Aedes albopictus* cells. *Virology* **107**, 497-508.

- Giorgi, C., Accardi, L., Nicoletti, L., Gro, M. C., Takehara, K., Hilditch, C., Morikawa, S. & Bishop, D. H. (1991). Sequences and coding strategies of the S RNAs of Toscana and Rift Valley fever viruses compared to those of Punta Toro, Sicilian Sandfly fever, and Uukuniemi viruses. Virology 180, 738-753.
- Giuffre, R. M., Tovel, D. R., Kay, C. M. & Tyrrell, D. L. J. (1982). Evidence for an interacton between the membrane protein of a paramyxovirus and actin. *Journal of Virology* 42, 963-968.
- Gliedman, J. B., Smith, J. F. & Brown, D. T. (1975). Morphogenesis of Sindbis virus in cultured Aedes albopictus cells. Journal of Virology 16, 913-926.
- Gonzalez-Scarano, F. (1985). La Crosse virus G1 glycoprotein undergoes a conformational change at the pH of fusion. Virology 140, 209-216.
- Gonzalez-Scarano, F., Beaty, B., Sundin, D., Janssen, R., Endres, M. J. & Nathanson, N. (1988). Genetic determinants of the virulence and infectivity of La Crosse virus. *Microbial Pathogenesis* 4, 1-7.
- Gonzalez-Scarano, F., Janssen, R. S., Najjar, J. A., Pobjecky, N., Nathanson, N. & (1985). An avirulent G1 glycoprotein variant of La Crosse bunyavirus with defective fusion function. *Journal of Virology* 54, 757-763.
- Gonzalez-Scarano, F. & Nathanson, N. (1990). Bunyaviruses. Fields Virology. New York, Raven Press. Second, ed. p. 1195-1228.
- Gonzalez-Scarano, F., Pobjecky, N. & Nathanson, N. (1984). La Crosse bunyavirus can mediate pH dependent fusion from without. Virology 132, 222-225.
- Gonzalez-Scarano, F., Pobjecky, N. & Nathanson, N. (1987). The fusion function of La Crosse bunyavirus is associated with the G1 glycoprotein. *The biology of negative strand viruses*. Amsterdam, Elsevier. p. 33-39.
- Gonzalez-Scarano, F., Shope, R. E., Calisher, C. H. & Nathanson, N. (1982). Characterization of monoclonal antibodies against the G1 and N proteins of La Crosse and Tahyna, two California serogroup bunyaviruses. *Virology* **120**, 42-53.
- Gonzalez-Scarano, F., Shope, R. E., Calisher, C. H. & Nathanson, N. (1983). Monoclonal antibodies against the G1 and nucleocapsid proteins of La Crosse and Tahyna viruses. *California serogroup viruses*. New York, Liss, Alan R. p. 145-156.
- Gonzalez-Scarano, F., Endres, M. J. & Nathanson, N. (1991). Pathogenesis. Bunyaviridae Current Topics in Microbiology and Immunology 169 Berlin, New York, Heidelberg, Springer-Verlag. p. 217-249.
- Goswami, K. K., Randall, R. E., Lange, L. S. & Russell, W. C. (1987). Antibodies against the paramyxovirus SV5 in the cerebrospinal fluids of some multiple sclerosis patients. *Nature* 327, 244-247.
- Gould, E. A., Chanas, A. C., Buckley, A. & Varma, M. G. R. (1983). Immunofluorescence studies on the antigenic interrelationships of the Hughes virus serogroup (Genus *Nairovirus*) and identification of a new strain. *Journal of General Virology* 64, 739-742.

- Grady, L. J., Sanders, M. L. & Campbell, W. P. (1987). The sequence of the M RNA of an isolate of La Crosse virus. *Journal of General Virology* 68, 3057-3071.
- Grady, L. J., Srihongse, S., Grayson, M. A. & Deibel, R. (1983). Monoclonal antibodies against La Crosse virus. *Journal of General Virology* 64, 1699-1704.
- Grimstad, P. R., Ross, Q. E. & Craig, G. B. J. (1980). Aedes triseriatus (Diptera: Culicidae) and La Crosse virus. II. Modification of mosquito feeding behaviour by virus infection. Journal of Medical Entomology 17, 1-7.
- Guild, G. M. & Stollar, V. (1977). Defective-interfering particles of Sindbis virus. V. Sequence relationships between SV_{STD} 42S RNA and intracellular defective viral RNA. Virology 77, 175-188.
- Hacker, D., Raju, R. & Kolakofsky, D. (1989). La Crosse virus nucleocapsid protein controls its own synthesis in mosquito cells by encapsidating its mRNA. *Journal of Virology* 63, 5166-5174.
- Hacker, D., Rochat, S. & Kolakofsky, D. (1990). Anti-mRNAs in La Crosse bunyavirusinfected cells. *Journal of Virology* 64, 5051-5057.
- Hardy, J. L. (1988). Susceptibility and resistance of vector mosquitoes. *The arboviruses:* epidemiology and ecology. Boca Raton, Florida, CRC Press. p. 87-126.
- Hardy, J. L., Houk, E. J., Kramer, L. D. & Reeves, W. C. (1983). Intrinsic factors affecting vector competence of mosquitoes for arboviruses. Annual Review of Entomology 28, 229-262.
- Harvey, L., Gray, T., Beneton, M. N., Douglas, D. C., Kanis, J. A. & Russell, R. G. G. (1982). Ultrastructural features of the osteoclasts from Paget's disease of bone in relation to a viral aetiology. *Journal of Clinical Pathology* 35, 771-779.
- Hewlett, M. J. & Chiu, W. (1991). Virion structure. Bunyaviridae. Current Topics in Microbiology and Immunology 169 London, Springer-Verlag. 79-90.
- Hewlett, M. J., Clerx, J. P., Clerx-van Haaster, C., Chandler, L. J., McLean, D. M. & Beaty, B. J. (1992). Genomic and biologic analyses of snowshoe hare virus field and laboratory strains. American Journal of Tropical Medicine and Hygiene 46, 524-532.
- Hewlett, M. J., Pettersson, R. F. & Baltimore, D. (1977). Circular forms of Uukuniemi virion RNA: an electron microscopic study. *Journal of Virology* 21, 1085-1093.
- Holland, J. J. (1986). Generation and replication of defective viral genomes. Virology. New York, Raven Press. p. 77-99.
- Holland, J. J. (1987). Defective interfering rhabdoviruses. *The Rhabdoviruses*. New York, Plenum press. p. 297-360.
- Holland, J. J. (1990). *Defective viral genomes*. Fields Virology. New York, Raven Press. p. 151-166.
- Holland, J. J., Kennedy, S. I. T., Semler, B. L., Jones, C. L., Roux, L. & Grabau, E. A. (1980). Defective-interfering RNA viruses and the host-cell response. Virus-host

Huang, A. S. (1977). Bacteriol. Rev. 41, 811-821.

•

interactions: Viral invasion, persistence, and disease. H. Fraenkel-Conrat and R. R. Wagner (Eds.). New York, Plenum Press. p. 137-192.

- Holtzer, H., Biehl, J., Yeoh, G., Meganathan, R. & Kajl, A. (1975). Effect of oncogenic viruses on muscle differentiation. *Proceedings of the National Academy of Sciences*, U.S.A. 72, 4051-4055.
- Horodyski, F. M. & Holland, J. J. (1980). Viruses isolated from cells persistently infected with vesicular stomatitis virus show altered interaction with defective-interfering particles. *Journal of Virology* **36**, 627-631.
- Horta-Barbosa, L., Fuccillo, D. A., Sever, J. L. & Zeman, W. (1969). Subacute sclerosing panencephalitis: isolation of measles virus from a brain biopsy. *Nature* 221, 974.
- Howard, M., Davis, N., Patton, J. & Wertz, G. (1987). Roles of vesicular stomatitis virus (VSV) N and NS proteins in viral RNA replication. *The biology of negative strand viruses*. Amsterdam, Elsevier. p. 134-149.
- Howard, M. & Wertz, G. (1989). Vesicular stomatitis virus RNA replication: a role for the NS protein. *Journal of General Virology* **70**, 2683-2694.
- Hsu, M., Scheid, A. & Choppin, P. W. (1981). Activation of the Sendai virus fusion protein (f) involves a conformational change with exposure of a new hydrophobic region. *Journal of Biological Chemistry* 256, 3557-3563.
- Hsu, M. T., Parvin, J. D., Gupta, S., Krystal, M. & Palese, P. (1987). Genomic RNAs of influenza viruses are held in a circular conformation in virions and in infected cells by a terminal panhandle. *Proceedings of the National Academy of Sciences*, U.S.A. 84, 8140-8144.
- Huang, A. S. (1988). Modulation of viral disease processes by defective interfering particles. RNA genetics. Boca Raton, Florida, CRC press. p. 195-208.
- Huang, A. S. & Baltimore, D. (1970). Defective viral particles and viral disease processes. *Nature* 226, 325-327.
- Huang, A. S. & Baltimore, D. (1977). Defective interfering animal viruses. Comprehensive Virology 10 New York, Plenum Press. p. 73-116.
- Hung, T., Xia, S. M., Zhao, T. X., Zhou, J. Y., Song, G., Liao, G. X., Ye, W. W., Chu, Y. L. & Hang, C. S. (1983). Morphological evidence for identifying the viruses of hemorrhagic fever with renal syndrome as candidate members of the Bunyaviridae family. Archives of Virology 78, 137-144.
- Igarashi, A. (1978). Isolation of a Singh's Aedes albopictus cell clone sensitive to Dengue and Chikungunya viruses. Journal of General Virology 40, 531-544.
- Igarashi, A., Koo, R. & Stollar, V. (1977). Evolution and properties of Aedes albopictus cell cultures persistently infected with Sindbis virus. Virology 82, 69-83.
- Igarashi, A. & Stollar, V. (1976). Failure of defective interfering particles of Sindbis virus produced in BHK or chicken cells to affect viral replication in *Aedes albopictus* cells. *Journal of Virology* **19**, 398-408.

- Ihara, T., Akashi, H. & Bishop, D. H. L. (1984). Novel coding strategy (ambisense genomic RNA) revealed by sequence analyses of Punta Toro phlebovirus S RNA. Virology 136, 293-306.
- Ihara, T., Matsuura, Y. & Bishop, D. H. (1985a). Analyses of the mRNA transcription processes of Punta Toro phlebovirus (Bunyaviridae). Virology 147, 317-325.
- Ihara, T., Smith, J., Dalrymple, J. M. & Bishop, D. H. (1985b). Complete sequences of the glycoproteins and M RNA of Punta Toro phlebovirus compared to those of Rift Valley fever virus. Virology 144, 246-259.
- Inglot, A. D., Albin, M. & Chudzio, T. (1973). Persistent infection of mouse cells with Sindbis virus: role of virulence of strains, autointerfering particles and interferon. Journal of General Virology 20, 105-110.
- Iroegbu, C. U. & Pringle, C. R. (1981). Genetic interactions among viruses of the Bunyamwera complex. Journal of Virology 37, 383-394.
- Ishihawa, A. & Nagata, K. (1988). Viral RNA polymerases. CRC Critical Reviews in Biochemistry 23, 27-76.
- Jacobson, S. & Pfau, C. J. (1980). Viral pathogenesis and resistance to defective interfering particles. *Nature* 283, 311-313.
- James, W. S. & Millican, D. (1986). Host-adaptive antigenic variation in Bunyaviruses. Journal of General Virology 67, 2803-2806.
- Janda, J. M., Davis, A. R., Nayak, D. P. & De, B. K. (1979). Diversity and generation of defective interfering influenza virus particles. *Virology* **95**, 48-58.
- Janssen, R. S., Nathanson, N., Endres, M. J. & Gonzalez-Scarano, F. (1986). Virulence of La Crosse virus is under polygenic control. *Journal of Virology* 59, 1-7.
- Jennings, P. A., Finch, J. T., Winter, G. & Robertson, J. S. (1983). Does the higher order structure of the Influenza virus ribonucleoprotein guide sequence rearrangements in influenza viral RNA? Cell 34, 619-627.
- Jin, H. & Elliott, R. M. (1991). Expression of functional Bunyamwera virus L protein by recombinant vaccinia viruses. *Journal of Virology* **65**, 4182-4189.
- Jin, H. & Elliott, R. M. (1992). Mutagenesis of the L protein encoded by Bunyamwera virus and production of monoclonal antibodies. *Journal of General Virology* 73, 2235-2244.
- Johnson, M. D. (1981). The characteristics required for a Sendai virus preparation to induce high levels of interferon in human lymphoblastoid cells. *Journal of General* Virology 56, 175-184.
- Johnston, R. E., Tovell, D. R., Brown, D. T. & Faulkner, P. (1975). Interfering passages of Sindbis virus: Concomitant appearance of interference, morphological variants and truncated viral RNA. *Journal of Virology* 16, 951-958.
- Kakach, L. T., Suzich, J. A. & Collett, M. S. (1989). Rift Valley fever virus M segment: phlebovirus expression strategy and protein glycosylation. *Virology* **170**, 505-510.

- Kang, C. Y., Weide, L. G. & Tischfield, J. A. (1981). Suppression of vesicular stomatitis virus defective interfering particle generation by a function(s) associated with human chromosome 16. *Journal of Virology* **40**, 946-952.
- Kaplan, G., Levy, A. & Racaniello, V. R. (1989). Isolation and characterization of HeLa cell lines blocked at different steps in the poliovirus life cycle. *Journal of Virology* 63, 43-51.
- Kappus, K. D., Monath, T. P., Kaminski, R. M. & et al. (1983). Reported encephalitis associated with California serogroup virus inections in the United States, 1963-1981. *California serogroup viruses*. New York, Liss, Alan R. p. 31-41.
- Karabatsos, N. (1985). International catalogue of arboviruses including certain other viruses of vertebrates. San Antonio, American Society of Tropical Medicine and Hygiene.
- Kascsak, R. J. & Lyons, M. J. (1978). Bunyamwera virus. II. The generation and nature of defective interfering particles. *Virology* 89, 539-546.
- Kauffman, R. S., Ahmed, R. & Fields, B. N. (1983). Selection of a mutant S1 gene during reovirus persistent infection of L cells: role in maintenance of the persistent state. Virology 131, 79-87.
- Kawai, A. & Matsumoto, S. (1977). Interfering and noninterfering defective particles generated by a rabies small plaque variant virus. *Virology* **76**, 60-71.
- King, C.-C., King, M. W., Garry, R. F., Wan, K. M.-M., Ulug, E. T. & Waite, M. R. F. (1979). Effect of incubation time on the generation of defective interfering particles during undiluted serial passage of Sindbis virus in *Aedes albopictus* and chick cells. *Virology* 96, 229-238.
- Kingsford, L. & Hill, D. W. (1981). The effects of proteolytic enzymes on structure and function of La Crosse G1 and G2 glycoproteins. *The replication of negative strand viruses*. New York, Elsevier. p. 111-116.
- Kingsford, L., Ishizawa, L. D. & Hill, D. W. (1983). Biological activities of monoclonal antibodies reactive with antigenic sites mapped on the G1 glycoprotein of La Crosse virus. *Virology* **129**, 443-455.
- Kirkegaard, K. & Baltimore, D. (1986). The mechanism of RNA recombination in poliovirus. Cell 47, 433-443.
- Klimas, R. A., Thompson, W. A., Calisher, C. H., Clark, G. G., Grimstad, P. R. & Bishop, D. H. L. (1981). Genotypic variaties of La Crosse virus isolated from different geographic regions of the continental United States and evidence for a naturally occurring intertypic recombinant La Crosse virus. American Journal of Epidemiology 114, 112-131.
- Kolakofsky, D., Bellocq, C. & Raju, R. (1987). The translational requirement for La Crosse virus S-mRNA synthesis. Cold Spring Harbor Symposium of Quantitative Biology 52, 373-379.
- Kolakofsky, D. & Hacker, D. (1991). Bunyavirus RNA synthesis: Genome Transcription and Replication. D.Kolakofsky. (Ed.). Berlin, Springer-Verlag. p. 143-159.

- Kormelink, R., Kitajima, E. W., De, H. P., Zuidema, D., Peters, D. & Goldbach, R. (1991). The nonstructural protein (NSs) encoded by the ambisense S RNA segment of tomato spotted wilt virus is associated with fibrous structures in infected plant cells. *Virology* 181, 459-468.
- Kozak, M. (1986). Bifunctional messenger RNAs in eukaryotes. Cell 47, 481-483.
- Kozak, M. (1986). Point mutations define a sequence flanking the AUG initiator codon that modulates translation by eukaryotic ribosomes. *Cell* 44, 283-292.
- Kramer, L. D., Hardy, J. L., Presser, S. B. & Houk, E. J. (1981). Dissemination barriers for western equine encephalomyelitis virus in *Culex tarsalis* infected after ingestion of low viral doses. *American Journal of Tropical Medicine and Hygiene* 30, 190-197.
- Krug, R. M., St, A. C., Broni, B. & Shapiro, G. (1987). Transcription and replication of influenza virion RNA in the nucleus of infected cells. *Cold Spring Harbor Symposium* of Quantitative Biology 52, 353-358.
- Kuismanen, E., Bang, B., Hurme, M. & Pettersen, R. F. (1984). Uukuniemi virus maturation: immunofluoresence microscopy with monoclonal glycoprotein-specific antibodies. *Journal of Virology* 51, 137-146.
- Kuismanen, E., Hedman, K., Saraste, J. & Pettersson, R. F. (1982). Uukuniemi virus maturation, accumulation of virus particles and viral antigens in the Golgi complex. *Molecular and Cellular Biology* 2, 1444-1458.
- Lai, M. M. C., (1990). Coronavirus: Organization, Replication and Expression of Genome. Annual Review of Microbiology 44, 303-333.
- Laughlin, L. W., Meegan, J. M., Strausbaugh, L. H., Morens, D. M. & Watten, R. H. (1979). Epidemic Rift Valley fever in Egypt: observations of the spectrum of human illness. *Transcripts of the Royal Society of Tropical Medicine and Hygiene*. 73, 630-633.
- Law, M. D. & Moyer, J. W. (1989). Physicochemical analysis of a serologically distinct tomato spotted wilt virus strain (Abstract). *Phytopathology* 79, 1157.
- Law, M. D. & Moyer, J. W. (1990). A tomato spotted wilt-like virus with a serologically distinct N protein. *Journal of General Virology* 71, 933-938.
- Law, M. D., Speck, J. & Moyer, J. W. (1991). Nucleotide sequence of the 3' non-coding region and N gene of the S RNA of a serologically distinct tospovirus. *Journal of General Virology* 72, 2597-2601.
- Law, M. D., Speck, J. & Moyer, J. W. (1992). The M RNA of Impatiens necrotic spot tospovirus (Bunyaviridae) has an ambisense genome organization. *Virology* 188, 732-741.
- Lazzarini, R. A., Keene, J. D. & Schubert, M. (1981). The origins of defective interfering particles of the negative strand RNA viruses. *Cell* 26, 145-154.
- Le Duc, J. W. (1979). The ecology of California group viruses. Journal of Medical Entomology 16, 1.

- Le Duc, J. W., Suyemoto, W., Eldrige, B. F., Russell, P. K. & Barr, A. R. (1975). Ecology of California encephalitis viruses on the Del Mar Va Peninsula. II. Demonstration of transovarial transmission. *American Journal of Tropical Medicine and Hygiene* 24, 124-126.
- Leake, C. J. (1984). Transovarial transmission of arboviruses by mosquitoes. Vectors in Virus Biology. London, Academic Press. p. 63-91.
- Lee, C.-H. & Schloemer, R. H. (1981). Identification of the antiviral factor in culture medium of mosquito cells persistently infected with Banzi virus. Virology 110, 445-454.
- Lee, C.-H. & Schloemer, R. H. (1981). Mosquito cells infected with Banzi virus secrete an antiviral activity which is of viral origin. *Virology* **110**, 402-410.
- Lee, H. W., French, G. R., Lee, P. W., Baek, L. J., Tsuchiya, K. & Foulke, R. S. (1981). Observation on natural and laboratory infection of rodents with the etiologic agent of Korean hemorrhagic fever. American Journal of Tropical Medicine and Hygiene 30, 477-482.
- Lee, H. W., Lee, P. W., Baek, L. J., Song, C. K. & Seong, I. W. (198b). Intraspecific transmission of Hantaan virus, etiologic agent of Korean hemorrhagic fever, in the rodent Apodemus agrarius. American Journal of Tropical Medicine and Hygiene. 30, 1106-1112.
- Lees, J. F., Pringle, C. R. & Elliott, R. M. (1986). Nucleotide sequence of the Bunyamwera virus M RNA segment: conservation of structural features in the Bunyavirus glycoprotein gene product. Virology 148, 1-14.
- Lehmann-Grube, F., Peralta, L. M., Bruns, M. & Lohler, J. (1983). Persistent infection of mice with lymphocytic choriomeningitis virus. *Comprehensive Virology* 18 New York, Plenum Press. p. 43-103.
- Leibovitz, A. (1963). The growth and maintenance of tissue cell cultures in free gas exchange with the atmosphere. *American Journal of Hygiene* 78, 173-180.
- Leppert, M., Kort, L. & Kolakofsky, D. (1977). Further characterization of Sendai virus DI RNAs: a model for their generation. *Cell* 12, 539-552.
- Leppert, M., Rittenhouse, L., Perrault, J., Summers, D. F. & Kolakofsky, D. (1979). Plus and minus strand leader RNAs in negative strand virus-infected cells. *Cell* 18, 735-747.
- Levine, B., Hardwick, J. M., Trapp, B. D., Crawford, T. O., Bollinger, R. C. & Griffin, D. E. (1991). Antibody-mediated clearance of alphavirus infection from neurons. *Science* 254, 856-860.
- Levine, B., Huang, Q., Isaacs, J. T., Reed, J. C., Griffin, D. E. & Hardwick, J. M. (1993). Conversion of lytic to persistent alphavirus infection by the *bcl-2* cellular oncogene. *Nature* 361, 739-742.
- Logan, K. B. (1979). Generation of defective interfering particles of Semliki Forest virus in a clone of *Aedes albopictus* (mosquito) cells. *Journal of Virology* **30**, 38-44.

- Ludwig, G. V., Christensen, B. M., Yuill, T. M. & Schultz, K. T. (1989). Enzyme processing of La Crosse virus glycoprotein G1: A bunyavirus-vector infection model. *Virology* 171, 108-113.
- Ludwig, G. V., Israel, B. A., Christensen, B. M., Yuill, T. M. & Schultz, K. T. (1991). Monoclonal antibodies directed against the envelope glycoproteins of La Crosse virus. *Microbial Pathogenesis* 11, 411-421.
- Ludwig, G. V., Israel, B. A., Christensen, B. M., Yuill, T. M. & Schultz, K. T. (1991). Role of La Crosse virus glycoproteins in attachment of virus to host cells. *Virology* 181, 564-571.
- Lyons, M. J. & Heyduk, J. (1973). Aspects of the developmental morphology of California encephalitis virus in cultured vertebrate and arthropod cells and in mouse brain. *Virology* 54, 37-52.
- MacPherson, I. & Stoker, M. (1962). Polyoma transformation of hamster cell clones-an investigation of genetic factors affecting cell competence. *Virology* **16**, 147-151.
- Madoff, D. H. & Lenard, J. (1982). A membrane glycoprotein that accumulates intracellularly: cellular processing of the large glycoprotein of La Crosse virus. Cell 28, 821-829.
- Mahy, B. W. (1985). Strategies of virus persistence. British Medical Bulletin 41, 50-55.
- Mandahar, C. L. (1981). Virus transmission through seed and pollen. *Plant diseases and vectors, ecology and epidemiology*. New York, Academic Press. 241-292.
- Marion, P. L. & Robinson, W. S. (1983). Hepadna viruses: Hepatitis B and related viruses. Current Topics in Microbiology and Immunology 105, 99-121.
- Marriott, A. C., Ward, V. K. & Nuttall, P. A. (1989). The S RNA segment of Sandfly Fever Sicilian virus: evidence for an ambisense genome. *Virology* 169, 341-345.
- Martin, M. L., Lindsey-Regnery, H., Sasso, D. R., McCormick, J. B. & Palmer, E. (1985). Distinction between Bunyaviridae genera by surface structure and composition with Hantaan virus using negative stain electron microscopy. Archives of Virology 86, 17-28.
- Matsuoka, Y., Ihara, T., Bishop, D. H. & Compans, R. W. (1988). Intracellular accumulation of Punta Toro virus glycoproteins expressed from cloned cDNA. Virology 167, 251-260.
- McLintock, J. (1978). Mosquito-virus relationships of American encephalitides. Annual Review of Entomology. 23, 17-37.
- Meegan, J. M. & Bailey, C. L. (1988). Rift Valley fever. The arboviruses: epidemiology and ecology. Boca Raton, Florida, CRC Press. p. 51-76.
- Meegan, J. M. & Shope, R. E. (1981). Emerging concepts on Rift Valley fever. Perspective Virology 11, 267-287.

- Meier, E., Harmison, G. G., Keene, J. D. & Schubert, M. (1984). Sites of copy-choice replication involved in generation of vesicular stomatitis virus defective-interfering particle RNAs. *Journal of Virology* 51, 515-521.
- Meinkoth, J. & Kennedy, S. I. T. (1980). Semliki Forest virus persistence in mouse L929 cells. Virology 100, 141-155.
- Messing, J. & Vieira, J. (1982). A new pair of M13 vectors for selecting either DNA strand of double-digest restriction fragments. *Gene* 19, 269-276.
- Messing, J. (1979). A multi-purpose cloning system based on the single-stranded DNA bacteriophage M13. Recominant DNA Technical Bulletin, N. I. H. Publications 79-99, 43-48.
- Miller, B. R. (1983). A variant of La Crosse virus attenuated for Aedes triseriatus mosquitoes. American Journal of Tropical Medicine and Hygiene. 32, 1422-1428.
- Miller, M. L. & Brown, D. T. (1992). Morphogenesis of Sindbis Virus in three subclones of Aedes albopictus (mosquito) cells. Journal of Virology 66, 4180-4190.
- Miller, R. B., DeFoliart, G. R. & Yuill, T. M. (1977). Vertical transmission of La Crosse virus (California encephalitis group): Transovarial and Filial infection rates in *Aedes triseriatus* (Diptera: Culicidae). *Journal of Medical Entomology*. 14, 437-440.
- Mills, B. G. & Singer, F. R. (1976). Nuclear inclusions in Paget's disease of bone. Science 194, 201-202.
- Mills, B. G., Singer, F. R., Weiner, L. P. & Holst, P. A. (1981). Immunohistological demonstration of respiratory syncytial virus antigens in Paget's disease of bone. *Proceedings of the National Academy of Sciences, U.S.A.* 78, 1209-1213.
- Mills, B. G., Singer, F. R., Weiner, L. P., Suffen, S. C., Stabile, E. & Holst, P. A. (1984). Evidence for both respiratory syncytial virus and measles virus antigens in the osteoclasts of patients with Padget's disease of bone. *Clinical Orthopaedics and Related Research* 183, 303-311.
- Milne, R. G. & Francki, R. I. B. (1984). Should tomato spotted wilt virus be considered as a possible member of the family Bunyaviridae? *Intervirology* 22, 72-76.
- Mitchell, C. J. (1983). Mosquito vector competence and arboviruses. Current topics in vector research. New York, Praeger. p. 63-92.
- Mizzen, L., Cheley, S., Rao, M., Wolf, R. & Anderson, R. (1983). Fusion resistance and decreased infectability of coronavirus persistence. *Virology* **128**, 407-417.
- Monroe, S. S. & Schlesinger, S. (1983). RNAs from two independently isolated defectiveinterfering particles of Sindbis virus contain a cellular tRNA sequence at their 5' ends. *Proceedings of the National Academy of Science, U.S.A.* 80, 3279-3383.

Moscona, A. (1991). Defective interfering particles of human parainfluenza virus type 3 are associated with persistent infection in cell culture. *Virology* **183**, 821-824.

- Murphy, F. A., Harrison, A. K. & Whitfield, S. G. (1973). Bunyaviridae: morphologic and morphogenetic similarities of Bunyamwera serologic supergroup viruses and several other arthropod-borne viruses. *Intervirology* 1, 297-316.
- Murphy, F. A., Whitfield, S. G., Sudia, W. D. & Chamberlain, R. W. (1975). Interactions of vector with vertebrate pathogenic viruses. *Invertebrate Immunity: Mechanisms of Invertebrate Vector-Parasite Relations*. Orlando, Academic Press. p. 25-37.
- Nayak, D. P., Chambers, T. M. & Akkina, R. H. (1989). Structure of defective-interfering RNAs of influenza viruses and their role in interference. *The Influenza viruses*. New York, Plenum press. p. 269-317.
- Nayak, D. P., Sivasubramanin, N., Davis, A. R., Cortini, R. & Sung, J. (1982). Complete sequence analyses show that two defective interfering influenza viral RNAs contain a single internal deletion of a polymerase gene. *Proceedings of the National Academy of Sciences*, U.S.A. **79**, 2216-2220.
- Newton, S. E., Short, N. J. & Dalgarno, L. (1981). Bunyamwera virus replication in cultured *Aedes albopictus* (mosquito) cells: establishment of a persistent infection. *Journal of Virology* 38, 1015-1024.
- Newton, S. E. & Dalgarno, L. (1983). Antiviral activity released from Aedes albopictus cells persistently infected with Semliki Forest virus. Journal of Virology 47, 652-655.
- Nicoletti, L. & Verani, P. (1985). Growth of the Phlebovirus Toscana in a mosquito (Aedes pseudoscutellaris) cell line (AP-61): establishment of a persistent infection. Archives of Virology 85, 35-45.
- Nishiyama, Y., Ito, Y. & Shimokata, K. (1978). Properties of viruses selected during persistent infection of L cells with VSV. *Journal of General Virology* **40**, 481-484.
- Nuesch, J., Krech, S. & Siegl, G. (1988). Detection and characterization of subgenomic RNAs in hepatitis A virus particles. *Virology* 165, 419-427.
- Nuesch, J. P., de Chastonay, J. & Siegl, G. (1989). Detection of defective genomes in hepatitis A virus particles present in clinical specimens. *Journal of General Virology* 70, 3475-3480.
- O'Brien, R. L., Brinster, R. L. & Storb, U. (1987). Somatic hypermutation of an immunoglobulin transgene in kappa transgenic mice. *Nature* 326, 405-409.
- Obijeski, J. F., Bishop, D. H. L., Murphy, F. A. & Palmer, E. L. (1976). Structural proteins of La Crosse virus. Journal of Virology 19, 985-997.
- Obijeski, J. F., Bishop, D. H. L., Palmer, E. L. & Murphy, F. A. (1976). Segmented genome and nucleocapsid of La Crosse virus. *Journal of Virology* 20, 664-675.
- Obijeski, J. F., McCauley, J. & Skehel, J. J. (1980). Nucleotide sequences at the termini of La Crosse virus RNAs. *Nucleic Acids Research* 8, 2431-2438.
- Obijeski, J. G. & Murphy, F. A. (1977). Bunyaviridae: recent biochemical developments. Journal of General Virology 37, 1-14.

- Odagiri, T. & Tobita, K. (1990). Mutation in NS2, a nonstructural protein of influenza A virus, extragenically causes aberrant replication and expression of the PA gene and leads to generation of defective interfering particles. *Proceedings of the National Academy of Sciences*, U.S.A. 87, 5988-5892.
- Odagiri, T. & Tobita, K. (1990). Nucleotide sequence of the PA gene of influenza A/WSN/33(H1N1). Nucleic Acids Research 18, 654.

Oldstone, M. B. (1989). Viral alteration of cell function. Scientific American 261, 42-48.

Oldstone, M. B. (1989). Viral persistence. Cell 56, 517-520.

- Oldstone, M. B. (1991). Molecular anatomy of viral persistence. Journal of Virology 65, 6381-6386.
- Oldstone, M. B. A. (1984). Virus can alter cell function without causing cell pathology: disordered function leads to imbalance of homeostasis and disease. *Concepts in Viral Pathogenesis*. New York, Springer-Verlag. p. 202-209.
- Ozden, S. & Hannoun, C. (1978). Isolation and preliminary characterization of temperature-sensitive mutants of Lumbo virus. *Virology* 84, 210-212.
- Ozden, S. & Hannoun, C. (1980). Biochemical and genetical characterization of Germiston virus. Virology 103, 232-234.
- Pardigon, N., Vialat, P., Gerbaud, S., Girard, M. & Bouloy, M. (1988). Nucleotide sequence of the M segment of Germiston virus: comparison of the M gene product of several bunyaviruses. Virus Research 11, 73-85.
- Pardigon, N., Vialet, P., Girard, M. & Bouloy, M. (1982). Panhandles and hairpin structures at the termini of Germiston virus RNAs (Bunyavirus). Virology 122, 191-197.
- Parker, M. D., Smith, J. F. & Dalrymple, J. M. (1984). Rift Valley fever virus intracellular RNA: a functional analysis. Segmented negative strand viruses. Orlando, Academic press. p. 21-28.
- Parrington, M. A. & Kang, C. Y. (1990). Nucleotide sequence analysis of the S genomic segment of Prospect Hill virus: comparison with the prototype Hantavirus. Virology 175, 167-175.
- Patel, A. & Elliott, R. M. (1992). Characterization of Bunyamwera virus defective interfering particles. *Journal of General Virology* 73, 389-396.
- Patrican, L. A., DeFoliart, G. R. & Yuill, T. M. (1985). La Crosse viremias in juvenile, subadult and adult chipmunks (*Tamias striatus*) following feeding by transovariallyinfected Aedes triseriatus. American Journal of Tropical Medicine and Hygeine 34, 596-602.
- Patterson, J. L., Holloway, B. & Kolakofsky, D. (1984). La Crosse virions contain a primer-stimulated RNA polymerase and a methylated cap-dependent endonuclease. *Journal of Virology* 52, 215-222.

- Patterson, J. L. & Kolakofsky, D. (1984). Characterization of La Crosse virus smallgenome segment transcripts. *Journal of Virology* **49**, 680-685.
- Pattnaik, A. K. & Abraham, G. (1983). Identification of four complementary RNA species in Akabane virus-infected cells. *Journal of Virology* 47, 452-462.
- Pelet, T., Curran, J. & Kolakofsky, D. (1991). The P gene of bovine parainfluenza virus 3 expresses all three reading frames from a single mRNA editing site. *EMBO Journal* 10, 443-448.
- Pelletier, I., Couderc, T., Borzakian, S., Wyckoff, E., Crainic, R., Ehrenfeld, E. & Colbere, G. F. (1991). Characterization of persistent poliovirus mutants selected in human neuroblastoma cells. Virology 180, 729-737.
- Peluso, R. W. & Moyer, S. A. (1988). Viral proteins required for the in vitro replication of vesicular stomatitis virus defective interfering particle genome RNA. Virology 162, 369-376.
- Penn, C. R. & Mahy, B. W. (1985). Novel polypeptides encoded by influenza virus subgenomic (DI type) virion RNAs. Virus Research 3, 311-321.
- Pennington, T. H., Pringle, C. R. & McCrae, M. A. (1977). Bunyamwera virus-induced polypeptide synthesis. *Journal of Virology* 24, 397-400.
- Pensiero, M. N., Jennings, G. B., Schmaljohn, C. S. & Hay, J. (1988). Expression of the Hantaan virus M genome segment by using a vaccinia virus recombinant. *Journal of* Virology 62, 696-702.
- Perbal, B. (1988). A practical guide to molecular cloning. John Wiley and sons. pp. 811
- Perrault, J. (1981). Origins and replication of defective interfering particles. Current Topics in Microbiology and Immunology 93, 151-207.
- Persson, R. & Pettersson, R. F. (1991). Formation and intracellular transport of a heterodimeric viral spike protein complex. *Journal of Cell Biology* **112**, 257-266.
- Pesonen, M., Kuismanen, E. & Petterson, R. F. (1982). Monosaccharide sequence of protein-bound glycans of Uukuniemi virus. Journal of Virology 41, 390-400.
- Peter, B., Man, Y. M., Begg, C. E., Gall, I. & Leader, D. P. (1988). Mouse cytoskeletal gamma-actin: analysis and implications of the structure of cloned cDNA and processed pseudogenes. *Journal of Molecular Biology* 203, 665-675.
- Pettersson, R. F. (1974). Effect of Uukuniemi virus infection on host cell macromolecular synthesis. *Medical Biology* 52, 90-97.
- Pettersson, R. F., Kaariainen, L., von Bonsdorff, C.-H., Oker-Blom, N. & (1971). Structural components of Uukuniemi virus, a noncubical tick-borne arbovirus. Virology 46, 721-729.
- Pettersson, R. F., Kuismanen, E., Ronnholm, R. & Ulmanen, I. (1985). mRNAs of Uukuniemi virus, a Bunyavirus. Viral messenger RNA: transcription, processing, splicing and molecular structure. Y. Becker (Ed.). Boston, Nijhoff Publishing. p. 283-300.

- Pettersson, R. F. & von Bonsdorff, C.-H. (1975). Ribonucleoproteins of Uukuniemi virus are circular. Journal of Virology 15, 386-392.
- Pettersson, R. F. & von Bonsdorff, C.-H. (1987). Bunyaviridae. Animal virus structure. Amsterdam, Elsevier. p. 147-157.
- Pettersson, R. F., Hewlett, M. J., Baltimore, D. & Coffin, J. M. (1977). The genome of Uukuniemi virus consists of three unique RNA segments. *Cell* 11, 51-63.
- Pifat, D. Y., Osterling, M. C. & Smith, J. F. (1988). Antigenic analysis of Punta Toro virus and identification of protective determinants with monoclonal antibodies. *Virology* 167, 442-450.
- Pobjecky, N., Nathanson, N. & Gonzalez-Scarano, F. (1989). Use of a resonance energy transfer assay to investigate the fusion function of La Crosse virus. *Genetics and pathogenicity of negative strand viruses*. New York, Elsevier.
- Poidinger, M., Coelen, R. J. & Mackenzie, J. S. (1991). Persistent infection of Vero cells by the flavivirus Murray Valley encephalitis virus. *Journal of General Virology* 72, 573-578.
- Preble, O. T. & Youngner, J. S. (1975). Temperature-sensitive viruses and the etiology of chronic and inapparent infection. *Journal of Infectious Diseases* 131, 467-473.
- Pringle, C. R., Clark, W., Lees, J. F. & Elliott, R. M. (1984). Restriction of sub-unit reassortment in the Bunyaviridae. Segmented negative strand viruses. Orlando, Academic Press. 45-50.
- Pringle, C. R. & Iroegbu, C. U. (1982). A mutant identifying a third recombination group in a bunyavirus. *Journal of Virolog* 42, 873-879.
- Pringle, C. R., Lees, J. F., Clark, W. & Elliott, R. M. (1984). Genome sub-unit reassortment among bunyaviruses analysed by dot hybridization using molecularly cloned complementary DNA probes. *Virology* 135, 244-256.
- Raju, R. & Kolakofsky, D. (1986). Inhibitors of protein synthesis inhibit both La Crosse virus S-mRNA and S genome syntheses in vivo. Virus Research 5, 1-9.
- Raju, R. & Kolakofsky, D. (1987). Translational requirement of La Crosse virus S-mRNA synthesis: *in vivo* studies. *Journal of Virology* 61, 96-103.
- Raju, R. & Kolakofsky, D. (1987). Unusual transcripts in La Crosse virus-infected cells and the site for nucleocapsid assembly. *Journal of Virology* **61**, 667-672.
- Raju, R. & Kolakofsky, D. (1988). La Crosse virus infection of mammalian cells induces mRNA instability. *Journal of Virology* 62, 27-32.
- Raju, R. & Kolakofsky, D. (1989). The ends of La Crosse virus genome and antigenome RNAs within nucleocapsids are base paired. *Journal of Virology* 63, 122-128.
- Raju, R., Raju, L. & Kolakofsky, D. (1989). The translational requirement for complete La Crosse virus mRNA synthesis is cell-type dependent. *Journal of Virology* 63, 5159-5165.

- Ramseur, J. M. & Friedman, R. M. (1977). Prolonged infection of interferon-treated cells by VSV: possible role of temperature sensitive mutants and interferon. *Journal of General Virology* 37, 523-533.
- Ranki, M. & Pettersson, R. F. (1975). Uukuniemi virus contains an RNA polymerase. Journal of Virology 16, 1420-1425.
- Razzaque, A., Mizusawa, H. & Seidman, M. M. (1983). Rearrangements and mutagenesis of a shuttle vector plasmid after passage in mammalian cells. *Proceedings of the National Academy of Sciences, U.S.A.* 80, 3010-3014.
- Re, G. G. & Kingsbury, D. W. (1986). Nucleotide sequences that affect replicative and transcriptional efficiencies of Sendai virus deletion mutants. *Journal of Virology* 58, 578-582.
- Re, G. G. & Kingsbury, D. W. (1988). Paradoxical effects of Sendai virus DI RNA size on survival: inefficient envelopment of small nucleocapsids. *Virology* 165, 331-337.
- Re, G. G., Morgan, E. M. & Kingsbury, D. W. (1985). Nucleotide sequences responsible for generation of internally deleted Sendai virus defective interfering genomes. *Virology* 146, 27-37.
- Rebel, A., Baslé, M., Pouplard, A., Kouyoumdjian, S., Filmon, R. & Lepatezaw, A. (1980). Viral antigens in osteoclasts from Paget's disease of bone. *Lancet* 16, 344-346.
- Rebel, A., Malkani, K., Baslé, M. & Bregeou, C. (1977). Is Pagets disease of bone a viral infection? *Calcified Tissue Research (Supplement)* 22, 283-286.
- Reddy, E. P., Sandberg-Wollheim, M., Mettus, R. V., Ray, P. E., DeFreitas, E. & Koprowski, H. (1989). Amplification and molecular cloning of HTLV-I sequences from DNA of multiple sclerosis patients. *Science* 243, 529-533.
- Renz, D. & Brown, D. T. (1976). Characteristics of Sindbis virus temperature-sensitive mutants in cultured BHK-21 and Aedes albopictus (mosquito) cells. Journal of Virology 19, 775-781.
- Riedel, B. & Brown, D. T. (1977). Role of extracellular virus in the maintenance of the persistent infection induced in *Aedes albopictus* (mosquito cells)cells by Sindbis virus. *Journal of Virology* 23, 554-561.
- Riedel, B. & Brown, D. T. (1979). Novel antiviral activity found in the media of Sindbis virus-persistently infected mosquito (*Aedes albopictus*) cell cultures. *Journal of* Virology 29, 51-60.
- Roizman, B. & Sears, A. E. (1987). An inquiry into the mechanisms of herpes simplex virus latency. Annual Review of Microbiology 41, 543-571.
- Ron, A., Tattersall, P. & Tal, J. (1984). Formation of a host range mutant of the lymphotrophic strain of minute virus of mice during persistent infection in mouse L cells. *Journal of Virology* 52, 63-69.

- Ron, D. & Tal, J. (1985). Coevolution of cells and virus as a mechanism for the persistence of lymphotropic minute virus of mice in L-cells. *Journal of Virology* 55, 424-430.
- Ron, D. & Tal, J. (1986). Spontaneous curing of a minute virus of mice carrier state by selection of cells with an intracellular block of viral replication. *Journal of Virology* 58, 26-30.
- Ronnholm, R. & Pettersson, R. F. (1987). Complete nucleotide sequence of the M RNA segment of Uukuniemi virus encoding the membrane glycoproteins G1 and G2. Virology 160, 191-202.
- Rossier, C., Patterson, J. & Kolakofsky, D. (1986). La Crosse virus small genome mRNA is made in the cytoplasm. *Journal of Virology* 58, 647-650.
- Rossier, C., Raju, R. & Kolakofsky, D. (1988). La Crosse virus gene expression in mammalian and mosquito cells. *Virology* 165, 539-548.
- Rothman, J. E. (1981). The Golgi apparatus: two organelles in tandem. Science 213, 1212-1218.
- Roux, L. & Holland, J. J. (1979). Role of defective interfering particles of Sendai virus in persistent infections. Virology 93, 91-103.
- Roux, L., Simon, A. E. & Holland, J. J. (1991). Effects of defective interfering viruses on virus replication and pathogenesis in vitro and in vivo. Advances in virus research 40, 181-211.
- Rowlands, D. J., Clarke, B. E., Carroll, A. R., Brown, F., Nicholson, B. H., Bittle, J. L., Houghten, R. A. & Lerner, R. A. (1983). Chemical basis of antigenic variation in foot-and-mouth disease virus. *Nature* 306, 694-697.
- Rozhon, E. J., Gensemar, P., Shope, R. E. & Bishop, D. H. L. (1981). Attenuation of virulence of a bunyavirus involving an L RNA defect and isolation of LAC/SSH/LAC and LAC/SSH/SSH reassortants. Virology 111, 125-138.
- Russell, W. C. (1983). Paramyxoviruses and morbillovirus infections and their relationship to neurological disease. *Progress in Brain Research* 59, 113-132.
- Russell, W. C., Randall, R. E. & Goswami, K. K. (1989). Multiple sclerosis and paramyxovirus. *Nature* 340, 6229.
- Ruusala, A., Persson, R., Schmaljohn, C. S. & Pettersson, R. F. (1992). Coexpression of the membrane glycoproteins G1 and G2 of Hantaan virus is required for targeting to the Golgi complex. Virology 186, 53-64.
- Sakimura, K. (1962). The present status of thrips-borne viruses. *Biological transmission of disease agents*. New York, Academic Press. 33-40.
- Samso, A., Bouloy, M. & Hannoun, C. (1975). Presence de ribonucleoproteines circulaires dans le virus Lumbo (Bunyavirus). C. R. Academy of Science (III) D280, 779-782.

- Samso, A., Bouloy, M. & Hannoun, C. (1976). Mise en evidence de molecules d'acide ribonucleique circulaire dans le virus Lumbo (Bunyavirus). C. R. Academy of Science (III) D282, 1653-1655.
- Scheefers-Borchel, U., Scheefers, H., Edwards, J. & Brown, D. T. (1981). Sindbis virus maturation in cultured mosquito cells is sensitive to actinomycin D. Virology 110, 292-301.
- Scheid, A. & Choppin, P. W. (1977). Two disulphide-linked polypeptide chains constitute the active F protein of paramyxoviruses. *Virology* 80, 54-66.
- Schlesinger, S. (1988). The generation and amplification of defective interfering RNAs. RNA Genetics. Boca Raton, Florida., CRC Press. p. 167-185.
- Schmaljohn, C. S. (1990). Nucleotide sequence of the L genome segment of Hantaan virus. Nucleic Acids Research 18, 6728.
- Schmaljohn, C. S., Arikawa, J., Hasty, S. E. & et al (1988). Conservation of antigenic properties and sequences encoding the envelope proteins of prototype Hantaan virus and two virus isolates from Korean haemorrhagic fever patients. *Journal of General* Virology 69, 1949-1955.
- Schmaljohn, C. S., Chu, Y. K., Schmaljohn, A. L. & Dalrymple, J. M. (1990). Antigenic subunits of Hantaan virus expressed by baculovirus and vaccinia virus recombinants. *Journal of Virology* 64, 3162-70.
- Schmaljohn, C. S. & Dalrymple, J. M. (1983). Analysis of Hantaan virus RNA: evidence for a new genus of Bunyaviridae. *Virology* **131**, 482-491.
- Schmaljohn, C. S. & Dalrymple, J. M. (1984). Biochemical characterization of Hantaan virus. Segmented negative strand viruses. Orlando, Academic Press. p. 117-124.
- Schmaljohn, C. S., Hasty, S. E., Rasmussen, L. & Dalrymple, J. M. (1986). Hantaan virus replication: effects of monensin, tunicamycin and endoglycosidases on the structural glycoproteins. *Journal of General Virology* 67, 707-717.
- Schmaljohn, C. S., Jennings, G. B., Hay, J. & Dalrymple, J. M. (1986). Coding strategy of the S genome segment of Hantaan virus. Virology 155, 633-643.
- Schmaljohn, C. S., Parker, M. D., Ennis, W. H., Dalrymple, J. M., Collett, M. S., Suzich, J. A. & Schmaljohn, A. L. (1989). Baculovirus expression of the M genome segment of Rift Valley fever virus and examination of antigenic and immunogenic properties of the expressed proteins. *Virology* 170, 184-192.
- Schmaljohn, C. S. & Patterson, J. L. (1990). Bunyaviridae and their replication. Part II. Replication of Bunyaviridae. *Fields Virology*. New York, Raven Press. p. 1175-1194.
- Schmaljohn, C. S., Schmaljohn, A. L. & Dalrymple, J. M. (1987). Hantaan virus M RNA: coding strategy, nucleotide sequence, and gene order. *Virology* **157**, 31-39.
- Schneider-Schaulies, S., Liebert, U. G., Baczko, K., Cattaneo, R., Billeter, M. & V., t. M. (1989). Restriction of measles virus gene expression in acute and subacute encephalitis of Lewis rats. Virology 171, 525-534.

- Sekellick, M. J. & Marcus, P. I. (1980). Viral interference by defective particles of vesicular stomatitis virus measured in individual cells. *Virology* **104**, 247-252.
- Sharief, M. K., Hentges, R. & Ciardi, M. (1991). Intrathecal immune response in patients with the post-polio syndrome. *New England Journal of Medicine* 325, 749-755.
- Shenk, T. E., Koshelnyk, K. A. & Stollar, V. (1974). Temperature-sensitive virus from Aedes albopictus cells chronically infected with Sindbis virus. Journal of Virology 13, 439-447.
- Sheshberadoran, H. & Norrby, E. (1984). Three monoclonal antibodies against measles virus F protein cross-react with cellular stress proteins. *Journal of Virology* 52, 995-999.
- Shope, R. E. (1985). Bunyaviruses. Fields Virology. New York, Raven Press. p. 1055-1082.
- Shope, R. E. & Casey, O. R. (1962). Further studies on the serological relationships of group C arthropod-borne viruses and the application of these relationships to rapid identification of types. American Journal of Tropical Medicine and Hygiene 11, 886-887.
- Shope, R. E., Rozhon, E. J. & Bishop, D. H. L. (1981). Role of the middle-sized bunyavirus segment in mouse virulence. Virology 114, 273-276.
- Simizu, B. & Maeda, S. (1981). Growth patterns of temperature-sensitive mutants of Western equine encephalitis virus in cultured Aedes albopictus (mosquito) cells. Journal of General Virology 56, 349-361.
- Simons, J. F., Hellman, U. & Pettersson, R. F. (1990). Uukuniemi virus S RNA segment: ambisense coding strategy, packaging of complementary strands into virions, and homology to members of the genus Phlebovirus. *Journal of Virology* 64, 247-255.
- Simons, J. F. & Pettersson, R. F. (1991). Host-derived 5' ends and overlapping complementary 3' ends of the two mRNAs transcribed from the ambisense S segment of Uukuniemi virus. *Journal of Virology* 65, 4741-4748.
- Singh, K. R. P. (1967). Cell cultures derived from larvae of Aedes albopictus (Skuse) and Aedes aegypti (L.). Current Science 36, 506-508.
- Sivasubramanian, N. & Nayak, D. P. (1983). Defective intefering influenza RNAs of polymerase 3 gene contain single as well as multiple internal deletions. Virology 124, 232-237.
- Skehel, J. J., Bayley, P. M., Brown, E. B., Martin, S. R., Waterfield, M. D., White, J. M., Wilson, I. A. & Wiley, D. C. (1982). Changes in the conformation of influenza virus hemagglutinin at the pH optimum of virus-mediated membrane fusion. Proceedings of the National Academy of Sciences, U.S.A. 79, 968-972.
- Smith, J. F. & Pifat, D. Y. (1982). Morphogenesis of sandfly viruses (Bunyaviridae family). Virology 121, 61-81.
- Sobrino, F., Dávila, M., Ortin, J. & Domingo, E. (1983). Multiple genetic variants arise in the course of replication of foot-and-mouth disease virus cell culture. Virology 128, 310-318.
- Stalder, J., Reigel, F. & Koblet, H. (1983). Defective viral RNAs in Aedes albopictus C6/36 cells persistently infected with Semliki Forest virus. Virology 129, 247-254.
- Stark, C. & Kennedy, S. I. T. (1978). The generation and propagation of defectiveinterfering particles of Semliki Forest virus in different cell types. Virology 89, 285-299.
- Steinhauer, D. A. & Holland, J. J. (1987). Rapid evolution of RNA viruses. Annual Review of Microbiology 41, 409-433.
- Stephenson, J. R. & ter Meulen, V. (1982). A comparative analysis of measles virus RNA by oligonucleotide fingerprinting. Archives of Virology 71, 279-290.
- Stevens, J. G. (1989). Human herpesviruses: a consideration of the latent state. Microbiology Reviews 53, 318-332.
- Stohwasser, R., Giebel, L. B., Zoller, L., Bautz, E. K. & Darai, G. (1990). Molecular characterization of the RNA S segment of nephropathia epidemica virus strain Hallnas B1. Virology 174, 79-86.
- Stoker, M. & MacPherson, I. (1961). Studies on transformation of hamster cells by polyoma virus in vitro. Virology 14, 359-370.
- Stollar, V. (1980). Togaviruses in cultured arthropod cells. The Togaviruses: Biology, Structure, Replication. New York, Academic Press. p. 584-621.
- Stollar, V., Harrap, K. A., Thomas, V. & Sarver, N. (1979). Observations related to cytopathic effect in Aedes albopictus cells infected with Sindbis virus. Arctic and Tropical Arboviruses. New York, Academic Press. p. 277-296.
- Straver, P. J. & van Bekkum, J. G. (1972). Plaque production by carrier strains of footand-mouth disease virus in BHK-monolayers incubated at different temperatures. Arch. Gesamte Virusforsch 37, 12-18.
- Struthers, J. K., Swanepoel, R. & Shepherd, S. P. (1984). Protein synthesis in Rift Valley fever virus-infected cells. Virology 134, 118-124.
- Sundin, D. R. & Beaty, B. J. (1988). Interference to oral superinfection of Aedes triseriatus mosquitoes infected with La Crosse virus. American Journal of Tropical Medicine and Hygeine 38, 428-432.
- Sundin, D. R., Beaty, B. J., Nathanson, N. & Gonzalez, S. F. (1987). A G1 glycoprotein epitope of La Crosse virus: a determinant of infection of Aedes triseriatus. Science 235, 591-593.
- Suzich, J. A., Kakach, L. T. & Collett, M. S. (1990). Expression strategy of a phlebovirus: biogenesis of proteins from the Rift Valley fever virus M segment. *Journal* of Virology 64, 1549-1555.

- Takehara, K., Min, M. K., Battles, J. K., Sugiyama, K., Emery, V. C., Dalrymple, J. M. & Bishop, D. H. (1989). Identification of mutations in the M RNA of a candidate vaccine strain of Rift Valley fever virus. *Virology* 169, 452-457.
- Takemoto, K. K. & Habel, K. (1959). Virus-cell relationship in a carrier culture of Hela cells and coxsackie A9 virus. Virology 7, 28-44.
- Talmon, Y., Prasad, B. V. V., Clerx, J. P. M., Wang, G. J., Chin, W. & Hewlett, M. J. (1987). Electron microscopy of vitrified-hydrated La Crosse virus. *Journal of Virology* 61, 2319-2321.
- Tesh, R. B. (1980). Experimental studies on the transovarial transmission of Kunjin and San Angelo viruses in mosquitoes. *American Journal of Tropical Medicine and Hygiene* **29**, 657-666.
- Tesh, R. B. & Beaty, B. J. (1983). Localisation of California serogroup viruses in mosquitoes. *California serogroup viruses*. New York, Alan R. Liss. p. 67-75.
- Tesh, R. B. & Gubler, D. J. (1975). Laboratory studies of transovarial transmission of La Crosse and other arboviruses by Aedes albopictus and Culex fatigans. American Journal of Tropical Medicine and Hygiene 24, 876-880.
- Tesh, R. B. & Shroyer, D. A. (1980). The mechanism of arbovirus transovarial transmission in mosquitoes: San Angelo virus in *Aedes albopictus*. American Journal of Tropical Medicine and Hygiene 29, 1394-1404.
- Thompson, W. & Beaty, B. (1977). Venereal transmission of La Crosse (California encephalitis) arbovirus in Aedes triseriatus mosquitoes. Science 196, 530-531.
- Thompson, W. H. & Beaty, B. J. (1978). Venereal transmission of La Crosse virus from male to female Aedes triseriatus. American Journal of Tropical Medicine and Hygiene 27, 187-195.
- Tooker, P. & Kennedy, S. I. T. (1981). Semliki Forest virus multiplication in clones of *Aedes albopticus* cells. *Journal of Virology* 37, 589-600.
- Turell, M. J. (1988). Horizontal and vertical transmission of viruses by insect and tick vectors. *The arboviruses: epidemiology and ecology*. Boca Raton, Florida, CRC Press. p. 127-152.
- Turell, M. J. (1989). Effect of environmental temperature on the vector competence of *Aedes fowleri* for Rift Valley fever virus. *Research in Virology* **140**, 147-154.
- Turell, M. J. & Hardy, J. L. (1980). Carbon dioxide sensitivity of mosquitoes infected with California encephalitis virus. *Science* 209, 1029-1030.
- Turell, M. J., Hardy, J. L. & Reeves, W. C. (1982). Sensitivity to CO₂ in mosquitoes infected with California serogroup arboviruses. American Journal of Tropical Medicine and Hygiene 31, 389-394.
- Turell, M. J., Hardy, J. L. & Reeves, W. C. (1982). Stabilised infection of California encephalitis virus in *Aedes dorsalis*, and its implications for viral maintenance in nature. *American Journal of Tropical Medicine and Hygiene* 31, 1252-1259.

- Turell, M. J. & Le Duc, J. W. (1983). The role of mosquitoes in the natural history of California serogroup viruses. *California serogroup viruses*. New York, Alan R. Liss. p. 43-55.
- Turell, M. J., Reeves, W. C. & Hardy, J. L. (1982). Transovarial and trans-stadial transmission of California encephalitis virus in Aedes dorsalis and Aedes melanimon. 31, 1021-1029.
- Turell, M. J., Rossignol, P. A., Spielman, A., Rossi, C. A. & Bailey, C. L. (1984). Enhanced arboviral transmission by mosquitoes that concurrently ingested microfilariae. *Science* 225, 1039-1041.
- Tyrrell, D. L. J., Taechaudomtavorn, W., Lund, G. A. & Tovell, D. R. (1987). The mechanism by which Paramyxoviruses break tolerance to cellular actin. *The biology of negative strand viruses*. p. 304-308.
- Ulmanen, I., Seppala, P. & Pettersson, R. F. (1981). *In vitro* translation of Uukuniemi virus-specific RNAs: identification of a non-structural protein and a precursor to the membrane glycoproteins. *Journal of Virology* 37, 72-79.
- Ushijima, H., Clerx-van Haaster, C. M. & Bishop, D. H. L. (1981). Analyses of Patois group bunyaviruses evidence for naturally occuring recombinant bunyaviruses and existence of immune precipitable and non-precipitable nonvirion proteins induced in bunyavirus-infected cells. *Virology* **110**, 318-332.
- Vandvik, B. & Norrby, E. (1989). Paramyxovirus SV5 and multiple sclerosis. *Nature* 338, 769-771.
- Varma, M. G. R., Pudney, M. & Leake, C. J. (1979). Methods in mosquito cell culture. Practical tissue culture applications. New York, Academic Press, Inc. p. 331-350.
- Verani, P., Nicoletti, L. & Marchi, A. (1984). Establishment and Maintenance of persistent infection by the Phlebovirus Toscana in Vero cells. *Journal of General Virology* 65, 367-375.
- Vezza, A. C., Rejik, P. M., Cash, P. & Bishop, D. H. (1979). In vivo transcription and protein synthesis capabilities of bunyaviruses. Wild-type snowshoe hare virus and its temperature sensitive group I, group II and group I/II mutants. Journal of General Virology 31, 426-436.
- Vogt, M. & Dulbecco, R. (1958). Properties of Hela cell culture with increased resistance to poliomyelitis virus. Virology 5, 425-434.
- Volkmer, N., Soares, M. C. M. & Rebello, M. A. (1983). Autointerference of Maribuba virus (Bunyaviridae) in mouse L cells by defective interfering particles. *Intervirology* 20, 108-113.
- von Magnus, P. (1947). Studies on Interference in experimental influenza. I. Biological observations. Arch. Kemi. Mineral. Geol. 24, 1.
- Vorndam, A. V. & Trent, D. W. (1979). Oligosaccharides of the California encephalitis virus. Virology 95, 1-7.

- Walker, D. L. (1964). The viral carrier state in animal cell culture. *Progress in Medical Virology* 6, 111-148.
- Walker, D. L. (1968). Persistent viral infection in cell cultures. *Medical and Applied Virology*. p. 99-110.
- Wang, G. J., Hewlett, M. & Chiu, W. (1991). Structural variation of La Crosse virions under different chemical and physical conditions. *Virology* 184, 455-459.
- Ward, V. K., Marriott, A. C., el Ghorr A. A. & Nuttall, P. A. (1990). Coding strategy of the S RNA segment of Dugbe virus (Nairovirus; Bunyaviridae). Virology 175, 518-524.
- Watret, G. E. & Elliott, R. M. (1985). The proteins and RNAs of St. Abb's Head virus, a Scottish Uukuvirus. *Journal of General Virology* 66, 1001-1010.
- Watret, G. E. & Elliott, R. M. (1985). The proteins and RNAs specified by Clo Mor virus, a Scottish Nairovirus. *Journal of General Virology* 66, 2513-2516.
- Watret, G. E., Pringle, C. R. & Elliott, R. M. (1985). Synthesis of bunyavirus-specific proteins in a continuous cell line (XTC-2) derived from Xenopus laevis. Journal of General Virology 66, 473-482.
- Watts, D. M., Pantuwatana, S., DeFoliart, G. R., Yuill, T. M. & Thompson, W. H. (1973). Transovarial transmission of La Crosse virus (California encephalitis group) in the mosquito Aedes triseriatus. Science 182, 1140-1143.
- Wechsler, S. & Meissner, H. C. (1982). Measles and SSPE viruses: Similarities and differences. *Progress in Medical Virology* 28, 65-95.
- Weinberg, R. A. (1980). Integrated genomes. Annual Review of Biochemistry 49, 197-226.
- Weiss, B., Levis, R. & Schlesinger, S. (1983). Evolution of virus and defective-interfering RNAs in BHK cells persistently infected with Sindbis virus. *Journal of Virology* 48, 676-684.
- Weiss, B., Rosenthal, R. & Schlesinger, S. (1980). Establishment and maintenance of persistent infection by Sindbis virus in BHK cells. *Journal of Virology* 33, 463-474.
- Welsh, R. M. & Pfau, C. K. (1972). Determinants of lymphocytic choriomeningitis interference. Journal of General Virology 14, 177-187.
- Wickens, M. P. & Dahlberg, J. E. (1987). RNA-protein interactions. Cell 51, 339-342.
- Wilfert, C. M., Buckley, R. H., Mohanakumar, T., Griffith, J. F., Katz, S. L., Whisnant, J. K., Eggleston, P. A., Moore, M., Treadwell, E., Oxman, M. N. & Rosen, F. S. (1977). Persistent and fatal central nervous system echovirus infections in patients with agammaglobulinemia. *New England Journal of Medicine* 296, 1485-1489.
- Wyatt, H. (1973). Poliomyelitis in hypogammaglobulinemics. Journal of Infectious Diseases 128, 802-806.

- Yoo, D. W. & Kang, C. Y. (1987). Nucleotide sequence of the M segment of the genomic RNA of Hantaan virus 76-118. *Nucleic Acids Research* 15, 6299-6300.
- Younger, J. S. & Preble, O. T. (1980). Viral persistence: Evolution of viral populations. Comprehensive Virology 16 H. Fraenkel-Conrad and R. Wagner (Eds.). New York, Plenum Press. p. 73-135.
- Yuill, T. M. (1983). The role of mammals in the maintenance and dissemination of La Crosse virus. *California serogroup viruses*. C. H. Calisher and W. H. Thompson (Eds.). New York, Liss, Alan R. p. 77-87.
- Zweerink, H. J. & Joklik, W. K. (1970). Studies on the intracellular synthesis of reovirusspecified proteins. *Virology* **41**, 501-518.

